Market Diffusion of Plug-in Electric Vehicles and their Charging Infrastructure

Till Gnann





Fraunhofer Institute for Systems and Innovation Research ISI

Book Series »Innovation Potentials«

Till Gnann

Market Diffusion of Plug-in Electric Vehicles and their Charging Infrastructure

FRAUNHOFER VERLAG

Contact: Fraunhofer-Institute for Systems and Innovation Research ISI Breslauer Strasse 48 76139 Karlsruhe Phone 0721 6809-0 Fax 0721 689152 E-Mail info@isi.fraunhofer.de URL www.isi.fraunhofer.de

D 90

Zugl.: Karlsruhe, KIT, Diss., 2015

Bibliographic information published by Die Deutsche Nationalbibliothek
Die Deutsche Nationalbibliothek lists this publication in the Deutsche Nationalbibliografie;
detailed bibliografic data is available in the Internet at www.dnb.de.
ISSN: 1612-7455
ISBN: 978-3-8396-0933-0

Printing and Bindery: Mediendienstleistungen des Fraunhofer-Informationszentrum Raum und Bau IRB, Stuttgart

Printed on acid-free and chlorine-free bleached paper.

© by FRAUNHOFER VERLAG, 2015

Fraunhofer Information-Centre for Regional Planning and Building Construction IRB P.O. Box 80 04 69, D-70504 Stuttgart Nobelstrasse 12, D-70569 Stuttgart Phone +49 (0) 7 11/970-2500 Fax +49 (0) 7 11/970-2507 E-Mail : verlag@fraunhofer.de URL www.verlag.fraunhofer.de

All rights reserved; no part of this publication may be translated, reproduced, stored in a retrieval system, or transmitted in any form or by any means, electronic, mechanical, photocopying, recording or otherwise, without the written permission of the publisher.

Many of the designations used by manufacturers and sellers to distinguish their products are claimed as trademarks. The quotation of those designations in whatever way does not imply the conclusion that the use of those designations is legal without the consent of the owner of the trademark.

Market diffusion of plug-in electric vehicles and their charging infrastructure

GENEHMIGTE DISSERTATION

zur Erlangung des akademischen Grades eines Doktors der Sozial- und Wirtschaftswissenschaften (Dr. rer. pol.) von der Fakultät für Wirtschaftswissenschaften des Karlsruher Instituts für Technologie (KIT)

von

Dipl.-Wi.-Ing. Till Gnann

Tag der mündlichen Prüfung: 27.07.2015 Referent: Prof. Dr. rer. pol. Martin Wietschel Korreferent: Prof. Dr. rer. pol. Wolf Fichtner

Für meinen Vater

Abstract

Plug-in electric vehicles (PEVs) are a means to reduce greenhouse gas emissions from the transportation sector and diminish the dependency on fossil fuels. An often stated barrier to the diffusion of PEVs is the lack of charging options which prevents potential users from purchasing PEVs. However, the limited number of PEV users impedes a profitable operation of public charging points and thus their installation and roll-out. This thesis analyzes this potential lock-in effect for PEVs.

An agent-based model is developed to simulate the joint diffusion of PEVs and their charging infrastructure in Germany until 2030. An individual analysis of several thousand real-world driving profiles from the private sector and several hundred from commercial vehicles collected for this thesis allows to determine the individual utility optimizing drive train. The utility of each drive train is calculated based on its total costs of ownership, favoring and obstructing factors of PEVs, and takes the current charging infrastructure into account. The subsequent joint simulation of PEV users' driving and charging at public charging points permits to decide on the profitability of public charging stations and a potential further construction.

The results show that the dissemination of public slow-charging options has no influence on the diffusion of PEVs in Germany until 2030 from a techno-economical point of view. Instead, domestic charging points are a prerequisite for private PEVs and commercial charging points for commercial PEVs. These charging facilities already cover the charging needs of most vehicles, while additional charging points at work could further increase the number of private PEVs. Furthermore, the annual PEV registrations are dominated by commercial plug-in hybrid electric vehicles which only occasionally recharge public slow-charging facilities. Hence, public slow-charging options cannot become economically viable until 2030 – even with a growing number of PEVs. Further research is needed to quantify the psychological need of public charging points and their implications for the diffusion of PEVs.

This thesis is based on my research conducted at the Fraunhofer Institute for Systems and Innovation Research ISI under the supervision of Professor Dr. Martin Wietschel at the Institute for Industrial Production (IIP) at the Karlsruhe Institute of Technology.

Kurzfassung

Elektrofahrzeuge können einen Beitrag leisten, den Ausstoß von Treibhausgasen und die Abhängigkeit von fossilen Brennstoffen im Transportsektor zu reduzieren. Häufig wird dabei die fehlende Ladeinfrastruktur als Hemmnis für die Verbreitung von Elektrofahrzeugen genannt, das potenzielle Nutzer davon abhält, Elektrofahrzeuge zu kaufen. Jedoch verhindert die beschränkte Anzahl an Elektrofahrzeugen einen profitablen Betrieb öffentlicher Ladepunkte und einen damit verbundenen Auf- und Ausbau. In dieser Arbeit wird dieser potenzielle Lock-in-Effekt untersucht.

Für die gekoppelte Diffusion von Elektrofahrzeugen und ihrer Ladeinfrastruktur wird ein agentenbasiertes Modell entwickelt, das Deutschland bis 2030 betrachtet. Mithilfe einer individuellen Analyse von mehreren tausend privaten und einigen hundert, eigens für diese Arbeit erhobenen, gewerblichen Fahrprofilen von Fahrzeugen, wird die nutzenoptimale Antriebsart für jedes Fahrzeug bestimmt. Der Nutzen wird auf Basis der Total Cost of Ownership des Fahrzeugs, sowie hemmenden und fördernden Faktoren berechnet, wobei die gegebene Ladeinfrastruktur berücksichtigt wird. Anschließend wird eine simultane Ladesimulation der Elektrofahrzeuge durchgeführt, welche eine Rentabilitätsrechnung für öffentliche Ladepunkte und die Entscheidung über einen Ausbau ermöglicht.

Die Ergebnisse zeigen, dass die Verbreitung öffentlicher Langsamladepunkte in Deutschland aus techno-ökonomischer Sicht keinen Einfluss auf die Marktdiffusion von Elektrofahrzeugen bis 2030 hat. Ladepunkte an privaten Stellplätzen für Privatfahrzeuge oder auf dem Firmengelände für gewerbliche Fahrzeuge sind eine Voraussetzung für die Verbreitung von Elektrofahrzeugen. Diese Lademöglichkeiten decken den Ladeinfrastrukturbedarf für die meisten Fahrzeuge ab; eine zusätzliche Lademöglich-keit für Privatfahrzeuge am Arbeitsplatz kann die Nutzerzahl zudem erhöhen. In den simulierten Neuzulassungen der Elektrofahrzeuge überwiegen jedoch gewerbliche Plug-in Hybridfahrzeuge, die nur in selten öffentlich laden. Aus diesen Gründen können sich öffentliche Ladepunkte bis 2030 nicht amortisieren – auch nicht mit einer zunehmenden Zahl an Elektrofahrzeugen. Weiterer Forschungsbedarf ist notwendig, um den psychologischen Bedarf an öffentlicher Ladeinfrastruktur zu quantifizieren und dessen Implikation auf die Diffusion von Elektrofahrzeugen zu bestimmen.

Diese Arbeit wurde im Rahmen meiner Forschungsarbeit am Fraunhofer-Institut für System- und Innovationsforschung ISI erstellt und betreut von Prof. Dr. Martin Wietschel am Institut für industrielle Produktion (IIP) am Karlsruher Institut für Technologie (KIT).

Contents

1	Introduction										
	1.1	Motiv	ation	1							
	1.2	Resear	rch question and course of action	3							
2	Bac	Background									
	2.1	Background of alternative fuel vehicles									
		2.1.1	Overview of alternative fuel vehicles	5							
		2.1.2	Charging infrastructure needs of plug-in electric vehicles	7							
	2.2	.2 Literature review of previous modeling approaches									
		2.2.1	Stylized facts of vehicle and infrastructure market diffusion	10							
		2.2.2	Review of models for joint vehicle and infrastructure								
			diffusion	15							
		2.2.3	Discussion of presented co-diffusion models	20							
3	Veh	Vehicle usage data 2									
	3.1	Long	observation periods of vehicle usage data and different user groups	27							
	3.2 Private vehicle usage data										
		3.2.1	Overview of private vehicle usage data sets	30							
		3.2.2	Comparison of private vehicle usage data sets	32							
	3.3	Comm	nercial vehicle usage data	35							
		3.3.1	Overview of commercial vehicle usage data sets	35							
		3.3.2	Comparison of commercial vehicle usage data sets	36							
4	Mo	del dev	velopment and techno-economical parameters	43							
	4.1	Model overview									
	4.2	4.2 Mathematical description of the model									
		4.2.1	Individual plug-in electric vehicle simulation	48							
		4.2.2	Determination of utility	49							
		4.2.3	Stock model and joint simulation	52							
		4.2.4	Charging point operator	54							
	4.3	Techn	o-economical parameters	56							
		4.3.1	Scenario definition	56							
		4.3.2	Cost for charging infrastructure	59							
		4.3.3	Vehicle market and vehicle cost parameters	60							
		4.3.4	Willingness to pay more for plug-in electric vehicles	62							
		4.3.5	Limited plug-in electric vehicle availability	64							
		4.3.6	Adaptations for the geographical simulation	65							
	4.4	Discus	sion	68							

		$\begin{array}{c} 4.4.1 \\ 4.4.2 \\ 4.4.3 \end{array}$	Validation	. 68 . 71 . 75		
5	Moo 5.1	lel resu Market 5.1.1 5.1.2 5.1.3 5.1.4	ults c potentials of private and commercial plug-in electric vehicles Comparison of private and commercial vehicle usage PEV market potentials in different user groups	77 . 79 . 79 . 82 . 85 . 85		
	5.2	Market 5.2.1 5.2.2 5.2.3 5.2.4 5.2.5	building of plug-in electric vehicles Market diffusion of plug-in electric vehicles until 2030 Sensitivity analyses Increasing range with non-public charging or rental cars Discussion Summary	. 87 . 88 . 90 . 94 . 98 . 101 . 104		
	5.3 5.4	Interact 5.3.1 5.3.2 5.3.3 5.3.4 5.3.5 5.3.6 Synops	tive plug-in electric vehicle and infrastructure market diffusion Static analysis of geographically distributed driving behavior Public charging infrastructure evolution	 105 105 107 113 119 121 125 126 		
6	Con 6.1 6.2	clusion Summa Discuss	ary and further research ary and conclusions	129 . 129 . 132		
A	ppei	ndices	3	134		
A	Veh A.1 A.2 A.3	icle usa Extrac Furthe Collect	age datation of private driving profilesr information about MOPSion of commercial vehicle usage data	135 . 135 . 141 . 142		
в	Tech	nno-eco	onomical parameters	145		
Da	Danksagung 1					
Lis	st of	Abbre	viations and Variables	151		

List of Figures	158
List of Tables	160
Bibliography	161

Chapter 1

Introduction

1.1 Motivation

The reduction of greenhouse gas (GHG) emissions and the dependency on fossil fuels are two of the most fundamental challenges of the 21st century [Kahn Ribeiro et al., 2007, IEA, 2010]. This is particularly important for the transport sector which is a major contributor to global GHG emissions (23% of global GHG emissions in 2004 [Kahn Ribeiro et al., 2007], 22% of global CO₂ emissions in 2012 [IEA, 2014]). For this reason, the European Union has set the goal for a reduction of 60% of all GHG emissions from the transportation sector by 2050, compared to 1990 [Peters et al., 2013], and the European Commission (EC) reacted with EC directive 443/2009 [EC, 2009]. This directive forces car makers to reach an average CO_2 emission of 95 g CO_2/km by 2020 in all European markets. Especially longterm goals, like a reduction of vehicle emissions to 20 g CO_2/km necessary to reach the 2°C goal in 2050, are not attainable with conventional fuels [McKinsey, 2009, Schade, W., 2010, Skinner et al., 2010, Peters et al., 2013]. Alternative fuel vehicles (AFVs) in general and plug-in electric vehicles (PEVs) or fuel cell electric vehicles (FCEVs) in particular could provide a means to reduce emissions from the transportation sector. Although the new attention paid to alternative fuels is often considered as a new peak of Gartner's hype cycle [Bakker, 2010, Jun, 2012, Konrad et al., 2012], the situation for PEVs is different compared to the 1980's in California or 1990's in Germany [Chan, 2007]. Apart from the political interest and rising prices for fossil energy [IEA, 2014]¹, PEVs are expected to become technically and economically competitive to conventional fuel vehicles in the near future due to technological improvements, like rising energy densities, and falling prices of Lithium-ion batteries [Nykvist and Nilsson, 2015, Thielmann, 2011, Thielmann et al., 2012].

Germany as one major car market and supplier of vehicles [OICA, 2014] is particularly interested in future technological improvements. German car makers want to obtain their position on the global car market, while policy makers aim at maintaining jobs in the automotive industry and at keeping up with developments in "green" technologies.² Hence, the German government has set a goal has set a goal of reaching a stock of one million plug-in electric vehicles by 2020 and 6 million by 2030 [NEP, 2009, RegProg, 2011, BMWi

¹Although currently the oil price is at a five-year-minimum, the long-term growth rate has been positive for several decades [IEA, 2014].

²Green technologies are sometimes denoted as the sixth Kondratieff wave [Naumer et al., 2010, Naumer et al., 2013].

and BMU, 2010]. Further, Germany intends to become a lead market and lead supplier of PEVs [NEP, 2009, RegProg, 2011, BMWi and BMU, 2010]. In order to monitor these goals and to propose further courses of action in Germany, the National Platform for Electric Mobility has been installed, consisting of policy makers, scientists as well as representatives from non-governmental organizations and the automobile industry (see [NPE, 2014] for their latest report). At the moment, the market for PEVs, in Germany and worldwide, is at an early stage and several scientific studies have been carried out to forecast their future market development.³ A common way to model the future evolution of PEVs is based on the total cost of ownership (TCO) of a vehicle⁴ and vehicle usage data as user behavior⁵. Yet, studies in this area of research incorporate several insufficiencies: (1) the commercial vehicle sector which accounts for more than half of all vehicle registrations in several major car markets⁶ is often neglected in scientific studies [Hacker et al., 2011b, Redelbach et al., 2013, Schühle, 2014]. (2) Also the vehicle buying decision is not solely based on cost⁷, but also on attributes like vehicle size, brand or safety, and costbased analyses should be extended (see e.g. [Eppstein et al., 2011]). Further, (3) most models use average driving patterns⁸ although driving varies largely between drivers and days⁹. When modeling the market diffusion of PEVs (and their charging infrastructure), shortcomings 2 and 3, and for Germany also shortcoming 1, should be reflected.

In addition to other factors, an important barrier to the adoption of PEVs is the so-called limited range anxiety [Tate et al., 2008], partly caused by the current lack of charging infrastructure [Dütschke et al., 2011b, Egbue and Long, 2012, Steinhilber et al., 2013]. Users wish to have a sufficient charging infrastructure before they buy a plug-in electric vehicle [Dütschke et al., 2011b, Egbue and Long, 2012, Steinhilber et al., 2013]. For this reason, the European Directive on the deployment of alternative fuels infrastructure [EC, 2014] forces member states to propose a national action plan on public charging infrastructure development and set up a minimum amount of public charging points. For Germany, the directive suggests 150,000 public charging points by 2020 [EC, 2014]. First actions in Germany have been taken by the Federal Ministry of Economic and Energy (BMWE) by defining the power levels and suggesting a first number of charging spots that should be publicly available (35,000 public charging points by 2020) [BMWE, 2015]. These amounts differ significantly, since there is no commonly accepted approach to determine the number of charging points needed and most scientific studies focus on the placement of charging points [Lam et al., 2013, Dong et al., 2014, Namdeo et al., 2014, Sathaye and Kelley, 2013]. Studies analyzing the economy or use of charging points often find that extensive usage is necessary to make charging points economically viable [Kley, 2011] while usage in several model projects is comparatively low [Ecotality and INL, 2012, Bruce et al., 2012]. Occupancy rates depend on the number of PEVs in the vehicle stock and should thus be analyzed jointly to overcome a potential lock-in [Lin and Greene, 2011, Dong et al., 2014, NPE, 2012, Kalhammer et al., 2007, BCG, 2009, Ma et al., 2014, Chen et al., 2013].

³See [Al-Alawi and Bradley, 2013] for a review on market diffusion models of electric vehicles.

⁴See e.g. [Thiel et al., 2010, McKinsey, 2011, Pfahl et al., 2013, Wu et al., 2015].

⁵See e.g. [Dagsvik et al., 2002, Santini and Vyas, 2005, Keles et al., 2008, Lamberson, 2008, Mock et al., 2009, Nemry and Brons, 2010, Wansart and Schnieder, 2010, Shepherd et al., 2012].

⁶See e.g. [KBA, 2014b, UK DoT, 2013, JP-StatBureau, 2013].

⁷See e.g. [Mueller and de Haan, 2009, de Haan et al., 2009].

⁸See e.g. [Dagsvik et al., 2002, Santini and Vyas, 2005, Lamberson, 2008, Mock et al., 2009, Wansart and Schnieder, 2010, Shepherd et al., 2012].

⁹See e.g. [Smith et al., 2011, Amjad et al., 2011, Neubauer et al., 2012].

This interaction and co-diffusion of AFVs and their refueling infrastructure is a field of research for several types of vehicles¹⁰, yet it has not been comprehensively analyzed for PEVs and their charging infrastructure.

1.2 Research question and course of action

Based on the three identified research gaps, (i) insufficiencies in PEV market diffusion modeling, (ii) missing quantitative simulation results for public charging options and (iii) lack of a joint analysis for PEVs and charging infrastructure diffusion, the main research question of this thesis is:

How do the diffusion of plug-in electric vehicles and the diffusion of their charging infrastructure mutually influence each other?

This research question has, to the best of the author's knowledge, not yet been comprehensively analyzed. Since PEVs are a new technology and data about vehicle registrations or the charging infrastructure stock is rare, a modeling approach shall be developed to answer the research question for Germany until 2030. This implies several consecutive questions:

Can the co-diffusion of PEVs be modeled in the same way as for other alternative fuel vehicles? Since the introduction of automobiles, several types of vehicles have diffused into vehicle stocks worldwide. A refueling infrastructure was set up for the gasoline vehicle as the first mass market vehicle. The first mass market alternative fuel vehicle¹¹ - the diesel vehicle - profited from the gasoline infrastructure that could be extended to diesel [Karlsch and Stokes, 2003, Grube, 2004]. The same holds for other fuels, for which refueling stations could be expanded. However, for PEVs, currently planned charging facilities exceed the number of refueling stations largely [BMWE, 2015, EC, 2014] because of technical limitations of the vehicles. Thus, when analyzing the co-diffusion of PEVs and their charging infrastructure, modeling approaches have to be investigated and extended for PEV specialties.

What are adequate data sources to model the driving and charging behavior of private and commercial PEVs? The technical limitations of PEVs necessitate a detailed analysis of driving behavior to estimate their market potential. Such analyses are often performed with vehicle usage data of conventional vehicles¹². As vehicle usage data is publicly available in many countries [US-DoT, 2009, ENTD, 2009, MOP, 2010, Auto21, 2011, Karlsson, 2013], adequate German data sets for the analysis of PEVs have to be determined that cover private and commercial vehicles. The latter is important so that the high share of commercial vehicle registration can be considered.

What influences the adoption of PEVs apart from (public) charging infrastructure? The market diffusion of PEVs is not only influenced by the diffusion of charging infrastructure, but also by a variety of other factors like energy or vehicle prices, the availability

¹⁰See e.g. [Huétink et al., 2010, Janssen et al., 2006, Melaina, 2003, Meyer and Winebrake, 2009, Schwoon, 2006, Stephan and Sullivan, 2004, van der Vooren and Alkemade, 2010].

¹¹Plug-in electric vehicles were available already in 1830, yet they could not compete with the mass production of gasoline vehicles in the early 1900's [Chan, 2007].

¹²See e.g. [Dagsvik et al., 2002, Santini and Vyas, 2005, Mock et al., 2009, Nemry and Brons, 2010, Shepherd et al., 2012].

of PEVs on the market or common parking spots. These should be retrieved and analyzed to put the influence of charging infrastructure into context. This allows to answer a last relevant question for policy makers: What should an adequate charging infrastructure construction look like from a technical, economical and user behavioral point of view?

In this thesis, the focus on Germany until 2030 allows to determine the early evolution of PEVs in a major market, after which a critical mass should be reached and the co-diffusion should have become self-sustaining [Allen, 1988, Mahler and Rogers, 1999]. Results in this area may be generalized and transferred to other countries. The outline of this thesis runs as follows: Chapter 2 contains a brief overview of plug-in electric vehicles and their charging infrastructure which points out the main differences to conventional vehicles and refueling stations (2.1). Thereafter existing studies on the co-diffusion of AFVs and their refueling infrastructure are reviewed to retrieve stylized facts that models should cope with and best fitting modeling approaches for the co-diffusion (2.2). Chapter 3 contains a presentation of vehicle driving data sets of private and commercial vehicles. The driving profiles later used for simulation are compared to other data sets and tested for their representativity. In Chapter 4 the agent-based simulation model ALADIN (Alternative Automobiles Diffusion and Infrastructure) is developed (Sections 4.1 and 4.2). Scenarios for the simulation are presented and all techno-economical assumptions, e.g. for vehicle characteristics, vehicle market or energy prices, are described (Section 4.3). The results of the simulations are presented in Chapter 5 which is divided into three parts: First, the market potential of commercial fleet vehicles is determined to show the differences of private and commercial vehicle driving and purchasing behavior (Section 5.1). Second, the market diffusion of PEVs with non-public charging infrastructure is determined. Here, special focus is put on the influence of framework conditions to show their impact on the PEV market diffusion (Section 5.2). Lastly, the joint diffusion of public charging points and PEVs is determined. Here, several variations of infrastructure configurations and subsidies are analyzed (Section 5.3). Finally, the thesis is summarized, conclusions are drawn and suggestions for further research are made in Chapter 6.

Chapter 2

Diffusion models of alternative fuel vehicles and their refueling infrastructures

Introduction

This chapter aims at demonstrating the need for research in the field of PEVs and their charging infrastructure diffusion and serves to point out what can be learned from earlier works when modeling the interaction of PEVs and charging infrastructure. It contains two sections: The first Section 2.1 holds a brief overview of AFVs and defines the distinctions of PEVs to other AFVs which are considered in this work. In addition, the differences of charging infrastructure of PEVs compared to refueling stations of conventional fuels are shortly described. In the second Section (2.2) a structured literature review of AFV and refueling infrastructure diffusion models is performed to gain insights from current modeling approaches.¹³

2.1 Background of alternative fuel vehicles

2.1.1 Overview of alternative fuel vehicles

Alternative fuel vehicles have been a field of research for several decades (see e.g. [Greene, 1985, Sperling and Kurani, 1987, Greene, 1996, Chan, 2007]). Research on PEVs started in the 1980's in California, while fuel cell electric vehicles (FCEVs) came into focus of research and politics in the 1990's. Since 2000 both vehicle groups have gained worldwide attention by car makers, politicians and the media, especially because of rising emissions and the dependency on fossil fuels in the transport sector [IEA, 2010, Kahn Ribeiro et al., 2007]. Besides these two vehicle groups, there are three more types of AFVs that have already gained significant market shares in some national vehicle markets: vehicles powered with biofuels, natural or synthetic gas vehicles and hybrid electric vehicles, although the latter are often not considered as AFV but as a further development of conventional vehicles (CV).

¹³Literature reviews are often performed to identify research gaps and before the starting new modeling processes [Doebling et al., 1996, Garcia and Calantone, 2002, Kemp and Volpi, 2008, Hacker et al., 2009].



Figure 2.1: Alternative fuel vehicles considered in this thesis. CV=conventional vehicle, HEV=hybrid electric vehicle, AFV=alternative fuel vehicle, NGV=natural gas vehicle, LPGV=liquefied petroleum gas vehicle, PEV=plug-in electric vehicle, FCEV=fuel cell electric vehicle, PHEV=plug-in hybrid electric vehicle, BEV=battery electric vehicle.

The largest group of alternative fuel vehicles of the worldwide vehicle stock is flexfuel vehicles with about 35 million vehicles and more than 23 million vehicles in Brazil. They can burn gasoline, ethanol and methanol or any mixture of them in an internal combustion engine. Since the latter two fuels are mostly produced from biomass, flex-fuel vehicles can be counted to biofuel powered vehicles. Vehicles solely fueled by ethanol are mainly registered on the Brazilian vehicle market and account for about six million vehicles worldwide.¹⁴ Natural gas vehicles (NGVs) are propelled by an internal combustion engine that burns gaseous or compressed liquid natural gas (CNG). Together with vehicles that consume liquefied petroleum gas (liquefied petroleum gas vehicles (LPGVs)) or any other synthetic gas, they make up about 35 million vehicles powered with gases worldwide. About 20 million are NGVs with the largest markets in Iran (3.5m), China (3.3m), Pakistan (2.8m) and Argentina (2.5m), while LPGVs' largest markets are Turkey (3.9m), Russia (3m), Poland (2.8m) and South Korea (2.4m) totaling to 25 million LPGVs registered until 2014 [NGV-Journal, 2015, WLPGA, 2014]. Hybrid electric vehicles (HEVs) use an internal combustion engine for propulsion. In addition, a small battery is used for start-stop operation and recovers the energy from braking in a battery which is called recuperation. The energy that can be used for an additional stimulus of propulsion determines whether the vehicle is a mild (battery is only used for start-stop operation) or a full hybrid (battery is also used for "boosting"). Up to now, about 8.2 million hybrids have been sold worldwide with Japan (4m) and the US (3.4m) as their major markets [Marklines, 2015]. Fuel cell electric vehicles contain an electric drive train that is powered by a fuel cell unit where hydrogen is converted into electricity. A small buffer battery cannot be recharged at the electricity grid, but permits recuperation of braking energy. Currently, FCEVs are not recorded in any sales statistic since only very few FCEVs are available on the vehicle market at the moment, yet they may gain importance in the future.¹⁵

PEVs can be divided into three subgroups that are distinguished according to their level of electrification (see Figure 2.1 for distinctions and e.g. [Michaelis et al., 2013b, Wietschel et al., 2011] for definitions). Battery electric vehicles (BEVs) work completely

¹⁴Since there is no publicly available data on the registrations and sales of biofuels, which may stem from varying definitions about the mix of fuels, these registrations are based on press releases [Calmon, F., 2013, Motavalli, J., 2012].

¹⁵According to [Marklines, 2015] there are 135 FCEVs registered in stock in Japan and the US as of 1st of January 2015.

electric and store their propulsion energy in a Lithium-Ion battery.¹⁶ They can only be charged via the electricity grid and thus their tank-to-wheel emissions depend on the electricity mix charged. The second option of PEVs is the plug-in hybrid electric vehicle (PHEV) which can also be refueled at conventional refueling stations. PHEVs can be subdivided into parallel and serial hybrids, while the position of the fuel-powered and electric machine determine the distinction. Parallel PHEVs contain two drive trains that can be used independently or at the same time. An independent discharge of the battery is called charge-depleting mode, whereas during a joint use of both drive trains the vehicle is driven in charge-sustaining mode. In serial PHEVs, the vehicle comprises only the electric drive train that discharges the battery which can be recharged through a fuel-powered generator during driving. Serial PHEVs are often called range-extended electric vehicles (REEV) or extended range electric vehicles (EREV).¹⁷ Table 2.1 shows the market shares of PEVs in the major car markets in 2013 as well as their stock in 2014. Currently, PEVs do not account for a significant market share in any major car market, although smaller markets like Norway or the Netherlands reach up to 6% [Mock and Yang, 2014]. Today, the largest car market for PEVs are the US (223,000 PEVs) whereof the majority is driven in California.

Country	DE	US	JP	FR	CN
PEV market shares 2013 (new car sales) ^{a}	0.2%	1.3~%	0.6~%	0.8~%	0.1~%
PEV new car sales $2013^{a,b}$	7,000	95,000	25,000	9,000	12,000
PEV stock $01/01/2014^{c}$	24,000	223,600	88,500	$37,\!100$	29,100
PEV charging stations during $2014^{c,d}$	4,800	$15,\!200$	$5,\!000$	8,000	8,100

a: [Mock and Yang, 2014]; b: [ICCT, 2014]; c: [NPE, 2014]; d: Public & semipublic charging points for Germany; DE=Germany, US=United States, JP=Japan, FR=France, CN=China

This thesis focuses on PEVs and their charging infrastructure, since, together with FCEVs, they are the only means to reach EU and national emission targets for 2050 [McK-insey, 2009, Schade, W., 2010, Skinner et al., 2010]. With a certain number of PEVs available on the vehicle market, PEVs will be the near-term solution to cope with these targets [NPE, 2014]. To reach the current limit values for CO₂ emissions in 2020, however, a certain number of PEVs is necessary [Peters et al., 2013]. Apart from emission targets, plug-in electric vehicles could help to integrate intermittent renewable energies into the energy system [Dallinger and Wietschel, 2012, Dallinger et al., 2013]. The most important characteristics of PEVs and their charging infrastructure that are different from other (alternative) fuel vehicles are pointed out in the following section.

2.1.2 Charging infrastructure needs of plug-in electric vehicles

The major barriers for a large adoption of PEVs are their cost, their limited range and their associated insufficient charging infrastructure [Dütschke et al., 2011b, Götz et al., 2011, Globisch et al., 2013]. While PEV price reductions can be expected by economies

¹⁶Earlier PEVs were also based on nickel metal hydrid batteries (see e.g. [Watabe and Mori, 2011]).

¹⁷Serial and parallel hybrid electric vehicles are not distinguished in this thesis since their recharging or refueling behavior can hardly be distinguished. When both vehicle options are considered in the further analyses, they only vary in technical and economical parameters.

of scale, the limited range of PEVs is a matter of battery technology. Currently, batteries have an energy density that is about 50 times lower than for conventional fuels (see e.g. [Fischer et al., 2009]). R&D activities focus on improving the energy density of battery technologies, e.g. with new materials, but current Lithium-ion technology will be used in PEVs until at least 2025 [Thielmann et al., 2012, Thielmann et al., 2014]. The lower energy density of Lithium-ion batteries necessitates higher recharging frequencies for PEV users if today's driving behavior should be maintained.

The insufficiency of charging infrastructure seems to be unproblematic when comparing the current PEV stock and PEV charging stations in Table 2.1. The ratio of PEVs per (public) PEV charging stations (5-20) is much lower than the average ratio of vehicles per refueling stations for NGV suggested by [Janssen et al., 2006] (1,000). However, apart from the higher recharging frequency, the observed necessity of additional charging stations arises from their technical restraints.

Charging facilities can be described according to a number of attributes (see [Kley et al., 2011] and Figure 2.2). In this work, only conductive (wired) charging facilities are considered since inductive and charging options as well as battery exchange systems are very expensive [Kley, 2011]. Inductive charging facilities suffer from high energy losses when the vehicle and charging facility are not very close to each other Schraven et al., 2011. Also investments of charging facilities integrated into streets or parkings are significantly higher than for conductive charging points [Wietschel et al., 2009, Kley et al., 2010]. Investments for battery exchange stations would be even higher [Kley et al., 2010] and would force car makers to standardize their batteries. Apart from that, insurance policies for batteries are one more issue to solve which was not feasible for the only company offering this service [manager magazin, 2013]. For load shifting and the storage of intermittent renewable energies, a significant number of PEVs is required for a cost efficient operation. Only the power and capacity of a vehicle fleet is comparable to power or energy for needed for load shifting on a national level [Dallinger and Wietschel, 2012, Heinrichs, 2014, Babrowski et al., 2014]. Hence, only unidirectional connections without information flow are considered. This work focuses on very simple charging facilities to determine the amount of charging infrastructure for an early market evolution. For simplicity, pay per use is considered for billing which is metered at the charging station. To sum up, in this thesis, charging facilities are only distinguished by their accessibility and their power.

Four different types of accessibility of charging infrastructure are distinguished: domestic, commercial, semipublic and public charging facilities. Private or domestic locations are only accessible to private cars of one household. Commercial charging facilities offer the possibility to recharge commercial vehicles at the company site and are thus a pendant to domestic charging points. Semipublic charging facilities offer access to a limited number of people¹⁸, e.g. the members of a sports club or the workplace, whereas public charging options are available to everyone [Becker, 2009, Kley et al., 2010]. A lot of charging facilities are already available to potential PEV users with domestic charging spots, although at low power rates (level 1 and 2 (see below)). According to [infas and DLR, 2002, MOP, 2010, Behrends and Kott, 2009] about 60% of private vehicles in

¹⁸In [Reinke, 2014], semipublic charging facilities belong to a private owner, but offer access to the public, while publicly owned charging facilities with access to a limited number of people are special cases of public charging infrastructure. In this thesis, the definition for semipublic places is consistent with [Reinke, 2014] while the latter is considered as public charging infrastructure.



Figure 2.2: Characteristics of charging infrastructure considered in this thesis. Illustration based on [Kley et al., 2011].

Germany are parked in garages overnight (see also Section 3.3.2 and Figure 2.3).¹⁹ Also commercial charging facilities could most often be easily installed for vehicles belonging to a company.

The power at the charging facilities determines the time for recharging. Usually three levels of alternating current (AC) power connections are distinguished which depend on each country's electricity grid. In Germany these are 3.7 kW at level 1, 11–22 kW at level 2 and above 22 kW at level 3. A fourth level allows on high-voltage direct current (DC) power above 50 kW [Kley et al., 2011, Kley, 2011, Jochem et al., 2014]. Hence, charging a battery with a capacity of 25 kWh completely would take almost 7 hours at the lowest power level 1 in Germany. Higher power reduces the time to charge, yet also with a 100 kW charger it would take 15 minutes to recharge and thus longer than with conventional fuels.²⁰ With higher accessibility and power, the technical and organizational requirements of setting up charging facilities rise [Kley et al., 2011] and therewith the costs to install charging facilities.

Thus, the following major differences for the charging infrastructure of PEVs to other propulsion technologies can be identified:

- The technical limitations of current batteries and charging stations lead to higher refueling frequencies and durations compared to conventional fuels and to a higher need of charging facilities.
- However, the availability of some charging facilities in private, commercial or semipublic places reduces the necessity of a massive public charging station roll-out. This distinction of charging access types is different from conventional fuels.

Hence, models that focus the co-diffusion of PEVs and their charging infrastructure should consider these specialties.

¹⁹Similar shares of garages are also found for the US [Lin and Greene, 2011, Vyas et al., 2009].

 $^{^{20}}$ Costs for these charging options will be presented in Section 4.3.



Figure 2.3: Availability of garages in German vehicle stock. Garage availability and city sizes from [infas and DLR, 2002], total number of vehicles from [KBA, 2014a].

2.2 Literature review of previous modeling approaches²¹

Modeling the interaction in diffusion of AFVs and their refueling infrastructure has been a field of research for some decades, yet PEVs and their charging infrastructure interaction has not received much attention up to now. Therefore, this literature review of AFVs aims at identifying important factors and insights for further modeling undertakings. The following subsection identifies stylized facts and requests to models that should be considered when the co-diffusion of AFVs and their refueling infrastructure is analyzed in general (2.2.1). Models that treat the co-diffusion for different AFVs are presented and classified in Section 2.2.2. Their research questions and coverage of stylized facts are discussed in Section 2.2.3, while conclusions for further research are drawn at the end of this chapter.

2.2.1 Stylized facts of vehicle and infrastructure market diffusion

In the following, stylized facts of AFV market diffusion that should be included in models are identified. It is common practice in economical research to analyze important empirical regularities that are stable across different studies, markets or sectors [Kaldor, 1957, Easterly and Levine, 2001, Cont, 2001]. These empirical regularities are coined stylized facts [Kaldor, 1957] and helpful to summarize the present state of knowledge as well as to identify directions of further research [Easterly and Levine, 2001]. This subsection is

²¹This section is based on [Gnann and Plötz, 2015].

structured as follows: The background analysis starts with a review of empirical studies of NGVs as an AFV type that has diffused into markets in several countries in order to derive general stylized facts for AFV modeling. Thereafter user acceptance studies on AFVs and their refueling infrastructure are regarded to retrieve how potential users evaluate vehicles propelled by new fuels. These findings are summed up as factors that should be considered by models in the last part of this subsection. They build the basis for the model that is described in Chapter 4.

Empirical studies on natural gas vehicles and their refueling infrastructure

While for gasoline an initial infrastructure built-up was necessary, it was different for diesel vehicles and NGVs, since gasoline refueling stations were already in place and could be extended for these fuel types with minor modifications. For a comparison of several countries, their adoption of NGVs and their refueling infrastructure Janssen et al. (2006) introduced the so-called vehicle to refueling station index (VRI), defined as the number of refueling stations times 1,000 divided by the number of vehicles [Janssen et al., 2006]. The index shows the market development phases with the ratio being below one for early markets and above one for more mature markets (see [Janssen et al., 2006]).

Extending the work of Janssen et al. (2006) with newer data [VDA, 2010, eldia.com, 2011, BFS, 2013, ANP, 2012, KBA, 2013b, KBA, 2013a, MWV, 2013, Eurostat, 2013, GVR, 2012, Erdöl-Vereinigung der Schweiz, 2008, europia, 2012, UP, 2012], the historical evolution of NGV stocks is illustrated in Figure 2.4 while NGVs comprise vehicles propelled with liquefied petroleum gas (LPG) or CNG and their charging infrastructure for six different countries. Abscissa and ordinate show the market shares of stocks of NGVs (number of NGVs in stock divided by total number of vehicles in stock) and of refueling stations respectively (as number of refueling stations selling gas for these vehicles divided by the total number of refueling stations for CVs). Displayed are market shares for NGVs and CNG refueling stations for Argentina (AG), Brazil (BR), Germany (DE), Italy (IT), Sweden (SE) and Switzerland (CH) as well as LPGVs and LPG refueling stations for Germany and the Netherlands (NL). The blue line indicates equal market shares of vehicles and refueling stations. The diagram has two logarithmic axes to compare large values, e.g. of Brazilian market shares (share of CNG vehicles 2009: 7.5%), to small ones, e.g. in Switzerland (share of CNG 1999: 0.002%). Here, market shares are used instead of absolute values in order to be independent of the factor that has to be multiplied by vehicles (1,000 in [Janssen et al., 2006]), which allows a direct comparison of different vehicle markets.²² Although this illustration does not consider the charging station capacities (in terms of nozzles or amount of fuel sold) it serves as a proxy for the development of early markets.

All countries except Brazil show a share of NGV refueling stations higher than the share of NGVs. The Netherlands, Brazil and Argentina are more mature markets (more on the upper right corner) and seem to develop an initial refueling infrastructure before vehicles gain larger market shares. All countries' developments have positive slopes and for all countries except Brazil, the slopes are smaller than one, i.e. the stock share of vehicles is growing faster than the market share of refueling stations; for Brazil the slope is almost one. This might derive from the market maturity: younger markets with little

²²Since the ratio of refueling stations selling gas and the total number of refueling stations is displayed, the density of the refueling station network of CVs for each country is implicitly considered as well.



Figure 2.4: Stock market share of different gas vehicles and corresponding refueling stations for different countries and years. Data from [VDA, 2010, eldia.com, 2011, BFS, 2013, ANP, 2012, KBA, 2013b, KBA, 2013a, MWV, 2013, Eurostat, 2013, GVR, 2012, Erdöl-Vereinigung der Schweiz, 2008, europia, 2012, UP, 2012].

market shares for vehicles seem to converge to equal market shares of vehicles and refueling stations.

This illustrative example suggests the following relationships:

- Market shares for NGV refueling infrastructure are higher than market shares for NGVs in the beginning of a market diffusion.
- The ratio of both market shares tends to develop to one for early markets and to higher market shares for vehicles for more mature markets.

These findings are confirmed by studies on NGVs in several countries. Flynn (2002) analyzed NGVs in Canada in the years 1984 to 1986 [Flynn, 2002]. He studied policy measures, focusing on the barriers to adoption. His main findings with respect to the interaction of vehicle and infrastructure were that (1) infrastructure has to be available to customers as soon as there are vehicles on the market and (2) refueling stations have to become profitable soon to sustain investments into a further infrastructure development. Yeh (2007) studied the adoption of NGVs in Argentina, China, Italy, Pakistan, Brazil, India, New Zealand and the United States [Yeh, 2007], using the vehicle-to-refueling-station index VRI suggested in [Janssen et al., 2006]. In her work she found that the following conditions were decisive for a market diffusion of NGVs in all countries: (1) The prices for natural gas should be 40-50% lower than for conventional fuels, (2) the average payback period should not be more than 3-4 years, and (3) "that successful NGV markets have the tendency to gravitate toward a VRI of 1" [Yeh, 2007], which supports the

findings of the example on NGVs. Collantes and Melaina (2011) studied the co-evolution of NGVs and their refueling infrastructure in Argentina [Collantes and Melaina, 2011]. They used a quantitative and qualitative approach and concluded that there was little political influence in the build-up of refueling infrastructure and private users invested when vehicles entered the market. However, they stated that lower fuel prices and several infrastructure standards were the main drivers for adoption.

From the evidences collected by these studies, the following findings can be added:

- There has to be some infrastructure as soon as there are vehicles on the market which approves the first observation in this subsection.
- Refueling stations have to economize in the short to medium term, so a model should reflect their profitability.
- Fuel prices for AFVs should be lower than for conventional vehicles.

For modeling the co-diffusion of PEVs and their charging infrastructure, this means that an initial amount of charging infrastructure has to be in place, while the profitability of charging stations should be included in the model. Also the profitability of electric versus conventional driving will be reflected in the model introduced in Chapter 4.

How does refueling infrastructure influence the adoption of AFVs?

Several studies analyze the adoption of AFVs and their refueling infrastructure in general whereof some are presented in the following.

Sperling and Kurani (1987) investigated diesel vehicles and their corresponding refueling network in California [Sperling and Kurani, 1987]. They conducted a survey of diesel vehicle buyers in 1986 that addressed the influence of fuel availability in the vehicle buying decision. They compared buyers that bought their vehicles prior to or after 1982 and tested whether those buyers were less concerned about finding refueling stations who bought their car in later years. All interviewees were asked after they bought their car. From the responding 535 participants of this study they concluded that: (1) diesel car owners had the same or less difficulty finding refueling sites than expected, (2) the level of concern was about the same for earlier and later buyers. (3) Furthermore the authors stated that a minimum level of 15-20 % of all refueling stations has to offer the "new" fuel to meet users' needs [Sperling and Kurani, 1987]. Another survey performed by Sperling and Kitamura (1986) showed that an initial refueling network should be "about 1/10th of the size of the gasoline retail network [...] to relegate refueling concerns to a relatively insignificant role in the vehicle-purchase decision" [Sperling and Kitamura, 1986]. Greene (1996) supports the results of [Sperling and Kurani, 1987] regarding the minimum percentage of refueling coverage in a US-wide survey about flex-fuel vehicles conducted in 1996 with about 2,000 participants interviewed by phone [Greene, 1996]. He also found that if 15-20% of conventional refueling stations offered the new fuels, it would be sufficient to remove the adoption barrier of fuel availability.

Dütschke et al. (2011) reviewed user acceptance studies on natural gas vehicles in Germany and put them into international context to find out how acceptance for plug-in electric vehicles could be increased [Dütschke et al., 2011a]. Additionally, they conducted a survey of 142 NGV drivers in Germany in 2010. The main barriers to NGV adoption identified in this comprehensive analysis were cost and the lack of infrastructure, although the infrastructure concern "was not referred to very often by the interviewees" [Dütschke et al., 2011a]. Furthermore, concerns about lacking infrastructure for NGV-interested people were lower than for non-interested car drivers [Dütschke et al., 2011a].

Peters et al. (2011) studied plug-in electric vehicles and their possible early adopters with an online survey including 969 participants and tried to find out how Rogers' consumer groups [Rogers, 1962] could be characterized and identified for PEVs [Peters et al., 2011b]. They pointed out that consumers regard conventional infrastructure as highly superior to PEV charging infrastructure, but a significant influence of infrastructure in the purchase decision was not observed [Peters et al., 2011b]. The largest user acceptance study for PEVs in Germany up to now was conducted in the German pilot regions of E-mobility with more than 2,300 participants [Dütschke et al., 2011b]. According to Dütschke et al. (2012) the main factors consumers wish to be focused on for plugin electric vehicles and charging infrastructure are to increase the vehicle range and to decrease their charging duration [Dütschke et al., 2011b]. Users also wish that public charging infrastructure is set up by companies or public authorities which is supported by a study of Continental (2011) [Continental, 2011]. Götz et al. (2011) also identified the missing charging infrastructure as one of the major barriers to adoption [Götz et al., 2011]. Globisch et al. (2013) detected that charging their vehicles at home is perceived as positive by users of PEVs while charging duration and frequency of PEVs are clear disadvantages [Globisch et al., 2013]. With a questionnaire distributed in the German pilot regions, they compared private and commercial users and found that the availability of charging infrastructure is perceived as insufficient, while commercial users rate charging infrastructure as even less sufficient [Globisch et al., 2013].

The evidence presented in this section suggests that:

- There has to be a minimum level of infrastructure for first users to adopt (15-20 % of conventional refueling stations for fuels similar to conventional fuels). This confirms the first finding in the empirical studies section.
- Concerns about infrastructure of non-interested people are higher than for interested ones (or early adopters) who tend to have almost no concerns. Thus, a model for AFV and infrastructure adoption should be able to distinguish different user groups.
- Actual users of AFVs have less difficulty to find refueling stations than they thought prior to purchase. Hence, decreasing concerns with AFV use should be reflected by models as well.
- As political action for refueling infrastructure construction is demanded by users, the integration of policy options into models should be considered.

Transferring these findings to PEVs, the initial charging infrastructure that has to be in place are domestic charging facilities in garages, which can already be used. Further, the model described in Chapter 4 will distinguish user groups with large sets of vehicle driving profiles. Also decreasing concerns regarding the use of PEVs as well as the integration of policy options will be considered in the model.

Summary of stylized facts

In this subsection a number of stylized facts from empirical and user acceptance studies on AFVs are identified that should be considered in models for vehicle and refueling infrastructure diffusion. From the evidence collected in this subsection, it can be concluded that the following general factors should be integrated into models of combined AFV and AFV refueling infrastructure market diffusion:

- (A) An initial amount of AFV refueling or recharging infrastructure
- (B) AFV and AFV refueling infrastructure market shares
- (C) profitability of refueling or charging stations
- (D) fuel prices for conventional and alternative fuels
- (E) different user groups
- (F) decreasing user concerns with AFV use
- (G) potential policy measures

In Section 2.1.2 some specialties of PEVs and their charging infrastructure were pointed out that should be incorporated by models for PEV and charging infrastructure diffusion. Thus, apart from the general factors models should integrate, there are two specific factors relevant for PEV and infrastructure market diffusion models:

(H) refueling duration and frequencies can differ between PEV and conventional vehicles

(I) multiple types of infrastructure have to be differentiated by accessibility

Thus, nine factors are proposed which should be considered by models treating the interaction PEV and charging infrastructure diffusion (see also Table 2.3, first column). In the following section, existing models of AFV and AFV refueling infrastructure are tested whether they obey to these requests and find out if their modeling approach should be adopted. The requests to the model proposed in this thesis will be discussed in Section 4.4.

2.2.2 Review of models for joint vehicle and infrastructure diffusion

The studies presented thus far provide evidence that there is an interaction between AFV market diffusion and AFV refueling infrastructure market diffusion. This subsection holds a general scheme of model classification followed by a summary of model approaches used in literature to evaluate the interaction between AFV and AFV infrastructure.

Classification of models for vehicle and infrastructure diffusion

Generally, models can be clustered in many different ways and there is no common classification for all types of models (see e.g. [Zeigler, 1976, Geroski, 2000, Fleiter et al., 2011, Al-Alawi and Bradley, 2013, Jebaraj and Iniyan, 2006, Tran and Daim, 2008]). For the present case of models for AFV and their infrastructure market diffusion, model classification schemes from the market diffusion literature (such as [Geroski, 2000, Al-Alawi



Figure 2.5: Classification of models based on [Dreher, 2001, Sensfuss, 2008, Fleiter et al., 2011]

and Bradley, 2013]) could be used as well as model classification schemes from energy economics (such as [Fleiter et al., 2011, Ventosa et al., 2005]) for energy-related infrastructures such as refueling stations networks.

In the following, a classification based on [Dreher, 2001, Sensfuss, 2008, Fleiter et al., 2011] is used (Figure 2.5). This is a common classification for energy system models which copes with dynamic effects such as the interaction of vehicle and refueling infrastructure market diffusion (see also [Dreher, 2001, Ventosa et al., 2005, Sensfuss, 2008]).²³ This classification distinguishes between two model philosophies: bottom-up and top-down models. These model philosophies and subordinated model classes are briefly described in the following for a better understanding.

Top-down models are based on at least one main assumption or development which is decomposed in the analysis. They are generally used for macroeconomic coherences which study industries in relation to national economies [Dreher, 2001,Sensfuss, 2008]. Top-down models can be further subdivided into three model classes: input-output models, generalequilibrium models and macro-econometric models. *Input-Output models* are adapted to evaluate changes in economy through exogenous changes in the sectoral demand [Kemfert, 1998, Sensfuss, 2008] while *Computable General Equilibrium* models assume long-term equilibria and model the economy based on equations. They are frequently used to analyze policies and their impacts [Sensfuss, 2008]. *Macro-econometric models* are often applied to evaluate past events empirically and derive prognoses thereof. For that reason imperfect market behavior is estimated based on economic data [Sensfuss, 2008, Kemfert, 1998].

Bottom-up models combine several detailed assumptions which are composed to an overall picture. They are used if detailed technological and economical information about all necessary subgroups is available [Sensfuss, 2008, Fleiter et al., 2011]. Bottom-up models can be further divided into optimization models, simulation models and accounting frameworks [Fleiter et al., 2011]. Optimization models perform a demand and supply matching and often try to maximize the economic surplus. While the maximization can derive from all factors in the objective function there is always a number of constraints that have to be respected [Sensfuss, 2008, Fleiter et al., 2011]. Simulation models are not based on equilibria. Instead, a set of rules is applied for mechanisms that define behavioral processes in the model which return a step wise development of the whole system [Sensfuss, 2008]. Sensfuss (2008) also subdivides simulation models into agent-based models and

²³For a more general approach on the diffusion of innovations, refer to [Rogers, 1962, Geels, 2002]. In this study, a classification based on mathematical solution methods is preferred.

system dynamics models [Sensfuss, 2008]. Accounting frameworks model several sectoral outcomes and aggregate them for a full development [Fleiter et al., 2011].

With this classification, model approaches may be categorized and evaluated by their quality in fitting to the interaction of vehicle and refueling infrastructure diffusion for AFVs in general and PEVs in particular. This classification is used to categorize models from literature in the following.

Models for AFV and AFV infrastructure diffusion

In a literature review, ten studies were identified that jointly analyze the market diffusion of AFVs and their refueling infrastructure. An overview is given in Table 2.2, their approaches and main findings are presented in the following and summarized in Table 2.4 and Table 2.5.

Reference	AFV-type	Country	Time horizon*	
[Hu and Green, 2011]	LPGV, FCEV	several	2000 - 2030	
[Huétink et al., 2010]	FCEV	Netherlands	2010 - 2100	
[Janssen et al., 2006]	NGV	Switzerland	2000 - 2030	
[Köhler et al., 2010]	FCEV	EU27+2	2010 - 2040	
[Melaina, 2003]	FCEV	United States	2010 - 2030	
[Meyer and Winebrake, 2009]	FCEV	none	0 - 50	
[Schwoon, 2006]	FCEV	Germany	2010 - 2030	
[Schwoon, 2007]	FCEV	Germany	0 - 20	
[Stephan and Sullivan, 2004]	FCEV	none	0 - 20	
[van der Vooren and Alkemade, 2010]	LPGV	none	0-200	

Table 2.2: Overview of analyzed models for AFVs and their refueling infrastructure

*years or time steps

Hu and Green (2011) examined the co-diffusion of LPGVs and their infrastructure and drew conclusions for FCEVs in a macro-econometric approach [Hu and Green, 2011]. In a multi-national analysis, country-specific break-even-distances for LPGVs²⁴ were analyzed in an elasticity model and several break-even-distances were calculated to find out if the system could become self-sustaining. After this ex-post analysis on LPGVs, findings were transferred to FCEVs keeping as many aspects from LPGVs as possible. The main conclusions were that (1) there is a strong connection between market penetration and break-even-distance of LPGVs, (2) infrastructure increases trigger market penetration of vehicles and vice versa, not enough, however, for a self-sustaining growth. Moreover, (3) a mix of financial and non-financial support policies is required for a market penetration of FCEVs.

Huétink et al. (2010) investigated FCEVs and their refueling infrastructure in the Netherlands [Huétink et al., 2010]. An agent-based simulation model was used to study the "relation between initial refueling infrastructure and [...] hydrogen vehicles" with special focus on user behavior [Huétink et al., 2010]. In their work Huétink et al. (2010) applied

 $^{^{24}}$ In [Hu and Green, 2011], the break-even-distance is defined as the distance after which a LPGV is more economic than a CV.

Rogers' perceived innovation attributes (relative advantage, compatibility, complexity, trialability and observability [Rogers, 1962]) to FCEVs. They derived the FCEV purchase price, the availability of hydrogen and social learning as main drivers for FCEV and infrastructure diffusion. Based on these drivers Huétink et al. (2010) formed consumer and refueling station agents that interact in a predefined fictive area which allowed to test different initial infrastructure set-ups. This led to the overall result that network effects have a significant impact with three consecutive findings: (1) diffusion is slower than in benchmark patterns if network effects are considered, (2) a nationwide infrastructure roll-out is better for diffusion than an urban strategy and (3) the social network structure also has an influence, i.e. small networks have favorable conditions.

Janssen et al. (2006) studied natural gas vehicles with a focus on policy development in a system dynamics approach [Janssen et al., 2006]. After an examination of previous international experiences, a system dynamics model with three reinforcing loops (emerging market loops for vehicles and fueling stations as well as a market forecasting loop) and two balancing loops (market saturation loops of vehicles and fueling stations) was established. Several policies were compared to a base scenario for the Swiss vehicle market in order to test the possibility of stimulating the latter. They found five indicators for a degree of market penetration that were drawn from the analysis: (1) the ratio between NGVs per CNG refueling station, (2) the type coverage, which is the offer of vehicle models presented to potential customers, (3) the NGV investment pay-back time, (4) the sales per NGV type and (5) the subsidies per vehicle.

Köhler et al. (2010) examined FCEVs and their refueling infrastructure in Germany using a system dynamics approach to retrieve the initial amount of infrastructure needed and its political and macroeconomic implications [Köhler et al., 2010]. They extended an existing transportation model by an economic policy analysis within the transition towards hydrogen in the transport sector. The main findings were that (1) there is only a small subsidy needed for the initial set-up of refueling infrastructure, (2) "the overall impact on the economy is positive", and (3) both FCEVs and their infrastructure need political support during its introduction while a full transition to FCEVs will take a long time [Köhler et al., 2010]. Besides, the "provision of a hydrogen distribution (as well as production) is not a major economic barrier to the adoption of hydrogen vehicles" [Köhler et al., 2010, p.1238], which is not in line with findings of other studies [Ball et al., 2009, Offer et al., 2010].

Melaina (2003) analyzed FCEVs in the US [Melaina, 2003]. A threefold simulation approach was used to determine the initial number of hydrogen stations necessary to trigger a self-sustaining market diffusion for the US market. In the first approach, he determined the number of hydrogen stations as a percentage of gasoline stations based on [Sperling and Kurani, 1987] and found 4,600 to 17,700 stations. The second approach divided metropolitan land areas into parts which resulted into 1,600 to 4,500 stations. The third and finally preferred approach regarded arterial roads in the US with corresponding 4,500 to 9,200 hydrogen stations. With these estimated numbers he explored the cost of hydrogen infrastructure in the early FCEV adoption process and found that a highly coordinated political effort would generate the lowest cost.

Meyer and Whinebrake (2009) also investigated FCEVs and their infrastructure in the US [Meyer and Winebrake, 2009]. Based on the theory of complementary goods, a system dynamics model was created on three attributes: the FCEV adoption, FCEV refueling stations and hydrogen market conditions with four reinforcing and two balancing loops.

Unlike [Janssen et al., 2006] Meyer and Whinebrake (2009) used a reinforcing loop for the causal combination of FCEVs and hydrogen stations and one for conventional vehicles and their refueling stations. Two more reinforcing loops covered the scrapping of FCEVs and conventional vehicles. The balancing loops covered the construction of hydrogen and conventional stations. Like in [Köhler et al., 2010], the individual user decides whether to buy a FCEV based on a utility function in which the station density attractiveness is included. They found that scenarios in which both vehicles and infrastructure were subsidized, markets could reach momentum and become self-sustaining, while focusing on one market only would not. Hence, they concluded that policies would have to consider both complementary goods.

Schwoon (2006) analyzed hydrogen vehicles in Germany with a multi-agent simulation approach and differentiated between three types of interacting agents: consumers, car producers and refueling station owners [Schwoon, 2006]. Moreover, he used a utility function for each consumer which was extended by a term for the fuel availability. Unlike in system dynamic approaches, weights for the infrastructure values individually change over time due to network effects. In his work Schwoon (2006) focused on different taxing and refueling station set-up scenarios with the following findings: (1) a diffusion of FCEVs can take place without a dense infrastructure if conventional vehicle owners pay taxes for their driving and (2) car manufacturers could benefit from own investments in infrastructure or cooperation with its constructors.

Schwoon (2007) studied hydrogen vehicles and refueling infrastructure in an agentbased simulation model combined with a spatial GIS-model for Germany [Schwoon, 2007]. Extending his previous work, which dealt with the adoption and also regarded the car producers explicitly, [Schwoon, 2007] focused on the initial infrastructure development based on geographic information. For each individual driver a certain "don't worry distance" was introduced, which is the distance between two refueling stations that is enough for the user, and tested in several scenarios if the infrastructure development was sufficient to reach the tipping point²⁵. He concluded that (1) there have to be some initial hydrogen refueling stations for a market penetration kick-off, (2) the German HyWay-ring may be a good starting point, but (3) the optimal setting of the initial refueling stations depends on the "don't worry distance".

Stephan and Sullivan (2004) used an agent-based approach in a fictive area (central business district and surrounding rural area) for hydrogen vehicle and infrastructure market penetration [Stephan and Sullivan, 2004]. Two types of agents were considered: hydrogen retailers and (about 400) vehicle owners. Like most of the other agent-based approaches, consumers in [Stephan and Sullivan, 2004] also optimized a utility function, which in this case includes a worry factor for the availability of hydrogen like in [Schwoon, 2007]. They found that (1) cost for vehicles and infrastructure is an important factor and (2) the initial placement of refueling stations is crucial.

Van der Vooren and Alkemade (2010) had a more general approach to large technological systems in an accounting model [van der Vooren and Alkemade, 2010]. Their main research question was how the rise and fall of technologies that depend on infrastructure could be explained (cf. [Grübler, 1990]). They concluded: (1) the timing of a competing system has a large influence on the technological diffusion of the regarded system, (2) reaching momentum with infrastructure does not necessarily mean that the

²⁵When the co-diffusion system has reached the tipping point or critical mass, it becomes self-sustaining and no more external stimuli are needed [Allen, 1988, Mahler and Rogers, 1999].

corresponding technology has to reach momentum and (3) lock-in occurs only when both reach momentum.

When comparing the results of these ten studies, there are three main findings which several studies support: (1) political support is needed for the initial infrastructure setup [Hu and Green, 2011, Köhler et al., 2010, Melaina, 2003, Meyer and Winebrake, 2009]. (2) According to several studies the system can become self-sustaining [Köhler et al., 2010, van der Vooren and Alkemade, 2010], although some studies disagree [Hu and Green, 2011]. (3) An initial amount of refueling infrastructure is necessary for a self-sustaining system [Köhler et al., 2010, Schwoon, 2007, Stephan and Sullivan, 2004] which supports request A from the previous subsection. Furthermore, a variety of factors are identified which confirm the previously conducted requests.

While these models are compared according to the coverage of the extracted factors in the following section, their findings will be discussed with model results in Chapter 5. These findings are summed up in Table 2.4 and Table 2.5 at the end of this chapter.

2.2.3 Discussion of presented co-diffusion models

General discussion

Some models analyze vehicles powered by natural gas (three out of ten), while most analyze FCEVs (eight out of ten). To the best of the author's knowledge, there is no diffusion model that explicitly models the interaction of PEVs and their charging infrastructure. The large number of models considering FCEVs on the one hand and only a few with NGV on the other indicate the novelty of a combined diffusion approach in this field.

Beside the various results and implementation ideas covered by the models, there is a clear focus on bottom-up (nine out of ten) and simulation models (eight out of ten). Within the simulation models there are four agent-based approaches and three system dynamics models. While the bottom-up approaches require a large effort for data collection, top-down models do not operate on a similar level of detail. Of course, the required level of detail in modeling depends to a certain extend on the specific research question of each study.

There are three groups of research questions which are investigated by the models: (I) Is there a tipping point or how high is the tipping point? The majority of models tries to answer this question by estimating the number of refueling stations and users that have to adopt [Hu and Green, 2011, Köhler et al., 2010, Melaina, 2003, Meyer and Winebrake, 2009, Schwoon, 2007, Stephan and Sullivan, 2004]. (II) The second group of research questions focuses on policy options and infrastructure roll-out strategies that are often compared to the number of refueling stations [Huétink et al., 2010, Janssen et al., 2006, Schwoon, 2006]. (III) Finally, the last group of studies aims for theoretical insights [Hu and Green, 2011, van der Vooren and Alkemade, 2010]. When model classes and research questions are compared, no model class seems to be particularly well suited for one specific research question. Still, since the majority of the observed models favors simulation approaches, this model class seems appropriate for the specific requests in the papers.

Coverage of stylized facts

Table 2.3 summarizes which factors identified in Section 2.2.1 are covered by the ten models from literature. The symbol ' \checkmark ' indicates whether a specific factor is integrated in a model, '-' if it is not and '(\checkmark)' when the integration is not clear, but is presumed.

Table 2.3: Analysis of different models for AFVs and their refueling infrastructure. Indication if request is reflected in the model with ' \checkmark ', if not with '-', otherwise with '(\checkmark)' and footnote.

										_
(rear) (S) (Year) Factors to be modeled	[Hu and Green, 2011]	[Huétink et al., 2010]	[Janssen et al., 2006]	[Köhler et al., 2010]	[Melaina, 2003]	[Meyer and Winebrake, 2009]	[Schwoon, 2006]	[Schwoon, 2007]	[Stephan and Sullivan, 2004]	[van der Vooren and Alkemade, 2010
(A) Consideration of initial refueling	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	_
infrastructure										
(B) Eval. of market shares for AFVs and infrastructure	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
(C) Observation of profitability of charging stations	_	\checkmark	\checkmark	\checkmark	\checkmark	$(\checkmark)^a$	_	_	_	$(\checkmark)^b$
(D) Reflection of ratio of fuel prices	\checkmark	\checkmark	$(\checkmark)^c$	$(\checkmark)^a$	$(\checkmark)^a$	\checkmark	\checkmark	$(\checkmark)^d$	\checkmark	$(\checkmark)^b$
(E) Contemplation of different user groups	—	\checkmark	$(\checkmark)^f$	$(\checkmark)^f$	_	\checkmark	\checkmark	\checkmark	\checkmark	$(\checkmark)^b$
(F) Integration of decreasing user concerns with use	_	\checkmark	_	_	_	\checkmark	\checkmark	_	$(\checkmark)^a$	_
(G) Modeling of policy options	$(\checkmark)^e$	\checkmark	\checkmark	\checkmark	_	$(\checkmark)^a$	\checkmark	_	_	_
(H) Consideration of refueling time and frequency	_	_	_	—	_	-	_	_	_	_
(I) Reflection of different infrastructure	—	_	_	_	_	_	_	_	_	_

a: unclear, but supposedly yes; b: unclear, since too general; c: externally defined; d: supposedly individual reflection; e: policy recommendation, calculation unclear; f: logit for user differentiation

None of the models covers all aspects derived from the literature in Section 2.2.1. While all general AFV questions may be answered with 'yes' for [Huétink et al., 2010], there is one model where only once can be affirmed for sure and three times with doubts [van der Vooren and Alkemade, 2010]. Thus, if the affirmations are summed up, all models are within a narrow range and this comparison is not considered to be quantitative. However, (1) bottom-up simulation models answer more questions with 'yes' than top-down models and (2) there is no difference in coverage between system-dynamics and agent-based models. Hence, this indicates that a bottom-up simulation approach is not

only the most frequently used but also the most promising one to reproduce the stylized facts from Section 2.2.1.

Taking a closer look at the at the previously determined general factors, it can be stated that most models cover the general factors quite well, while user behavioral aspects (E and F) are best-covered by simulation models. Different user groups (E) are a basic ingredient of agent-based models and the integration of decreasing user concerns with time of use (F) is often covered by neighboring effects or social interaction [Bonabeau, 2002]. In the system dynamics models, users are often differentiated with logit-functions ([Janssen et al., 2006], in the part concerning the model ASTRA in [Köhler et al., 2010]) while in other models users individually decide on the basis of the logit-function [Meyer and Winebrake, 2009]. However, no correlation is to be found between factors and research questions.

Concerning the specific factors of PEVs there is no model covering any of the two aspects. Neither refueling duration and frequency (H), nor the different infrastructure owners (I) are covered. This might correspond to the fact that these two aspects result from technical and organizational challenges linked to PEV requirements which are not analyzed in these studies. Yet, this also means that current approaches cannot be transferred to plug-in electric vehicles without adaptations. For this reason, a new model approach is proposed in Chapter 4 that covers the extracted stylized facts as well as special PEV requests.

Summary and conclusions for further research

In this chapter, plug-in electric vehicles (PEVs) were defined and differentiated against other alternative fuel vehicles (AFVs). In addition, their differences concerning refueling behavior compared to conventional fuels were pointed out (Section 2.1). Thereafter, models for the interaction of the market diffusion of AFVs and their refueling infrastructure were analyzed in a structured literature review (Section 2.2). Stylized facts and factors to be integrated into models were retrieved from empirical analyses, studies on user behavior of AFVs and the specific characteristics of PEVs of Section 2.1. The identified factors were discussed for ten models. This collection comprises, to the best of the author's knowledge, the current models that cope with combined AFVs and refueling infrastructure market diffusion. Three conclusions and suggestions for the further procedure of this thesis can be condensed:

- 1. There is a chicken-and-egg-problem for AFVs, i. e. a lock-in effect where potential consumers do not buy vehicles when there is no refueling infrastructure and designated refueling infrastructure suppliers do not set up facilities as there are no customers. To overcome this situation, some initial refueling infrastructure has to be installed. Then the system may reach a tipping point from which on it can become self-sustaining. The tipping point is the most frequent research topic in this field. In the further proceeding of this work, this tipping point for PEVs will be analyzed.
- 2. Existing models on NGVs and FCEVs can reproduce historical market evolutions. The demands from the stylized facts where covered by most models analyzed (2.2.3).

Most of them are able to answer economical questions and to analyze some policy options. However, user adoption behavior and its change over time could be better integrated. The most common approaches are system dynamics or agent-based simulations. Hence, in Chapter 4, a simulation model for the co-diffusion of PEV and their charging infrastructure is proposed which is based on real-world driving profiles presented in Chapter 3.

3. The co-diffusion models analyzed in this section have to be strongly adapted for an integration of currently uncovered PEVs. With a widely available electricity grid and a large number of domestic charging facilities (at home in many major car markets) the chicken-egg-problem is different for potential PEV users that could recharge their vehicles at home. The longer charging duration and higher frequency as well as the lack of public charging options probably leads to a higher psychological need for charging facilities than the proposed 15–20% of conventional refueling stations. Currently, there is no model that treats the co-diffusion of plug-in electric vehicles and their charging infrastructure and none of the models analyzed here could be used to integrate PEVs without major adaptations. This attests the necessity of a new model for PEVs and their charging infrastructure which explicitly considers the charging duration, frequency and ownership (see Chapter 4).
| Author(s)
(Year) | Model
class | Research question | Novelty /
Highlights | Major findings |
|---------------------------|--|--|--|---|
| Hu and
Green
(2011) | Macro-
Economet
Model | existence or
richeight of
tipping
point (I),
theoretical
insights (III) | break-even-
distances from
LPGV transferred
to FCEV | market penetration
and break-even-
distance connected system not self-
sustaining mix of support
policies required |
| Huétink et
al. (2010) | Agent-
based
Simula-
tion | policy
options and
infrastruc-
ture roll-out
strategies
(II) | application of
Rogers' perceived
innovation
attributes to
FCEVs and
derivation of main
drivers for FCEV
and infrastructure
diffusion | inclusion of network
effects leads to slower
diffusion nationwide roll-out
preferable to urban
strategy structure of social
network influences
results |
| Janssen et
al. (2006) | System
Dynam-
ics
Simula-
tion | policy
options and
infrastruc-
ture roll-out
strategies
(II) | development of
policy options
and indicators for
degree of market
penetration | indicators: ratio between vehicles
and refueling stations offer of models pre-
sented to customers vehicle investment
pay-back time sales per vehicle type subsidies per vehicle |
| Köhler et
al. (2010) | System
Dynam-
ics
Simula-
tion | existence or
height of
tipping
point (I) | combination of
existing transport
model with model
for economic
policy analysis | only small subsidy
needed for initial hy-
drogen infrastructure overall economy im-
pact is positive political support for
vehicles and infrastruc-
ture needed infrastructure not a
major barrier for vehi-
cle diffusion |
| Melaina
(2003) | Various
Simula-
tions | existence or
height of
tipping
point (I) | threefold
approach for
determination of
initial charging
infrastructure
amount | number of refueling
stations varies with ap-
proach (1,600 to 17,700
hydrogen refueling sta-
tions for the US) coordinated politi-
cal set-up with lowest
costs |

Table 2.4: Analysis of models for AFVs and their refueling infrastructure - part 1 $\,$

Author(s) (Year)	Model class	Research question	Novelty / Highlights	Major findings
Meyer and Whine- brake (2009)	System Dynam- ics Simula- tion	existence or height of tipping point (I)	model based on theory of complementary goods	both vehicles and in- frastructure have to be financially supported
Schwoon (2006)	Agent- based Simula- tion	policy options and infrastruc- ture roll-out strategies (II)	changing weight for infrastructure in individual utility function due to network effects	- FCEV diffusion possible without dense infrastructure network if conventional cars are additionally taxed - car manufacturers could benefit from investments in infra- structure
Schwoon (2007)	Agent- based Simula- tion	existence or height of tipping point (I)	combination of agent-based model and GIS-simulation and introduction of "don't worry distance"	 initial amount of fill- ing stations necessary HyWay-ring in Ger- many as starting point optimal initial net- work depends on "don't worry distance"
Stephan and Sullivan (2004)	Agent- based Simula- tion	existence or height of tipping point (I)	combination of agent-based model and geographic information aspects	 cost is an important factor in vehicle and refueling infrastruc- ture diffusion modeling initial placement is crucial for successful market penetration
van der Vooren and Alkemade (2010)	Accountin Model	ng theoretical insights (III)	general view on large technological systems	 timing has a large influence on technology diffusion co-diffusing technologies do not have to reach momentum at the same time lock-in only occurs when both reach momentum

Table 2.5: Analysis of models for AFVs and their refueling infrastructure - part 2 $\,$

Chapter 3

Vehicle usage data

Introduction

The aim of this chapter is to describe and compare the main German vehicle usage data sets and to determine the ones to use in this thesis. First, the term driving profile is defined, then, the importance of a long observation period for an analysis of PEVs is shown and the differences of user groups are displayed (3.1). In the following, different data sets of private and commercial driving behavior are presented and analyzed whether they are representative for German vehicle registrations (Sections 3.2 and 3.3). While private driving profiles with a long observation period are publicly available for Germany, there are no commercial driving profiles with an observation period of more than one day. For this reason, a data collection of commercial vehicle usage data with an average observation period of three weeks has been performed by the author which is described in Section 3.3 as well.

3.1 Long observation periods for individual PEV analyses and the distinction of user groups

Vehicle usage data is an important means to study the driving behavior of persons and vehicles and are used for several detailed analyses (see e.g. Santini and Vyas, 2005, Kley, 2011, Amjad et al., 2011, Smith et al., 2011, Plötz et al., 2014a]). In this thesis, the term driving profile (often also referred to as driving pattern) comprises all trips of one vehicle within a certain time horizon and will be used equivalently to vehicle usage data. While several studies use average vehicle usage data [Dagsvik et al., 2002, Santini and Vyas, 2005, Keles et al., 2008, Köhler et al., 2010, Meyer and Winebrake, 2009, Nemry and Brons, 2010, Schade, 2008, Lamberson, 2008, Shepherd et al., 2012, Wansart and Schnieder, 2010, Orbach and Fruchter, 2011, they neglect the large variations in driving between drivers or between days of individual drivers [Amjad et al., 2011, Smith et al., 2011, Neubauer et al., 2012]. The inclusion of individual user behavior is relevant to identify niches (see also Section 2.2.1) and is particularly important for PEVs since their ability to perform their driving profile completely with a BEV or with high amounts of electric driving in a PHEV strongly depends on the individual vehicle usage and the regularity of his driving behavior [Karlsson and Kullingsjö, 2013, Plötz et al., 2014a]. The electric driving share of PHEVs essentially determines a its TCO. For example, if there are two vehicles whose

driving profiles have been collected for two days, and vehicle A drove 10 km the first and 130 km the second day while vehicle B drove 70 km on both days. The average daily mileage is 70 km for both vehicles, however a BEV with an electric range of 100 km would only be able to perform driving profile B without recharging during the day. If a PHEV with an electric range of 50 km is assumed to be able to recharge once per day, its electric driving share would be 71% for profile B, but only 43% for driving profile A.

The connection between technical PEV potentials and observation period was analyzed systematically in [Gnann et al., 2012a], where driving profiles with an observation period of seven days were tested in order to determine whether BEVs could perform them completely with varying battery capacities. The simulation of seven days was compared to the feasibility of every single day in the driving profiles while the latter showed a higher share of profiles that could performed with a BEV. Thus, the feasibility of a driving profile can be largely overestimated if short time driving profiles are used. Furthermore, an analysis of the influence of the observation period on electric driving shares was performed in which the confidence band of each vehicles' electric driving share with respect to the number of observation days was examined [Plötz et al., 2014a].²⁶ With an observation period of five days the 95%-confidence band of the individual electric driving shares showed a median at 19.1%, i. e. the error (due to observation period) on the electric driving share of half of the users was larger than 19.1%. This error decreased with 20 days of observation where the median of the 95%-confidence bands was at 9.5%. Hence, a precise determination of an individual's electric driving share requires a long observation period. If BEV feasibility and PHEV electric driving share are considered in a market potential analysis, results are affected by the observation period as well [Gnann et al., 2012c]. Since in this thesis the market diffusion of PEVs and their charging infrastructure is analyzed. the usage of driving profiles with a long observation period is mandatory (see also Greene, 1985, Neubauer et al., 2012). Yet, only a few models consider this premise (see e.g. [Pearre et al., 2011, Khan and Kockelman, 2012).

As a matter of fact, data collections are designed for special purposes and are subject to limited funding. So, data collections very often focus either on a long observation period to the account of the number of observations or on a large number of observations at the expense of a shorter observation period. Figure 3.1 shows the number of observations and the observation period of several international driving profile data sets [US-DoT, 2009, ENTD, 2009, WVI et al., 2010, MOP, 2010, Auto21, 2011, Karlsson, 2013, Fraunhofer ISI, 2014]. While there are some data sets with more than 30,000 individual driving profiles (NHTS (United States): 180,000; KiD2010 (Germany): 47,114; MiD2008 (Germany): 34,601), they were collected for only one day to provide cross-sectional data of the national driving behavior. However, as described earlier, an analysis of individual profiles is accompanied by errors if only one day is used. An analysis of individual driving behavior without large errors due to variation between days can be performed with very long observation periods like in [Auto21, 2011] (see e.g. [Blum, 2014]). Yet, the small sample of 75 vehicles in [Auto21, 2011] does not allow to generalize results for any subgroup. Hence, this thesis uses driving profiles of a certain length and a relevant number of observations which allows to use both advantages (individual analysis and possibility to generalize). These will be described and their characteristics will be discussed in the following sections.

²⁶For this analysis the electric driving share of each individual profile was assumed to be Gaussian distributed, so that the confidence interval was given by $\Delta s_i(T) = t_{(1-\alpha/2,T-1)} \cdot \sigma_i(T)/\sqrt{T}$ with T being the number of observation days and $t_{(x,n)}$ the Student's t-distribution for n degrees of freedom.



Figure 3.1: Comparison of different driving profile data sets with respect to observation period and number of observations. Data from [US-DoT, 2009, ENTD, 2009, WVI et al., 2010, MOP, 2010, Auto21, 2011, Karlsson, 2013, Fraunhofer ISI, 2014]

Furthermore, this work distinguishes between private and commercial driving profiles. This is important to mention since several studies neglect commercial vehicle owners (see e.g. [Schühle, 2014, Santini and Vyas, 2005, Kley, 2011, Shafiei et al., 2012, Hacker et al., 2011b]), although they account for more than 60% of annual passenger car registrations in Germany [KBA, 2014b].²⁷ German commercial and private vehicles differ in a number of characteristics (cf. Table 3.1). In contrast to vehicle registrations, the German vehicle stock is dominated by private vehicles which results from a shorter holding period of commercial vehicles. Table 3.1 also shows that the average driving distance of commercial cars is larger than for private vehicles which, paired with a regular driving, is favorable for PEVs. The statistical significance of differences of average distance and regularity of driving will be analyzed in Section 5.1.1.

However, the group of commercial users has to be further subdivided, since it comprises vehicles which are only used for commercial purposes as well as vehicles with mixed (private and commercial) use. In the following, the solely commercially used vehicles are called fleet vehicles, while vehicles with mixed use are so-called company cars. According to [Pfahl, 2013], the total number of company car registrations is about the same as for fleet vehicles [Pfahl, 2013]. Yet, the exact number of company cars is difficult to determine, since the German Federal Motor Authority does only distinguish natural and legal car holders, i. e. the distinction in Table 3.1. For the further procedure of this work, the following is assumed: The commercial vehicle stock is distributed equally to fleet vehicles and company cars based on [Pfahl, 2013].

Summing up, this section showed that (1) a long observation period is decisive for the analysis of PEVs and (2) the commercial vehicle sector has to be considered separately.

²⁷The commercial passenger car sector is very heterogeneous and no unique definition exists for it [Steinmeyer, 2007, Deneke, 2005]. Here, a commercial vehicle is defined to be licensed to a legal person or public entity, while a private vehicle is licensed to a private person.

criteria	private	commercial	reference
Stock (2014-01-01)	39,363,889	4,487,341	[KBA, 2014a]
Share of stock	89.8%	10.2%	[KBA, 2014a]
Registrations (2013)	$1,\!120,\!125$	$1,\!832,\!306$	[KBA, 2014b]
Share of registrations	37.9%	62.1%	[KBA, 2014b]
Avg. vehicle holding period [a]	6.2	3.8	[VCD, 2008, DAT, 2011]
Avg. motor size [ccm]	$1,\!638$	1,994	[KBA, 2014a]
Avg. VKT on weekday [km]	40.1	76.8	[IVS et al., 2002]
Avg. VKT on Sat./Sun. [km]	28.8	29.3	[IVS et al., 2002]

Table 3.1: Privately and commercially licensed passenger cars in Germany

These two conditions are considered when the driving profiles for the further analysis are determined in the following.

3.2 Private vehicle usage data

3.2.1 Overview of private vehicle usage data sets

In Germany, there are two large studies on the travel behavior of households publicly available: Mobility in Germany (MiD, [infas and DLR, 2002, infas and DLR, 2008]) and the German Mobility Panel (MOP, [MOP, 2010]). A third study on travel behavior (not publicly available) is similar to MOP yet restricted to the region of Stuttgart in southern Germany (MOPS, [Hautzinger et al., 2013]). These data sets are shown in Table 3.2.

Attribute	MiD2002	MiD2008	MOP	MOPS
Reference	[infas and	[infas and	[MOP, 2010]	[Hautzinger
	DLR, 2002]	DLR, 2008]		et al., 2013]
Data collection design	Questionnaire	Questionnaire	Questionnaire	Questionnaire
Time of collection	2002	2008	$1994 - 2010^{a}$	2012
Number of households	$26,\!848$	25,922	$12,\!812$	$992,\!584$
Number of vehicle	$33,\!293$	34,601	6,339	$1,\!312,\!817$
profiles	$(17,773)^b$	$(20,927)^b$		
Number of vehicle trips	61,645	$73,\!552$	$172,\!978$	$19,\!100,\!429$

Table 3.2: Overview of private driving profiles for Germany

a: collection is still ongoing, b: vehicles with movement

The data collection for MiD was performed twice in 2002 and 2008 and MiD is a household travel survey. Thus, not only trips with vehicles were collected, but also trips by foot, bike or public transport were recorded. The collection of 2002 was performed for 26,848 households (2008: 25,922) which had to report all their trips on one reporting day. Trips were collected during the whole year 2002 (and partially in 2003) whereof 33,293 vehicles can be extracted (2008: 34,601). Apart from the travel behavior, there is a large number of socio-demographic information about the households and drivers (e.g. income, education, sex) as well as the vehicles (e.g. vehicle size, fuel type, brand).

MOP is an ongoing household travel survey that has been performed since 1994 with

1,000 households every year.²⁸ Users report all their trips during one week which also include trips by foot, bike or public transport. By using information from 1994 until 2010, data of 12,812 households can be analyzed. In contrast to MiD, the allocation of trips to vehicles is not available in the initial data set. Thus, this allocation has to be performed where unambiguously possible based on the following assumptions (see also [Kley, 2011, Chlond et al., 2014]):

- If there is only one vehicle in the household, trips of all household members as vehicle driver are assigned to the vehicle. In this case the socio-demographic information of the first driver is assigned to the vehicle profile.
- If the number of vehicles exceeds the number of drivers, the trips of the first household member are assigned to the first vehicle, those of the second driver to the second vehicle until the last driver's trips are assigned. This might overestimate the driving of single vehicles.
- If the number of vehicles is smaller than the number of household members, the vehicles will be exempted from the further analysis, since the allocation of trips to vehicles is unknown.

Since the data structure changes over the years, an allocation algorithm has to be adapted for every year (cf. A.1 for details). This returns 6,339 vehicle driving profiles with about 170,000 single trips. Apart from the trip distances, the purposes of all trips are collected, which allows to distinguish between trips to private locations²⁹, to the work place³⁰ and trips with public destinations (other purposes) equal to the accessibility types of charging infrastructures in Section 2.1.2. Also, socio-demographic and vehicle information is available for this data set, such as the vehicle ownership which distinguishes natural and legal persons. Thus, this data set does not only contain private vehicles, but also company cars which are analyzed separately in the following.

For the MOPS-data, a seven-day mobility survey was performed with about 5,000 households in the region of Stuttgart (i.e. the six districts: Stuttgart, Göppingen, Ludwigsburg, Rems-Murr-Kreis, Esslingen and Böblingen – see Figure 3.2). Based on this survey, socio-demographic data of the region and trip matrices, the data set was extrapolated to the whole region of Stuttgart. Thus, this sample contains trips for 2.7 million persons, including all trips by foot, public transport or bike including their starting and stopping zones. Those zones are different in size and smaller the closer they are to the city center (central station of Stuttgart). There are also zones outside the observation area with starting and stopping points of trips, although the home of all users is within the observation area (see right panel of Figure 2). For more details on these zones refer to Table 3.3 and Figure A.1.

As only vehicle trips are of interest in this study, an allocation of personal trips to vehicles is performed where unambiguously possible (same assumptions as for MOP). A focus on 15 min-intervals further reduces the sample size and complexity. Socio-demographic information of vehicle owners and vehicles is rare in this data set since it is synthetically

²⁸Some households are chosen to participate multiple, yet not more than three, times.

²⁹Purposes 7 and 77 according to the codebook of MOP.

³⁰Purposes 2 and 4 according to the codebook of MOP.



Figure 3.2: Observation area of MOPS on German map (left), in detail (center) and divided into zones including surrounding outer area (right). Own display with data from [Hautzinger et al., 2013].

generated. Yet, the information about trip locations permits simulating the driving of PEVs simultaneously and geographically and testing whether charging points could be occupied by other PEVs in a charging simulation [Kuby et al., 2013]. The additional information available besides the trip distances, times, and purposes are the household sizes, age, sex and occupation of the driver as well as the locations of their homes and work places.

Table 3.3: Geographic zones of observation area of MOPS.

Attribute	Value
Surface of inner area	$3,652 \text{ km}^2$ (1% of Germany)
Zones in inner area	1,014
Average surface of inner area zones	$3.8 \text{ km}^2 \text{ (SD}=6.1 \text{ km}^2)$
Surface of outer area	$13,186 \ {\rm km^2}$
Zones in outer area	$140 \ (+20 \ \text{distant zones})$
Average surface of outer area zones	97.0 km ² (SD=92.9 km ²)

3.2.2 Comparison of private vehicle usage data sets

In this thesis, each driving profile is analyzed separately to test whether it is technically feasible and economically sensible as PEV. As explained in Section 3.1 this requires an observation period of multiple days. Hence, the data sets MOP and MOPS are used for further analyses. However, the information of MiD serves for validation of the other data sets in terms of driving behavior, vehicle sizes and garage availability.

Figure 3.3 shows the distributions of daily and annual vehicle kilometers traveled (VKT) in the data sets. On the left panel, the cumulative distribution functions of the daily trips are displayed. Shown are the data sets MiD2002 (dotted green) and MiD2008 (dotted black) with their mileage on the day of observation. Thus, only the vehicles with movement are displayed here. For MOP and MOPS two curves are displayed: the average daily distance for all drivers (i. e. the sum of the trips during the observation week divided by seven; displayed in dashed blue for MOP and dash-dotted black for MOPS) and the daily distances of all drivers on all driving days (dashed red for MOP and dash-dotted yellow for MOPS). By dividing the sample sizes, it is visible that a vehicle is moved on



Figure 3.3: Comparison of daily and annual VKT in different private driving profile data sets. *Left panel:* Cumulative distribution function of daily VKT. *Right panel:* Cumulative distribution function of annual VKT. Data from [infas and DLR, 2002, infas and DLR, 2008, MOP, 2010, Hautzinger et al., 2013].

6.46 days on average in MOP and on 4.63 days in MOPS.³¹ The figure contains four interesting findings: (1) The distribution daily VKTs of MiD2002 and MiD2008 is almost equal. This confirms the robustness of samples with a large number of observations when analyzing the whole sample. (2) When all individual daily distances of MOP or MOPS are compared to MiD, the shares of MOP and MOPS are slightly lower for shorter distances (73% of all vehicle trips in MiD and 67% of all vehicle trips in MOP are lower than 50 km). Thus, daily driving distances may be slightly overestimated by using the MOP data. (3) The share of users with an average daily vehicle distance below 120 km is lower in MOP than in MiD and higher above 120 km. This confirms the slight overestimation of driving distances when using the MOP data. (4) Driving distances for MOPS are lower than in all other data samples, especially long-distance trips are not included in the sample. These findings suggest a careful interpretation of results.

The right panel of Figure 3.3 uses the same display as the left panel, although here the cumulative distributions of the annual VKT are displayed. Shown are the reported annual VKT for MiD2002 (dash-dotted green), MiD2008 (dotted black) and MOP (dashed red). Due to incomplete responses in the questionnaires, the sample sizes are reduced here. Two main findings can be extracted: (1) The three curves are almost equal to each other, thus the sample size of MOP is sufficient for the further analysis of the VKT. (2) Several steps can be found which can be explained by the estimated annual VKT in the questionnaires. The statistical relevance of these differences will be analyzed hereafter.

Since driving behavior is connected with vehicle size, Table 3.5 shows the mean annual VKT, their standard deviations and subsample sizes of the three data sets MiD2002, MiD2008 and MOP distinguished by car sizes. For this distinction, a small vehicle is defined to have a motor with a cubic capacity of less than 1,400 ccm, a medium sized vehicle between 1,400 ccm and 2,000 ccm and a large vehicle with 2,000 ccm and more. As MOPS does not contain any information about car sizes it is not displayed in this table.

³¹This is not comparable to the share of vehicles in motion in MiD since not all trips in MOP could be allocated to vehicles and vehicles without movements were extracted.

Vehicle size	attribute	MiD2002	MiD2008	MOP
small	mean [km]	$11,\!280$	$11,\!826$	10,466
	stdev [km]	9,910	$13,\!384$	9,020
	number	$5,\!442$	6,751	966
medium	mean [km]	14,791	$14,\!805$	$14,\!676$
	stdev [km]	11,759	15,209	$11,\!199$
	number	13,222	20,517	2,165
large	mean [km]	17,839	15,965	16,204
	stdev [km]	$16,\!637$	16,912	11,746
	number	$5,\!536$	5,519	596

Table 3.4: Annual VKT in different vehicle sizes of private driving profiles for Germany. Data from [infas and DLR, 2002, infas and DLR, 2008, MOP, 2010]

In all three data sets medium sized vehicles' means are equal at an annual mileage of about 14,700 km per year with statistically insignificant differences.³² For small and large vehicles the average annual VKT differs slightly more. Based on statistical tests, only large vehicles of MiD2008 and MOP are in good accordance while all other samples are significantly different.³³ Significant differences are found between all three data sets. As all three data sets comprise a large number of participants, it is unclear which one is best. Thus, MOP is considered representative for private driving behavior.

Apart from driving behavior the availability of garages or a parking close to the own grounds is important when analyzing PEVs with respect to their charging infrastructure. As mentioned in Section 2.1.2 the availability of a garage could simplify the (individual) infrastructure set up largely. The questionnaires of MiD and MOP contain questions regarding this topic. However, their formulation is slightly different for all three. In MiD2002 users were asked for the common parking spot of every vehicle with the following responses: in a garage (58.5% of all vehicles), at a parking on own property (29.2%), at parking close to own property or apartment (7.9%) and at varying parking spots (3.9%)(see also Figure 2.3 and [Gnann et al., 2013] for a further distinction of city sizes). For MiD2008, responses were changed to the question of the common parking spot: on the own property (71.3%), at parking directly close to own property or apartment (25.2%), farther away from own property or apartment (1.0%) and at varying parking spots (2.3%). In MOP the share of vehicles with garages is 66.2%. Thus, the responses in MiD2008 do not allow to determine if the vehicle parking spot is a garage, and it is only possible to compare the garage availability from MOP to MiD2002 finding it slightly higher in MOP. Yet, the availability of garages in both samples is around 60%, which is confirmed by [Behrends and Kott, 2009].

In summary, MOP and MOPS will be used for further analyses because of their long observation period. Furthermore, in terms of driving behavior and garage availability there are small differences in the data sets. Their influence on results will be discussed in the further proceeding of this work.

 $^{^{32}}$ A two-sided t-test with unequal sample sizes and unequal variances rejects the null hypothesis with p=66.0% for a comparison of MiD2002 and MOP, p=62.3% when MiD2008 and MOP are compared and p=92.4% for MiD2002 and MiD2008 (see e.g. [Fahrmeir et al., 2011]).

³³A two-sided t-test with unequal sample sizes and unequal variances rejects the null hypothesis for small vehicles with p=1.1% (MiD2002-MOP), p < 0.01% (MiD2008-MOP) and p=0.97% (MiD2002-MiD2008). For large vehicles p=0.21% (MiD2002-MOP), p=65.3% (MiD2008-MOP) and p < 0.01% (MiD2002-MiD2008) are found.

Response option	MiD2002	MiD2008	MOP
Garage	58,8%	-	66.2%
on the own property	29.2%	71.3%	-
on parking directly close to own property or apartment	-	25.2%	-
on parking close to own property or apartment	7.9%	-	-
farther away from own property or apartment	-	1.0%	-
at varying parking spots	3.9%	2.3%	33.8%

Table 3.5: Common parking spot of private vehicles for Germany based on [infas and DLR, 2002, infas and DLR, 2008, MOP, 2010].

3.3 Commercial vehicle usage data³⁴

Commercial vehicles comprise about 60% of annual German vehicle registrations and are thus a relevant market for PEVs. Available vehicle usage data sets are discussed in this section.

3.3.1 Overview of commercial vehicle usage data sets

Until 2011, the only publicly available vehicle usage data set for German commercial vehicles was "Motor Traffic in Germany" (KiD, [IVS et al., 2002, WVI et al., 2010]). This data set only comprises vehicle driving profiles with an observation period of one day. For this reason a data collection for commercial vehicles has been designed and performed by the author in the ongoing project "Regional Eco Mobility 2030" resulting in the "REM2030 driving profiles" (REM2030, [Fraunhofer ISI, 2014]).³⁵ A subsample for the region of Stuttgart is named REM2030S and Table 3.6 gives an overview of the commercial data sets.

Criteria	KiD2002	KiD2010	REM2030	$\operatorname{REM2030S}$	
Reference	[IVS et al.,	[WVI	[Fraunhofer ISI, 2014]		
	2002]	et al., 2010]			
Collection design	Questie	onnaire	GPS-tracking		
Time of collection	2001-2002	02 2010 2		$2011-2014^a$	
Observation period	1 d	lay	21.0	days	
Number of vehicles $(\text{profiles})^b$	76,797	70,249	$522 \ (498)^c$	164	
	$(32, 171)^b$	$(24,958)^b$			
Total number of vehicle trips	$163,\!108$	330,293	$71,\!338$	$13,\!374$	

Table 3.6: Overview of commercial driving profiles for Germany

a: data collection ongoing; b: profiles with movement; c: profiles with at least seven days of observation

KiD was collected twice, in 2002 and 2010. The design of the study was similar to MiD, although no household records but only those of company vehicles were collected in KiD. A questionnaire was distributed for about one year and vehicle trips were reported

³⁴This section which is based on [Gnann et al., 2015a].

 $^{^{35}}$ For a description of their collection and preparation refer to Section A.3.

on one observation day. KiD2002 comprises the information of 76,797 passenger cars of which 32,171 were in motion on the observation day. In total, the data set comprises 163,108 vehicle trips. In the 2010 edition of KiD 330,293 vehicle trips of 25,958 vehicles in motion were collected. However, the observation period in KiD is only one day per vehicle. Since the time horizon of the used data collection has a significant influence on the upscale to VKT as well as on the technical feasibility and potential electric driving share, a single day data base might result in a strong bias (see e.g. [Gnann et al., 2012a, Plötz et al., 2014a] and Section 3.1).

For this reason, a data collection of conventional vehicle profiles with a time horizon of about three weeks was performed by the author in the on-going project "REM2030" [Fraunhofer ISI, 2014]. The REM2030 data was collected with GPS-trackers over 21 days on average and currently contains 522 vehicle driving profiles of which 498 have an observation period more than six days and are analyzed in the following.³⁶ The 373 passenger cars and 125 light commercial vehicles (LCVs) in REM2030 with an observation period of at least one week perform about 53,000 or 19,000 trips respectively. This yields a daily average of 6.7 trips and 73 km per day for passenger cars and 6.8 daily trips and 67 km on average for LCVs which is in line with [WVI et al., 2010]. Regarding the distinction of solely commercially used (fleet vehicles) and partly privately used commercial vehicles (company cars), the REM2030 data mainly contains company fleet vehicles which are used for commercial purposes only.³⁷

Since the collection design does not allow to distinguish vehicle trip purposes, the average beeline to the company site is calculated to determine a trip to the company site. With this calculation a stopping point is assumed to be commercial when the distance of the beeline is lower than 500 meters and public otherwise. This follows the distinction of possible charging facilities in Section 2.1.2 and allows to determine commercial and public locations in which potential PEVs could be charged. Furthermore, since an annual mileage is not part of the small questionnaire distributed with the GPS-trackers, it is calculated by the mean of the daily distance means times 365^{38} .

REM2030S is a subsample of profiles and comprises 164 profiles whose company sites are in the region of Stuttgart. This subsample is extracted to maintain a data set with the same spatial focus as MOPS. MOPS and REM2030S are used for the simultaneous simulation of users at charging locations. Their representativity is discussed in the following, while their impact on results will be discussed in Section 5.3.5.

3.3.2 Comparison of commercial vehicle usage data sets

Like for private vehicles the arguments of a long observation period also hold for commercial driving profiles. Thus, the REM2030 and REM2030S profiles will be used in the further analyses, while KiD2002 and KiD2010 are exploited for comparison of driving behavior, vehicle sizes and affiliation to commercial branches.

³⁶For a description of their collection and preparation refer to Section A.3.

³⁷Although the vehicle usage is only available for data collected as of 1st of January 2014, it shows a majority of fleet vehicles. Prior to to 2014, companies were requested to put the GPS-trackers in fleet vehicles.

 $^{{}^{38}}VKT^a = 365 \cdot 1/7 \cdot 1/K_i \sum_{i=mon}^{sun} \sum_{k=1}^{K_i} VKT_{ik}$ with *i* as weekday, VKT_{ik} as the vehicle kilometers traveled on the *k*th weekday of *i* and K_i as frequency of this weekday in the driving profile.



Figure 3.4: Comparison of daily VKT in different commercial driving profile data sets. Data from [IVS et al., 2002, WVI et al., 2010, Fraunhofer ISI, 2014]

Figure 3.4 shows the cumulative distribution functions of daily VKT for these commercial driving profiles. Like for MiD2002 and MiD2008 in the left panel of Figure 3.3 for private users, the daily VKT for KiD2002 (green dash-dotted) and KiD2010 (black dotted) are the reported driving distances on the day of observation. Hence, the number of daily VKT corresponds to the number of vehicles with movement on the day of observation. For this figure only passenger cars and LCVs are considered.³⁹ Like MOP for private vehicles, the REM2030 data is displayed in two ways: the average daily driving as explained earlier in this section (solid blue) is displayed as well as the single driving days (red dashed). For an average number of 21.0 days with observation in REM2030, an average number of 15.7 days with driving can be determined which could result from five working days per week for most commercial branches. There are three points to retain for discussion in the following analyses: (1) The distributions for KiD2002 and KiD2010 are almost equal to each other. (2) The distribution of the single days of observation of REM2030 is also very close to the distributions of KiD2002 and KiD2010. (3) However, the distribution of calculated average daily distances for REM2030 shows less long distances (above 100 km) than the other distributions. An interpretation may be that the low frequency of long-distances per vehicle is compensated by a high frequency of shorter trips within a profile. For the following analyses the similarity of single daily VKT in REM2030 to KiD2002 and KiD2010 suggests that the REM2030 profiles are an adequate choice for commercial driving behavior.

Further, the differences in driving between vehicle sizes in KiD2002, KiD2010 and REM2030 are evaluated. Table 3.7 shows the mean daily VKT, its standard deviation and number of driving profiles for the four sizes classes: small (cubic capacity (CC)<1,400 ccm), medium (1,400 ccm \leq CC<2,000 ccm), large (2,000 \leq CC) and LCVs (weight below 3.5 tons).

³⁹The SQL code for this query runs as follows: SELECT * FROM pkw WHERE v01 IN (2,3).

Vehicle size	attribute	KiD2002	KiD2010	REM2030
Small	mean [km]	28.27	31.34	42.00
	stdev [km]	63.87	62.06	29.01
	number of profiles	4,551	5,076	113
Medium	mean [km]	53.22	60.73	82.32
	stdev [km]	112.31	123.59	63.24
	number of profiles	$14,\!570$	$14,\!427$	198
Large	mean [km]	54.62	60.73	104.05
	stdev [km]	123.85	117.70	94.89
	number of profiles	9,958	8,716	56
LCV	mean [km]	33.67	42.33	66.68
	stdev [km]	84.21	98.81	64.37
	number of profiles	40,851	$25,\!573$	131

Table 3.7: Average daily driving of commercial driving profiles for Germany differentiated by vehicle sizes. Data from [IVS et al., 2002, WVI et al., 2010, Fraunhofer ISI, 2014].

While the daily VKT is always larger in REM2030, insignificant differences of daily VKT are found among almost all subsamples.⁴⁰ Moreover, the overestimation of driving behavior could result from the different calculations of average daily VKT (as explained earlier in this section) as well as from the collection design. Companies that took part in the data collection of REM2030 were allowed to choose the vehicles whose driving was recorded. While standard deviations are almost twice the mean values for KiD2002 and KiD2010, mean and standard deviation are about equal for REM2030 within a subsample. Thus, the sample values are broader distributed for KiD than for REM2030 (which also stems from the inclusion of several vehicles that are not moved during the day of observation in KiD). For the further proceeding of this work, the tendency of REM2030 to overestimate driving will be kept in mind.

Another important distinction of driving profiles in the commercial passenger car sector is the commercial branch [Ketelaer et al., 2014]. The commercial branches are clustered according to [Eurostat, 2008]. Since the REM2030 data collection is intended to be representative for commercial registrations regarding commercial branches, this premise is tested in the following. In Table 3.8 the number of annual vehicle registrations [KBA, 2014b], the average daily VKT, its standard deviation and the number of driving profiles within a commercial branch for data sets REM2030 and KiD2010 are presented. Further, the p-value of a two-sided t-test for mean values with unequal sample sizes and unequal variances is shown in the last column.

First, it has to be mentioned that about 90% of all commercially licensed vehicles are newly registered in four groups: G (Wholesale and Trade), C (Manufacturing), N (Administrative services), and S (Other services).⁴¹ About 89% of vehicles within section G (Wholesale and Trade) are licensed to companies that work within vehicle trade, with

⁴⁰A two-sided t-test with unequal sample size and unequal variances rejects the null hypothesis at p=7.1% for large vehicles and p=1.7% for small vehicles when KiD2002 and KiD2010 are compared and p<1% for all other comparisons.

 $^{^{41}}$ When comparing registrations with vehicle stock, it is found that the holding periods in sectors G (Wholesale and Trade), C (Manufacturing) and N (Administrative services) are much lower than average (1.1 to 1.7 years compared to 3.8 years).

m 11 c	0	$\mathbf{D} \cdot \cdot$	1	1	· 1 1	1	• 1	1 1
Table -	5.8:	Driving	distances	distingi	ushed	bv	commercial	branches
10010 0			0100011000			~,	commence eren	or our or or

NACE section		K	iD2010	b	R	EM2030) ^c	
	\mathbf{s}^{a}						-	
	tion			КT		<u> </u>	КТ	
	tra		TK_	N		TK_	\sim	
	Sis.	ze		aily	ze	N N	aily	
	l re	e si	ail	q	Si	ail	q	e^d
	iua	plqı	р.	lev.	plqı	р.	lev.	alu
	ann	san	avg	$\operatorname{st.d}$	san	avg	$\operatorname{st.d}$	b-v-d
G - Wholesale and trade	699,506	3,068	45.20	106.10	45	64.19	34.83	0.00
C - Manufacturing	380,367	2,559	70.11	142.13	101	78.67	45.66	0.11
N - Administrative and	$357,\!835$	958	56.09	108.51	43	78.61	66.09	0.04
support service activities								
S - Other service activities	265,926	$8,\!457$	60.06	129.09	51	62.09	43.81	0.75
Q - Human health and social	33,391	1,064	42.39	79.86	67	51.74	31.52	0.04
work activities								
F - Construction	$31,\!150$	2,559	59.00	121.25	38	47.13	29.23	0.03
O - Public administration,	28,546	1,910	41.46	93.67	66	29.05	18.15	0.00
defense, social security		,						
H - Transportation and	27,269	890	52.31	103.38	45	202.84	96.92	0.00
storage								
K - Financial and insurance	$18,\!582$	238	59.61	98.83	5	48.83	25.53	0.44
activities								
J - Information and	$16,\!271$	146	47.41	92.45	10	65.56	37.51	0.22
communication								
M - Professional, scientific,	12,065	10	95.46	82.55	4	60.20	20.94	0.24
technical activities								
D - Electricity, gas, steam,	$7,\!452$	376	52.52	99.66	16	32.47	21.63	0.01
air conditioning								
I - Accommodation and food	$5,\!495$	226	43.16	110.73	0	0.00	n.n.	n.n.
service								
L - Real estate activities	4,419	1	0.00	n.n.	0	0.00	n.n.	n.n.
E - Water, sewery, waste,	$3,\!938$	419	59.24	119.85	7	37.81	38.89	0.21
remediation								
R - Arts, entertainment and	$3,\!541$	119	36.67	79.55	0	0.00	n.n.	n.n.
recreation								
A - Agriculture, forestry,	2,963	543	39.96	98.57	0	0.00	n.n.	n.n.
shipping								
P - Education	2,134	61	60.64	93.95	0	0.00	n.n.	n.n.
U - Extraterritorial	1,418	33	75.33	232.21	0	0.00	n.n.	n.n.
organizations and bodies								
B - Mining and quarrying	$1,\!192$	100	61.60	118.62	0	0.00	n.n.	n.n.
Total	1,903,460	0 23,737	55.46	119.10	498	71.50	65.00	0.00

a: [KBA, 2014b]; b: [WVI et al., 2010]; c: [Fraunhofer ISI, 2014]; d: p-value for two-sided t-test of difference of means with unequal sample sizes and unequal variances.

vehicle parts and vehicle maintenance [KBA, 2013]. In section C (Manufacturing) another 74% of vehicles are registered to vehicle construction while car rentals sum up to 85% of vehicles in section N (Administrative services) [KBA, 2013]. Thus, in total 63% of all commercial passenger car registrations are directly related to the automotive industry. This is important to know since a large number of these vehicles might be showroom cars that are hardly driven during first registration or company cars with a high amount of driving. The fourth largest car registration group is section S (Other services) which contains about 14% of all commercial registrations. 99% of the vehicles in this group comprise membership organizations, trade unions as well as political and religious organizations [KBA, 2013]. In the further analysis special emphasis will be put on these four vehicle groups.

Furthermore, some branches within the single data sets show high daily VKT, for example the branches H (Transport), C (Manufacturing) and N (Administrative services). These average daily VKT between the different commercial branches can be compared by a t-test assuming log-normal distributed daily VKT.⁴² Statistical tests for two data sets of commercial driving show the average daily VKT in branches H (Transport), O (Public administration), A (Agriculture, forestry, shipping), and D (Energy) to differ from at least two third of the other branches within one data set.⁴³

However, for the representativity for driving in commercial branches, the average daily VKT between REM2030 and KiD2010 within each branch is tested (p-values in the last column). The null hypothesis (values differ from each other) is rejected with the probability p, thus small p-values can be interpreted as significant differences while large p-values show statistically insignificant differences. p-values smaller than 1% are found for commercial branches G (Wholesale and Trade), O (Public administration) and H (Transport). Since branch G (Wholesale and Trade) accounts for about 30% of annual registrations this does not favor a detailed analysis of driving behavior within commercial branches. However, the differences between REM2030 and KiD concerning driving behavior are insignificant for commercial branches that are responsible for two thirds of annual registrations. For the further procedure of this work, the REM2030 data can be considered representative for annual registrations in large parts, however, conclusions for single commercial branches will be drawn carefully.

For this work, the following conclusions for commercial driving behavior can be drawn: Since there was no data set with more than one day of observation for commercial vehicles, the REM2030 driving profiles have been collected for this purpose and will be used for the further analyses. A comparison of driving distances distinguished by vehicle size, by commercial branch, and without distinction shows that the REM2030 profiles tend to slightly overestimate driving distances. However, a comparison of the other available data sources does not show good accordance either. Thus, for the following, analyses concerning vehicle sizes, commercial branches, and commercial driving will be discussed in light of these findings.

⁴²The distribution of daily driving distances in the samples is right-skewed. Thus, using the logarithms of the daily VKT, the normality premise for a t-test is approximately fulfilled. This corresponds to the assumption that average daily driving distances of randomly chosen vehicles are log-normal distributed (see [Greene, 1985,Lin et al., 2012,Dong et al., 2014,Kagerbauer, 2010] for a similar discussion.

 $^{^{43}}$ A two-sided t-test for mean annual VKT with unequal variances and unequal sample sizes is significant at p < 1%.

Summary

The aim of this chapter was to present the main data sources for this thesis and to discuss their main characteristics with respect to other data sets. The main data source is vehicle driving profiles which comprise all vehicle trips performed in a certain observation period. Private and commercial driving profiles that are publicly available or were collected for this thesis were presented and their differences were discussed. The following important findings can be summed up:

- 1. A long observation period is decisive for the analysis of individual driving profiles. Plug-in electric vehicles face the problem that they have to drive a significant amount of kilometers electrically to be able to economize while they are technically restricted by their electric range. An analysis of individual driving profiles allows to identify potential early adopters with respect to their driving. However, driving profiles with short observation periods tend to overestimate electric driving and market potentials subsequently.
- 2. In Germany there are two large private data sets with one week of observation that will be used in the further analyses: the German Mobility Panel (MOP) and a data set similar to the Mobility Panel for the region Stuttgart (MOPS). When compared to other large German data sets (MiD2002 and MiD2008) daily and annual vehicle kilometers traveled as well as garage availability are only slightly different between the data sets (slightly lower driving in MOPS).
- 3. As there is no data set with more than one day of observation, a data collection of more than 500 vehicle driving profiles for commercial vehicles has been performed for this thesis, the so-called REM2030 driving profiles. Driving in this data set is slightly higher than in large data sets that serve for comparison (KiD2002 and KiD2010) whose influence will be discussed in the further analyses. Yet, the data set can be considered as representative for commercial vehicle registrations.

Chapter 4

Model development and technoeconomical parameters⁴⁴

Introduction

The aim of this chapter is to introduce a new model for the co-diffusion of plug-in electric vehicles and their charging infrastructure. In the following Section 4.1, an overview of the model, its main assumptions and an argumentation for the approach are presented. Section 4.2 comprises a mathematical description of the model. The parameters needed to model the market diffusion of PEVs and charging infrastructure for Germany until 2030 are described in Section 4.3 followed by a discussion of the model in Section 4.4.

4.1 Model overview

In Chapter 2, the literature on models for the co-diffusion of alternative fuel vehicles and their infrastructure was presented to retrieve aspects that could be learned from earlier modeling approaches. The main findings of that chapter were that: (1) There is a tipping point for the co-diffusion of AFVs and their refueling infrastructure beyond which the system becomes self-sustaining. (2) Existing models on NGVs and FCEVs can reproduce historical market evolutions, for which simulation models were the most promising approaches. And, (3) PEV specialties compared to other vehicles like the different charging duration, frequency and charging infrastructure ownership cannot be integrated in the existing approaches without major adaptations. Hence, in this chapter a new approach for the simulation of the co-diffusion of plug-in electric vehicles and their charging infrastructure is proposed.

Based on the literature analysis in Chapter 2, an agent-based simulation model is developed since simulation models best fulfill the requirements for the co-diffusion of PEVs and their charging infrastructure extracted in Section 2.2.1.⁴⁵ In agent-based models (ABM) a number of agents interact based on a set of rules over a certain time [Bonabeau, 2002] while the complexity rises with the number of agents, rules and the complexity of rules that are integrated. According to [Bonabeau, 2002] an ABM is useful "when

⁴⁴Parts of this chapter are based on Plötz, Gnann and Wietschel (2014) [Plötz et al., 2014a] and [Gnann et al., 2015b].

 $^{^{45}\}mathrm{For}$ a classification of models refer to Section 2.2.2.

the interactions between the agents are complex, nonlinear, discontinuous, or discrete. [...] When the population is heterogeneous, when each individual is (potentially) different" [Bonabeau, 2002]. However, the topology of interactions or the behavior (e.g. learning) can be complex [Bonabeau, 2002]. An ABM incorporates several issues that have to be taken into account: Since the interaction of human behavior is modeled, several soft factors have to be integrated, yet "soft factors in decision making processes are often difficult to quantify, calibrate, and sometimes justify" [Bonabeau, 2002]. Also, like every bottom-up model, ABM is data intensive [Sensfuss, 2008] and can become computation intensive as well because of the individual agent behavior [Bonabeau, 2002]. The main challenge is to simplify user behavior and reduce data within the right level of detail to still gain valuable outcomes. It is common practice to distinguish between agent-based simulation (ABS) and multi-agent simulations (MAS) [Hare and Deadman, 2004]. The first type of ABMs is used for the simulation of individuals with different characteristics and those individuals interacting with each other (also called individual-based simulation [Huston et al., 1988]). The latter group of models additionally assumes that agents learn from each other and from their surroundings while interacting Wooldridge and Jennings, 1995, Steinbach, 2015].

This is exactly the case for vehicle usage profiles as described in the previous Chapter 3. Individual vehicles differ from each other in their driving behavior and their sociodemographic characteristics (see Chapter 3) as well as their willingness to pay more for PEVs [Peters et al., 2011a]. An interaction at public charging stations is thus not predictable without the diffusion of PEVs and their individual driving behavior. They interact when several users arrive at the same time and only one vehicle is able to recharge. To model the market diffusion of PEVs and their charging infrastructure, the agent-based simulation model ALADIN (Alternative Automobiles Diffusion and Infrastructure) is developed in this thesis.

The model is structured as depicted in Figure 4.1. There are four main model steps, (1) the individual PEV simulation, (2) the individual utility calculation, (3) the aggregation in the stock model and joint PEV simulation and (4) the optimal charging infrastructure setup by the charging point operator. Within these steps there are certain parts where actual user behavior is integrated. The PEV simulation is based on driving profiles in an infrastructure scenario. Furthermore, the cost for infrastructure, the willingness to pay more and a brand loyalty of each individual user are incorporated into the utility calculation. While the first two model steps are performed individually for every vehicle driving profile, the stock model aggregates the preceding results to a market diffusion. A joint simulation of the vehicle stock allows the charging point operator to determine the public charging price and the infrastructure setup for the subsequent individual simulation.

This model includes social interaction, but the agents are not able to learn from each other. Individual agents are used since buying decisions for passenger cars are complex [Klöckner, 2014] and many factors play a role, both in private [Mueller and de Haan, 2009, de Haan et al., 2009] and commercial vehicle purchase decisions [Globisch and Dütschke, 2013, Sierzchula, 2014]. Based on a survey of private passenger car buyers [Peters and de Haan, 2006], Figure 4.2 gives an overview of factors ranked first in private users' decision making processes. Size, price and safety can be identified as the most important factors in the purchase decision.

The importance of the different vehicle attributes motivates to model the PEV pur-



Figure 4.1: Overview of the model ALADIN - Alternative Automobiles Diffusion and Infrastructure. Based on individual driving data from private, commercial and company cars (left panel) and using techno-economical parameters (bottom), the market shares of different propulsion technologies are determined in four steps (central panel): (1) each driving profile is simulated as PEV and conventional vehicle; (2) based on the vehicle TCO, the cost for individual charging, the limited choice of PEV makes and models and the individual willingness to pay more, the utility maximizing vehicle option is chosen; (3) the vehicle choices are aggregated to a PEV stock and jointly simulated at public charging points; (4) the charging point operator decides about the public charging price and construction based on the amount of public charging from the previous model step.

chase decision as maximization of utility among several vehicle alternatives. For the future market diffusion of PEVs the model determines the users' utility obtained from vehicle size, price, brand, fuel consumption and fuel type, and to a certain extend engine power, emissions and acceleration. Since the focus is on the vehicles' propulsion technology, safety, gear shift and four-wheel drive are disregarded. The potential utility of each technology is calculated for each user (or agent) individually with these factors. Furthermore, three user groups are distinguished that differ in their purchase decisions: (1) private car buyers, (2) commercial vehicles used in commercial vehicle fleets only, and (3) company cars used by employees for both commercial and private purposes. For Germany each group amounts to about one third of the annual passenger car registrations [Pfahl, 2013]. Furthermore, four vehicle sizes are distinguished: small, medium, large and LCVs. To take the importance of vehicle size in the vehicles' utility into account, PEVs are only considered for the same vehicle size as the conventional ones in the profiles, i.e. there is no switch between vehicle sizes in the buying decision. That is, every user is assumed to buy a vehicle of the same size as his current vehicle. Purchase price and fuel consumption of a vehicle are aggregated to the vehicle's TCO. The consumption costs strongly depend on the annual VKT and the individual driving pattern, in particular the regularity of



Figure 4.2: Criteria ranked first in the decision making process for passenger car purchase differentiated by vehicle size. Figure is based on data from [Peters and de Haan, 2006].

driving. For a reliable estimate, each user's driving profile is simulated as a vehicle with each of the propulsion systems (BEV, PHEV, diesel and gasoline) and the resulting fuel costs are calculated.

Fuel type, emission standards and acceleration are different for conventional internal combustion engine vehicles or plug-in electric vehicles. Furthermore, many consumers are willing to pay a price premium for a new technology [Rogers, 1962] in general and for PEVs in particular [Wietschel et al., 2012]. The positive factors of PEVs such as reduced noise, dynamic driving experience, their novelty and innovativeness are integrated in the model proposed here as willingness to pay more of some users [Peters et al., 2011b, Dataforce, 2011]. Other factors are difficult to model and are assumed to be comparable between conventional and plug-in electric vehicles, such as design, safety and engine power. Factors like design and safety can be quantified in relation to other attributes with conjoint analyses (see e.g. [Kreyenberg et al., 2013]), however, this is not the focus of this thesis and quantified data on these matters is not available to the author.

Apart from the positive image of plug-in electric vehicles as a new technology, there are some obstacles to overcome. One important factor is the need for frequent recharging caused by the limited electric range of PEVs [Tate et al., 2008, Kalhammer et al., 2007]. To address this issue, the cost for individual charging options is integrated into the buying decision while the cost for shared charging points are allocated to their users through a public charging price that includes the cost for public charging stations. In addition, the choice of PEVs in terms of brands and models as offered by manufacturers is still limited and likely to remain so for the next years. This will certainly restrain some users from buying a PEV despite their potential benefits. This effect of a limited choice of brands is integrated into the model by a two-step process: First, users are assumed to stick to their current vehicle brand if possible. Second, if a PEV would maximize the user's individual utility but is not available from his current manufacturer, then a share of users (equal to the share of brands offering PEVs in that year) is assumed to choose a PEV from another manufacturer and the rest of the users are assumed to choose their second best vehicle option.

One of the most important aspects of ALADIN is the usage of real-world driving profiles. This is a major improvement over existing models and has to the author's knowledge not been used comprehensively in a co-diffusion model for PEVs and their charging infrastructure so far. As explained in the previous chapter, the distribution and regularity of trip lengths varies strongly between different users and influences the TCO and potential use of PEVs significantly. Consequently, driving profiles of at least one week are analyzed (cf. Section 3.1). Based on the individual driving profile, each vehicle profile is simulated as BEV and PHEV based on the existing charging infrastructure. The resulting electric driving share and annual VKT are used to calculate the individual TCO of each driving profile and vehicle option. Based on the individual TCO and the additional positive and negative factors integrated in the model as user specific utility, the utility maximizing propulsion technology for each driving profile is chosen. Thus, in each user group, a share of driving profiles will correspond to PEVs. This share is then extrapolated to the annual registrations of vehicles in this user group. The first model outputs are the individual utility of each vehicle technology and the individual purchase decision in a given year. The technological and economical parameters vary over time and the decision process is repeated for each year. The annual registrations are built up to a stock of PEVs via a stock model.

To determine the need for public charging spots, the PEV agents in stock are simulated to calculate the amount of energy charged at public charging points. This is different to the first simulation where vehicle buyer agents are simulated individually for registrations in comparison to the PEV stock simulation for the usage of charging points. Based on the public charging cost, their batteries' state of charge and the availability of a free charging spot, PEV agents decide to charge in a joint simulation. Thereafter, the charging point operator (agent) decides about a new public charging price that covers price for electricity and charging points and the optimal (de-) construction of public charging points. The market diffusion of charging points on the other hand is based on economical assumptions and the return on invest since charging stations have to become profitable soon after they are built (see argument C in Section 2.2.1). In the following simulation run, these charging points can be used in the individual simulation of vehicle driving profiles to obtain higher electric driving shares and thus a higher PEV utility. Since model steps 2, 3 and 4 are based on earlier steps that can be performed independently the results section is divided into three sections that address the model steps individually. The mathematical details of the model will be described in the following section.

4.2 Mathematical description of the model

The four model steps of ALADIN are explained in the following subsections: the individual PEV simulation (4.2.1), the determination of the individual utility (4.2.2), the stock model and joint simulation of PEVs (4.2.3) and setup of charging points by the charging point operator (4.2.4).⁴⁶

In the following, let i be a vehicle driving profile of user group u (private, fleet, company vehicle) and vehicle size r (small, medium, large, LCV), s the propulsion technology regarded (Gasoline, Diesel, PHEV, BEV) and t the year of observation to name the main indices. All costs are VAT exempted for commercially licensed vehicles and include VAT for privately owned cars. All units of variables are shown in the abbreviations at the beginning of this thesis.

⁴⁶The description of the first three subsections is based on [Plötz et al., 2014a, Gnann et al., 2015b].



Figure 4.3: Example for battery simulation. Assumed is an electric consumption of 0.18 kWh/km and overnight charging with 3.7 kW. Distances in blue on left abscissa, battery SOC in green on right abscissa.

4.2.1 Individual plug-in electric vehicle simulation

In the PEV simulation, the battery's state of charge (SOC) of BEVs and PHEVs is simulated for each driving profile to determine whether it could be performed by a BEV or which electric driving share would result for a PHEV of the same vehicle size. More specifically, the SOC is calculated for each point in time τ as

$$\operatorname{SOC}_{i}(\tau + \Delta\tau, t) = \begin{cases} \operatorname{SOC}(\tau, t) - d(\Delta\tau) \cdot c_{r,s}^{e}(t) & d(\Delta\tau) > 0\\ \min\{\operatorname{SOC}(\tau, t) + \Delta\tau \cdot P_{l}(\tau, t), C_{r,s}(t)\} & d(\Delta\tau) = 0. \end{cases}$$
(4.1)

where the initial value for each year t is given by $\text{SOC}_i(\tau_i^0, t) = C_{r,s}(t)$. $C_{r,s}(t)$ is the net capacity of the battery analyzed, calculated as the gross capacity multiplied by its maximum depth of discharge (DoD), and τ_i^0 is the starting time of the driving profile. $\text{SOC}_i(\tau, t)$ denotes the state of charge at time τ in year t. The distance driven between τ and $\tau + \Delta \tau$ is given by $d(\Delta \tau)$. $c_{r,s}^e(t)$ is the consumption of electric power in kWh/km, depending on car size r and propulsion technology s. $P_l(\tau, t)$ in kW describes the power for charging at the location where car i was parked at τ and year t. If no charging infrastructure is available, $P_l(\tau, t) = 0$. The locations l of $P_l(\tau, t)$ are private, work or public grounds for charging facilities, while for public charging $P_p(\tau, t) = (P_{p,z_{min}}(t), ..., P_{p,z_{max}}(t))^T$ and $P_{p,z}(t)$ signifies the power for public charging in zone z at time t. Figure 4.2.1 holds an example of the PEV simulation.

For public charging there are additional conditions integrated: Since public charging is always considered less convenient and more expensive than charging at home or work, a BEV is only recharged when (1) its battery capacity is below 50% (to return home) and (2) there is a predefined minimum number of charging points available within the area where the vehicle is parking. The minimum number of charging points will be determined in Section 5.3.1. Furthermore, (3) for PHEV also the cost for electric driving has to be lower than for conventional driving since the vehicle could drive with conventional fuel otherwise.

With this simulation, it is possible to determine the VKT with positive SOC divided by the distance of all VKT for each profile, i. e. the electric driving share

$$s_{i,s}(t) = \frac{d_{i,s}^{el}(t)}{d_i}$$
 (4.2)

Here, $d_{i,s}^{el}(t)$ is the VKT in the driving profile that is driven in electric mode by vehicle i with propulsion technology s and $d_{i,s}$ is the total VKT in the profile. Note, that not the number of trips with positive SOC, but the distances are considered for the electric driving share. When analyzing a BEV, the electric driving share $s_i(t)$ must be 100% to fulfill the whole profile and to be considered in the utility analysis. For a PHEV, the electric driving share is an important measure to determine its variable cost, since cost for electric and conventional driving is significantly different. For both PEVs, regular driving favors a high electric driving share $s_{i,s}(t)$ while a significant amount of (electric) kilometers has to be performed because of the PEV's economics (see also Chapter 3).

4.2.2 Determination of utility

Based on this first model step, the most beneficial vehicle type from the four propulsion technologies s (Gasoline, Diesel, PHEV and BEV) for every user i is determined by:

$$u_{i,s}^{a}(t) = -\operatorname{TCO}_{i,s}^{a,\operatorname{veh}}(t) - \operatorname{TCO}_{i,s}^{a,\operatorname{CI}}(t) + \operatorname{WTPM}_{i,s}^{a}(t)$$

$$(4.3)$$

That is, the utility function consists of the TCO of the vehicle $\text{TCO}_{i,s}^{a,\text{veh}}(t)$, the TCO of the individual charging infrastructure $\text{TCO}_{i,s}^{a,\text{CI}}(t)$, and the willingness to pay more (WTPM) WTPM_{i,s}^a(t); the latter being added to the first two terms that are subtracted and all terms are discounted to an annual value. In this function monetary and non-monetary factors are combined in a utility function measured in EUR/yr. The inclusion of the vehicle's TCO assumes that users weigh the purchasing costs as important as the operating costs, which is a common approach for PEV market diffusion models (see [Plötz et al., 2014a, sec.4]). Utility is calculated for each vehicle type. Note that, some terms can also be zero, e.g. the charging infrastructure cost or the WTPM if conventional vehicles are considered.

The vehicle's annual TCO are calculated as

$$\operatorname{TCO}_{i,s}^{a,\operatorname{veh}}(t) = \underbrace{a_{i,s}^{\operatorname{veh,capex}}(t) + a_{i,s}^{\operatorname{veh,opex}}(t) - dep_{i,s}^{\operatorname{veh,capex}}(t) - dep_{i,s}^{\operatorname{veh,opex}}(t)}_{(1)} + \underbrace{g_{i,s}^{\operatorname{veh}}(t)}_{(2)} \quad (4.4)$$

They consist of (annual) capital expenditures $a_{i,s}^{\text{veh},\text{capex}}(t)$ and operating expenditures $a_{i,s}^{\text{veh},\text{opex}}(t)$ that may be reduced for commercial vehicles by depreciation allowances for capital $dep_{i,s}^{\text{veh},\text{capex}}(t)$ and operating expenditures $dep_{i,s}^{\text{veh},\text{opex}}(t)$. For company cars the tax that has to be paid by the driver for using a company car (a so-called fringe benefit) is added $g_{i,s}^{\text{veh}}(t)$. These additional costs are added to consider the company's (1) as well as the user's vehicle buying decision (2) which is in line with other approaches [Pfahl, 2013].

The discounted cash-flow method with resale values⁴⁷ is used to calculate the investment annuity for user i and propulsion technology s as

$$a_{i,s}^{\text{veh,capex}}(t) = \left(\text{LP}_{r,s}(t) \cdot (1 + z_u(t))^{T_u^{\text{veh}}(t)} - \text{SP}_{i,s}(t) \right) \cdot \frac{z_u(t)}{(1 + z_u(t))^{T_u^{\text{veh}}(t)} - 1}$$
(4.5)

Here, the list price $LP_{r,s}(t)$ (for vehicle and battery)⁴⁸ is multiplied by the annuity factor consisting of the interest rate $z_u(t)$ and the investment horizon $T_u^{\text{veh}}(t)$. $SP_{i,s}(t)$ denotes

⁴⁷For an introduction to accounting, see [Wöhe and Döring, 2002].

 $^{{}^{48}\}mathrm{LP}_{r,s}^{PEV}(t) = p_{r,s}^{car}(t) + \kappa_{r,s}(t) \cdot p_s^{batt}(t)).$

the sale price of vehicle *i* for resale after $T_u^{\text{veh}}(t)$ years and depends on the vehicle's annual VKT, its age and its list price. The resale value is calculated for each vehicle *i* with the individual annual VKT_i. To determine $\text{SP}_{i,s}(t)$, results of [Dexheimer, 2003, Linz et al., 2003] are used with $\text{SP}_{i,s}(t) = \exp \left[\alpha_1 + 12 \cdot \beta_1 T_u^{\text{veh}}(t) + \beta_2 \text{VKT}_i/12\right] \cdot \text{LP}_{r,s}(t)^{\beta_3}$ where the parameters $\alpha_1 = 0.97948$, $\beta_1 = -1.437 \cdot 10^{-2}$, $\beta_2 = -1.17 \cdot 10^{-4}$ and $\beta_3 = 0.91569$ have been obtained by regression (see [Dexheimer, 2003, Linz et al., 2003] for details).⁴⁹ $T_u^{\text{veh}}(t)$ depends on the user group *u* to reflect the different average holding times for private and commercial users.

The operating expenditure of vehicle i for one of the propulsion technologies (s) is calculated as

$$a_{i,s}^{\text{veh,opex}}(t) = \text{VKT}_i \cdot \left(C_i^e(t) + (1 - s_i(t)) \cdot c_{r,s}^c(t) \cdot k_{r,s}^c(t) + k_{r,s}^{\text{OM}}(t) \right) + k_{r,s}^{\text{tax}}(t)$$
(4.6)

The individual annual VKT_i are multiplied by the costs for driving in electric mode $C_i^e(t)$ plus the costs for driving in conventional mode and the costs for operations and maintenance $k_{r,s}^{\text{OM}}(t)$. The costs for conventional driving consists of the share of conventional driving $(1 - s_i)$, the conventional consumption $c_{r,s}^c(t)$ and the costs for conventional fuel $k_{r,s}^c(t)$. The annual vehicle taxes $k_{r,s}^{\text{tax}}(t)$ are independent of VKT_i.

The cost for electric driving is the sum of cost for energy charged at different charging locations:

$$C_i^e(t) = \sum_{l \in L} p_{i,l}(t) \cdot W_{i,l}(t) = \sum_{l \in L} p_{i,l}(t) \cdot \sum_{\tau = \tau_i^0}^{\tau_i^{max} - 1} W_{i,l}(\tau, \tau + 1, t)$$
(4.7)

Since four types of accessibility $L = \{\text{domestic, commercial, work, public}\}$ are considered, the costs for electric driving consist of the energy charged $W_{i,l}(t)$ at home multiplied by the domestic electricity price $p_{i,l}(t)$, the energy charged at work times the price for charging at work and the public charging price multiplied by the energy charged publicly for private and company vehicles. For fleet vehicles only the commercially and publicly charged energy is multiplied with their respective prices.

For commercially licensed vehicles, there are depreciation allowances which reduce the vehicle costs [BMF, 2001]. The value of capital assets in companies may be reduced by

$$dep_{i,s}^{\text{veh,capex}}(t) = LP_{r,s}(t) \cdot \frac{(1 + z_u(t))^{T^{dlim}} \cdot z_u(t)}{(1 + z_u(t))^{T^{dlim}} - 1} \cdot DR$$
(4.8)

Thus, the vehicle's list price $LP_{r,s}(t)$ can be depreciated linearly over a certain time horizon T^{dlim} , which is currently six years for vehicles in Germany [BMF, 2001]. Since this reduces the taxes for profits a company has to pay, the depreciation rate DR is similar to a company tax rate. Also for the operating expenditures, a company can reduce its profits and reinvest the resulting lower tax payments: $dep_{i,s}^{\text{veh,opex}}(t) = a_{i,s}^{\text{veh,opex}}(t) \cdot DR$.

⁴⁹Please note that the regression results for the PEV resale values imply a higher absolute resale price SP but lower relative or percentage resale value RV \equiv SP/LP. If the vehicle's age and annual VKT are assumed as fixed at average values, the sales price is given as SP = $c \cdot LP^{\beta_3}$. Thus the relative resale value will be given as RV \equiv SP/LP = $c \cdot LP^{\beta_3-1}$ and accordingly RV^{PEV}/RV^{ICE} = $(LP^{ICE}/LP^{PEV})^{1-\beta_3} < 1$ since $LP^{ICE} < LP^{PEV}$ and $0 < \beta_3 < 1$.

The last term in equation 4.4 is the tax for company cars that users have to pay [BMF, 2015]:

$$g_{i,s}^{\text{veh}}(t) = (LP_{r,s}^G(t) - TE_{r,s}^{PEV}(t)) \cdot (0.01 + 0.0003 \cdot ACD_i(t)) \cdot ITR_i(t) \cdot 12$$
(4.9)

Starting point for this tax is the gross list price of the vehicle $LP_{r,s}^G(t)$ which is not VAT-exempted like all other costs for commercial vehicles. This price is multiplied by 1% (so-called 1%-rule) plus 0.03% times the average commuting distance $ACD_i(t)$ to receive the monthly taxable value for the vehicle. By multiplying the income tax rate of the owner of vehicle *i*, $ITR_i(t)$, and multiplying it by twelve, the annual costs for the user of a company car can be calculated. Company PEVs profit from a correction factor of the gross list price $TE_{r,s}^{PEV}(t)$ of up to 10,000 EUR, since company car drivers do not profit from the lower operating cost when they own a fuel card [BMF, 2015]. The term for company cars is added to the annual vehicle TCO in equation 4.4.

The second term in equation 4.3 is the cost for individual charging points and is zero for conventional vehicles since the cost for refueling stations of conventional vehicles is included in the fuel prices. This cost is integrated as plug-in electric vehicles will be charged at individual domestic, commercial or work charging spots - there is a one to one allocation of charging spot per vehicle. The cost for the individual charging points is calculated as their annual capital expenditures and their operating cost $a_{i,s}^{\text{CI,opex}}(t)$.

$$TCO_{i,s}^{a,CI}(t) = I^{CI}(t) \cdot \frac{(1+z_u(t))^{T_u^{CI}(t)} \cdot z_u(t)}{(1+z_u(t))^{T_u^{CI}(t)} - 1} + a_{i,s}^{CI,opex}(t)$$
(4.10)

By adding the infrastructure cost to the TCO calculation, the fact that users must have at least one charging point to charge their vehicle regularly is addressed. Private users and company car owners that own a garage are assumed to install a simple charging spot without large financial effort while users without garages need a more expensive solution [Gnann et al., 2013]. The garage availability is part of the socio-demographic information in the driving profiles.

The third term of the utility function (4.3) is the WTPM for a PEV discounted to one year which is simply calculated as

$$WTPM_{i,s}^{a} = wtpm_{i,s}(t) \cdot LP_{r,s}(t) \cdot \frac{(1 + z_u(t))^{T_u^{veh}(t)} \cdot z_u(t)}{(1 + z_u(t))^{T_u^{veh}(t)} - 1}$$
(4.11)

The WTPM is based on a percentage wtpm_{*i*,*s*}(t) that users are willing to pay more for a plug-in electric vehicle compared to a conventional one and will further be described in Section 4.3.4. This percentage is multiplied by the investment of a comparable conventional vehicle ($LP_{r,s}(t)$, powered with gasoline for small and medium, diesel for large vehicles and LCV). There is no WTPM for conventional cars.

In summary, the best vehicle option for each vehicle buyer agent is determined based on the individual utility which is determined by vehicle cost as well as infrastructure cost for individual charging points and WTPM for PEVs. Costs for commercial vehicles contain some specialties concerning taxes and they are VAT exempted in all calculations. Results for individual analyses of driving profiles are shown for the commercial passenger car sector in Section 5.1. The individual best vehicle choices are aggregated in the stock model. Figure 4.7 shows an example of different vehicle utilities.

4.2.3 Stock model and joint simulation

Stock model

The PEV simulation and utility calculation above are performed for every driving profile. Three different user groups (private, commercial fleet, company car) and four vehicle sizes (small, medium, large and LCV) are distinguished, where LCV are almost exclusively purchased by commercial fleets and accordingly neglected for the other user groups. Thus $m = 3 \cdot 3 + 1 = 10$ vehicle groups are considered in the stock model. To derive the share of driving profiles of individuals $f_{m,s}(t)$ that are assumed to buy a PEV according to their individual utility $f_{m,s} = \frac{\sum_i \{f_{i,m,s} | s \in \{PHEV, BEV\}\}}{\sum_i f_{i,m,s}}$ is multiplied by the number of vehicles in the corresponding user group and vehicle size $n_m(t)$. The annual PEV registrations are calculated as:

$$N_{m,s}(t) = f_{m,s}(t) \cdot n_m(t).$$
(4.12)

However, the vehicles that were purchased in a given year do not remain in stock forever. Instead vehicles will be scrapped with an age-dependent probability $P^{\text{scrap}}(a)$ and commercial vehicles diffuse into the private vehicle stock after the first registration period $(T_u^{\text{veh}}(t); \text{ second-hand car market})$. The survival probability $L(a) = 1 - \int_0^a P^{\text{scrap}}(a') da'$ for a vehicle to survive until age a. With this distribution at hand, the stock $S_{m,s}(t)$ of PEVs ($s \in \{PHEV, BEV\}$) of vehicle group m in year t can be written as the sum of PEVs purchased in earlier years $N_{m,s}(t')$ that survived until year t:

$$S_{m,s}(t) = \sum_{t'=t_0}^{t} N_{m,s}(t')L(t-t').$$
(4.13)

The survival probability has been obtained from the official German vehicle statistics (see [Plötz et al., 2012] for details). A lifetime distribution for the vehicles to remain in stock is needed for the stock model introduced above. Data for the complete German vehicle fleet is used and the age dependent scrapping probability over ten years. These probabilities are calculated from the age structure of the German vehicle stock since 2001 by computing the change between adjacent ages in subsequent years for all years available. The Weibull distribution for the survivor function is given by $L(t) = e^{-(t/\theta)^{\beta}}$, where the parameters $\theta = 14.7$ for scale and $\beta = 3.5$ for shape have been obtained from a least square fit (see [Bain and Englehardt, 1991, Lawless, 1982] for the justification of the Weibull distribution as a survivor function). These parameters imply an average age for scrapping of 13.8 years and an average age of the vehicles in stock of 7.3 years, both in good agreement with other studies of the German passenger car stock [Plötz et al., 2013]. This distribution will be used for the stock model of the German vehicle fleet.

Since PEVs are in an early market phase, the choice of models and brands is and will remain limited for the next years. This fact slows down the market diffusion of PEVs since brand and design are vehicle purchase criteria (cf. Figure 4.2). The limited choice of brands and models is included in the PEV market diffusion model in two steps: In a first step, the present and near-future choice of PEVs is collected (from press announcements), with announcements for up to two years in the future being available. Based on this data and a number of relevant brands (the 20 most sold brands in the German vehicle market) within each vehicle segment for normalization, a logistic regression of the upcoming brands



Figure 4.4: Algorithm for the inclusion of the limited availability.

is performed. The resulting logistic availability function is then extrapolated into the future (see Section 4.3.5 for details). The availability is integrated into the purchase decision as follows: If a PEV is utility optimal for a vehicle profile of brand b_i and this brand has announced a vehicle for the year under consideration (or earlier) the PEV will be bought by that user. If the user's brand b_i does not offer a PEV, then some of the users choose a PEV of a different brand $(lim_{m,s}(t) \text{ according to the logistic availability function})$ and the rest chooses the second best TCO option (see Figure 4.4 for graphical explanation of the algorithm). With the inclusion of the limited availability, one driving profile may be split into shares for each propulsion technology $f_{i,m,s}(t)$ with $f_{i,m,s}(t) \in [0,1]$ and $\sum_s f_{i,m,s}(t) = 1$, whereas before $f_{i,m,s}(t) \in \{0,1\}$. The inclusion of a limited availability (incl. brand loyalty) is only possible if the vehicle brand is available, which is only the case for MOP. Since REM2030 does not contain information about the vehicle's brand, the test for vehicle brand is skipped for REM2030.

To this point, the first three model steps form a market diffusion model for PEVs in which charging is only possible at domestic, commercial and work charging spots or at public charging points without public charging costs. Results based on these assumptions will be presented in Section 5.2. The inclusion of a public charging infrastructure diffusion is described in the following section.

Joint simulation

For an economic operation of public charging infrastructure, a sufficient occupancy rate by PEVs in the vehicle stock is decisive. Hence, for the simulation of the vehicle stock, not only the number but also the driving profiles of PEVs within stock are needed. Further, the occupancy of public charging spots is analyzed in a joint simulation of PEVs that interact when arriving at a charging point. Thus, the arrival of two or more vehicles at a charging point is to be simulated with an analysis of spatial driving behavior as well. This analysis is possible with the geographical information within the driving profiles MOPS



Figure 4.5: Example for PEV stock simulation when multiple users arrive at a charging point. Three users (A...C) arrive at two charging points (X_1, X_2) in zones 2 and 5.

and REM2030S.

The charging behavior of the PEV stock determines the total electricity consumed at public charging points. Here, the same charging rules as in the individual simulation (equation 4.1) apply except for the charging point density at public charging points, which is replaced by a real availability of charging points: a user may charge his PEV only if a charging point is not in use at his arrival. While in the individual simulation, every user performs a simple forecast of his driving behavior and estimates his charging shares based on his usual routes and his impression of charging stations available to him, in the simulation of the PEV stock the usage of individual charging points is simulated. Figure 4.5 shows an example where user A can recharge his battery at charging station X_1 , but users B and C both arrive at charging point X_2 and only the user arriving first may recharge. Whenever a BEV arrives at a charging point which is not in use and the BEV's SOC is below 50%, the vehicle is recharged. The same holds for PHEVs, where in addition electric driving with the current public charging price has to be cheaper than conventional driving. The outputs of this model step are the amount of vehicles and their energy consumption distinguished by accessibility types for each year. The total amount of public charging $W_{pc}(t) = \sum_{i} W_{i,l}(t)$ and l =public is the main input for the consecutive model step.

4.2.4 Charging point operator

Based on the energy consumption at all public charging spots $W_{pc}(t)$, the number of public charging points and the price for public charging in the next period is determined in the fourth model step. Equation 4.14 shows the relationship between prices, charging points and public energy consumed:

$$p_{pc}(t) := p_{el}(t) + p_{cp}(t) = p_{el}(t) + \frac{n_{cp}(t) \cdot a_{cp}(t)}{W_{pc}(t-1)}.$$
(4.14)

The public charging price $p_{pc}(t)$ consists of a price for electricity $p_{el}(t)$ and a price for charging points $p_{cp}(t)$. The number of charging points $n_{cp}(t)$ multiplied by their annual cost $a_{cp}(t)^{50}$ and divided by the total energy consumed at public charging points $W_{cp}(t)$. While the energy consumed is derived within the PEV stock simulation, the price for electricity and the annual cost for charging infrastructure are exogenously defined.

⁵⁰As in equation 4.10 the annual cost for public charging points is defined by its discounted investment and operating expenditure $a_{cp} = I^{CP}(t) \cdot \left((1 + z_u(t))^{T_u^{CI}(t)} \cdot z_u(t) \right) / \left((1 + z_u(t))^{T_u^{CI}(t)} - 1 \right) + a_{i,s}^{CP,opex}(t).$

Since the consumption of energy charged at public charging points changes with an increasing number of PEVs, the charging point operator will build new charging points based on the current public charging price $p_{pc}(t)$ and electricity price $p_{el}(t)$, but based on the new cost for public charging points $a_{cp}(t+1)$:

$$n_{cp}(t+1) := \frac{p_{pc}(t) - p_{el}(t)}{a_{cp}(t+1)} \cdot W_{pc}(t).$$
(4.15)

Note that increasing prices for electricity p_{el} or charging points p_{cp} may also lead to a decreasing number of public charging points, i.e. a shut down of several public charging points.

With the number of charging points, the electricity price and the charging point costs in the next period as well as the energy consumed at public charging stations, the public charging price for the next period $p_{pc}(t)$ is calculated with formula 4.14 and the simulation can start at the first step again.⁵¹

The mechanism to determine the zones in which charging points should be built, is shown in Figure 4.6. First, the number of charging points in the next and the current period determine the construction of charging points $\Delta n_{cp} = n_{cp}(t+1) - n_{cp}(t)$. If this delta is negative (there are fewer charging points in the following period), the charging stations with the lowest usage are put out of service. That is, the zone z^* with the minimal use per charging point (CP) is determined with

$$min_z use'_z(t+1) = use_z(t)/n_{CP,z}(t+1)$$
(4.16)

and one charging point is taken out of service in zone $z^{*,52}$ This procedure is repeated until Δn_{cp} iterations are completed and Δn_{cp} charging points are taken out of service (see lower bound of Figure 4.6).

The construction of charging points in case of a positive delta is performed in two phases: At first, charging infrastructure is built in areas with a high *vehicle* occupancy until the minimum number of public charging points per zone is reached. Here, the zone z^{**} for a construction is determined by $max_z occ'_z = occ_z \cdot (1 - (n_{CP,z}(t+1)/CPN_z))$. Thereafter, the charging infrastructure is built in places z^{***} with a high PEV occupancy to cover a higher need for public charging points $(max_z occ'_{z,PEV}(t+1) = occ_{z,PEV}(t)/n_{CP,z}(t+1))$. This two-step approach assures a minimum coverage at the beginning moving to a useroriented approach after a minimal coverage is given [Funke et al., 2015].

To sum up, in the last part of the model the charging point operator decides on the public charging price as well as the (de-)construction of public charging points. Based on this simulation potential PEV users may buy a PEV based on the increased utility through new charging points. Thus, the new vehicle registrations are dependent on the charging point stock which completes the joint simulation of PEV and charging infrastructure simulation. Results for all four model steps will be shown in Section 5.3.

⁵¹Note again, that in formulas 4.14 and 4.15, the price for electricity $p_{el}(t)$ and the cost for public charging points $a_{cp}(t)$ are externally defined, while the amount of public charging $W_{pc}(t)$ is a simulation result. Both the price for electricity and the annuity of public charging points may include a contribution margin.

⁵²The number of charging points per zone are initialized by $\forall z : n_{CP,z}(t+1) := n_{CP,z}(t)$.



Figure 4.6: Algorithm for the construction and deconstruction of public charging points.

4.3 Techno-economical parameters

The calculations require a variety of parameters for modeling the interaction of PEVs and their charging infrastructure. The most important framework parameters are varied and combined in scenarios in the following Section 4.3.1. Thereafter, technical and economical parameters for vehicles and the vehicle market (4.3.3), the assumptions for the WTPM (4.3.4) and the limited availability (4.3.5) as well as several adaptations for the data sets with geographical information, MOPS and REM2030S (4.3.6) are presented. All costs are given in EUR₂₀₁₄ and real values for the future.

4.3.1 Scenario definition

In this subsection scenarios for the future development of several important framework conditions as well as for charging infrastructure are defined for Germany until 2030. Scenarios are used for the most important parameters since their future development is uncertain and their influence is large. Scenarios allow to evaluate outcomes based on these developments, yet they are no prognoses and no probability for their realization is given. However, the determination of consistent scenarios permits to show the range of potential results [Dieckhoff et al., 2014].

The market diffusion of plug-in electric vehicles is influenced by both the framework conditions in general and the parameters depending on the vehicles in particular. The framework conditions include the number of new car purchases divided into segments and user groups forming the general potential for electric cars. Other parameters like the oil price and the electricity price are almost independent from an early PEV diffusion which has not reached a mass market level. Vehicle dependent parameters such as purchase price or fuel consumption form the basis for the utility calculation for each segment and user group.

For illustrative purposes, Figure 4.7 shows a potential utility composition of parame-



Figure 4.7: Composition of vehicle costs for different drive trains in 2020. Assumed are 15,000 kilometers traveled per year, 60% electric driving share for PHEV and garage ownership. Resale value exempted from vehicle and battery cost. All cost parameters for private user in medium scenario with VAT in EUR₂₀₁₄.

ters for all four drive trains in 2020. The vehicle is assumed to travel 15,000 kilometers per year with an electric driving share of 60% for a PHEV and the vehicle owner is assumed to hold a garage. Further a gasoline price of 1.65 EUR/l, a diesel price of 1.58 EUR/l, an electricity price of 0.29 EUR/kWh and a battery price of 330 EUR/kWh are assumed.⁵³ A WTPM of 15% for PEVs is shown as price increase for conventional vehicles instead of decrease for plug-in electric vehicles for an easier reading.

The example in Figure 4.7 shows that some factors have a greater influence on vehicle utility than others. While vehicle taxes play a minor role, the cost for operations and maintenance for all drive trains are very close to each other. The largest differences between vehicle types can be found in capital costs (for vehicle and battery), costs for energy consumption and the WTPM which is in line with other studies [Hacker et al., 2011b, Mock, 2010, ESMT, 2011]. For this reason, battery and energy prices are varied in different scenarios. The influence of the WTPM is analyzed separately in Section 5.2.2.

For battery prices, as well as electricity and fuel prices, three scenarios are defined, which are summarized in Table 4.1. The first scenario makes rather optimistic assumptions with regard to the market success of plug-in electric vehicles (pro-EV scenario), the second more pessimistic assumptions (contra-EV scenario), and the assumptions made in the third scenario for Germany up to 2030 lie in-between these two (medium scenario). The battery prices for all three scenarios decrease exponentially from values up to 550 EUR/kWh in 2015 (pro-EV, medium, contra-EV) to below one third in 2030 (all values without VAT) [Pfahl, 2013, Plötz et al., 2013]. The prices for batteries used for the simulations were discussed with several experts of the German automotive industry [Plötz et al., 2013]. Battery costs are not distinguished between PEV types, although several

⁵³Energy prices are taken from the later described medium scenario and resale values for vehicles and batteries are exempted from vehicle cost. All other values are described in the following Section 4.3.

studies suggest this due to different battery use patterns (see e.g. [Nelson et al., 2009, Santini et al., 2010]). Instead, a differentiation via depths of discharge was used. Long-term price estimates are at the upper end of estimates (see e.g. [Rousseau et al., 2012, Nykvist and Nilsson, 2015]), yet they present prices to the consumer, not at cell or pack-level. Prices for diesel and gasoline are equal for all scenarios in 2015 based on [IEA, 2013]. The development of fuel prices until 2030 is based on the New Policy Scenario in [IEA, 2013] for the medium scenario with an additional increase up to 2.00 EUR/l (including VAT) for gasoline in the pro-EV scenario and a decrease to 1.50 EUR/l in the contra-EV scenario in 2030. This corresponds to an oil price of 183 \$/bbl in 2030 in the pro-EV scenario, 149 \$/bbl in the medium scenario and 115 \$/bbl in the contra-EV scenario for a constant mineral oil tax. These bandwidths are chosen since the future development of fuel prices is unclear. A relatively constant oil price until 2030 in the contra-EV scenario, an increasing price based on international experts in the medium scenario and a further increasing oil price based on the developments in the last years in the pro-EV scenario demonstrate the bandwidth of options.

Parameter	year	Pro-EV	medium	Contra-EV
Diesel price ^{a}	2015		1.45	
[EUR/kWh]	2030	1.90	1.65	1.42
Gasoline price^{a}	2015		1.52	
[EUR/kWh]	2030	2.00	1.75	1.50
Oil price ^{a}	2015		118	
[EUR/bbl]	2030	183	149	115
Electricity price $private^b$	2015		0.29	
[EUR/kWh]	2030	0.27	0.32	0.35
Electricity price commercial ^{b}	2015		0.21	
[EUR/kWh]	2030	0.20	0.22	0.25
Battery price^{c}	2015	450	500	550
[EUR/kWh]	2030	235	266	295

Table 4.1: Scenario-specific parameters used in ALADIN. All prices with VAT in EUR₂₀₁₄.

a: [IEA, 2013] and own assumptions; b: own assumptions based on [McKinsey, 2012, BCG, 2013]; c: [Pfahl, 2013]

In Germany, several studies predict a further increase of electricity prices in the future [Schlesinger et al., 2011, McKinsey, 2012, BCG, 2013]. For this thesis the following is assumed: The average wholesale price for electricity rises only slightly until 2020 and remains stable until 2030 due to the increase of renewable energies (in line with [BDEW, 2014, Capros et al., 2013]). Further, the investments for grid expansion slightly raise the electricity price (0.005 EUR/kWh in 2030). The revision of the renewable energy law of 2014 performs well and leads to price decreases of 0.04 EUR/kWh in 2030 in the pro-EV scenario, to 0.02 EUR/kWh in the medium scenario and to no change in the contra-EV scenario. Vehicle-to-grid operation allows to reduce prices by about 0.005 EUR/kWh between 2020 and 2030 in the pro-EV scenario [Dallinger and Wietschel, 2012]. This results in electricity prices as shown in Table 4.1.

These scenarios combine favorable and unfavorable conditions for plug-in electric vehicles which are consistent with other studies. These parameters are chosen to be integrated into scenarios as they contribute strongly to the individual vehicle utility (see Figure 4.7).

4.3.2 Cost for charging infrastructure

As mentioned in Chapter 3, three user groups from two data sets are distinguished. In the PEV simulation, it is assumed that private and company cars can charge with 3.7 kW whenever they are at home. The trip purpose "home trip" is used to decide about the parking spot of the vehicle [MOP, 2010]. For fleet vehicles, the trip purposes are unknown but the GPS-location allow to determine the distance from the company location [Fraunhofer ISI, 2014]. Thus fleet vehicles charge with 3.7 kW during the day when they are not further than 500 m away from their main company location.⁵⁴ In addition, they can charge overnight, assuming that the vehicle can be plugged in, no matter if it is parked at a private household or at the company site. In the socio-demographic information of [MOP, 2010], the common overnight parking spot of private and company cars is available, so vehicles with and without garage can be differentiated. Users of vehicles that are parked in a garage are assumed to buy a wallbox for charging, while non-garage-owners have to pay for a simple on-street charging facility (similar to [Plötz et al., 2013]). For the latter, the cheapest charging facility available is chosen – a charging point integrated into a lantern – and the investment and running cost are split up between two users, assuming they could share one charging point [Plötz et al., 2013]. With these assumptions for private car holders without a garage an upper limit for their PEV ownership is obtained. Investment and running cost for both solutions as well as investment horizons are given in Table 4.2.

Table 4.2 :	Cost for	charging	infrastructure	options.	All prices	without	VAT in	$1 EUR_{2014}$.
---------------	----------	----------	----------------	----------	------------	---------	--------	------------------

Parameter	unit	2015	2020	2025	2030
invest wallbox ^{<i>a</i>}	EUR	404	370	343	323
operating cost wallbox ^{a}	$\mathrm{EUR/yr}$	0	0	0	0
invest domestic on-street charging $point^b$	EUR	657	537	444	372
operating cost domestic on-street charging $point^b$	$\mathrm{EUR/yr}$	287	246	211	182
invest public charging point $(3.7 \text{ kW})^c$	EUR	$1,\!314$	1,074	888	744
operating cost public charging point $(3.7 \text{ kW})^c$	EUR/yr	574	292	422	364
invest public charging point $(22 \text{ kW})^c$	EUR	$5,\!281$	4,694	4,107	3,521
operating cost public charging point $(22 \text{ kW})^c$	$\mathrm{EUR/yr}$	795	712	628	544

a: 3.7 kW, domestic users with garage, commercial and work charging facilities; b: 3.7 kW, domestic users without garage; c: in public places. All cost assumptions based on [Kley, 2011, Plötz et al., 2013].

The one-to-one allocation of charging points is also assumed for private and company car users at work: in scenarios where charging at work is allowed, every vehicle owner pays for his individual work charging point. As the installation is assumed to be simple, the cost for wallboxes in Table 4.2 is considered. For public charging points, two slow-charging solutions are considered by the charging point operator: a 3.7 kW lantern charging option with one charging point and a 22 kW charging station with two charging points. While the first solution is a low-cost option, the second is an average charging point that is currently most common in Germany [Lemnet, 2014]. For all charging options an investment horizon of 15 years is assumed [Kley, 2011].

 $^{^{54}\}mathrm{A}$ distinction of different distances with respect to company size shows no influences on scenario results.
4.3.3 Vehicle market and vehicle cost parameters

New vehicle registrations in Germany have been almost constant with 3.1 million vehicles over the past seven years [KBA, 2012a,KBA, 2012b]. Segment shares within the new registrations that did not change significantly [KBA, 2012a,KBA, 2012b]. Approximately 30% of all new purchased cars in Germany are company cars [NPE, 2011b,NPE, 2011a,Pfahl, 2013]. Combined with [NPE, 2012], the number of new purchased cars per car size and user group can be obtained (see Table 4.3). For the simulations, the number and segmentation of vehicle sales are assumed constant until 2030. Here and in the following, the simplified segmentation (see Chapter 3), compared to statistics of the German Federal Motor Authority (KBA) [KBA, 2012a,KBA, 2012b], predefined by the driving profiles is used. Table 4.3 also shows general cost parameters like the interest rate of 5%, which is considered for private and commercial users for vehicles and charging infrastructure. The investment horizons are based on [Pfahl, 2013] and equal to the first vehicle registration period.

Vehicle registrations	unit	value 2015 – 2030
vehicle registrations private small^a	-	$475,\!309$
vehicle registrations private medium ^{a}	-	$694,\!275$
vehicle registrations private $large^a$	-	$143,\!309$
vehicle registrations fleet small ^{a}	-	233,240
vehicle registrations fleet $medium^a$	-	$454,\!998$
vehicle registrations fleet $large^{a}$	-	46,339
vehicle registrations fleet LCV^b	-	204,000
vehicle registrations company small ^{a}	-	106,996
vehicle registrations company medium ^a	-	497,681
vehicle registrations company large ^{a}	-	$244,\!563$
General cost parameters	unit	value 2015 – 2030
interest rate $private^c$	-	0.05
interest rate commercial ^{c}	-	0.05
investment horizon private vehicles $(T_1)^c$	years	6.2
investment horizon commercial vehicles $(T_1)^c$	years	3.8

Table 4.3: Vehicle registrations and general cost parameters.

a: [KBA, 2012a], b: [KBA, 2012b], c: [Pfahl, 2013]

The cost differences between electric and conventional vehicles are significantly driven by the differences in purchase prices of the varying technologies. The purchase price of plug-in electric vehicles consists of two parts: a relatively constant price for the chassis and drive train and a price for the battery system. All other electric components, like the electric motor or power electronics are highly developed vehicle parts which are not assumed to profit from further economies of scale. The battery system which accounts for the majority of investments in a PEV is assumed to further reduce in price [Nykvist and Nilsson, 2015]. The net purchase prices (without battery) are taken from [NPE, 2011a] and extrapolated until 2030. For conventional vehicles, prices are increased until 2030 based on [Mock et al., 2013] to reflect the additional cost for the required efficiency gains to achieve EU fleet targets for CO_2 emissions [EC, 2009]. The combinations of drive trains and segments missing in [NPE, 2011b, NPE, 2011a] are calculated with the existing ratios of gasoline/diesel technology to alternative technology [NPE, 2011a]. This leads, for instance, to slightly higher chassis prices (medium size) of BEVs with 18,000 EUR compared to 17,500 EUR of gasoline vehicles in 2020 (all values in Tables B.1-B.4).

The total purchase price of PEVs is determined by the battery size and price of the battery. In combination with the maximum depth of discharge, limits the electric range of the PEV. Battery sizes result from of a combination of several studies (cf. [Hacker et al., 2011b, Gnann et al., 2012a, Linssen et al., 2012, Pfahl, 2013]), yet values are slightly different compared to [Plötz et al., 2014a, Gnann et al., 2015b, Plötz et al., 2013]: (1) The maximum depths of discharge (=percentage of usable battery capacity) is assumed to be 90% for BEVs and 80% for PHEVs. (2) Battery sizes for medium sized BEVs increase from 24 kWh in 2015 to 40 kWh in 2020, because of economies of scale paired with the desire for more electric range, and remain stable thereafter. (3) Batteries for large BEVs start at 55 kWh in 2015 and increase up to 80 kWh in 2020 and afterwards (see Tables B.1-B.4).

All values on fuel consumption are based on [Helms et al., 2011]. The main assumption for future development of fuel consumptions is an efficiency gain (diesel, gasoline) of at least 1.5% per year to meet the EU emission targets [EC, 2009].⁵⁵ Compared to past efficiency developments [Mock et al., 2013], these assumptions seem moderate. Consumption values were adjusted in case of higher weight due to larger batteries. Values are average annual fuel consumptions and do not reflect extreme values in winter or summer.⁵⁶ Note that the values represent real consumptions and not driving cycle values. As the model calculates the TCO depending on individual driving behavior with different shares of electric driving for PHEVs, the illustrated conventional values in Tables B.1-B.4 represent a purely conventional operation after having fully depleted the battery, i. e. PHEVs are simulated in charge-depleting mode.

Maintenance costs also differ among technologies and are currently unknown for PEVs. The simulation of failure probabilities for each drive train components, performed in [Propfe et al., 2012b], leads to specific maintenance costs for large vehicles. Small deviations in battery size of BEV and PHEV between the model and [Propfe et al., 2012b] lead to minor adaptations. These maintenance costs also incorporate battery degradation due to cycling [Linden and Reddy, 2002]. Values for other size classes (gasoline and diesel) rely on [Frühauf, 2012, ADAC, 2013] and are transferred to the other technologies based on [Propfe et al., 2012b].

Vehicle taxes are calculated based on the current German tax legislation with a tax exemption for BEV owners [BMF, 2014]. PEVs are tax exempted for ten years if they are registered prior to the 1st of January 2016 and for five if they are registered thereafter although this is not legally fixed at the moment. A further exemption from tax after 2021 is considered. All technical and economical parameters for vehicles are shown in Tables B.1-B.4. For company car taxes only the gross list price $(LP_{r,s}^G(t))$ is changing over time according to Tables B.1-B.4 and the tax exemption $(TE_{r,s}^{PEV}(t))$ according to [BMF, 2015]. The modeling considers a reduction of the gross list price of 500 EUR multiplied by its battery capacity, but not more than 10,000 EUR in 2013. Thus, the gross list price for

⁵⁵This implies a slight hybridization of conventional drive trains.

⁵⁶For an analysis on deviations of consumption due to auxiliaries for heating and cooling see [Gnann, 2010], due to variations in driving in winter or summer see [Michaelis et al., 2013a], and due to variations due to driving aggression see [Funke and Plötz, 2014, Gnann et al., 2015a].

a BEV with 18 kWh battery capacity was reduced by $18 \cdot 500 \text{ EUR} = 9,000 \text{ EUR}$ in 2013 for the calculation. The deduction per kWh and the maximum deduction are reduced by 50 EUR/kWh and 1000 EUR per year, so that a PEV bought in 2020 with 18 kWh can reduce its gross list price by $18 \cdot 150 \text{EUR} = 2,700 \text{ EUR}$ which is still lower than the maximum of 3,000 EUR. For the other term in formula 4.2.2, fixed values are used: The average commuting distance $(ACD_f(t))$ is assumed to be 15 km and the income tax rate $(ITR_f(t))$ is equal to the maximum income tax rate in Germany (42%) since company cars are most often available to persons with high income.

4.3.4 Willingness to pay more for plug-in electric vehicles

An important aspect of a PEV's utility are its positive non-monetary characteristics. PEVs are perceived as new and innovative, as silent and environmentally friendly [Dütschke et al., 2011b, Peters et al., 2011b]. These positive aspects of PEVs are modeled with a will-ingness to pay more of some users, the magnitude of which depends on the users position in the adoption process [Rogers, 1962, Laroche et al., 2001]. Of course, a stated willingness to pay is not equal to the actual willingness to pay in a buying decision [Huang et al., 1997, Bradley and Daly, 1991]. However, the stated WTPM gives an indication for the appreciation of a new technology and an approximation of the actual WTPM. Using a WTPM is a common approach in market diffusion models for plug-in electric vehicles [Mock, 2010, Eppstein et al., 2011].

To assess a private user's position in the adoption process of PEVs and their individual WTPM, two empirical data sets are combined (see Peters and Dütschke, 2014, Wietschel et al., 2012, Peters et al., 2011a], cf. [GFK, 2012, Heupel et al., 2010, Knie, 1999]). The purpose of the original survey was to identify and characterize the different adopter groups in the adoption of PEVs according to Rogers' "Diffusion of Innovations" and they are ideal to assess the adopter groups' WTPM. In these studies the WTPM has been determined independently for four adopter groups with a different attraction to plug-in electric vehicles: (1) users of PEVs, identified as likely innovators, (2) attracted individuals with purchase intention in the near future, identified as likely early adopters, (3) attracted individuals without purchase intention, identified as likely early and late majority, (4) uninterested individuals, identified as likely laggards (cf. Table 4.4). The four adopter groups were formed by the participants' answers concerning their current vehicle usage, the interest in PEVs, and their intention to buy a PEV in the near future (see Peters et al., 2011b, Dütschke et al., 2011a, Plötz et al., 2014b] for details). For an individual analysis of vehicles and users, these survey results have to be combined with the driving profiles. Each driving profile is thereby assigned to one of the four adopter groups with their WTPM.

Members of the four adopter groups differ significantly in socio-economic variables like household income, employment status, household size, city size and the willingness to accept a higher price for a PEV [Peters and Dütschke, 2014, Wietschel et al., 2012, Peters et al., 2011b, Dütschke et al., 2011a]). As the data set also contains information about age, sex and education of the user groups, it is possible to assign each driving profile to one of the four groups according to their resemblance with the other group members (see [Plötz et al., 2014a, sec. 3.2] for a validation of this assignment). The participants stated an individual WTPM for PEVs. Here, the adopter group average WTPM is used to include the positive aspects of PEVs mentioned earlier. The percentage WTPM is Table 4.4: Definition of private adopter groups according to [Peters et al., 2011b] and their willingness to pay more. Participants in survey answered the indicated questions and were considered members of the four indicated adopter groups. A small number of respondents answered the questions as no, no, yes and have been excluded from further analysis. The numerical values for the WTPM are median values of the group members' answers.

Group definition					tributes
group	PEV user?	PEV interest?	purchase	share of	willingness-
label			intention?	$users^a$	to-pay-more ^{b}
innovators	yes	-	-	0.5%	30%
early adopters	no	yes	yes	1.5%	15%
majority	no	yes	no	48%	10%
laggards	no	no	no	50%	1%

a: [Wesche, 2013, Dütschke et al., 2013]; b: [Peters and Dütschke, 2014],

[Wietschel et al., 2012, Peters et al., 2011a]

converted to absolute monetary values by using the conventional reference vehicle for the vehicle size (Gasoline for small and medium sized vehicle, Diesel for large and LCVs). For the individual user, the positive aspects are finally included in the utility calculation by subtracting the absolute WTPM from the vehicle list price (eq. (4.3)). The specific values are summarized in Table 4.4.⁵⁷

Although the described data set contains about 1,000 respondents, it is not representative for the group sizes in Germany [Peters and Dütschke, 2014, Wietschel et al., 2012], users of PEVs and other PEV friendly groups are clearly overrepresented. It is still useful for the validity of the average WTPM in the groups. To correct the non-representative group sizes, a second survey representative for private German car buyers is used [Wesche, 2013, Dütschke et al., 2013]. The groups are defined in the same way, i.e. according to PEV ownership, interest in PEVs and purchase intention. Since the latter survey is representative, it is used to determine the relative size of the adopter groups. The resulting share of each adopter group is summarized in Table 4.4.

To assign each driving profile of MOP to one of the adopter groups with their WTPM, the following algorithm was used. For each driving profile, first the agreement in sociodemographic characteristics with each survey respondent was calculated. Matches were collected from seven variables: sex, age, employment status, education, household size, household income and city size (all variables were categorical). That is, a driving profile could achieve up to seven matches with each of the survey respondents from a known adopter group. The number of matches $m_{ijk} \leq 7$ of user *i* with adopter group member $j = 1, \ldots, L_k$ (out of the $k = 1, \ldots, 4$ groups) were collected and normalized $M_{ik} =$ $\sum_j m_{ijk}/(7L_k)$. The driving profile *i* should then be assigned to group *k* where the overlap was the largest $M_{ik} > M_{il} \forall l \neq k$. However, since the relative group size should be limited

⁵⁷The data on WTPM used here contains only positive values for WTPM. However, studies on willingness to pay for PEV range show that the limited range of PEVs a major drawback for potential users [Dimitropoulos et al., 2011]. Apart from methodological problems of many PEV range willingness to pay studies (the survey participants have not experienced PEVs) that are met in the WTPM data used here (a noteworthy share of respondents in [Peters and Dütschke, 2014] has used PEVs), the usage of only positive WTPM is valid for the suggested model since range limitations are explicitly included via the PEV simulation and the cost for individual charging points. That is, whatever a user's potential WTPM is, he is excluded from BEV purchase in the model if his driving exceeds the BEV range.

(the number of innovators is rather small), only the top 0.5% (cf. Table 4.4), i.e. those 0.5% with the largest overlap with the survey innovators, were considered as innovators. The other potential innovators were then assigned to their second best matching adopter group. The same procedure was applied to the following groups in descending order in the innovation process: innovators, early adopters, majority and laggards (see [Plötz et al., 2013, p. 182] for computational details). As a result of this algorithm, each driving profile has a position in the adoption process according to its socio-demographic variables with an associated WTPM. The validity of this assignment is analyzed in [Plötz et al., 2014a, sec. 3.2].

To assess the WTPM of commercial vehicle fleets, results from a survey of approximately 500 German fleet managers is used [Dataforce, 2011]. About half the fleet managers stated a WTPM with an average of 10%. Again, this WTPM needs to be assigned to individual commercial vehicle driving profiles [Fraunhofer ISI, 2014]. The company size (measured as number of employees) was used as a proxy for the position in the adoption process. Since larger companies seem more likely to engage early in innovative technologies [Dataforce, 2011], commercial vehicles from companies with more than 250 employees were assigned a WTPM of 10%. About 50% of the driving profiles are from such a company in agreement with the results from [Dataforce, 2011]. However, a sensitivity analysis showed that the assignment of WTPM to other groups of commercial car owners had no strong effect on the model results. No reliable data was available for WTPM of company car buyers, thus company car buyers are assumed to have zero WTPM in the model.

Further, the WTPM was determined in 2011 and the newness of a technology diminishes over time. As there are, to the best of the author's knowledge, no publicly available studies about the decline of a WTPM over time, it is assumed that the WTPM declines linearly to 60% of its value until 2020 based on [Plötz et al., 2013] and to zero until 2030.

4.3.5 Limited plug-in electric vehicle availability

The diffusion of innovations and new technologies typically follows an S-shaped curve, well described by a logistic function [Rogers, 1962, Geroski, 2000, van der Vooren and Alkemade, 2010, Massiani, 2010, Meade and Islam, 2006]. It is assumed that the availability of PEVs from different brands can be described by a logistic function, too. That is, the share of brands per segment that offer a PEV grows logistically over time $A(t) = [1 + e^{-(t-t_0)/\eta}]^{-1}$. t_0 denotes the point in time when 50% of the brands in a given segment offer a PEV and η is the time scale of change of PEV availability. Technically, PEV announcements were collected from different brands and the cumulative number of brands per year that already offer or have announced to offer a PEV in the given year were calculated (see Plötz et al., 2013, Ch. 7.4). This cumulative number of brands has been divided by the number of brands active in that segment for normalization. For the case of Germany, all brands with non-zero new registrations in 2011 were defined as active (26 brands in the small segment, 32 in medium and 29 in large). The parameters of the logistic function to estimate future availability of PEVs were obtained by least-squares regression and assumed to be partly equal between the groups (see [Plötz et al., 2013] for details). The results for future availability of PEVs from different brands in Germany are summarized in Figure 4.3.5 and Table B.5.



Figure 4.8: Limited availability of PEVs based on press announcements for different brands (see [Plötz et al., 2013, Ch. 7.4]).

4.3.6 Adaptations for the geographical simulation

For the simulation of charging stations and the PEV stock, geographical information about the driving behavior is needed for a simultaneous simulation of PEVs at charging facilities. For this reason, the data sets MOPS and REM2030S, a subset of driving profiles of REM2030, are used. However, these profiles lack of several information that is required for the previous model steps: (1) there is no information about the car holder, i. e. a distinction of private users and company cars is not possible, (2) the socio-demographic variables are not sufficient to assign the WTPM as for MOP and (3) information about garage ownership is not provided. Since this information is important for the simulation, several steps are performed to assign this information to the MOPS driving profiles. Since this might influence results, the following chapter contains two sections without public charging where MOP and REM2030 are used for simulation (Section 5.1 and 5.2). The inclusion of public charging points in Section 5.3 necessitates driving profiles with geographic information - MOPS and REM2030S. Differences will be discussed in Section 5.3.5.

The distribution of registrations to user groups in this region is different to that of Germany in total. The federal state Baden-Wuerttemberg, of which Stuttgart is the capital city, has a slightly higher share of commercially licensed vehicles [KBA, 2014a], while total registrations per area and capita are even higher in the region of Stuttgart than in Baden-Wuerttemberg and Germany [KBA, 2014c]. Since a data set with both attributes (user groups and rural districts) is not available, the share of commercial vehicles for the region of Stuttgart is assumed to be equal to the share of commercial vehicles for Baden-Wuerttemberg [KBA, 2014a] and the share of company cars and fleet vehicles in commercial passenger car registrations is assumed to be equal based on [Pfahl, 2013]. These numbers are shown in Table 4.5.

Table 4.5 shows that the number of profiles in company cars is equal to the registrations, which is an assumption of the author with the following argumentation. The purpose of the MOPS data is to provide a data set that is representative for the vehicle stock [Hautzinger et al., 2013]. While the German vehicle stock is distributed 90% to private vehicles and 10% to commercial vehicles, it is unclear how the commercial vehicle

attribute	private	company	fleet
	vehicles	cars	vehicles
registrations in observation area ^{a}	63,772	$39,\!391$	39,391
vehicle driving profiles in MOPS	$1,\!273,\!426$	$39,\!391$	164
driving profiles used for simulation	$15,\!943$	$9,\!848$	164
driving profiles multiplier	4	4	240
registrations			
a: [KBA, 2014c, KBA, 2014a, Pfahl, 2	2013]		

Table 4.5: Vehicle registrations in region of Stuttgart and corresponding driving profiles.

stock is distributed to company cars and fleet vehicles. It is thus assumed that company cars have a shorter holding time than fleet vehicles and their share in the commercial vehicle stock is about one quarter. Hence, the share of company cars in the MOPS-profiles should be 2.5%/(90%+2.5%)=2.8% of all vehicle profiles which is almost equal to its registrations.⁵⁸ Whenever the vehicle stock is needed, all profiles of MOPS are considered.

For the determination of company cars the following procedure is performed: The company car owners of MOP are significantly different to private car owners in variables sex, occupation, household size, cars in household and driving behavior - expressed as μ and σ of an assumed individual log-normal distribution of driving days (see Section 5.1.1 and [Plötz, 2014]). Hence, these variables are used to describe the company car owners. Now, all driving profiles of MOPS are compared to these company cars of MOP and the ones with the equally weighed lowest squared differences were considered assigned company cars in MOPS. Table 4.6 shows the mean, median and quartiles of company cars in MOPS (n=39,391).

A good agreement of mean and median values for company cars in the two data sets is found for variables household size, cars in household as well as μ and σ . Company cars seem to slightly differ in medians and means for the variables occupation and sex, although this might also result from equal weights for all variables. The smaller variation in MOPS (e.g. quartiles are equal to median in household size and cars in household) arises from the comparison to average values of MOP. These differences will be discussed in Chapter 5.

The WTPM was assigned to users from MOP using the attributes household income, sex, age, education, employment status, household size, and city size, which describe user groups in [Peters and Dütschke, 2014,Wietschel et al., 2012,Peters et al., 2011a] reasonably well. However, only the employment status, sex and age are available in MOPS, thus an assignment with the above described algorithm (Section 4.3.4) is not possible. For this reason, the WTPM is randomly assigned to driving profiles and the influence of a random assignment to results is tested with MOP in Section 5.2.2.

Lastly, information about the vehicles is not available for MOPS. Yet, the geographic information is a sufficient condition for the analysis of public charging infrastructure, which is the focus of this study. Since the availability of garages depends on the city sizes or settlement structures (see [Gnann et al., 2013] for a display of garage availability and city sizes of [infas and DLR, 2002]), the settlement structures of the profiles are considered

⁵⁸Since only a share of vehicles is used for simulation due to limited computing capacities, this assumption is not decisive.

attribute	data set	25%	median	75%	mean
		quartile		quartile	
sex^a	MOP	1	1	2	1.303
	MOPS	1	2	2	1.526
occupation ^{b}	MOP	1	1	2	1.975
	MOPS	1	2	2	1.634
household size ^{c}	MOP	2	3	4	3.082
	MOPS	3	3	3	3.057
cars in household ^{d}	MOP	1	2	2	1.639
	MOPS	2	2	2	1.809
individual log-normal μ	MOP	3.437	3.909	4.300	3.856
	MOPS	3.465	3.726	3.980	3.895
individual log-normal σ	MOP	0.572	0.922	1.284	0.964
	MOPS	0.550	0.757	0.993	0.779

Table 4.6: Comparison of assignment of company cars in MOPS and MOP.

a: non-metric scale with 1=male, 2=female; b: ordinal scale with 1=fully occupied, 2= partially occupied, 3=not occupied, 4=student, 5=trainee, 6=housewife/househusband; c: number of household members; d: number of vehicles in household

for an assignment of garage availability based on [infas and DLR, 2002]: The rural district of Stuttgart has the highest settlement structure based on [infas and DLR, 2002], 45.1% of all profiles of this area get assigned to own a garage. All other rural districts in the observation area are in the second highest category of settlement structures and 56.8% of profiles from these areas are appropriated with garages. As vehicle sizes are not available in MOPS, all vehicles are considered medium sized.

The inclusion of the limited availability is slightly adjusted since both data sets do not contain vehicle brand information and because of the necessary information about the vehicle usage of PEVs in the vehicle stock: Vehicles are only registered up to the limited availability $N_{m,s} = s_{m,s}(t) \cdot n_m(t) \cdot lim_{m,s}(t)$. The driving profiles that are part of the PEV stock simulation are randomly chosen from the registrations of each group $N_{m,s}$.

Since model results should be available for Germany, results have to be scaled up from the region of Stuttgart. Many factors should be considered like the population density, income or car ownership. All these factors are reflected when the vehicle registrations are used for up-scaling as they include the earlier mentioned factors. Hence, results from the region of Stuttgart are multiplied by 20.54 to retrieve results for Germany [KBA, 2014c]. The initial number of charging stations for the region of Stuttgart is extracted from [Lemnet, 2014]. 374 charging points with 3.7 kW and 289 charging points with 22 kW can be found. For the calculations, these charging points are assumed to be of equal power levels. With these adaptions of the input data and all previous assumptions, a simulation with ALADIN is possible.

4.4 Discussion

The introduction of a new model is based on a variety of assumptions, since models are always a simplification of reality for a special purpose [Hartmann and Frigg, 2006]. While the need for a new model approach was discussed in Section 2.2.3 and the data in Chapter 3, this section holds a discussion of the modeling approach. It is divided into three parts: a validation of the model (4.4.1), a general discussion of the approach (4.4.2) and the coverage of the stylized facts of Section 2.2.1 (4.4.3).

4.4.1 Validation

Since the sales for PEVs are still limited, it is not possible to compare model results to actual PEV registrations. A model that is able to reproduce market shares of different historical propulsion technologies seems better suited to predict future market shares than a model that does not reproduce these historical market shares. However, there is data publicly available for some early PHEV-users [Voltstat, 2014]. Their electric driving shares serve for comparison as a first validation. Further, an assessment of historical diesel market shares is included as a validation for the inclusion of a TCO calculation proposed here. Also, the validity of the WTPM assignment was analyzed in [Plötz et al., 2014a, sec. 3.2]. Since a comparison of model results for market diffusion with real-world sales data is not possible, the important influence factors will be discussed in sensitivity analyses in Chapter 5.

Comparison of electric driving shares

The first validation comprises a comparison of calculated electric driving shares with real electric driving shares. For this analysis, data from MOP and REM2030 is used for simulation and data from [Voltstat, 2014]⁵⁹ for real-world electric driving shares of PHEVs. The simulation is performed for medium sized vehicles in the medium scenario in 2020 (as then sample sizes are reasonably large for simulated PHEVs) and results are shown in Figure 4.9.⁶⁰

Figure 4.9 shows the electric driving shares over the annual vehicle mileage as a scatter plot. Results are shown for users for which PHEV receive the highest utility in REM2030 (red) and MOP (blue). Further, the real-world data from [Voltstat, 2014] is displayed with green crosses. On the one hand, it can be observed that simulation results reach from very small to very large values for similar annual VKT which may result from driving that does not allow to regularly recharge the vehicle and respectively perform a high amount of VKT electrically. On the other hand, simulated and actual users obtain high electric driving shares and perform a relevant amount of annual VKT, although not more than technically possible. A window for plug-in electric vehicles can be identified, the boundaries of which are determined by cost (for the lower boundary) and technical limitations of battery and recharging facilities (for the upper boundary). The 10%, 25%,

⁵⁹Data for this section was retrieved by the authors of [Plötz et al., 2015] who kindly permitted to use it in this thesis.

 $^{^{60}\}mathrm{A}$ comparison of data from US and Germany seems reasonable since their driving behavior does not vary much [Gnann et al., 2012a].



Figure 4.9: Electric driving shares over annual VKT for medium sized vehicles of three different samples: PHEV-users in 2020 of MOP with blue crosses (n=131), PHEV-users in 2020 of REM2030 with red crosses (n=17) and real users of Chevrolet Volts from [Voltstat, 2014] with green crosses (n=1,831).

75% and 90%-quantiles of these samples as well as their median, mean and standard deviations are shown in Tables 4.7.

Firstly, it should be mentioned, that a lot of users of both samples obtain high electric driving shares (median and mean above 65% for MOP and above 75% for REM2030). Secondly, the electric driving shares for simulated vehicles with PHEV as utility maximizing option are significantly higher than in the full sample⁶¹ (above 80% for both samples), which is not surprising since a high electric driving share favors the adoption of PEVs. Thirdly, the observed electric driving shares of [Voltstat, 2014] are also high and very similar to the values obtained from the simulation for MOP and REM2030. Thus, the simulation results can reproduce the buying behavior of PHEVs in terms of electric driving share.

electric driving share	q-10%	q-25%	median	mean	q-75%	q-90%	stdev
MOP	0.29	0.44	0.67	0.66	0.91	1.00	0.27
MOP PHEV	0.67	0.74	0.82	0.81	0.90	0.98	0.13
REM2030	0.39	0.57	0.84	0.76	1.00	1.00	0.25
REM2030 PHEV	0.70	0.74	0.85	0.83	0.93	0.96	0.10
Chevrolet Volt	0.56	0.69	0.82	0.79	0.90	0.96	0.15
Annual VKT	q-10%	q-25%	median	mean	q-75%	q-90%	stdev
MOP	$5,\!683$	8,760	12,775	16,201	19,710	$29,\!930$	$13,\!288$
MOP PHEV	$16,\!615$	$18,\!980$	20,805	$21,\!673$	24,820	$28,\!229$	4,974
REM2030	$6,\!256$	8,103	$13,\!976$	18,076	$22,\!878$	$37,\!583$	$13,\!197$
REM2030 PHEV	$17,\!115$	19,569	$22,\!159$	22,814	$25,\!811$	$27,\!997$	4,694
Chevrolet Volt	8,359	$12,\!079$	16,313	$17,\!418$	$21,\!582$	$27,\!579$	8,265

Table 4.7: Comparison of electric driving shares and annual VKT⁶².

 $^{61}\mathrm{A}$ two-sided t-test with unequal sample sizes and unequal variances rejects the null hypothesis at p =2.5% for REM2030 and p <1% for MOP.

Results are similar for PHEV and their annual VKT. Also here, a good agreement of medians, means and upper sample limits is found. However, in [Voltstat, 2014] there are also vehicles with lower annual VKT than in the simulated samples. An explanation might be, that these vehicles were bought even though they are not economical for the user or the WTPM is of these users might be higher than for others. However, also vehicle taxes and fuel prices are different in the US which is the origin of this data. Still, this analysis shows that real electric driving shares of PHEV users can be reproduced by the model.

Reproduction of Diesel market shares

The reproduction of diesel market shares may serve for validation of the TCO-based part of the model approach. For this comparison, neither costs for individual charging points nor a limited PEV availability applies. To rule out an eventual willingness to pay more, the commercial fleet vehicle market is considered whose buying decisions largely base on cost [Golob et al., 1997, Globisch and Dütschke, 2013, Laroche et al., 2001]. This supports the general proposal of including TCO as one important factor in the purchase decision for passenger cars. The fact that commercial fleets are only one of the three user groups under consideration here is acknowledged. However, it is responsible for about one third of the annual registrations of passenger cars in Germany and thus an important market.

For this analysis a large sample of German commercial passenger cars is studied which was collected in 2002 [IVS et al., 2002] and has already been described in Section 3.3. This data set is used as REM2030 does not contain information about the actual fuel type of the vehicles which is available in this data set. For each vehicle in the database that has been used on the day of the survey, the lengths of all daily trips are summed up and multiplied by the average number of working days in Germany (which is 220 days per year) to obtain an estimate for the vehicles annual VKT. In this case, the latter was not part of the survey and thus had to be calculated. For each vehicle, the TCO as gasoline and diesel car is calculated and the vehicle is assigned the fuel type with lower TCO. For the validation purpose, only medium sized vehicles are studied and a purchase price of 19,560 Euro for gasoline and 21,560 Euro for the diesel vehicle is assumed. The average fuel prices in 2002 have been taken from [Plötz et al., 2012] and were 1.34 Euro/liter for gasoline and 1.26 Euro/liter for Diesel fuel. The assumptions for the fuel consumption for passenger cars in German commercial fleets of 2002 are 7.6 liters/100km for gasoline and 6 liters/100km for Diesel fuel. Furthermore, average operation and maintenance costs amounted to 0.025 Euro/km for Gasoline and 0.023 Euro/km for Diesel fuel whereas vehicle taxes were 114 Euro/a for Gasoline and 242 Euro/a for Diesel vehicles. The share of diesel vehicles in the different commercial branches estimated by the TCO model proposed here are shown in Figure 4.10 together with the actual market share as stated in the corresponding survey.⁶³

⁶³In the display of Figure 4.10 the estimates of diesel market shares in German commercial sectors are shown with confidence bands based on [Plötz et al., 2014a]. Shown are the $\alpha = 0.1, 1, 5, 10, 30 \%$ confidence bands (from light to dark blue), i.e. the "true" value should lie within the confidence band in 99.9, 99, 95, 90, 70 % of the cases where confidence bands are estimated. The width of the confidence bands increases with decreasing sample size (shown in parentheses in the abscissa). In most cases the observed market share is within or close to the range of the confidence bands. Thus, the TCO calculation seems to capture important aspects of the purchase decision. Also, the calculated confidence bands help to distinguish purely statistical uncertainty from possible systematic inaccuracies.



Figure 4.10: Diesel market shares within different commercial branches in Germany. Shown are the actual values from a large-scale survey (solid line) and the estimate from a simple TCO calculation (dashed line) together with confidence bands (in blue) from the finite sample sizes (given in parentheses). See text for details of the calculation.

Figure 4.10 shows that the estimated and actual market shares of commercial diesel passenger cars in Germany are 40–60% in the major commercial branches and 20–70% over all branches. In Figure 4.10 the commercial sectors are sorted by sample size which roughly follows the registrations of passenger cars in these segments.⁶⁴ In most cases, the estimated market shares are very close to the actual market shares with significant deviations in the sectors HJ (Transport and Telecommunications), A (Agriculture and Forestry), and K (Finance). Even if the share of diesel in a sector is well reproduced, one could still question whether the individual vehicle assignments are correct. In total, 54.2% of the individual assignments are found to be correct with a lowest success rate of 38% in branch of industry K (Finance) and the highest rate of 66% in branch B (Mining). Thus, it is concluded that the proposed model is in principle able to reproduce the market shares of diesel passenger cars of German commercial fleet vehicles. TCO is thus one important aspect of the purchase decision and accordingly part of many market diffusion models for PEVs.

4.4.2 Discussion of modeling approach

Generally the modeling approach, scenarios and techno-economical parameters as well as the adaptations for Stuttgart have to be discussed.

For the co-diffusion of PEVs and their charging infrastructure a simulation model is proposed. This is not only the most common approach for the interaction of AFVs and their refueling infrastructure, but also permits to identify niches when individual

⁶⁴The commercial branches were introduced in Section 3.3.2. In 2002, the classification was slightly different, thus several commercial branches were combined to be similar to [Eurostat, 2008].

user behavior is analyzed (see Chapter 2). The variations in driving behavior between users and days within individual profiles favors an agent-based simulation approach with individual agents and their vehicle purchase decisions that depend on their vehicle usage (see Section 5.1.1). Driving profiles for a large number of vehicles for private users are publicly available and commercial vehicle driving profiles were collected for this thesis, both with an observation period of at least one week. Yet, the vehicle purchase behavior does not only depend on cost, but also on several non-monetary factors (see Figure 4.2) which were partly monetized in an individual utility function (eq. 4.3). This is the favoring WTPM for a new technology that was explicitly collected for PEVs [Peters and Dütschke, 2014, Wietschel et al., 2012, Peters et al., 2011a] and integrated into the model as well as the obstructing factors charging infrastructure which is integrated as cost for individual charging points. The interaction between individual agents is needed when the PEV stock and their public charging point usage is simulated and a charging point operator decides about the public charging price and infrastructure stock of the following period.

ABMs incorporate several issues that have to be taken into account. Firstly, they are very data intensive [Bonabeau, 2002] like many bottom-up models (see Section 2.2.2). With the publicly available private and for this purpose collected commercial driving profiles as well as the comprehensive collection of data for the WTPM Peters and Dütschke, 2014, Wietschel et al., 2012, Peters et al., 2011a, data sets are available in sufficient amount and were discussed in detail in the previous chapter and Section 4.3.4. Secondly, the simulation of ABMs is computation intensive [Bonabeau, 2002]. Also this issue could be resolved with a parallel simulation of some 15,000 agents in the individual simulation that has to be run for 16 times between 2015 and 2030 and up to 200,000 agents in the joint simulation that could not be parallelized. A simulation of 16 years takes about 16 hours on a 32 kernel server with 384 GB RAM. Thirdly, soft factors like the WTPM often are difficult to quantify [Bonabeau, 2002]. The large and comprehensive data collections of [Peters and Dütschke, 2014, Wietschel et al., 2012, Peters et al., 2011a] allow to retrieve quantifiable and statistically reliable data for user behavior on the WTPM. For driving behavior the earlier mentioned data collections permit a detailed and statistically sound analysis of vehicle usage data with a large amount of additional information about vehicle and vehicle owners (see Chapter 3). Quite often, such data is unavailable and the collection of data and the connection of the different data sources requires real effort. However, driving data with limited observation time is available for many industrialized countries and the interest in PEVs has triggered driving data collections over long time spans [Karlsson and Kullingsjö, 2013, Smith et al., 2011]. Thus, more driving data is becoming available and can be used for modeling in the future. Fourthly, "another issue has to do with the very nature of the systems one is modeling with ABM in the social sciences: they most often involve human agents, with potentially irrational behavior, subjective choices, and complex psychology - in other words, soft factors, difficult to quantify, calibrate, and sometimes justify. Although this may constitute a major source of problems in interpreting the outcomes of simulations, it is fair to say that in most cases ABM is simply the only game in town to deal with such situations." [Bonabeau, 2002] For this reason, a new approach to model the co-diffusion of PEVs and their charging infrastructure is proposed in a detailed, user specific and empirical way. The model decision for different propulsion system is based on an individual utility calculation resulting from a PEV simulation descending in the vehicle's TCO extended by a willingness to pay more for new and environmental friendly vehicles of some vehicle buyers and the cost for individual charging infrastructure reflecting the current lack of public charging infrastructure and the corresponding range anxiety. A joint simulation of the PEV stock where vehicle agents interact permits a charging point operator to decide about public charging cost and (de-)construction based on the public charging point usage. Note, that the model is an agent-based simulation and no multi-agent system, since, e.g. neighboring effects, could not be incorporated as there is no data publicly available in the desired quality.

The distinctive features of the present model are the individual utility maximization based on a detailed analysis of many individual driving profiles as well as in the inclusion of commercial vehicles and company cars. The user specific analysis allows to cover a wide range of usage scenarios and to study specific user groups such as commercial drivers or potential early adopters [Plötz and Gnann, 2013, Plötz et al., 2014b]. Furthermore, the large number of driving profiles allows the modeler to use statistical methods to assess the statistical quality of the model results. Additionally, PEV specific purchase decision factors such as the limited electric range and the need for frequent recharging are addressed by the model proposed here.

The individual model steps also depend on assumptions that can be discussed. The driving data of private users in the model extends over one week which might not contain rare long-distance trips that seem important for PEV adoption. The robustness of the model can be tested by calculating the number of days per year with more daily VKT than the BEV range and including the cost for a substitute vehicle following the methodology of [Plötz, 2014]. This will be analyzed in Section 5.2.3.

The individual PEV simulation is probably more abstract or mathematical than the actual purchase decision of private users. Yet it covers the important aspect of the regularity of an individual users' driving behavior. Users are aware of PEVs' limited electric range and understand the general economics of low operating costs for electric driving [Dütschke et al., 2011b]. Similarly, the TCO calculation of Eq. (4.4) is rather complex but the purchase and operation costs of a vehicle are an important aspect in the purchase decision both for private [Peters and de Haan, 2006] and commercial buyers [Dataforce, 2011]. This is indicated by the average annual VKT for diesel vehicles (22,300 km) and gasoline vehicles (11,800 km) in Germany [Follmer et al., 2010] – reflecting the average fuel economy under the German conditions of both propulsion technologies. Accordingly, TCO calculations are a part of many PEV market diffusion models [ESMT, 2011, NPE, 2011b, Plötz et al., 2012, Peters et al., 2012, Mock, 2010, McKinsey, 2011]. Along the same direction, recent studies pointed out that the costs of PEVs are a major influence in the purchase decision [Götz et al., 2011, Peters and Dütschke, 2014, Wietschel et al., 2012, Knie, 1999, Gnann et al., 2015b].

Although the TCO are an important factor in the vehicle buying decision, they alone cannot explain purchase decisions of car users, neither for private nor commercial car purchases [Peters and de Haan, 2006]. Furthermore, private buyers of hybrid and conventional vehicles seem to lack knowledge necessary for a TCO-based decision [Turrentine and Kurani, 2007]. An analysis of the potential early adopters of PEVs in Germany shows that more criteria than only the vehicle's TCO are important [Peters and Dütschke, 2014, Wietschel et al., 2012, Plötz et al., 2014b]. Accordingly, the proposed model covers further important aspects of the purchase decision: The need of frequent recharging is addressed in the model by adding the cost for individual charging options to the vehicle's TCO and the WTPM of some user groups has been derived from surveys and is added to

the driving profiles based on the vehicle owner's socio-demographic characteristics. Overall, it is attempted to make the most important factors in the PEV buying decision explicit and measurable. They have been included in the model in an empirical way that allows updates or corrections when more data on WTPM or choice of models become available in the future, especially the evolution of the WTPM could be of interest. In this approach for the utility calculation, neither a potential downsizing effect is reflected since vehicles are always purchased in the same vehicle size as the driving profile, nor is car-sharing which might reduce the number of vehicles in total. A change of user behavior in the future is not considered in this approach. However, there is no reliable data for these effects that could be incorporated into the model. Also the modeling approach for company cars is slightly different to private and fleet vehicles: the purchase of the company as vehicle owner is reflected with the first four terms of equation 4.3 while the decision process of the vehicle owner is presented by the fifth factor. With this approach, the preselection of the company that provides a limited offer to the vehicle driver is reflected as well as the vehicle driver's decision when also reflecting the cost he has to pay monthly for this car. Since a company car is often also a status symbol and the purchase decision may not only depend on cost, this might be the minority of cases, since also other studies rely on the same approach [Pfahl, 2013].

The limited choice of brands and models is included according to the current share of brands offering PEVs. It is retrieved from press announcements of PEVs, yet it could have been externally defined based on historical evolutions of other technologies. However, the derivation from future vehicle announcements puts these curves on a data-based foundation that permits to determine location and scale of the assumed logistic curves based on the technology itself.

The charging point operator is designed to behave like a company that is cost-oriented and wants profits from its investments. The total energy charged at public charging stations retrieved from the PEV stock simulations is a key performance indicator that will be technically available for every company working with charging points. Basing the construction and public charging price on this figure would be a common approach for an economical decision making process (Section 2.2.3). An annual change might be discussible, yet the variations in the public charging price mainly depend on the change in charging point cost (see formula 4.10) which changes only slightly and the availability of charging stations is the main focus of this research. An inclusion of potential subsidies through a charging point operator or state will be discussed in Section 5.3. Further, the rules for charging point placements are questionable as well, yet they base on reasonable assumptions: As long as only a few PEVs are on the roads, charging points are erected in areas where a lot of vehicles are parked - also to gain attention. When a minimal geographical coverage is fulfilled, highly PEV-frequented charging stations will be extended. These two rules could also be applied at the same time in reality, yet the amounts of conventional vehicles and PEVs differ too much to compare setup needs to each other. Lastly, the deconstruction of public charging points could be scrutinized, yet the operating cost of public charging points is considerably high. For this reason, charging points are taken out of service until they are needed to reduce sunk costs.

4.4.3 Coverage of stylized facts

In Section 2.2.1 a number of stylized facts were retrieved that models treating the codiffusion of PEVs and their charging infrastructure should consider. These are are listed and discussed for the presented model approach in the following.

- (A) \checkmark An initial amount of AFV refueling or recharging infrastructure: The initial amount of (public) charging infrastructure is incorporated into the model. For private households the availability of garages is considered and for public charging points the currently available ones in the region of Stuttgart.
- (B) \checkmark AFV and AFV refueling infrastructure market shares: Market shares of PEVs are explicitly considered in the PEV stock simulation. Market shares for public slow charging points can be extracted, yet they are hardly comparable to current refueling stations.
- (C) \checkmark Profitability of refueling or charging stations: The profitability of public charging stations is a precondition of the charging point operator.
- (D) \checkmark Fuel prices for conventional and alternative fuels: Both fuel prices and prices for electricity are integrated in the utility function of each individual user. Also their ratio is considered in the decision to recharge for PHEVs when arriving at a public charging point.
- (E) \checkmark Different user groups: Different user groups are reflected with driving profiles that are individually analyzed. Also different configurations of the utility function are incorporated for private, commercial fleet and company car users. Further, different adopter groups are regarded.
- (F) (\checkmark) Decreasing user concerns with AFV use: Currently PEVs are only publicly recharged when their battery SOC is below 50%. Decreasing concerns for users could be modeled with a descending minimal SOC that users are willing to accept, yet this had to be externally defined, since networking effects are not incorporated into the model. The increasing range through efficiency gains in electric consumption could be argued as implicit consideration of this factor. Further, different minimal SOCs will be tested in a sensitivity analysis.
- (G) \checkmark Potential policy measures: Monetary policy options such as subsidies for vehicles and charging infrastructure can be incorporated into the model. These will be discussed in Section 5.3.
- (H) ✓ Refueling duration and frequencies can differ between PEV and conventional vehicles: The duration and frequency of refueling for PEVs are explicitly considered in the individual and joint PEV simulation.
- (I) \checkmark Multiple types of infrastructure have to be differentiated by accessibility: This distinction is a prerequisite of the model as well.

With eight out of nine factors, the model is considered well-fitted to simulate the co-diffusion of PEVs and their charging infrastructure.

Summary

The aim of this section was to introduce a model for the joint simulation of PEVs and their charging infrastructure. Also scenarios and further parameters were discussed. The following should be maintained:

- 1. The agent-based diffusion model ALADIN allows to determine the co-diffusion of PEVs and their charging infrastructure based on real-world vehicle usage data. An initial PEV simulation with BEV feasibility and PHEV electric driving share as outcomes followed by the utility maximizing propulsion technology based on vehicle and infrastructure TCO and WTPM permits to determine every user's best vehicle option. In a stock model that also incorporates a limitation of diffusion due to limited PEV availability, the PEV stock is determined and simulated thereafter to provide the public charging point usage. With this usage a charging point operator decides on the (de-)construction of the PEV stock and thus the electric driving cost for the consecutive individual simulations.
- 2. A validation of the model shows that electric driving shares of PHEVs when compared to real-world electric driving shares and the historic market share of diesel vehicles can be reproduced. A further validation of the model is not possible, since market shares for PEVs are still low.
- 3. Three scenarios for framework conditions (fuel, battery and electricity prices) were defined with positive conditions for PEVs (pro-EV), negative conditions (contra-EV) and conditions in between (medium). These are no extreme scenarios, but show the range of results. The availability of charging infrastructure increases in the calculations in the next chapter: While in Section 5.1 only home charging (domestic and commercial) is considered, additional charging at work is reflected in Section 5.2. In Section 5.3 also public charging is analyzed.
- 4. Several adaptations are necessary for the driving profiles with geographic information (MOPS and REM2030S): Company cars are assigned based on the similarity to the average company car owners in MOP, the WTPM is randomly assigned and garage ownership is randomly assigned after a distinction of settlement structures. For this reason, MOP and REM2030 will be used for analysis in Section 5.1 and 5.2, MOPS and REM2030S in 5.3. The influence of these random allocations will be tested in the following chapter.

Chapter 5

Model results

Introduction

The market diffusion of plug-in electric vehicles is a current and important field of research [Al-Alawi and Bradley, 2013] and a relevant topic to decrease the dependency from fossil fuels and reduce GHG emissions [Kahn Ribeiro et al., 2007, IEA, 2010]. A large number of models study driver behavior to predict a future market evolution of PEVs⁶⁵, yet they suffer from several insufficiencies: (1) In studies for Germany, the commercial vehicle sector is often neglected⁶⁶, although it accounts for more than half of the vehicle registrations. (2) Several studies analyze driving behavior of only one day⁶⁷, which could lead to inaccuracies, since driving behavior varies between users and days [Smith et al., 2011, Amjad et al., 2011, Neubauer et al., 2012]. (3) The vehicle buying decision is often modeled with vehicles' TCOs⁶⁸, although it comprises a variety of factors to integrate. (4) Charging infrastructure is often discussed as a key barrier to market diffusion of PEVs⁶⁹ and a co-diffusion of PEVs and charging infrastructure is demanded⁷⁰. However, this co-diffusion is not modeled up to now.

The model ALADIN which was introduced in the previous Chapter 4 addresses these issues explicitly: More than 500 driving profiles with an observation period of 21 days for the commercial vehicle sector have been collected for this thesis. They are analyzed individually in a PEV simulation and consecutive utility comparison of different drive trains. The vehicle buying decision is based on the TCO for a vehicle and is extended by the WTPM for PEVs as a favoring aspect and the cost for charging infrastructure as well as the limited availability of PEVs as obstructing factors. For the simulation, driving profiles with an observation period of at least seven days are considered. And, the charging behavior of PEVs at various charging points is analyzed in a joint simulation of PEVs in stock.

In the following, the model will be applied to determine the market diffusion of PEVs

⁶⁵See e.g. [Dagsvik et al., 2002, Santini and Vyas, 2005, Keles et al., 2008, Lamberson, 2008, Mock et al., 2009, Nemry and Brons, 2010, Wansart and Schnieder, 2010, Shepherd et al., 2012].

⁶⁶See e.g. [Hacker et al., 2011b, Redelbach et al., 2013, Schühle, 2014].

⁶⁷See e.g. [Dagsvik et al., 2002, Santini and Vyas, 2005, Lamberson, 2008, Mock et al., 2009, Wansart and Schnieder, 2010, Shepherd et al., 2012].

⁶⁸See e.g. [Thiel et al., 2010, McKinsey, 2011, Pfahl et al., 2013, Wu et al., 2015]

⁶⁹See e.g. [Dütschke et al., 2011b, Egbue and Long, 2012, Steinhilber et al., 2013].

⁷⁰See e.g. [Lin and Greene, 2011, Dong et al., 2014, NPE, 2012, Kalhammer et al., 2007, BCG, 2009, Ma et al., 2014, Chen et al., 2013]

78

and their charging infrastructure for Germany until 2030. The chapter is divided into three parts (see Table 5.1): First, the driving behavior of commercial and private passenger car owners and their market potentials in 2020 are analyzed (5.1). In this analysis, the first two model steps of ALADIN are used and only private or commercial charging in the medium scenario is considered to focus on the differences between private and commercial driving behavior and resulting PEV market potentials. Second, the market evolution of PEVs until 2030 is analyzed by including the stock model in Section 5.2. Here, all three scenarios for framework conditions are considered while domestic, commercial and work charging is allowed. The diffusion of PEVs without public charging infrastructure and their main influence factors shall be tested in this section to be able to put the influence of public charging infrastructure into context. Third, the interaction of PEVs and charging infrastructure diffusion is considered in Section (5.3). Here, all infrastructure options are considered in the medium scenario in Section 5.3 focusing on the influence of different charging options on PEV market diffusion. While in Section 5.1 and Section 5.2, the data sets MOP and REM2030 are used for simulation, public charging analyses necessitates geographic information available in the data sets for the region of Stuttgart - MOPS and REM2030S.⁷¹ The differences of results due to data sets, based on the adaptations made in Section 4.3.6, will be discussed in Section 5.3.5.

Options	Section 5.1	Section 5.2	Section 5.3
ALADIN modeling steps			
Individual PEV simulation	\checkmark	\checkmark	\checkmark
Individual utility maximization	\checkmark	\checkmark	\checkmark
Stock model		\checkmark	\checkmark
Charging point operator			\checkmark
Scenarios			
$\operatorname{contra-EV}$		\checkmark	
medium	\checkmark	\checkmark	\checkmark
pro-EV		\checkmark	
Availability of charging infrastru	icture		
domestic & commercial	\checkmark	\checkmark	\checkmark
work		\checkmark	\checkmark
public			\checkmark

Table 5.1: Consideration of scenarios in results.

Table 5.1 gives an overview of modeling steps, framework scenarios and charging infrastructure consideration in the following sections. Each part contains an individual discussion and summary of results while a synopsis at the end of this chapter (Section 5.4) merges results of all three section.

⁷¹See Chapter 3 for a description of the vehicle usage data sets.

5.1 Market potentials of private and commercial plug-in electric vehicles⁷²

The commercial vehicle market is important for PEVs since it comprises more than 60% of annual vehicle registrations in Germany (cf. Section 3.1 and [Gnann et al., 2015a]). When market potentials are calculated, the commercial vehicle market is often neglected (see e.g. [Redelbach et al., 2013, Schühle, 2014]). Only some studies for Germany include commercial vehicles explicitly⁷³ or as a part of their analysis⁷⁴. The driving profiles of MOP and REM2030 introduced in Chapter 3 permit to analyze the regularity of driving behavior in different user groups as well as an individual analysis of BEV feasibility and electric driving shares of PHEVs. Because of the observation period of more than one day, it is possible to compare private and commercial driving behavior on a statistically sound data basis (see Section 3.1).

In the next subsection, the private and commercial driving profiles introduced in Chapter 3 are compared to each other regarding their individual daily distance and regularity of their driving which is important for PEVs (Section 5.1.1. Thereafter, the focus is on the market potential of PEVs in the different user groups (Section 5.1.2). Market potentials are calculated by using the first two model steps (cf. Section 4.2.1 and 4.2.2). A discussion (Section 5.1.3) and summary (Section 5.1.4) complete the first part of model results. Calculations in this section are performed with techno-economical parameters for 2020 as subsamples then are large enough for statistically sound conclusions.

5.1.1 Comparison of private and commercial vehicle usage

The economics of PEVs suggest they should drive many kilometers in order to economize but at the same time not too many because of the limited range of PEVs. This issue can be resolved by driving behavior that is very regular with higher than average daily VKT. In the present section, the regularity of driving and daily VKT of German commercial passenger cars is analyzed and compared to private passenger car driving. The regularity of daily driving is measured by the standard deviation of the logarithm of daily VKT. For each vehicle, the daily VKT r_{ij} by vehicle *i* on day *j* are analyzed. The logarithm of these daily VKT $\ln(r_{ij})$ is studied since daily VKT are right-skewed and their logarithms are approximately Gaussian distributed [Plötz, 2014, Lin et al., 2012]. For each vehicle *i*, the typical scale of daily driving $\mu_i = \frac{1}{n} \sum_{j=1}^n \ln(r_{ij})$ and the variation, i. e. standard deviation, in daily driving $\sigma_i = [\frac{1}{n-1} \sum_{j=1}^n (r_{ij} - \mu_i)^2]^{1/2}$ are calculated. The latter measures the individual regularity of daily driving. The former measures the typical scale of driving since the vehicle's median daily VKT is given by $r_{\text{med},i} = \exp(\mu_i)$ and the mean daily VKT by $\bar{r}_i = \exp(\mu_i + \sigma_i^2/2)$.

As explained in the previous chapters, there are three user groups under observation: private vehicles, company cars and commercial fleet vehicles. For the first two groups, the data of MOP is used, while fleet vehicles are based on REM2030. The scales and variances

⁷²This section is based on [Gnann et al., 2015a].

 $^{^{73}\}mathrm{See}$ [Berg, 1985,
Golob et al., 1997,
Gnann et al., 2012c,
Ketelaer et al., 2014,
Hacker et al., 2015,
Gnann et al., 2015a]

⁷⁴See [Hacker et al., 2011a, Kihm and Trommer, 2014, Wietschel et al., 2014b] and [Al-Alawi and Bradley, 2013] for a recent review on market diffusion models for hybrid electric vehicles and PEVs.



Figure 5.1: Mean and variation of log-normal distribution for private driving profiles. Data from [MOP, 2010]. The color indicates the number of profiles in specific bin.



Figure 5.2: Mean and variation of log-normal distribution for company car driving profiles. Data from [MOP, 2010]. The color indicates the number of profiles in specific bin.



Figure 5.3: Mean and variation of log-normal distribution for commercial fleet driving profiles. Data from [Fraunhofer ISI, 2014]. The color indicates the number of profiles in specific bin.

of individual daily driving have been calculated for these vehicles. Figure 5.1, Figure 5.2 and Figure 5.3 show scatter plots and two-dimensional histograms of the individual scales μ_i and standard deviations σ_i for the three aforementioned data sets with more than one day of observation. The histograms indicate that commercial fleet and company vehicles show higher average daily VKT than private ones. Furthermore, both private and commercial driving behavior show a large variation of daily VKT between vehicles (the μ_i range from 1 to 6) and between the days of one individual vehicle (the σ_i range from 0 to 2.5). Note that several days of observation for each vehicle are required for the calculation of the individual variance of daily VKT.

It is now possible to compare the distances and regularity of daily driving distances between private, commercial fleet and company vehicles. The μ_i and σ_i are calculated for each vehicle from the three data sets. Table 5.2 shows the median, mean and standard deviation of the daily VKT for their vehicle usage. Commercial vehicles (fleet vehicles and company cars) show higher average daily distances since the median and mean of typical daily VKT are larger for commercial vehicles. However, commercial fleet vehicles show smaller variation in daily driving between different days compared to private vehicles and company cars. The median and mean of standard deviation of daily VKT are smaller for commercial fleet vehicles indicating a larger regularity of driving.

Table 5.2: Summary statistics for daily VKT of private and commercial vehicles. Data from [MOP, 2010, Hautzinger et al., 2013, Fraunhofer ISI, 2014]

	mean	of daily V	KT μ_i	SD o	f daily VK	T σ_i
vehicle group	$fleet^a$	$private^b$	$company^c$	$fleet^a$	$private^b$	$company^c$
data set	REM2030	MOP	MOP	REM2030	MOP	MOP
median	3.77	3.33	3.83	0.83	0.87	0.93
mean	3.79	3.32	3.81	0.91	0.90	1.00
standard deviation	0.81	0.73	0.72	0.50	0.43	0.49

a: n=498, b: n=6177, c: n=162

Statistical tests are performed to measure the significance of these differences and are presented in Table 5.3. Here, the means and medians of the average μ 's and σ 's of the three data sets are compared to each other. To compare the means, a t-test for unequal variances and unequal sample sizes for the means with the null hypothesis of equal means is performed. The medians are tested with a Wilcoxon rank-sum test for unequal variances and unequal sample sizes for the means with the null hypothesis of equal medians. The resulting p-values of these tests are shown in Table 5.3 while small values (p < 0.05) indicate significant differences.

Table 5.3: Statistical comparison of driving behavior in different user groups. Shown are the p-values for different statistical tests.

parameter	mear	mean of daily VKT μ_i			SD of daily VKT σ_i		
	private	private	fleet	private	private	fleet	
	vs.	vs.	vs.	vs.	vs.	vs.	
	fleet	company	company	fleet	company	company	
mean^a	0.000	0.000	0.811	0.829	0.021	0.055	
$median^b$	0.000	0.000	0.567	0.010	0.040	0.399	

a: t-test for unequal means and unequal sample sizes

b: Wilcoxon rank-sum test for unequal medians and unequal sample sizes

Starting with the daily distances (μ_i) , one finds highly significant results for the comparison of private and commercial fleet vehicles as well as for private vehicles and

company cars, both for the means and medians (p < 0.01). Thus, commercial vehicles drive significantly more than private ones. No significant differences can be found for commercial fleet vehicles and company cars. Turning to the right part of the table, the standard deviations (σ_i) between different days are analyzed. Here, significant differences are found between private vehicles and company cars, i.e. private vehicles drive more regularly than company cars. While the medians of private and fleet vehicles are also significantly different, their means are not (fifth column). Lastly, the regularity of driving between fleet vehicles and company cars is marginally significant with p = 0.055 while their means do not show a significant difference.

To summarize, the statistical tests confirm a significantly higher driving for fleet vehicles and company cars compared to private vehicles. Further, commercial fleet vehicles have the most regular driving, directly followed by private cars. Company cars have the most irregular driving of the three user groups. For the potential of PEVs in these user groups, fleet vehicles seem to best fit requirements - long distances and high regularity followed by private vehicles and company cars.

5.1.2 PEV market potentials in different user groups

Having tested the regularity of driving in the different user groups, the profiles are now simulated as PEVs with the first model step (see Section 4.2.1). The simulations are performed for 2020 for the commercial fleet vehicle profiles of REM2030 and for private and company car profiles of MOP. Each profile is simulated individually as BEV and as PHEV while the feasibility as a BEV and the electric driving share as PHEV is found in Table 5.4. Further, the table shows the market share of utility maximizing BEVs and PHEVs with and without limited availability based on the second model step (Section 4.2.2). Market shares of profiles are shown with confidence bands.⁷⁵

user group	fleet	private	company
number of profiles	498	$6,\!177$	162
share of technically feasible $BEVs^a$	$48.8\%{\pm}4.4\%$	$57.2\% \pm 1.2\%$	$28.4\%{\pm}6.9\%$
average share of feasible trips with BEV	90.9%	96.9%	93.8%
average electric driving share PHEV	64.0%	67.7%	47.0%
BEV utility optimizing ^{a}	$6.5\%{\pm}2.2\%$	$1.1\%{\pm}0.3\%$	$0.0\%{\pm}0.0\%$
PHEV utility optimizing ^{a}	$13.1\%{\pm}2.9\%$	$8.2\%{\pm}0.7\%$	$0.6\% \pm 1.2\%^b$
BEV utility optimizing & lim. availability ^{a}	$5.4\%{\pm}2.0\%$	$1.1\%{\pm}0.3\%$	$0.0\%{\pm}0.0\%$
PHEV utility optimizing & lim. availability ^a	$11.7\%{\pm}2.8\%$	$7.9\%{\pm}0.7\%$	$0.5\% \pm 1.1\%^{b}$

Table 5.4: PEV market potential of different user groups.

a: electric vehicle shares with 95% confidence intervals.

b: lower bound for market potential at 0%.

Starting with the technical replaceability of BEVs, the largest values can be found for private vehicles, followed by fleet vehicles and company cars. The shorter daily distances of private vehicles compared to the other user groups favor BEVs from a technical point of view. However, the higher regularity in driving of fleet vehicles returns a higher amount

⁷⁵The relative frequency of PEVs is the number of PEVs k divided by the number of all vehicles n as: p = k/n. The 95% confidence interval is calculated by the standard normal approximation (see e.g. [Fahrmeir et al., 2011]): $\Delta p \approx 1.96 \cdot [p \cdot (1-p)/n]^{1/2}$.

of driving profiles feasible with a BEV compared to company cars. This is also visible in the average share of feasible trips with a BEV, which indicates that some company car profiles may contain some exceptionally long trips that are not feasible for BEVs while the average share of feasible trips is higher than for fleet vehicles. That confirms the necessity to use driving profiles with long observation periods. Results slightly change for average electric driving shares of PHEVs. Here fleet and private vehicles obtain similar average electric driving shares of about 65% while those of company cars are distinctly lower. The reason for this circumstance are again the regularity and distances of the driving profiles. It is important to mention again that the profiles of REM2030 contain three weeks of observation and MOP one week, which make results of the fleet vehicles more reliable than those of the other two groups. The influence of the observation period on technical results was analyzed in [Plötz et al., 2014a].

Inferences change when turning to the results for the utility calculation. First of all, PHEVs tend to receive higher market shares than BEVs in all user groups. This derives from the large batteries used in BEVs which are not affordable for a large number of users, while several PHEVs profit from high electric driving shares (see also Figure 4.9). Further, the effect of the limited availability is already low in 2020, since results do not change largely when it is considered.⁷⁶ Moreover, PEVs have the largest market potential within fleet vehicles, followed by private vehicles and company cars.⁷⁷ This derives from two reasons: First, the driving distances of most private vehicles are too low to profit from the lower operating costs of PEVs compared to their investment. With the higher mileage of fleet vehicles the share of vehicles with PEVs as utility maximizing solution rises. They also profit from their large technical potentials due to a more regular driving compared to company cars. Second, differences in the TCO calculation within the utility values are a reason for the different market potentials which is discussed in the following.

Utility differences of plug-in electric and conventional vehicles⁷⁸

The utilities of each propulsion technology for the individual user with his driving behavior forms an important part of his buying decision. To demonstrate the wide range of utilities for different users and the importance of the individual user behavior, the utility gaps (differences between PEVs and CVs with highest utility) between the different drive trains are analyzed using results of the second model step (see Section 4.2.2). In Figure 5.4, the utility-gaps are plotted against their share of users with this or a smaller utility gap. While the left panel shows the utility gaps for medium sized vehicles differentiated by user groups, the right panel contains private users and distinguishes vehicle sizes.

To understand this display, it should be noted that there are many individual utility gaps (differences in the utility between the drive systems) due to the large number of driving profiles, some of which have very different driving patterns. With approximately 7,000 driving profiles and four drive trains, there would be about 42,000 individual utility differences to compare. However, most important for the decision in favor of or against a PEV is the utility difference between the utility maximizing conventional vehicle and

 $^{^{76}}$ See Section 5.2.2 for a further analysis of the influence of the limited availability.

⁷⁷Note that these market shares are calculated for new vehicle registrations and are, because of the secondhand car market, different to shares in the vehicle stock which is often shown in studies on PEV market diffusion. The difference will be discussed in Section 5.2.1.

⁷⁸This section is based on [Gnann et al., 2015b]

the utility maximizing PEV. In the following, utility gap or utility difference will denote the difference in utility between the utility maximizing PEV and the utility maximizing conventional vehicle (i.e. $\Delta u \equiv \min_{p \in PEV} u_p - \min_{p \in CV} u_p$ where PEV = {BEV, PHEV} and CV = {gasoline, diesel}).⁷⁹ Figure 5.4 shows these utility gaps in ascending order on the abscissa with the share of users (respectively driving profiles) on the ordinate which have this or a smaller utility gap. The graph corresponds statistically to a relative cumulative frequency distribution or empirical cumulative distribution function [Fahrmeir et al., 2011]. The cumulative distribution function of utility gaps in Figure 5.4 shows the broad range of individual utility gaps in a statistically robust representation. Furthermore, utility-based market shares - and thus the potential impact of subsidies - can be read off directly: The market share is the share of users with utility gap smaller than zero.

Figure 5.4 demonstrates that PEVs are utility optimal for some users already in 2020 $(\Delta u < 0)$ and display a rising tendency to be so in the future (increasing share of vehicles with $\Delta u < 0$ between 2020 and 2030). In this case, the annual mileage is decisive. At low mileages, gasoline cars continue to dominate because PEVs are not able to compensate for their higher purchasing costs via their lower variable costs per kilometer. At very high annual mileages, in contrast, diesel engines are the utility maximizing option, because PHEV have to use their combustion engines too often and battery electric vehicles are unfavorable because of their limited range. The electric driving share together with the annual mileage is decisive for the difference in utility of each user. Sufficient annual mileage alone is not enough.

When comparing the graphs of the three user groups (left panel of Figure 5.4), it is noticeable that the curve of private users has the steepest slope and that of commercial fleet vehicles has the flattest slope. There are several reasons for this: First, the effect of VAT and depreciation has to be mentioned. Because VAT has to be paid on fuels by private users, the consumption savings between PEVs and CVs per kilometer driven are higher for private than for commercial users. In addition, the depreciation options for commercial drivers have the effect that utility gaps shrink further. Second, commercial users tend to have more uniform driving profiles and undertake longer trips more rarely (cf. Section 5.1.1). As a result, the electric driving shares within this group tend to be more similar compared to private users, and especially when compared to company car drivers (see also [Plötz et al., 2014a, sec. 3.1]). Yet, the tax for company cars raises their utility difference curve when compared to fleet vehicles. Although there is a special allowance in Germany for PEVs to reduce this company car tax because of high battery costs, it still increases the cost for PEVs compared to conventional vehicles significantly. Also, a WTPM is not incorporated for company cars as there is no publicly available data for it. A comparison of the utility gaps in 2020 with those in 2030 reveals that a low potential for PEVs exists under the assumptions made in 2020, but it increases over time.

Turning to the right panel of Figure 5.4, it is apparent that large passenger cars have the highest possibility to increase the share of private users with PEVs as utilitymaximizing option.⁸⁰ There are significant market shares for large vehicles and LCVs (about 40%), while small and medium vehicles' market shares are around 5%. Large vehicles are able to economize faster because of the large advantages in operating cost of PEVs. Decreasing the utility for the utility maximizing PEVs by 1,000 EUR, market

 $^{^{79}\}mathrm{See}$ Section 4.2.2 for the calculation of utility values.

⁸⁰Results for commercial fleet vehicles and company cars are not shown with the same display, as their qualitative results are similar to those of private vehicles.



Figure 5.4: Differences between individual utilities of best plug-in electric and conventional vehicle including the cost for vehicle and charging infrastructure as well as the WTPM in the medium scenario. The difference is shown on the abscissa in EUR_{2014} while the share of driving profiles is given on the ordinate. *Left panel*: Results for medium sized cars. Graphs are shown as cumulative distribution functions with dashed lines for values of 2020 and and solid lines for 2030. Private vehicles in red, fleet vehicles in blue and company cars green. *Right panel*: Results for different sized private vehicles. Graphs are shown as cumulative distribution functions for values 2020. Small cars in red, medium sized cars in blue and large cars in green.

shares of small, medium and large vehicles would increase by 5%, 10% and 20%. Such a change might, e.g., result from an increase of the gasoline price by 15% if the gasoline vehicle was the cheapest conventional car.⁸¹

The analyses of the driving profiles and utility gaps show that several drivers could achieve comparatively high electrical driving shares and have higher utilities for a PEV than compared to the respective cheapest conventional vehicle. Many of these users can be expected to buy a PEV.

5.1.3 Discussion

The presented results are subject to several assumptions that have to be discussed: The approach to determine the PEV replacement potential might be questioned as well as assumptions for driving data and technical and economical parameters that are used in the analysis. Furthermore, results are discussed in light of other studies on this topic.

In this analysis, a comparison of different drive trains is performed for every single driving profile to determine the utility-optimizing vehicle option for each profile. The potential market share of PEVs is calculated by the share of PEVs within their user group. Calculating market shares with total cost of ownership is a common approach [Thiel et al., 2010, McKinsey, 2011, Pfahl et al., 2013], although the buying decision for a single vehicle is often not solely based on cost [Globisch and Dütschke, 2013, Dataforce, 2011, Sierzchula, 2014].⁸² Some studies indicate a willingness to pay more for alternative fuel vehicles which would increase PEV market shares [Eppstein et al., 2011, Laroche et al., 2001, Mock, 2010] and is not easy to quantify. Other factors that influence the commercial buying

⁸¹If the gasoline consumption for an average sized conventional car is 0.06 l/km, its annual mileage is 20,000 km and the initial gasoline price is 1.34 EUR/l, an increase of 15% would be calculated as: $0.15 \cdot 1.34$ EUR/l $\cdot 0.06$ l/km $\cdot 20,000$ km $\cdot 3.8$ a ≈ 1000 EUR.

⁸²This type of analysis also reflects the growing amount of leasing vehicles with leasing rates reflecting the vehicle TCO.

decision [Berg, 1985] or specific use cases (like overnight inner city logistics with reduced noise or car-sharing) were not considered here. Since in company car fleets, trips could be rescheduled, PEVs could reach higher market shares when trips within the electric range were allocated to PEVs and long-distance trips to conventional vehicles (see [Haendel et al., 2015]).

The analysis of commercial fleet vehicles is based on driving profiles collected with GPS-trackers for three weeks [Fraunhofer ISI, 2014]. Although the sample is limited, it especially covers the large commercial branches in terms of registrations in Germany (see Section 3.3). The observation period of three weeks is longer than in previous studies [IVS et al., 2002, WVI et al., 2010 and decisive for realistic estimates of the technical feasibility of BEVs [Gnann et al., 2012a], the electric driving shares of PHEVs [Plötz et al., 2014a] and the market potential [Gnann et al., 2012c] since driving varies largely between drivers and days (see [Neubauer et al., 2012] and Chapter 3). Since the REM2030 data was collected with GPS-trackers, the availability of a charging point after each trip is unknown. In this analysis, vehicles are assumed to park at the company site (where charging is possible) when their bee distance is not more than 500 meters from the main company site. A distinction of different distances depending on the company size (larger distances for companies with more employees) showed no considerable changes in results. For private and and company car profiles, a data set with seven days of observation allows a more detailed analysis of market potentials than with one day. Limited observation periods may not include some rare long-distance trips, e.g. for long travels, that may not be possible for BEVs. The necessity for such trips may occur less often for fleet vehicles than for private or company cars, yet the transferability of the BEV feasibility in one or three weeks to one year is limited. A variation of the calculation for BEVs with an estimated number of long-distance trips will be discussed in the following Section 5.2. Also, the data sets with a limited number of observations influence results. For this reason, the market diffusion of PEVs in the following Section 5.2 is shown with confidence bands due to the limited sample size.

Furthermore, the results are based on assumptions for technical and economical parameters. The main assumptions for parameters were discussed in Section 4.3 and parameters are based on a large study that discussed several further parameter options Plötz et al., 2013]. A further analysis of electric consumptions in [Gnann et al., 2015a, sec. 5.3] allowed to determine the impact of changes to these parameters. This analysis of electric consumptions showed that the average values chosen for the market potential analysis reflect the peaks of electric consumption distributions and are thus justifiable. The battery degradation resulting from an intense use is reflected in an auxiliary analysis which shows that all trips performed by PEVs are covered [Plötz et al., 2013, sec. 7.8]. Also, the choice of battery sizes has an impact on results. The technical feasibility of BEVs as well as the electric driving share for PHEVs rises with the battery capacity (see e.g. [Kley, 2011, Gnann et al., 2012a). Yet, market potentials might decrease when raising battery sizes above an affordable limit (see [Kley, 2011]). Also the battery sizes for PHEVs and BEVs influence each others market potentials [Zischler, 2011]. An analysis of commercial fleet vehicles with similar assumptions for technical and economical parameters, yet smaller batteries, showed lower PEV market shares [Gnann et al., 2015a].

5.1.4 Summary

In this section, results for the vehicle usage in different user groups and the market potential of PEVs within these groups were presented for Germany in 2020. A PEV simulation of individual vehicle profile was performed to determine the technical potential for PEVs followed by an individual calculation of the utilities of different drive trains. From the evidence presented in this section the following can be summed up:

- Vehicle driving profiles with long observation periods permit statistical analyses of driving distances and regularity. A comparison of the means and variations of individually calculated log-normal distributions show that, on average, commercial fleet vehicles and company cars drive significantly more than private vehicles. Further, fleet vehicles tend to drive more regularly than private vehicles, while both groups drive significantly more regularly than company cars. This confirms the necessity to distinguish between private and commercial driving profiles.
- The driving of private vehicles is technically favorable for PEVs, followed by fleet vehicles and company cars which tend to drive too irregular. However, PEV market shares based on the utility maximizing drive trains are highest for commercial fleet vehicles, followed by private and company cars. The higher amount of VKT of commercial fleet vehicles makes them more favorable for PEVs compared to private vehicles, since they are able to profit from the lower operating costs of PEVs. Company cars suffer from the high mileage paired with a more irregular driving.
- Apart from driving, the differences in market shares stem from the exemption of VAT and depreciation allowances for commercial vehicles (fleet vehicles and company cars) which decrease the differences in utilities. Company cars, however, cannot profit of these fiscal instruments since the company car tax that has to be paid by the vehicle user. The reduction of the gross list price by a certain amount connected with the battery price is not sufficient to make them affordable to users. The missing WTPM for PEVs as company cars further decreases their market potentials.

5.2 Market diffusion of plug-in electric vehicles⁸³

The first three parts of the model ALADIN form a market diffusion model of PEVs without the co-diffusion of public charging points. After analyzing the market potential for PEVs as commercial passenger cars, understanding their diffusion and timely evolution is the goal of the following section. A special focus is put on factors that influence the PEV diffusion apart from charging infrastructure to set their influence into context. The model is extended by the stock model for PEVs (see Table 5.1) and the obstructing and favoring factors are included: the cost for individual charging points, the limited availability as obstructing as well as the WTPM for PEVs as favoring factor. In this section the influence of charging at home and at work (for private users) and at commercial sites (for fleet users) on the market diffusion of PEVs are analyzed.

This section is structured as follows: Results for PEV market diffusion are presented in Section 5.2.1. The sensitivity of results on changes in monetary and non-monetary influence factors is presented in Section 5.2.2 followed by a comparison of ways to increase the range of PEVs (Section 5.2.3). A discussion (5.2.4) and summary (5.2.5) round up this section.

Plug-in electric vehicle market diffusion models

The diffusion of new technologies and PEVs in particular has received considerable attention in the literature (see [Al-Alawi and Bradley, 2013] for a recent review of PEV market diffusion models). Since the literature review in Chapter 2 did not comprise the PEV market diffusion models, they are introduced briefly in the following. Results will be compared to those of other market diffusion models of PEVs for Germany in the discussion section.

A general classification of market diffusion models was given by Geroski [Geroski, 2000]. He describes two groups of models for market diffusion of innovations: population and probit models. Since probit models are one classification of consumer choice models (see e.g. [Train, 2009]), it is referred to consumer choice models for the second group. The latter also includes the frequently used agent-based models. These two model classes are discussed briefly and some of the existing market diffusion models for PEVs are classified accordingly in order to categorize the first three parts of ALADIN.⁸⁴

Population models describe users or adopters not as individuals, but as groups. Population models assume for example that the rate of adoption is proportional to the number of adopters and the remaining population that has not adopted a technology yet. This leads to the well-known logistic differential equation and can be interpreted via the spread of information about a technology [Geroski, 2000]. Population models offer a simple structure and interpretation. They are usually applied by calibrating the market diffusion curve to existing market data or by assuming hypothetical growth rates. This procedure is rather sensitive in early market phases when little data is available. Furthermore, the heterogeneity of the individual buying decisions and preferences of users, for example reflected in the willingness to pay more for new technologies of some users as well as the individual economics of the driving behavior, cannot be incorporated explicitly into these models.

⁸³Several results in this section have been published in [Gnann et al., 2015b].

⁸⁴The classification of Geroski is appropriate for the diffusion of technologies, while the one used in Section 2.2.2 fits better if (energy) systems are analyzed.

Population models for PEV market diffusion or market diffusion of other alternative fuel vehicles can be found in [Duleep et al., 2011, Richter and Lindenberger, 2010, Keles et al., 2008, Köhler et al., 2010, Lamberson, 2008, Meyer and Winebrake, 2009, Nemry and Brons, 2010, Shepherd et al., 2012, Wansart and Schnieder, 2010, McManus and Senter Jr., 2009], which range from simple mathematical equations to complex system dynamics models.

The second group of market diffusion models, consumer choice and agent-based models, studies adopters individually. These models are often applied when the purchase decision is more complex or the technologies to be adopted are rather expensive. For example, a simple probit model for PEV adoption would calculate the average ownership cost difference between conventional and plug-in electric vehicles and estimate a PEV market share based on this difference. As fuel and battery prices change over time, these cost differences change and with them the estimated PEV market share. Thus, consumer choice models develop market diffusion bottom-up and acknowledge that individual users can be very different. This is particularly important to identify niche markets in early phases of market development. However, these models face the problem that consumer statements about their preferences for PEVs are often inaccurate. Given the current market shares of PEVs, the vast majority of users has never experienced a PEV and can hardly judge its utility. Accordingly, the majority of studies uses stated preference data to study AFV diffusion (see e.g. [Bočkarjova et al., 2014, Brownstone et al., 2000, Batley et al., 2004, Potoglou and Kanaroglou, 2007, Glerum et al., 2013]).

Consumer choice and agent-based models were used to model PEV market diffusion in [Eppstein et al., 2011, Hacker et al., 2011b, Mock et al., 2009, Propfe et al., 2012a, Zhang et al., 2011, Shafiei et al., 2012, Yabe et al., 2012, Sullivan et al., 2009, Elgowainy et al., 2012, Brown, 2013, Higgins et al., 2012, Kihm and Trommer, 2014] where modeling approaches range from determining user shares by stated preference experiments to agentbased models. Some of these models are based on driving behavior of conventional vehicles [Hacker et al., 2011b, Mock et al., 2009, Brown, 2013, Kihm and Trommer, 2014]. This would in principle allow one to analyze user behavior in more detail. However, the latter models use driving profiles of only one day which can cause severe inaccuracies on the individual level, as a single day might not represent the individual's typical driving (see [Neubauer et al., 2012, Gnann et al., 2012a, Plötz et al., 2013] and Section 3.1).

In summary, agent-based models offer the possibility to include several aspects of great relevance for the market diffusion of PEVs in the current market development phase: individual purchase preferences, individual driving behavior (to account for the limited range of PEVs and the VKT related usage costs), the need for frequent recharging and infrastructure as well as the limited choice of PEV brands and models. In ALADIN, these these factors are explicitly taken into account in an agent-based model with different user groups and their individual decision making processes. It is possible to determine and quantify the influence of external conditions as well as (monetary) policy options. Especially the user-specific decision making is a particular strength of the model that permits to identify niche markets in early market phases. Results will be compared to other ABMs from Germany, i. e. [Mock et al., 2009, Hacker et al., 2011b, Propfe et al., 2012a, Kihm and Trommer, 2014], in Section 5.2.4.

5.2.1 Market diffusion of plug-in electric vehicles until 2030

Three scenarios for framework conditions for market diffusion were defined in Section 4.3. The results for PEV market diffusion in the three scenarios can be found in Figure 5.5. It shows the total number of PEVs in German stock on the ordinate over the years from 2015 until 2030 on the abscissa. Results are shown for the contra-EV scenario in red, the medium scenario in blue and the pro-EV scenario in green with 10% to 90%-confidence bands due to limited sample size (cf. [Plötz et al., 2014a] for details).⁸⁵ Within the 10%confidence band 1.5 to 1.9 million PEVs for the contra-EV scenario ($\approx 4\%$ of the German passenger car stock in 2030), 4.5 to 5 million PEVs for the medium scenario ($\approx 11\%$) and 9.5 to 10.5 million PEVs for the pro-EV scenario ($\approx 23-24\%$) are found.⁸⁶ These broad ranges arise from limited data samples that are considered as well as error propagation over the years. The large differences between the three scenarios also show the influence of the input parameters. Even small changes within the main drivers may change results significantly. In the medium scenario, PEV penetration rates of more than 10% of car sales are found after 2022, which derives from the increased availability of PEVs from different brands as well as decreasing vehicle and energy prices. The compound annual growth rates of all three scenarios⁸⁷ are still within the range of other AFVs or new technologies [Wietschel et al., 2014a, p.13,14]. The increasing PEV sales can be observed in Figure 5.6.

Figure 5.6 shows the registrations in the medium scenario for all drive trains on the left panel and for PEVs differentiated by user groups on the right panel. Studying the registrations distinguished by drive train, a high share of gasoline vehicles (about 60% of registrations) can be observed which is slightly higher than the current German average (about 50%). This results from the assumed fuel price ratios and costs for conventional vehicles which require about 20,000 km for medium sized diesel vehicles to economize against gasoline cars in 2015. PEV registrations are dominated by PHEVs in the beginning, while BEVs gain almost equal market shares until 2030. The lower market shares

⁸⁵ The statistical uncertainty due to finite sample size is statistically expressed as confidence bands or confidence intervals [Fahrmeir et al., 2011]. More precisely, one estimates the sales share $p_{l\tau}$ of vehicle type l (e.g. propulsion technology or vehicle size or a combination of such distinctive characteristics) from the number of driving profiles $k_{l\tau}$ that fulfill the required condition (e.g. that should be PEVs) and the number of all driving profiles in that group K_l as $\hat{p}_{l\tau} = k_{l\tau}/K_l$. Here, the hat ^ indicates an estimate for the "real" market share $p_{l\tau}$. Given a confidence level $0 < \alpha < 1$, the confidence band "contains the real value of $p_{l\tau}$ in $(1 - \alpha) \cdot 100\%$ of all cases in which confidence intervals are estimated" [Fahrmeir et al., 2011]. For a given confidence level α an upper value $p_{l\tau}^+$ and a lower value $p_{l\tau}^-$ are calculated, such that $\hat{p}_{l\tau} \in [p_{l\tau}^-, p_{l\tau}^+]$ in $(1 - \alpha) \cdot 100\%$ of the cases. Using a conservative approach, often referred to as "exact", the upper and lower confidence interval boundaries are given by [Brown et al., 2001] $p_{l\tau}^- = \text{Beta}^{-1}(\alpha/2, k_{l\tau}, K_l - k_{l\tau} + 1)$ and $p_{l\tau}^+ = \text{Beta}^{-1}(1 - \alpha/2, k_{l\tau} + 1, K_l - k_{l\tau})$. Here, Beta⁻¹(x; a, b) denotes the inverse of the cumulative Beta distribution Beta $(x; a, b) = (B(a, b))^{-1} \int_0^x t^{a-1}(1 - t)^{b-1} dt$ with the Beta function $B(x, y) = \int_0^1 t^{x-1}(1 - t)^{y-1} dt$ for normalization. Additionally, the statistical uncertainty can easily be propagated to derived quantities. See [Plötz et al., 2014a, sec. 2.3.3] for details.

⁸⁶Again, it is important to mention that the scenario results do not represent exact forecasts of the future development, but allow a model-based assessment of the main influence factors which are combined in scenarios.

⁸⁷The compound annual growth rate (CAGR) is calculated as $CAGR(t_0, t_1) = (n_1/n_0)^{1/(t_1-t_0)} - 1$ with t_0 being the initial year and and t_1 the year of observation. n_0 and n_1 are the corresponding values to observe, here the vehicle registrations. For the contra-EV scenario a CAGR(2030,2015)=14% is found, 19% for the medium scenario and 20% in the pro-EV scenario. However, the CAGR depends on the initial year and the years of observation (see [Wietschel et al., 2014a]).



Figure 5.5: Results for the three PEV market diffusion scenarios for Germany. Shown are the years on the abscissa and the total PEV stock on the ordinate. Results are shown with 10%, 30%, 50%, 70% and 90% confidence bands (cf. [Plötz et al., 2014a]), contra-EV scenario in red, medium scenario in blue and pro-EV scenario in green.

of BEVs stem from the high investments for large batteries which cannot be economized for a large group of vehicles until the prices for batteries are low enough.

Turning to the right panel of Figure 5.6 a domination of fleet vehicles in PEV registrations can be observed until 2020 (about two thirds for fleet vehicles and one third for private cars). Thereafter, the share of private vehicles increases until 2030 (about 40%) and company cars start to enter the market in 2023 increasing up to 10 % in 2030. The dominating position of fleet vehicles in PEV registrations results from the favoring conditions for PEVs: a PEV favoring driving behavior (Section 5.1.1), reimbursement of VAT and depreciation allowances (see Section 5.1.2). In 2030 almost every fourth registered car is a PEV in the medium scenario. These results are similar for the other scenarios. Table 5.5 sums up the results of PEV stocks in 2030. Results are distinguished by scenario, user group, vehicle size and PEV type.

PEV STOCK 2030	$\operatorname{contra-EV}$	medium	pro-EV
private	1,080,000	$3,\!388,\!000$	7,193,000
fleet	387,000	$1,\!177,\!000$	$2,\!151,\!000$
company	52,000	$273,\!000$	$540,\!000$
small	$175,\!000$	921,000	$2,\!110,\!000$
medium	$251,\!000$	$1,\!807,\!000$	$4,\!565,\!000$
large	714,000	$1,\!327,\!000$	$1,\!937,\!000$
LCV	379,000	784,000	$1,\!272,\!000$
PHEV	$976,\!000$	2,946,000	$5,\!824,\!000$
BEV	$543,\!000$	$1,\!893,\!000$	4,060,000
TOTAL	1,519,000	4,839,000	9,884,000

Table 5.5: PEV stock in 2030 differentiated by scenario, user group, vehicle size and PEV type.

When taking a look at Table 5.5, one may question the distribution of user groups in stock and those of registrations in Figure 5.6 which clearly differ. As mentioned in Section 4.2.3, it is assumed that all PEVs diffuse into the second-hand car market after their first holding period. These holding periods are 3.8 years for commercial and 6.2 years for private vehicles after which they diffuse into private ownership until their scrapping. For this reason, the majority of PEVs in 2030, about 70%, are privately owned, a small share (3–5%) are company cars and about one quarter of the PEV stock are fleet vehicles, independent of the scenario. PEV potentials for company cars may be lower since no WTPM is considered for these vehicles and the additional hurdle of the company car tax has to be overcome. Yet, the regularity of driving of company cars is similar to private vehicles paired with higher distances travelled which does not favor the adoption of PEVs.

Concerning car size, the distribution in the three scenarios is different. Large PEVs are less affected by market conditions and double or triple their stock with more favoring conditions (from contra-EV to pro-EV scenario). All other vehicle sizes show large variations to parameter changes. The fastest growing group of private PEVs between contra-EV and pro-EV scenario are medium sized vehicles. At a first glance, these high numbers for large vehicles seem counter-intuitive, yet longer distances are driven by large cars. With comparable investment differences and varying differences in variable costs, higher mileages allow large vehicles to economize easier. For example, the investment for a large commercial PHEV in 2020 is $LP_{large, PHEV} = 38,017 \text{ EUR}^{88}$ while a large diesel vehicle's investment is $LP_{large,diesel} = 33,387$ EUR and the difference 4,430 EUR. For small commercial vehicles the investment is $LP_{small,PHEV} = 16,822 \text{ EUR}^{89}$ for PHEVs and $LP_{small,diesel} = 12,888$ EUR for diesel vehicles resulting in a difference of 3,934 EUR. Thus the differences of vehicle investments are very close and additional investments for PHEVs are slightly higher for large vehicles. Assuming an electric driving share of 60%for both size classes, fuel cost differences would amount to 0.013 EUR/km for small vehicles and to 0.022 EUR/km for large vehicles.⁹⁰ Thus, at about 200,000 km a large PHEV would amortize against a diesel vehicle in this example and a small PHEV at about 300,000 km. Not only that the needed mileage is lower for large vehicles, but also the average mileage performed by large vehicles is higher (cf. Table 3.5 and 3.7). Although this is only an example and in the model calculations more options have to be taken into account (e.g. BEVs and gasoline vehicles as other vehicle options, different resale prices, costs for operations and maintenance or vehicle taxes) the example shows that both conditions (necessary mileage to economize and driven mileage) allow larger PEVs to gain higher market shares than smaller ones.

As mentioned already for PEV registrations the more promising PEV group is PHEVs until 2020, while BEV shares increase until 2030. In the stock BEVs account for 33% (contra-EV) to 40% (pro-EV scenario). This stems from the large batteries for medium and large for BEVs which need very high mileages to economize. PHEVs instead can gain market shares if their electric driving shares are sufficient and their mileages are high enough which is the case for an increasing number of vehicles until 2030.

Lastly, one may take a look at the number of individual charging points which are

 $^{{}^{88}}LP_{large,PHEV} = 34,351 \text{ EUR} + 13 \text{ kWh} \cdot 282 \text{ EUR/kWh} = 38,017 \text{ EUR}$

 $^{{}^{89}}LP_{small,PHEV} = 14,566 \text{ EUR} + 8 \text{ kWh} \cdot 282 \text{ EUR/kWh} = 16,822 \text{ EUR}$

⁹⁰The fuel cost for PHEVs would consist of a share for electric and one for non-electric driving, i. e. $cons_{small,PHEV} = 0.6 \cdot 0.144 \text{ kWh/km} \cdot 0.185 \text{ EUR/kWh} + 0.4 \cdot 0.051 \text{ l/km} \cdot 1.262 \text{ EUR/l} = 0.042 \text{ EUR/km}$ for small PHEVs and $cons_{large,PHEV} = 0.6 \cdot 0.193 \text{ kWh/km} \cdot 0.185 \text{ EUR/kWh} + 0.4 \cdot 0.078 \text{ l/km} \cdot 1.262 \text{ EUR/l} = 0.061 \text{ EUR/km}$ for large PHEVs in the medium scenario. The fuel costs for diesel vehicles would be $cons_{small,diesel} = 0.043 \text{ l/km} \cdot 1.262 \text{ EUR/l} = 0.053 \text{ EUR/km}$ for small and $cons_{large,diesel} = 0.066 \text{ l/km} \cdot 1.262 \text{ EUR/l} = 0.083 \text{ EUR/km}$ for large diesel vehicles.



Figure 5.6: Simulation results for vehicle registrations in 2015–2030 in the medium scenario. *Left panel:* All vehicles distinguished by propulsion technologies. BEV in green, PHEV in blue, diesel vehicles in light grey and gasoline vehicles in dark grey. *Right panel:* PEVs distinguished by user groups. Company cars in turquoise, fleet vehicles in yellow and private vehicles in khaki.

necessary for these market penetrations. Although this seems simple, as every vehicle gets assigned one charging point, the second-hand car market and the decommissioning of vehicles complicate the approach. A charging point is sold to every vehicle that is bought in the first car market which in the beginning is dominated by fleet users. Yet, more vehicles of the vehicle stock are private cars after several years. Also, it remains unclear if the owner of a PEV, that is replaced, needs to buy another charging point. Hence, there are three different options to determine the charging point stock of individual charging points (see also Figure 5.7 for the medium scenario): (1) Every vehicle sold needs one charging point. This would not consider the charging points necessary for private users in the second car market. (2) New charging points are needed for every PEV at the first registration and in the second-hand car market. For the second-hand car market the ratio of simple wallboxes and expensive domestic charging points is assumed equal to the share of new private PEV registrations. (3) Every charging point that was installed once is re-used while calculation of the ratio of wallboxes and more expensive charging points is similar to option 2. While option 1 can be ruled out completely, since it does not consider the distribution of vehicles in stock, the real number of charging points must be between options 2 and 3. However, as framework conditions ameliorate for PEVs in all scenarios a profile for which a PEV has once been utility-maximizing will not switch back to a conventional vehicle in a later period. A user that once bought a PEV will always be replace his vehicle by a PEV, hence option 3 seems to be most appropriate for the charging point calculation. One large advantage of this approach is that total charging point numbers of private (incl. company cars) and fleet vehicles are equal to their vehicle stock (cf. right bar in Figure 5.7 and PEV stock in the medium scenario in Table 5.5). Thus, the charging point shares are equal to the number of PEVs in stock in the contra-EV and pro-EV scenario. The share of private users that are in the need of a more expensive charging point, as they do not own a garage, is 6% of all charging points in the contra-EV scenario, 8% in the medium scenario and 10% in the pro-EV scenario in 2030.

In summary, the main findings of the PEV market diffusion are: (1) PEV market diffusion is largely influenced by framework conditions. (2) A large number of private and fleet driving profiles for which PEVs are, according to the model, a utility maximizing



Figure 5.7: Different inventory calculations of individual charging point stock for the medium scenario in 2030. The first bar shows the stock if only charging points of first vehicle buyers are considered, the second if also the second-hand car buyers and their charging points are reflected. In the third bar a re-use of charging points is considered. *Approach considered in this thesis.

solution. This is resulting in high PEV market shares (5% of new vehicle registrations in the medium scenario in 2020, 20% in 2030). (3) PEV registrations are dominated by fleet vehicles until 2020 which remain important in registrations until 2030, while the PEV stock is dominated by private users. (4) Although the availability of models in the vehicle market is especially limited for large PEVs, the highest market potential lies within this vehicle class as it has the highest possibility to economize. (5) High shares of PHEVs compared to BEVs are found. This results from some exceptional long trips that users might not fulfill with BEVs or the lower investments of PHEVs which is easier to economize.

5.2.2 Sensitivity analyses

The influence of individual factors is important for a deeper understanding but obscured in scenarios in which several parameters change simultaneously. Thus, the individual influence of monetary factors (retrieved in 4.3) and non-monetary factors (described in sections 4.2 and 4.3) are analyzed. In this section, all analyses refer to the medium scenario for framework conditions and only home charging is considered.

Influence of monetary factors

In Section 4.3, the cost composition of utility values for different vehicles was shown for an example and the most important factors were aggregated to scenarios (cf. Figure 4.7). Since the capital costs for the vehicles vary largely between propulsion technologies, the influence of battery prices and interest rates are chosen for variation. Further, electricity and fuel prices are varied as variable cost is different between drive trains. The influence of non-monetary factors is analyzed thereafter.

Figure 5.8 shows the influence of single parameter changes in relation to the total PEV stock in 2020 (left panel) and 2030 (right panel). On the ordinate a parameter change from 75% of the original value up to 125% is shown, while the abscissa shows the resulting total PEV stock in 2020 (2030). Changes for electricity prices are shown in dashed green, fuel prices in solid red, battery prices in dash-dotted blue and interest rates in dotted light blue. In the sensitivity analysis, the value for 2015 was fixed and



Figure 5.8: Sensitivities to parameter variation. Displayed is the influence of a variation of one single parameter on the total PEV stock. The variation of parameters is shown as factor of assumed value for every single parameter on the ordinate and the total PEV stock on the abscissa. Electricity prices in green dashes, fuel prices in solid red, battery prices with blue dash-dots and interest rates in light blue dots. *Left panel:* Variations until 2020. *Right panel:* Variations until 2030.

values for 2020 (2030) were changed by the percentage shown. Values between 2015 and the last year were adjusted accordingly. Further, in the sensitivity for electricity prices the private *and* commercial electricity prices were changed, the diesel *and* gasoline price in the fuel price sensitivity and the interest rates for private *and* commercial users in the interest rate sensitivity.

Some clear and expected results are observable in both panels of Figure 5.8. Higher numbers of PEVs can be found with increasing fuel prices and lower PEV numbers when fuel prices decrease. For all other parameter changes, lower values lead to higher market shares, e.g. there are more PEVs in stock when batteries are cheaper or the electricity price is lower. As investments for PEVs are higher than for conventional vehicles, clearly a decreasing interest rate is favorable. More interesting is the magnitude of changes caused by parameter variation: Fuel prices have the highest influence in positive and negative direction both in 2020 and 2030. The small knee in the red curves at a variation of 0.9 might derive from a switch of conventional fuel vehicles (from gasoline to diesel) or PEVs (from BEV to PHEV) as corresponding benchmarks. The second most important input factors are the battery price and electricity prices. The lowest influence is found for changes in the interest rates. These parameters are externally defined and can only slightly be influenced by PEV sales (only the battery price might decrease with economies of scale). Further, the effect of changes is naturally stronger in 2030 than in 2020. A 25% decrease of battery prices leads to a PEV stock increase of 40% in 2020 and 55% in 2030.

A monte-carlo-simulation was performed to test the robustness of results. A random change of fuel, electricity and battery prices in [Gnann et al., 2015b] was run 1,000 times. In this simulation slightly different parameters were used and simulations ran until 2020. The qualitative results, however, can be transferred to this work. A normal distribution $\mathcal{N}(\mu, \sigma^2)$ was used for the parameter variation with the value of the medium scenario as μ and the maximum of the differences between the pro-EV and medium or contra-EV and medium scenario as σ for the random variation of parameters. Model results ranged from 50,000 to 2.2 million PEVs in 2020, while the average was about 750,000 and the median 644,000 PEVs. The first quartile was at 443,000 PEVs, the third at 995,000, resulting in an interquartile range of 552,000 PEVs. Thus the medium range of most results was


Figure 5.9: Monte-carlo-simulation for variation of scenario parameters. Shown is the plug-in electric vehicle stock on the ordinate and its absolute frequency on the abscissa resulting from a random variation of fuel, electricity and battery prices.

not as wide as suggested by the scenarios. Also the medium scenario (620,000 PEVs) was very close to the median of the simulations.

From the results in this subsection, one may conclude that (1) the market evolution of PEVs is susceptible to changes in fuel prices, followed by battery and electricity prices. (2) The sensitivity analysis shows that an increase or decrease of the main influence factors by 25% can result in doubling the PEV stock in 2030 or cutting it by about 60%. (3) The monte-carlo-simulation further showed that "good" conditions favor the PEV market evolution more than "unfavorable" conditions hamper it.

Influence of non-monetary factors

One major advantage of ALADIN compared to pure TCO-models is the inclusion of several "soft factors" into the individual buying decision. These are: the availability of individual charging points, the limited choice of PEVs as hampering factors and the WTPM for PEVs as favoring factors. The model construction allows to individually exclude these factors from the buying decision and thus to test their influence on results. Figure 4.7 underlines the influence of the WTPM, the distinction of private garage owners and curbside parkers as well as the limited availability on the market diffusion of PEVs. Hence, their influence is analyzed in the following. Figure 5.10 shows the differences to the medium scenario if the three non-monetary factors are switched off individually (individual charging points in dark blue, limited availability in green and WTPM in orange) and altogether (light blue). The absolute changes in the resulting PEV stock can be found in the left panel while the right panel shows the relative changes when compared to the case with all non-monetary factors included.

The highest absolute differences can be found for individual charging points and the WTPM in 2030 when factors are considered individually. The exclusion of the limited availability has a low absolute influence. This is slightly different for the relative influence of these factors. Here, the limited availability and the WTPM have a very high influence in the beginning which decreases until 2030. This is certainly an effect of the assumed



Figure 5.10: Influence of non-monetary factors on PEV market diffusion. Shown are the differences to the medium scenario when the cost for individual charging points is not considered (dark blue), without the limited PEV availability (green), without the WTPM (orange) and when all factors except for the vehicle TCO are not considered (light blue). *Left panel:* Absolute change of PEV stock. *Right panel:* Relative change of PEV stock.

phase out of the WTPM until 2030 (see Section 4.3.4) and the decreasing limitations of PEV availability (see Section 4.3.5). Yet, the stock model causes a postponement of a zero-difference between the base case and a model without the WTPM. The charging infrastructure gains relative importance over the years and in 2030, the effects almost cancel each other, i. e. PEV market diffusion results are almost equal when soft factors are considered or not.

While the influence of individual charging points is analyzed in the next subsection, a deeper look at the allocation of the WTPM was taken in [Plötz et al., 2014a, Gnann et al., 2015b]. In [Plötz et al., 2014a], the allocation method of the private WTPM data to the driving profiles was tested and concluded that it is slightly better than a random allocation. Since a random allocation may still influence results of PEV market diffusion, a monte-carlo simulation for a random allocation of the WTPM was performed in [Gnann et al., 2015b]. Slightly different parameters until 2020 were used for the random allocation of the WTPM and the simulation was conducted 1,000 times. A decrease of the PEV stock of about 10% in 2020 was found compared to the structured allocation. Users who have been assigned a high WTPM were generally better suited for PEVs. Thus, users that are willing to pay a price premium already have favorable conditions because of their driving behavior. Beside this interesting effect, that is still valid under changing parameter assumptions, the negative effect of a random allocation might be about 5% in 2030 when the phase-out of the WTPM on the left panel of Figure 5.10 is considered.⁹¹

To conclude, a noteworthy impact of the soft factors on market diffusion can be observed especially in the beginning of PEV market diffusion. This impact decreases over time. The limited PEV availability has the strongest effect in the beginning of the market diffusion, while the cost for individual charging points and the WTPM strongly influence the PEV market diffusion until 2030. The influence of a random allocation of the WTPM should be around 5% in 2030 which is important for the interpretation of results in Section 5.3.

 $^{^{91}}$ The influence of the WTPM of changes to market diffusion results is at 80% of the PEV stock of the medium scenario in 2020 and 40% in 2030, thus the influence of a random allocation should also halve.

5.2.3 Increasing range with non-public charging or rental cars

Additional infrastructure increases the effective range of BEVs. However, there are multiple other options that might serve to address the problem of a BEV's limited range and which several users might be willing to accept to fulfill their driving needs (see also [Kurani et al., 1994, Funke and Plötz, 2014]). These will briefly be discussed to put charging infrastructure options into context:

- Switch to PHEV: A potential BEV user could simply switch from BEV to PHEV. This solution is incorporated in the model, since BEVs are excluded from the further analysis, if their range is not sufficient to fulfill the full driving profiles. The switch becomes visible when results of Section 5.1 are compared to those of [Gnann et al., 2015a] where BEV batteries are smaller and the share of PHEVs is higher since not all trips could be fulfilled by BEVs.
- 2. Use substitute vehicles for unfeasible trips: Another option is to switch to other vehicle options for trips that exceed the range of BEVs. For fleet vehicles, this would be possible if company vehicle fleets are scheduled according to their abilities: E.g. BEVs for frequent medium distances that occur regularly, diesel vehicles for very long-distance trips and PHEVs of gasoline vehicles for shorter irregular trips. This multi-dimensional optimization problem is analyzed in [Haendel et al., 2015]⁹² finding that several company fleets could be optimized by sparing vehicles completely and several companies could replace their vehicles by PEVs with even higher economic benefits than with an individual optimization. For private vehicles, it would also be possible to use different cars in the household or use rental cars for days on which the vehicle ranges might be exceeded. While for the first option the data set is not sufficient, the latter is analyzed in [Jakobsson et al., 2014]⁹³ and briefly described in the following.
- 3. Raise charging options: The effective range of BEVs (and also PHEVs) could be increased by additional public or semi-public charging infrastructure. The influence of additional charging options at work will thus be analyzed in the following. The additional availability of public charging spots was analyzed e.g. in [Kley, 2011, Gnann et al., 2012a] and compared to battery size increases in [Gnann et al., 2012b]. However, these studies did neither permit to determine a number of charging points nor was any cost for public charging considered. This will be part of the following Section 5.3.
- 4. Increase BEV battery size: Larger batteries would logically raise BEVs' ranges, yet it is more difficult to economize because of currently high battery prices. The possibility to economize does not only depend on framework conditions, but also on technical parameters of concurrent technologies, such as PHEVs. A comparison of the dependence of market shares with differing battery sizes for BEVs and PHEVs was analyzed in [Zischler, 2011].

Since option 1 is a permanent part of the model, options 2 and 3 (work charging) will be analyzed in the following. Public charging as an option to increase range (3) and the variation of battery sizes (4) will be part of the next Section 5.3.

⁹²Haendel, Gnann and Plötz (2015,i.p.)

⁹³Jakobsson, Plötz, Gnann, Sprei, Karlsson (2014)

Rental or car-sharing vehicles for long-distance trips⁹⁴

When ranges of BEVs are exceeded, the occasional use of rental or car-sharing vehicles might be considerable. To include this calculation in ALADIN, the number of days on which a substitute car is needed has to be determined. Therefore, a method proposed in [Plötz, 2014] is used that allows to calculate the number of days requiring adaptation.⁹⁵ The individual daily VKT r_l are assumed to be independent and identically distributed (iid) random variables. Let f(r) denote the user-specific distribution of daily VKT. The probability of driving more than L km on a driving day is then given by $\int_{L}^{\infty} f(r) dr =$ 1 - F(L) where F(r) is the cumulative distribution function of f(r). Let n denote the number of driving days out of N days of observation such that $\alpha = n/N$ is the share of driving days. Thus, D(L) = 365(n/N)[1 - F(L)] is the number of days per year with more than L km of daily VKT. Accordingly, D(L) is the number of days requiring adaptation for a potential BEV user. Following [Plötz, 2014], the log-normal distribution $f(r) = \exp[-(\ln r - \mu)^2]/(r\sqrt{2\pi}\sigma)$ is used to model the random variation in daily VKT of the drivers. For each individual driver, the log-normal parameters for the typical scale of daily driving μ and the variation in daily VKT σ are obtained by maximum likelihood estimates. The number of days requiring adaptation is calculated as follows: For each driver, the share of driving days is estimated as n/N and the driver-specific log-normal parameters are estimated from likelihood maximization. These were already introduced in Section 5.1.1. Using the cumulative distribution function of the log-normal distribution $F(x) = 1/2[1 + \operatorname{erf}((\ln x - \mu)/\sqrt{2}\sigma])$ the user-specific number of days requiring adaptation $D_i(L)$ is calculated (erf(x) denotes the error function). This procedure is repeated for each driver in the data base. In very rare cases (37 out of 6339), there is no variation in daily driving distance between the days reported, i.e., $\sigma_i=0$. σ_i is set equal to the sample mean in this case. However, this has almost no effect on the results reported below.⁹⁶

Figure 5.11 shows the results when BEVs are simulated with an additional cost for every day on which the range of BEVs is exceeded in relation to results of the medium scenario. Figure 5.11 uses the same display as Figure 5.12. Absolute changes in relation to the medium scenario are found on the left, their relative changes on the right panel. Four options are considered that differ in cost for rental cars: In options A0 and A1, the daily costs for a rental car, in B0 and B1, costs for a car-sharing vehicle are assumed.⁹⁷ Options A0 and B0 do not consider the costs per kilometer driven, while A1 and B1 also integrate the driving-dependent costs (with diesel) for the substitute vehicle. Until 2025, all options show a similar behavior: a raise of the PEV stock until 2020 and a decline until 2025, which is only differentiated by the height resulting from price differences for substitute vehicles. After 2025, BEVs with a rental-car option seem to be too costly compared to alternatives (PHEV, gasoline, diesel) and the difference in total BEV stock turns even negative. Car-sharing vehicles may be an option for later years since they show a positive trend even in 2030. However, the relative changes (right panel) indicate that results are positively affected in the beginning of a market evolution. But, the influence after 2025

⁹⁴This section is based on Jakobsson, Plötz, Gnann, Sprei and Karlsson (2014) [Jakobsson et al., 2014]
⁹⁵See [Weiss et al., 2014] for another approach to determine the days above a vehicle's range.

⁹⁶Please note that this log-normal estimate is expected to be valid for different driving ranges L but seems to slightly overestimate the actual number of days requiring adaptation [Plötz, 2014].

⁹⁷The cost for rental cars and car-sharing were retrieved with different time horizons before car rental, duration of car rental and in different German cities on www.billiger-mietwagen.de and www.flexauto.de/pages/carsharing-vergleich.php.



Figure 5.11: PEV stock development including substitute vehicles for long-distance trips. Shown are the differences between options in which private users may use a rental car when the range of BEVs is exceeded and medium scenario. Options differ in cost for rental cars. A0 (blue): rental car cost for small vehicles 50 EUR/day, medium sized vehicles with 65 EUR/day and large vehicles with 75 EUR/day. A1 (red) with same costs plus the cost for driving with a conventional diesel vehicle. B0 (green) and B1 (violet) with half the rental car prices of A0 and A1. *Left panel:* Absolute change of PEV stock. *Right panel:* Relative change of total PEV stocks (including commercial PEVs).

is only incremental. This is also visible when units of the abscissae of Figure 5.12 and Figure 5.11 are compared. Under the assumed parameters, the option to offer alternative vehicles for days when BEVs ranges might be exceeded, seems to be interesting until 2020 where they can raise PEV market shares. When such options are considered until 2030, results do not change noteworthy compared to the standard PEV simulation of ALADIN.

Influence of non-public charging infrastructure

In the previous subsections every vehicle has been assigned an individual charging point at home (for private and company car owners) or at the company site (for commercial vehicles). Yet, private vehicles might have the opportunity to recharge their vehicle at work as well, and for some users, the charging point at work might be even more useful than at home. For this reason, the following charging options for private users are studied in this section: (1) home only charging, (2) work only charging, (3) home and work charging. All simulations are performed in the medium scenario for framework conditions. Commercial vehicles remain unaffected and private users have to pay for either one (case 1 and 2) or two (case 3) individual charging points.

Figure 5.12 shows the differences between charging options work only vs. home only (dash-dotted blue) and home and work vs. home only (dashed red) and their timely evolution. On the left panel the absolute and on the right panel the relative changes are displayed. Results show that charging only at work would reduce the number of PEVs to half of the vehicle stock in 2030. Thus, one charging point at home is a prerequisite for PEVs given the facts that half of the registrations are commercial PEVs. Adding an additional charging point at work would instead increase the number of PEVs in stock by 20% by 2030. Thus, the additional charging point is affordable for a large number of vehicles since they are able to perform a higher share of trips electrically.

Summing up, the results of this section show (1) that rental car options (or mobility



Figure 5.12: Influence of individual charging infrastructure on PEV market diffusion. Shown are the differences between charging options in which private users only charge at work compared to charging at home (dash-dotted blue) and between charging options in which private users may charge at home and at work vs. only at home (dashed red). Commercial users without changes. *Left panel:* Absolute change of PEV stock. *Right panel:* Relative change of total PEV stocks (including commercial PEVs).

guarantees) may increase PEV sales in the beginning, yet they are no long-term option from a techno-economical point of view, as results in 2030 are not significantly different to the base case. (2) Additional charging infrastructure at work may increase electric ranges at an affordable cost for a PEV-user. This option increases the PEV-stock. (3) Charging only at work is not sufficient for the majority of potential PEV users. The impact of public charging points as well as the variation of battery prices will be discussed in Section 5.3.

5.2.4 Discussion

The results of the model ALADIN are subject to a number of uncertainties: (a) an inherent model design uncertainty, (b) an uncertainty due to the data used in the model and (c) an uncertainty caused by parameters. The uncertainty due to model design in Section 4.4 and data uncertainty was discussed in Section 3. This part focuses on the parameters that were considered in the model and their related impacts on model results.

In Section 5.2.1 the results of market diffusion are shown for three scenarios: pro-EV, medium and contra-EV. Although small variations between scenarios were considered, the difference between results is noteworthy. This arises from small utility differences between propulsion technologies (cf. Section 5.2.1). Especially for fleet vehicles, market shares can vary largely when monetary or non-monetary factors change. The display of utility gaps allows to interpret the changes caused by small differences in monetary and non-monetary factors. Since the data sample is limited, results are shown with confidence bands which cover the uncertainty due to data. Furthermore, the exact numbers of market diffusion results should permit to understand the range of results and what influences them most, yet they are no exact forecast. Under the assumptions made for the scenarios, PEVs enter the market also in the contra-EV scenario. This might be different, if, e.g., battery prices do not decline as much as assumed when PEV sales are low [Nykvist and Nilsson, 2015]. Also the composition of combining favoring and hampering factors for PEVs in the scenarios might be discussible, yet the combination of positive and negative factors for PEVs should show the range of results.

All scenarios show a high share of fleet vehicles in registrations, which is slightly decreasing over the years, in favor of a rising share of private vehicles and, delayed, a small share of company cars. The large number of private PEVs in vehicle stock stems from their diffusion in the second and car market after their first holding time. This assumption is based on the current registrations in the second car market of conventional vehicles and may be different for PEVs. While the large potential of fleet vehicles to gain significant market shares has already been discussed in the previous Section 5.1, the small share of company cars has several reasons: (1) there is no publicly available data for the WTPM of company cars, hence the utilities for PEVs and their market shares might be slightly higher. (2) The buying decision of company cars is more complex since the company and the driver participate. This is incorporated into the model (Section 4.2.2), but may differ in reality. (3) The driving behavior of company car drivers is more erratic than for fleet vehicles, but driving distances are higher than for private cars (see Section 5.1). While a certain mileage is important for PEVs to economize against conventional vehicles, several long-distance trips which occur in the week of observation make it difficult for company cars to gain high electric driving shares for PHEVs or to perform the full driving profile with a BEV.

The last point to mention concerning vehicle distribution is the ratio of BEVs and PHEVs, since the high shares of large PEVs in registration and stock was already discussed in Section 5.2.1. In all scenarios, PHEVs gain higher market shares than BEVs in the early market diffusion, while BEVs catch up until 2030. This certainly stems from battery capacities assumed in this thesis which for BEVs are difficult to economize in the beginning. However, developments in several large car markets suggest a similar evolution, since PHEVs allow to perform the common driving profile without behavioral changes [Mock and Yang, 2014]. The re-use of individual charging points may be criticized as well, yet other options were discussed thoroughly in the text.

As framework conditions have a large impact on results, several parameters were varied separately in a sensitivity analysis in Section 5.2.2. The most influential factors in utility calculation were chosen for variation. Still, several robust results on PEV market evolution and their influence factors could be obtained as a monte-carlo-simulation demonstrated (cf. Section 5.2.2). These are summed up in the following Section 5.2.5. Aside from framework conditions, user acceptance has a large impact on market diffusion but is difficult to measure and predict. Surveys clearly show that some user groups are willing to pay a premium for new technologies in general and PEVs in particular. Thus, the inclusion of a higher or smaller willingness to pay of some users is necessary to model future market diffusion of PEVs. Future research could put emphasis on retrieving more quantifiable data for the WTPM, especially for its evolution over time. Moreover, the future development of battery technology is unclear and might change results significantly if new battery technology generations are introduced [Thielmann et al., 2012].

The influence of individual charging infrastructure is analyzed in Section 5.2.3 and shows that additional charging points at work for private vehicle users can increase market shares.⁹⁸ It is questionable, whether a vehicle owner would have to pay for the installation of a charging point at work, but the employer is very likely to turn over the cost to the employee. The individual choice per user to install one or two charging points (at home and at work), which was not modeled, could increase PEV market shares even more.

⁹⁸See [Kley, 2011, Björnsson and Karlsson, 2015, Gnann et al., 2012a] for similar results for additional charging points at work.

Furthermore, it is assumed that the electricity charged at work has not to be taxed by PEV users, although this might be considered as a fringe benefit of the company to the employee. Apart from additional individual charging points to increase the electric range of PEVs, the option to use rental cars for occasional long-distance trips has been analyzed in Section 5.2.3. Results show that the influence of such an offer decreases to almost zero until 2030. However, the assumed increasing battery sizes allow longer distances over the years. These were chosen as a reaction of car manufacturers to decreasing battery cost and users' demand for higher ranges. Further, results are based on techno-economical calculations and the effect of such an offer on vehicle purchase was not analyzed.

Lastly, results should be discussed in light of other studies on the same topic. In the introduction of Section 5.2 a comparison to [Mock et al., 2009, Propfe et al., 2012a, Redelbach et al., 2013, Hacker et al., 2011b, Kihm and Trommer, 2014] was suggested. In [Mock et al., 2009, Propfe et al., 2012a, Redelbach et al., 2013] the model VECTOR21 is used, a discrete choice model that bases on a large consumer survey. While assumptions for cost and vehicles differ in the three studies, some results are observable in all of them: a declining share of diesel vehicles and a larger share of PHEVs compared to BEVs in Mock et al., 2009, Redelbach et al., 2013]. This confirms the same findings in this thesis. Results are not directly comparable to those of this thesis as also NGVs and FCEVs are considered in the scenarios which also amount to a large share of new registrations and user groups are not explicitly distinguished in the papers. While results in [Propfe et al., 2012a] seem moderate and comparable to this thesis, market shares in [Redelbach et al., 2013] of AFVs in 2030 amount to 80% which seems moderate compared to other works. Hacker et al. (2011) studied the market penetration of PEVs based on [infas and DLR, 2008] in the project OPTUM [Hacker et al., 2011b]. Although this work does only consider private vehicles, some results are comparable too. Also in this study, PHEVs gain larger market shares than BEVs and vehicles with garages are better suited than those without.⁹⁹ Kihm and Trommer (2014) use the model TREMOD for the development of a PEV market penetration [Kihm and Trommer, 2014]. They find a higher share of PHEVs compared to BEVs as well, yet other results differ largely to this thesis: In [Kihm and Trommer, 2014 private households dominate the vehicle sales and also company cars gain relevant market shares. Apart from different assumptions for techno-economical parameters, their reasoning for a non-consideration of VAT exemption is unclear and no evidence for an introduced "eco-factor" is given. Further, the individual analysis of driving profiles is based on [WVI et al., 2010] and [infas and DLR, 2008] which contain only one day of observation. Since vehicles are simulated individually and electric driving shares as well as BEV feasibility are based on the one observation day, the results of Kihm and Trommer, 2014] can be considered uncertain.

The comparison to other model results shows that several results and trends of this thesis are confirmed by other research. Yet, a PEV simulation with the differentiation of user groups and the use of driving profiles with long observation periods has to the best of the author's knowledge not been performed to this point. This analysis permits to retrieve several new results summed up in Section 5.2.5.

⁹⁹Although this is a premise of this thesis and users without a garage have to pay a premium for charging infrastructure, the general assumption that the hurdle of buying a PEV when not owning a garage is confirmed here.

5.2.5 Summary

This section analyzed the market evolution of PEVs in Germany until 2030 with nonpublic charging infrastructure. An analysis of the main monetary and non-monetary influence factors was followed by the influence analysis of additional individual charging points. The following findings should be maintained:

- The results of market diffusion demonstrate a great deal of uncertainty regarding the market evolution of PEVs because it heavily depends on external framework conditions such as price developments for batteries, crude oil and electricity prices, which may double results via small changes (±25%) of relevant input parameters. Also non-monetary factors do have considerable influence as e.g. the willingness to pay more for a new technology may cut the PEV stock in 2020 to half when it is not reflected.
- Nevertheless, the target of the German government of one million PEVs by 2020 can be reached under favorable conditions for PEVs without monetary support for the purchase of PEVs. In 2030 about 20% of of the PEV-stock could be (partly) electrified with favorable assumptions for PEVs while also in unfavorable conditions about 1.5-2 million PEVs in stock are possible until 2030. The most promising user group for PEV adoption are commercial fleet vehicles, followed by private PEVs which gain equal market shares until 2030.
- Further, results suggest a high share of PHEVs. This might derive from the strict exclusion of BEVs, if they cannot perform all of their driving electrically. Yet, no significant changes in the results were found when adding the possibility to use rental cars for long-distance trips [Plötz, 2014]. Further analyses could involve carsharing vehicles or substitute vehicles in a multi-car household (see e.g. [Khan and Kockelman, 2012]).
- Additional charging points at work increase the number of PEVs in stock even when users have to pay for it. Charging only at work is not sufficient for the majority of private PEV users. Hence, a home or primary charging point is currently a premise for the operation of a PEV.¹⁰⁰

 $^{^{100}\}mathrm{See}$ [Zhang et al., 2013] for a similar evaluation.

5.3 Interactive plug-in electric vehicle and infrastructure market diffusion

While the diffusion of PEVs with non-public charging infrastructure was analyzed in detail in the previous section, this last part of results focuses on the interaction of PEV diffusion and public charging infrastructure. The influence of non-public charging points as well as framework parameters on PEV diffusion was performed with driving profiles of MOP and REM2030. These profiles also contain information about the vehicle size and several socio-demographic attributes of vehicle and driver. However, these driving profiles do not contain geographic information about the trips, like geographic coordinates of starting and stopping points, which makes it impossible to simulate a joint use and occupancy rate of public charging points. The data set MOPS and REM2030S (both introduced in Chapter 3) contain geographic information (departure and arrival zones of each trip) and will be used for the joint analysis in the following. Although these data sets only contain trips for the region of Stuttgart, all results for PEV and public CP market diffusion are converted to Germany as described in Section 4.3.6.¹⁰¹

In the literature analysis in Chapter 2, no model was found that treated the codiffusion of PEVs and their charging infrastructure which required special features (charging time and frequency as well as a distinction of charging infrastructure accessibility types) that are not considered in the discussed models. For this reason, a comparison to models from the literature is not possible and several sensitivity calculations are performed instead.

This section is organized in the following way: First, a static analysis of public parking spots, duration and frequency after each trip of the driving profiles is performed in Section 5.3.1. Second, a detailed analysis of three scenarios for subsidizing public charging points is conducted in Section 5.3.2. Several variations are performed to analyze the influence of different parameters (5.3.3) followed by a separate analysis for fast charging points (5.3.4). A discussion (5.3.5) and summary (5.3.6) rounds up this section. As described in Table 5.1, only the medium scenario for framework parameters will be used in this section.

5.3.1 Static analysis of geographically distributed driving behavior¹⁰²

Before starting the simulation a minimum need for infrastructure per zone in the region of Stuttgart is defined (cf. Section 4.2.4). In [Funke et al., 2015] differences in a geographical coverage and a user-oriented charging infrastructure set-up were discussed, finding that a user-oriented approach would need less charging infrastructure than an approach based on a predefined geographical coverage (defined number of charging points per square meter for three types of population densities). Still, if public authorities set up charging infrastructure because of their public supply mandate, a geographical coverage is of interest. Since there is information about user behavior and geography in the data sets, it is possible to combine both approaches. As the option to recharge publicly is given when a vehicle is parked in public places, the total vehicle minutes parked publicly per zone in

 $^{^{101}}$ This conversion will be discussed in Section 5.2.4.

¹⁰²This section is based on [Gnann et al., 2015c].



Figure 5.13: Specific zone occupancy and minimum number of public charging points in different zones in the region of Stuttgart. *Left panel:* Specific zone occupancy in different zones in relation to distance to city center. Every point corresponds to one zone. *Right panel:* minimum number of public charging points in different zones in relation to distance to city center. Every point corresponds to one zone. *Right panel:* minimum number of public charging points in different zones in relation to distance to city center. Every point corresponds to one zone.

the driving profiles divided by the area are summed up and defined as the specific zone occupancy. Thus, this indicator describes how many vehicles are parked how long over the full observation period while discrepancies in surface area are reflected. The indicator is shown on the left panel of Figure 5.13 with respect to the zone's distance to the city center (central station). It is visible that zones which are closer to the city center are more likely to have a higher zone occupancy. That implies the further one approaches the city center the more vehicles are parked publicly. A further analysis shows that most zones have a zone occupancy lower than 500,000 vehicle minutes parked/(km²·wk).

To transform this variation of specific zone occupancies to charging points, it is assumed that users wish for a charging point within every 300 meters. This assumption is based on the average distance people are willing to accept to walk to the next public transport stop, which is also 300 meters according to [KVV, 2006]. With three circles that intersect in one point, the highest coverage with lowest overlap is possible, which results in 4.28 charging points per km² (see e.g. [Rune, 2001]). When this average charging point necessity is multiplied by the total area, the result would be the minimum number of public charging points for the geographical coverage approach [Funke et al., 2015]. Instead the zone occupancy and area are used to weigh the minimum number of public charging points:

$$CPN_z = A_z \cdot \overline{CPN} \cdot \frac{occ_z}{\overline{occ}}$$
(5.1)

With CPN_z being the minimum number of public charging points in zone z, A_z the area of zone z and \overline{CPN} the above mentioned average minimum number of charging point, the vehicle occupancy occ_z of zone z and the average \overline{occ} include the user-oriented approach to the analysis. The result of this formula for each zone can be found on the right panel of Figure 5.13 with respect to its distance to the city center. To give an example: In zone z_1 , two vehicles are parked publicly for half an hour and the area of zone z_1 is 10 km². Zone z_2 covers 1 km² and one vehicle is parked for five hours. The specific zone occupancy occ_{z_1} for zone z_1 would be $occ_{z_1} = 2 \cdot 30 \min/10 \text{ km}^2 = 6 \min/\text{km}^2$ and for z_2 : $occ_{z_2} =$ $300 \min/1 \text{ km}^2$. The average zone occupancy \overline{occ} would be $\overline{occ} = 153 \min/\text{km}^2$. Using the same distance between two charging points, the minimum number of charging points in the two zones would be $CPN_{z_1} = 10 \text{ km}^2 \cdot 4.28 \text{ CP/km}^2 \cdot 6 \min/\text{km}^2/153 \min/\text{km}^2 =$ 1.67 CP/km ≈ 2 CP/km and $CPN_{z_1} = 8.37$ CP/km ≈ 8 CP/km which is much less than a full coverage with 4.28 CP/km² · 11 km² ≈ 47 CP.

One can clearly observe that zones which are further away from the city center (< 40 km) need less charging points than those which are 10-40 km away while small zones in the city center also need less charging points because of their size. When considering that zones are larger the further they are away from the city center (cf. Figure A.1), the low occupancy in the zones further away from the city center weighs larger than their area.¹⁰³ Also the total sum of charging points necessary for the observation area (3,168 charging points) is significantly lower than with the geographical coverage (15,632 charging points). This zone-specific minimum number of public charging points will be used in the individual battery simulation where users are expected to only recharge their vehicle when the number of charging points in the zone they are stopping is equal or higher than the minimum number of public charging points ($n_{CP_z} \ge CPN_z$). Note that this constraint is not considered in the PEV stock simulation where vehicles stop at a charging point and charge their vehicle if it is not in use (and the battery's SOC is below 50%).

5.3.2 Public charging infrastructure evolution

With this pre-analysis determining the minimum number of public charging points, the simulation can be performed. Three subsidy scenarios with home and public charging points with 3.7 kW are analyzed in detail in this section: In subsidy scenario S1 public charging points are not subsidized while in subsidy scenario S2 the charging points are subsidized until 2020 and in S3 until 2030. The annuities of charging point costs and their subsidies are summarized in Table 5.6. A large subsidy for each charging point is considered for the first five years in subsidy scenario S2 while in subsidy scenario S3 it is phased out linearly until 2030.

Table 5.6:	Cost and subsidies for	public CPs	with a	charging	power	of 3.7	kW in	three	subsidy
scenarios.	All costs in EUR_{2014} v	vithout VAT	•						

subsidy scenario	option	2015	2020	2025	2030
S1 - no subsidy	real CP annuity	700	596	508	434
	annual subsidy	-	-	-	-
S2 - subsidy until 2020	assumed CP annuity	100	100	508	434
	annual subsidy	600	496	-	-
S3 - subsidy until 2030	assumed CP annuity	100	100	267	434
	annual subsidy	600	496	241	-

Simulation results for the PEV and CP stocks in three subsidy scenarios for Germany can be found in Figure 5.14 (S1 in red, S2 in blue and S3 in green). This figure uses a double-logarithmic scale (like Figure 2.4) to compare small and large values more easily. Several years are shown to indicate the evolution over time - every marker is equal to one year in the simulation. The first interesting result is that public charging points are not

¹⁰³Zones in the outer area, where none of the vehicle users in the data sample lives, are not considered for a charging point setup, which explains the missing minimum numbers of charging points for zones further away than 60 km from the city center.



Figure 5.14: Simulation results for Germany for PEV and public CP stock with different subsidies. Axes with logarithmic scales. Results without subsidies with red squares, for subsidies until 2020 with blue circles and for subsidies until 2030 with green triangles. If the CP stock is 0, it is set to 1 because of logarithmic scales.

able to economize when they are not subsidized (subsidy scenario S1).¹⁰⁴ Already in 2018, all charging stations are taken out of order since they cannot economize. This is certainly due to the number of PEVs that are in stock at this time (about 200,000 PEVs), as the charging points have to be subsidized until a sufficient number of PEVs is in place.

Taking a look at subsidy scenario S2, the public CP stock is not falling as much as in subsidy scenario S1, however it is much lower (about 800 public CPs in 2020) than in the beginning (about 14,000 public CPs). After 2020, the slope is declining too, since the amount of charging by the PEV stock at these charging stations is not sufficient for a take up thereafter. Only some public charging points can be maintained for subsidy scenario S2 until 2030. From 2015 until 2030, about 11 million EUR would have to be paid for subsidies in S2. In the last subsidy scenario (S3) results differ in terms of public charging points. The number of public CPs is rising when subsidies are still in place until 2030. Although the slope of the curve is declining, it remains positive until 2030 and supposedly thereafter. Thus, a tipping point is reached in this subsidy scenario and the system becomes self-sustaining. Subsidies only double (26 million EUR) when compared to subsidy scenario S2 as the subsidy per charging point decreases after 2020. When comparing the three subsidy scenarios the difference in public CP stock is obvious, yet the number of PEVs in the vehicle stock does not differ at all. For all three subsidy scenarios, about 4 million PEVs diffuse into the PEV stock until 2030 independent of the public charging options. That implies an impact of the PEV stock on public charging points (a sufficient number of PEVs has to be reached to pay of public charging points), yet an influence of public charging points on the PEV market diffusion can not be found in results. Also, the share of PHEVs in stock is similar for all three subsidy scenarios at about 70%. These results are largely driven by the energy charged at the different charging options which will be analyzed next.

¹⁰⁴Since zeros are not possible to display on a logarithmic scale, results for the public CP stock are set to one when they are actually zero.

Distribution of energy charged

The number of public CPs is directly linked to the energy charged at these charging options. Figure 5.15 shows the total energy charged¹⁰⁵ at public charging points (abscissa) with respect to the number of public charging points (ordinate) for the three subsidy scenarios. In this display the number of charging points as well as the energy charged is shown for the observation area (the region of Stuttgart).

While all three subsidy scenarios start at 688 public charging points and about 25,000 kWh charged at public charging points (in one year), they differ already in 2016 where in the non-subsidized scenario only 15 charging points are affordable compared to 50-60 in subsidy scenarios S2 and S3. In subsidy scenario S1 only few vehicles can charge at the 15 charging points remaining in 2016 which leads to a decrease in energy charged publicly to about 1,000 kWh. With a public charging price of 0.40 EUR/kWh and an electricity price of 0.30 EUR/kWh that is used in Formula 4.15, 0.10 EUR/kWh remain to cover the price for public charging points. Since 1,000 kWh multiplied by 0.10 EUR/kWh is lower than the cost for one charging point per year in S1 in 2017 (656 EUR/yr), no charging points remain in the stock after 2017. This also means that no vehicles can charge publicly from 2017 on and the number of charging points cannot increase anymore. The subsidies for subsidy scenarios S2 and S3 allow to keep a certain number of charging points in the CP stock, so charging in public remains possible after 2017. However, the sudden suspension of subsidies in S2 in 2020 decreases the number of charging points (and the amount of energy charged).¹⁰⁶ Since the prices for public charging points decrease over the years the decrease of S2 to one charging point is not as sudden as in subsidy scenario S1. Also here, it is possible to keep one charging point until 2030 with a slightly increasing amount of energy charged at this charging point. In subsidy scenario S3 instead it is possible to keep maintain the slightly increasing cost for charging points after 2020 lower than the additional earnings from energy sold at public charging points. Remembering that formula 4.15 reads as follows:

$$n_{cp}(t+1) := \frac{p_{pc}(t) + p_{el}(t)}{a_{cp}(t+1)} \cdot W_{pc}(t).$$

For the above mentioned case, let the public charging price $p_{pc}(t)$ and the price for electricity $p_{el}(t)$ be constant, then the number of charging points n_{cp} stays stable if $\Delta n_{cp} = n_{cp}(t+1) - n_{cp}(t) = 0$ or $\Delta a_{cp} = \Delta W_{pc}$, the change in additional cost for charging points is compensated by the additional energy charged. It increases if the change in public energy charged is larger than the change in cost and subsidies respectively. However, this is no trivial connection as the amount of energy charged depends on the number of charging points and PEVs and it is not possible to draw a simplified form of this connection.

Further, slightly differing results between subsidy scenarios S2 and S3 between 2015 and 2020 can be noted. Although the cost for public charging points is equal in both

¹⁰⁵For a classification of amounts of energy: a medium sized BEV with a consumption of 0.2 kWh/km and an annual mileage of 20,000 km would need 4,000 kWh per year, an average household has an electricity consumption of about 3,200 kWh per year and the total electricity consumed in Germany in 2014 amounts to $578.5 \cdot 10^9$ kWh [AGEB, 2015].

¹⁰⁶In 2020 about 7,000 kWh are charged publicly in S2 which allows a number of charging points $n_{cp} = \frac{7,000 \text{ kWh/a} \cdot 0.10 \text{ EUR/kWh}}{100 \text{EUR/a}} = 7$ charging points.



Figure 5.15: Simulation results for the region of Stuttgart for energy consumed at public CPs and public CP stock with different subsidies. Axes with logarithmic scales. Results without subsidies with red squares, for subsidies until 2020 with blue circles and for subsidies until 2030 with green triangles. If the CP stock is 0, it is set to 1 because of logarithmic scales.

subsidy scenarios, there are differences in the amount of charging and charging stations. This stems from the two random processes in the simulation: (1) Because of the limited availability of PEVs, a limited number of PEVs is randomly chosen to diffuse into the vehicle stock every year (cf. Section 4.2.3). However, this limitation decreases over the years and (2) the random order of PEVs at public charging to charge may have a greater influence. Consider a vehicle that stops at a public charging point and needs to recharge 10 kWh which takes about 2.7 hours, yet the vehicle is parked there for 10 hours. At the same time more vehicles could have been charged while the charging point was blocked. This could have been the case for one of the simulations and the other way round for the other. Especially the second random process could lead to differences in charging point usage, yet these differences decrease when only reflecting the later years with higher amounts of PEVs and CPs where positive (higher usage of individual CPs) and negative (lower usage) effects should cancel each other. To sum up, the subsidy to charging points depends on their cost, the number of public CPs and the PEVs charging at them and no trivial logic can be extracted from results to determine this number.

However, the energy charged at public charging points in relation to the full energy consumed by PEVs in combination with the number of PEVs could be a good proxy. Figure 5.16 shows the energy consumed at domestic, commercial and public charging spots in subsidy scenario S3 for one week in 2030 in the observation area (region of Stuttgart). The left panel of Figure 5.16 shows the distribution to accessibility types as stacked area plots and the right panel as lines with domestic charging in red, commercial charging in green and public charging in blue. Although commercial PEVs in subsidy scenario S3 amount to one third of the PEV stock in 2030, their higher mileage results in about 56% of the total energy consumed. Domestic charging accounts for 41% and public charging for about 3% of total energy consumed. Further, the characteristic evening peak for private vehicles (when returning home) is visible as well as the decreased peaks on



Figure 5.16: Simulation results for the region of Stuttgart for energy consumed by PEV stock over one week in subsidy scenario S3 (subsidies until 2030) in 2030. Energy charged at domestic charging locations in red, at commercial charging spots in blue and at public locations in green. *Left panel:* Total energy consumed as stacked area plot. *Right panel:* Energy consumed at different locations (non-stacked).

the weekend (last two days/peaks) (see e.g. [Dallinger and Wietschel, 2012]). Commercial charging is slightly more erratic and distributed over the day while public charging also has the small peaks in the afternoons or early evenings. Since the locations of these peaks are similar to those of domestic charging spots, a price independent load shift due to public charging points cannot be drawn from results. However, comparing the height and amount of energy charged at public charging points and in total, the impact of public charging on the energy system is small compared to domestic and commercial charging according to these results.

Finally, it is possible to analyze if the location of charging points has an impact on energy consumed at public charging spots. In Figure 5.17 two results are shown: the number of charging points with respect to their distance to the city center on the left panel and the energy charged at these charging points in 2030 on the right panel. The left panel of Figure 5.17 shows the timely evolution of charging points over the years. While most charging points are built in the city center at the moment (see blue graph for 2015), first charging points in 2020 are built further away from the city center. In 2025, several charging points are created closer to the city center, however the majority remains at about 10–20 km away. The peaks for the graph of 2030 (turquoise) indicate this finding even more. As the optimization of charging point setup changes in 2027/2028 from covering the minimum number of public charging points to the setup of charging points with high PEV occupancy (cf. Section 4.2.4), these high numbers stem from the demand for charging points in these zones.

Turning to the right panel of Figure 5.17, the usage of charging points (in 2030) is considered as well. In this figure every charging point gets assigned the average energy consumed by the charging points in its zone and the zone's distance to the city center. For example, in zone z there are ten public charging points, the energy charged in zone z is 340 kWh in the simulated week and zone z is 13 km away from the city center. Accordingly, ten charging points with a 34 kWh/wk and 13 km away from the city center are used for the display of the right panel of Figure 5.17. A lot of charging points (about 200) are found within a distance 12-15 km away from the city center and an average energy of 20 kWh/CP in one week. Yet, which amount of energy is needed for charging stations to pay off? The simulation of subsidy scenario S3 returns a public charging



Figure 5.17: Simulation results for the region of Stuttgart for public CP stock over time and energy consumed at public CPs in 2030 with respect to distance to city center in subsidy scenario S3 (subsidies until 2030). *Left panel:* Simulation results for public CP stock with respect to distance to city center in subsidy scenario S3 (subsidies until 2030) in observation area. Results are shown for different years as contours of histograms. *Right panel:* Shown is the the absolute frequency (color bar) of charging points described by their average energy consumed per charging point [kWh/CP] and the zone's distance to the city center in kilometers in a 2D-histogram.

price of 0.44 EUR/kWh in 2030 (resulting from the energy charged at and number of all public charging points). With the electricity price of 0.32 EUR/kWh about 65 kWh/wk (3,338 kWh/yr) have to be charged for a public charging point to disburse. Several charging points are above this level, however, no best location for charging points can be found on the right panel of Figure 5.17. In combination with the left panel, the best locations should be 10–20 km away from the city center, since most charging points are built there and the usage would even be higher with a lower number of charging points in these zones.

Hence, it can be retained that charging points within a distance of 10–20 km away from the city center obtain the highest usage in the simulations and can pay off most easily. This may have to do with the home locations of first PEV users in the simulation who will be analyzed in the following.

Where do PEV users live?

In this last part of Section 5.3.2, the geographical location of homes of PEV users is analyzed. For this analysis only private PEV users are analyzed since the number of commercial driving profiles is too small for a further geographical distinction.

Figure 5.18 shows the distribution of homes of private PEVs (left panel) and all vehicles (right panel) in the vehicle stock with respect to the distance to the city center. While the left panel holds simulation results for several years, the right panel contains the distribution of homes of all vehicles in MOPS. Starting with the left panel of Figure 5.18, the simulation results of the private PEV stock are shown for 2018 until 2030 in three-year steps. While for 2018 no clear pattern can be found, the distribution is peaked at a distance of 20 km away from the city center in 2021. This peak is even more distinctive in later years with more than 50% of PEV user's homes between 10–30 km distance from the city center in 2020. When comparing the distributions to those of all private vehicles in stock (right panel of Figure 5.18), the image is even more clear: Potential PEV users living in a distance between 10–30 km away from the city center are better suited for



Figure 5.18: Simulation results for the region of Stuttgart for PEV and total vehicle stock with respect to distance of home (private) or company (commercial) to city center in subsidy scenario S3 (subsidies until 2030). Results are shown as contours of histograms. *Left panel:* PEV stock for different years. *Right panel:* Total vehicle stock.

PEVs than those living closer to the city center. Vehicle owners living further away from the city center perform higher mileages on average than users living in the inner city and high mileages are needed for PEVs to economize against conventional vehicles. This argumentation is supported by other studies as well [Plötz et al., 2014b, Newman et al., 2014]. Combining the results of the distances of PEV homes (1, Figure 5.18), CP locations (2, Figure 5.17) and vehicle zone occupancy public parking (3, Figure 5.13), the following picture seems conceivable: Most PEV users live further away from the city center (1). They drive into the center during the day (probably for work) (3) and go shopping or to sports on their way home. Then they probably recharge publicly (2) in some distance to the city center before returning home.

To sum up this section, the following findings should be recorded: (1) The availability of public charging points has neither an influence on PEV market diffusion, nor on the ratio of BEVs and PHEVs from a techno-economical point of view. (2) Public charging points cannot pay off when they are not subsidized until 2030 and some public charging points have to be retained until the number of PEVs and the energy charged publicly is sufficient. (3) Only 3% of the total energy for PEVs is charged at public charging points, the distribution over the day is similar to those of home charging points. (4) Public charging points with highest usage are within 10-30 km away from the city center as are the homes of most PEV users. The robustness of these results will be analyzed in the following subsection.

5.3.3 Variation of assumptions

The results presented in the previous section are subject to a number of assumptions. Several of these assumptions are varied in the following to test their influence. The following variations are tested and results are displayed for Germany: increasing charging power, changing the charging availability with additional charging at work, varying the individual limit to recharge, changing the initial charging price, varying the (de-)construction strategies for the CPO and changing the battery sizes.

Increasing charging power

The first variation is to increase the charging power. Two variations are compared to the subsidy scenario S3 with 3.7 kW chargers subsidized until 2030: (Power1) A variation with 22 kW chargers that are also subsidized until 2030 and (Power2) a variation with 50 kW charger and subsidies until 2030. In both variations, the assumed annual costs for both charging options are cut down to 100 EUR/yr and subsidies are calculated thereof.¹⁰⁷ Results for both variations and subsidy scenario S3 are displayed in Figure 5.19 which the PEV and CP stock using the same display as Figure 5.14.

Starting with the 22 kW variation, a slight decrease of charging points can be observed until 2017 which increases until 2021 and stays almost equal until 2030. Remembering the calculations for the differences of additional energy charged and change in subsidies, the amount of additional energy charged after 2021 is enough to compensate the higher cost of charging points. When comparing this variation to S3, a lower number of charging points is found in 2030 (34,000 vs. 78,000) and the fluctuation is not as high as in S3. However, the number of PEVs in stock is unaffected of this change in power.

Turning to the 50 kW charging option, a decrease in public charging points until 2018 can be observed with a slight increase until 2020. Thereafter, the number of charging points is not able to pay off, though subsidized until 2030, and decreases to zero in 2026. The high costs of the 50 kW charging points cannot be compensated by the amount of energy charged after 2020. Also in this variation, the number of charging points has no influence on the number of PEVs (and the share of PHEVs) and the subsidies for public charging points only determine the total number of public CPs. Hence, it can be stated that the power at public charging points does not influence the market diffusion of PEVs and only determines the number of public charging points due to their cost.

Additional charging at work

The second variation is to change the availability of charging options. Here, three different charging options are compared to subsidy scenario S3: Charging at home-only (and at commercial charging spots for fleet vehicles respectively), charging at home and at work for private vehicles and charging at home, work and in public. Since public charging infrastructure is only contained in subsidy scenario S3 and the variation with home, work and public charging, these are shown on the left panel of Figure 5.20 with the same display as in Figures 5.14 and 5.19. On the right panel, the total number of PEVs in stock in 2030 for all four options is visualized instead.

On the left panel of Figure 5.20 the green graph is used for subsidy scenario S3 and the orange one for the home, work and public charging variation, i. e. in the orange variation all private users are able to recharge at work additionally (cost for charging points at work in Table 4.2). All assumptions for subsidies to public charging points are equal in both variations. One observes a slightly higher starting point in terms of PEVs in 2015 for the orange graph due to the higher registrations of private PEVs. Also the number of public charging points in the home, work and public charging variation does not decrease as much as for home and public charging since the number of PEVs in 2030 is higher when additional charging at work is possible for private users (4.7 million vs. 3.9 million

¹⁰⁷See Table B.6 for all cost assumptions.



Figure 5.19: Simulation results for Germany for PEV and public CP stock with variation of charging power. Axes with logarithmic scales. Results for subsidy scenario S3 with 3.7 kW (subsidies until 2030) with green triangles, charging power of 22 kW (with subsidies until 2030) with blue diamonds and for 50 kW charging power (with subsidies until 2030) with red squares. If the CP stock is 0, it is set to 1 because of logarithmic scales.

PEVs in S3) and also the number of public charging points differs slightly with a higher number of public CPs in subsidy scenario S3. While the latter can be explained with the less frequent use of public charging points when additional work charging is available, the change in PEV stock has to be further analyzed. The question is, whether the additional charging spot at work or the combination of home, work and public charging is responsible for the change in the number of PEVs?

On the right panel of Figure 5.20 the amount of PEVs in 2030 for all four charging availability options is analyzed. Here, it is possible to compare whether public charging may help to increase the number of PEVs in stock. By comparing the green (subsidy scenario S3) and red bars (home-only charging), it can be confirmed that additional charging in public has no positive influence on the PEV stock when added to homeonly charging. It even decreases the number of PEVs slightly due to some users that may incorporate public charging in their buying decision although charging at home would have been more economical for them. Then, the utility values for PEVs for these users may be higher than for conventional fuels and the number of registrations decreases. Although this effect should not be overstated as variations are only low, the non-existence of an influence of public charging points can be approved. Considering the blue (home and work charging) and orange bar (home, work and public charging), the same effect is visible as for the other two variations - additional public charging slightly decreases the number of PEVs. For all PEV options the share of PHEVs does not change significantly. While the same explanation holds for this comparison as given earlier, there are two main results from this part of the analysis: (1) no influence of public charging points on PEV diffusion results can be found and (2) charging at work increases the number of PEVs.



Figure 5.20: *Left panel:* Simulation results for Germany for PEV and public CP stock with variation of infrastructure availability. Axes with logarithmic scales. Results for subsidy scenario S3 (subsidies until 2030) with green triangles, additional charging at work (home, work & public) with orange diamonds. *Right panel:* PEV stock in scenario S3 (home&public, green), and variations home-only (red), home-and-work (blue) and home, work & public (orange).

Varying individual recharging limits and changing the initial charging price

After analyzing the influence of charging power and charging point availability, two more model assumptions are checked for their influence of simulation results: the limit of PEVs to recharge and the initial charging price.

As explained in Section 4.2.1, BEVs are always assumed to recharge when their battery's SOC is below 50%. The same holds for PHEVs for which driving with publicly charged electricity has to be cheaper than driving with conventional fuels additionally. This limit is changed to 70% and to 30% for two more calculations, which can also be understood as the range anxiety level of PEV users. Subsidies to public CPs are equal to subsidy scenario S3. Results for this analysis are shown on the left panel of Figure 5.21.

Surprisingly, results are almost equal for both variations and all public charging points are taken out of service until 2018. How can these very different assumptions return the same results? Turning to the variation with a limit of 30% at first. This variation assumed that vehicles are only recharged publicly when their battery's SOC is below 30%. Considering a case where a BEV arrives at a charging point with the battery SOC at 45% and it would not recharge although this would not be sufficient to return home. This may exempt several users from buying a BEV, if the individual simulation a BEV could not fulfill all his trips. Thus, only users that could fulfill their trips without public recharging would buy a BEV in this variation in the beginning, leading to a low amount of energy charged publicly and a deconstruction of charging points. Also the PEVs in vehicle stock would recharge only at a SOC of less than 30%, hence the amount of energy charged publicly would decrease even further. The variation with a 70% recharging limit suffers from a different problem: Vehicles that could return home to recharge now charge at public charging points. This increases their TCO for BEVs and they buy a PHEV or conventional car instead. In fact, results for BEVs decrease in the early years of market diffusion which decreases the amount of public charging. A higher subsidy to charging points would have been needed to keep them in stock.

Another option to increase the profitability of public CPs would be an increase of the public charging price to cover the cost for public charging points. A variation of an initial public charging price of 0.50 EUR/kWh is tested and results are shown on the right panel of Figure 5.21. Results are even more drastic than before, as now all PHEVs do



Figure 5.21: Simulation results for Germany for PEV and public CP stock with variation of recharging level when vehicles are expected to recharge and initial public charging price. Axes with logarithmic scales. Left panel: Variation of recharging level. Results for subsidy scenario S3 (subsidies until 2030) with green triangles where PEVs are assumed to recharge publicly at an SOC of less than 50%, an increased limit of 70% with red squares and a decreased limit with blue diamonds. *Right panel:* Variation of initial public charging price. Results for subsidy scenario S3 (subsidies until 2030) with green triangles where the initial public charging price is at 0.40 EUR/kWh and an increased price of 0.50 EUR/kWh with blue diamonds. If the CP stock is 0, it is set to 1 because of logarithmic scales.

not charge in public anymore since it is cheaper to drive with conventional fuels. This simply decreases the amount of public charging and the number of public charging points close to zero in the first years. A change to less than 0.4 EUR/kWh is not considered as the difference between the price for public charging $(p_{cp}(t))$ and electricity $(p_{el}(t))$, used to pay of public charging points, is already small.

These analyses showed that the initial public charging price as well as the level to recharge for public charging points have a major impact on the number of public charging points, yet the number of PEVs diffusing into the vehicle stock remains unaffected.

Changing the CPO strategies for charging point deconstruction and setup

As in all variations analyzed so far the deconstruction of public charging points has a major influence on the number of charging points. Hence, two further variations are tested in which a deconstruction is not considered: a variation without any subsidies for charging points (shown in blue on the left panel of Figure 5.22) and with subsidies until 2030 and a recharging level of 70% (in red). Results for the variation without subsidies show that the public CP stock stays equal until 2030 without changing the number of PEVs. This means that the energy charged at the public charging points is not sufficient for them to economize until 2030. The total energy charged at public charging points in the observation area is 2.5 million kWh/yr which is sufficient for 570 charging points (12,000 public CPs in Germany) to economize in 2030.¹⁰⁸ Thus again, public charging points do not diffuse into market without subsidies. This is also visible in the second variation without deconstruction that incorporates subsidies until 2030 and a higher SOClimit (or level of concern) to recharge (70%). In this variation, the number of charging points increases for the first time in 2023 and reaches a higher number of public charging

¹⁰⁸With a public charging price of 0.42 EUR/kWh and the price for electricity of 0.32 EUR/kWh, 0.10 EUR/kWh can be used to cover the cost for public charging points. This leads to n_{cp} = $\frac{0.10 \text{EUR/kWh} \cdot 2.5 \text{GWh}}{434 \text{EUR/(CP yr)}} = 576 \text{CP}.$



Figure 5.22: Simulation results for Germany for PEV and public CP stock with variation of charging point setup strategies. Axes with logarithmic scales. *Left panel:* Results for subsidy scenario S3 (subsidies until 2030) with green triangles, a variation where a deconstruction is not considered (no subsidies) with blue diamonds and a variation with non-consideration of deconstruction, necessity to recharge at 70% SOC and subsidies until 2030 with red squares. *Right panel:* Results for subsidy scenario S3 (subsidies until 2030) with green triangles and variation with a changed setup strategy with violet crosses.

points (185,000) than subsidy scenario S3 (78,000) because of the changed SOC-limit to recharge which increases the amount of public charging.

Although these results suggest a very clear picture, the location of initial charging points may not be well chosen and thus the amount of energy charged publicly could depend on the initial CP placement. Hence, on the right panel of Figure 5.22 a variation with a changed setup strategy is tested shown in comparison to subsidy scenario S3. In this variation, the minimum number of public charging points per zone is set to one. This allows a faster setup in zones with high demand for charging points and a lower minimum level of service. This variation shows similar results for the PEV stock like other variations as it does not change compared to subsidy scenario S3. In contrast to that, the CP stock is much higher (188,000 CPs) than in other variations (e.g. 78,000 CPs in S3). Thus, a different strategy for the setup of charging points could lead to a higher number of charging points or to a lower need for subsidies. However, it has no influence on the market diffusion of PEVs.

Variations of BEV battery sizes

Lastly, the model results are tested with a variation of battery sizes. Since the level to recharge also depends on the battery size, this assumption may influence results for charging infrastructure as well. While in subsidy scenario S3 a battery size of 40 kWh is assumed, it is changed to 24 kWh in the first variation and increasing from 24 kWh (2014) to 40 kWh in 2020 and remaining stable afterwards in variation 2. In all variations, charging points are subsidized until 2030 and results are shown on the left panel of Figure 5.23.

A higher starting point of PEVs in stock can be observed in both variations and the typical decrease in the number of public charging points (until 2018/2019) as well. However the number of PEVs is higher in the variation with a 24 kWh BEV battery as a higher share of potential users can afford a PEV. In this variation, the number of PEVs in 2030 is 6.5 million PEVs and the number of charging points is higher as well (315,000 CPs). This stems from the affordability of BEVs for a large number of users



Figure 5.23: Simulation results for Germany for PEV and public CP stock with variation of BEV battery sizes. Axes with logarithmic scales. *Left panel:* Results for subsidy scenario S3 (subsidies until 2030) with green triangles where the BEV battery capacity is 40 kWh, a variation with BEV battery capacities of 24 kWh and CP subsidies until 2030 with blue diamonds and a variation with BEV battery capacities of 27 kWh in 2015 rising to 40 kWh in 2020 and remaining stable afterwards plus CP subsidies until 2030 with red squares. If the CP stock is 0, it is set to 1 because of logarithmic scales.

and the PEV stock changes ratios from two thirds PHEVs in S3 to two thirds BEVs in the 24 kWh BEV battery variation. The variation with changing battery sizes returns the same number of PEVs as subsidy scenario S3 although the number of public CPs is different. The PEV stock changes from mostly BEVs in this variation in the beginning to a higher number of PHEVs which might explain the variation in the CP stock.

Having found a variation with more PEVs, the influence of public charging points should be tested again. On the right panel of Figure 5.23 three different subsidies are shown for the 24 kWh battery: (B1) no subsidies for public CPs (with red squares), (B2) with subsidies until 2020 (blue diamonds) and (B3) with subsidies until 2030 (with violet crosses). The results are similar to those shown in Figure 5.14: (1) The height and timing of subsidies changes the amount of public charging points in 2030. (2) Subsidies have to be granted also after 2020 to reach a tipping point where they economize on their own. Although the number of charging points varies (3) no influence of public charging infrastructure on the market diffusion of PEVs can be found as results for PEVs in stock are equal in 2030.

Having changed various assumptions for charging infrastructure and PEVs, it can be retained that the market diffusion of public charging points has no influence on the market diffusion of PEVs as well as the ratio of PHEVs and BEVs. The number of PEVs instead determines the profitability of public charging points. For this reason, public charging infrastructure has to be subsidized until a critical mass of PEVs with the respective amount of public energy charged is in the vehicle market. According to various calculations this tipping point is after 2020.

5.3.4 Estimation of fast charging need

In the previous subsection, the influence of different power levels was analyzed. Charging stations with 50 kW were analyzed and could not economize until 2030. However, in the combined market diffusion approach for PEVs and their charging infrastructure, it is assumed that PEVs are connected to the charging points whenever they are parked.

Since the charging time at 3.7 kW or 22 kW chargers is frequently exceeding the parking time this assumptions seems to be in order for so-called slow-charging options [BMWE, 2015]. For fast charging options, like the 50 kW charging stations, the charging time per charging event often exceeds the time connected to a charging point. With a higher amount of energy necessary to amortize, this approach could underestimate the potential for fast charging stations. Further, the possibility to interrupt a trip and recharge if the battery capacity is sufficient, is not considered in this approach.

In Section 5.2.3, an approach was introduced in which BEVs could replace their vehicles by rental or car-sharing cars if the battery capacity was exceeded. This approach bases on the methodology of [Plötz, 2014] that permits to estimate the days on which the BEV battery capacity is exceeded. Instead of rental cars, long-distance trips could be covered by recharging at fast charging points as well (cf. [Lin and Greene, 2011] for a similar approach). A simple assumption is used for a rough calculation: For every day on which the electric range of a BEV is exceeded, a potential BEV buyer would once have to pay 10 EUR (for 20 kWh of public charging, i. e. a public charging price of 0.50 EUR/kWh) to cover his mobility need. The additional range from recharging of about 100 km is sufficient for most profiles. PEV simulations with ALADIN are run with parameters of the medium scenario and the profiles of MOP to test this option.

The total number of PEVs increases with the fast charging approach by about 15% from 4.8 million PEVs to 5.5 million PEVs, thus this approach returns more PEVs. The BEVs in stock need to refuel 30 times per year on average, thus the total amount of money that could be earned with such an offer would sum up to 2.0 billion EUR between 2015 and 2030. While of the 0.50 EUR/kWh, 0.30 EUR/kWh would have to be subtracted for electricity, the remaining 0.20 EUR/kWh or 800 million EUR would serve to cover the cost for charging infrastructure. In [Plötz et al., 2013, p.158,159] the average terminal value¹⁰⁹ for a fast charging point is on average 70,000 EUR (between 2015 and 2030), so about 12,000 such charging points could be covered with the assumed costs.¹¹⁰ Thus, the fast charging option can be considered a more economic charging option than the slow-charging point. Yet, what would the usage rate of such charging facilities be?

The vehicle stock in 2030 contains about 1.2 million BEVs in the variation calculated with additional cost for fast charging. If these vehicles recharge 30 times per year at public fast charging stations, about 36 million fast charging events would take place over the year or about 96,000 per day. If these were equally distributed to the fast charging stations that contain three charging spots, then at every charging spot, 3.2 BEVs would recharge per day taking 1.25 hours. This charging time seems possible, yet it is based on the assumption that fast charging events are equally distributed over the year and between all charging stations. Since this approach does not analyze any simultaneous arrival at charging stations, it can only be considered a rough estimate. However, in contrast to slow charging stations, the PEV stock can be increased with such fast charging options and the economics seem more favorable than for slow charging points.

¹⁰⁹Here the terminal value of the cost is considered which combines investments and running costs of charging points to the TCO and transfers them to their end point.

¹¹⁰For charging points with a 50 kW, costs are taken from [Plötz et al., 2013, p.152].

5.3.5 Discussion

In the following, the data used for the analysis, several specific assumptions for public charging and the main results with respect to other studies are discussed.

For the co-diffusion of PEVs and their charging infrastructure a set of driving profiles has been used that contains geographic information of the vehicle trips. This information is decisive to determine the usage of public charging points, however, the data set lacks some socio-demographic attributes that could have been used in the simulations or for the assignment of a WTPM. For this reason, the following adjustments were made (cf. Section 4.3.6): (1) The WTPM was randomly assigned to private driving profiles. (2) Company car users were assigned to the private vehicle profiles based on the attributes: sex, occupation, household size, cars in household and driving behavior. (3) Garage ownership was randomly assigned to driving profiles based on the settlement structures of their homes and [infas and DLR, 2002]. By comparing market diffusion results of simulations with the two data sets and home-only charging, the influence of these assignments can be tested. Table 5.7 shows the results of the German PEV stock in 2030 for simulations with home-only charging for both data sets.

Table 5.7: Comparison of results with data for Germany and Stuttgart. All results for PEVs in vehicle stock in Germany in 2030 based on the medium scenario and home-only charging.

	German data		Data for region of Stuttgart			
data sets	MOP & REM2030		MOPS & REM2030S			
total number of PEVs	4,839,000	100%	$3,\!991,\!000$	100%		
private PEVs	$3,\!388,\!000$	70%	$2,\!685,\!000$	67%		
company car PEVs	273,000	6%	0	0%		
private PEV garage owners	3,039,000	63%	2,219,000	56%		
PHEVs	2,946,000	61%	$2,\!830,\!000$	71%		

Results for both data sets differ in terms of total PEVs in stock in 2030 while the share of private vehicles is about equal. Company car PEVs are not found in the results simulated with MOPS & REM2030S at all and private PEVs are less frequently bought by garage owners for the data from Stuttgart. In contrast to that, the share of PHEVs is higher with the data from Stuttgart. How can these changes be explained? As shown in Section 5.2.2, a random allocation of the WTPM to private driving profiles decreases results until 2020 by about 10% and 5% in 2030. Thus, decreasing the total number of PEVs in stock in 2030 by 5% (of private vehicles) would return 4,577,000 PEVs. By further subtracting all company car PEVs (273,000 PEVs) the total number reduces to 4,304,000 PEVs in 2030 which is very close to the results with the data from the region of Stuttgart and certainly within the 90%-confidence bands due to limited sample sizes (cf. Figure 5.5). Thus, the results for PEVs presented in Section 5.3 should be about 20%higher, however, the conclusions do not change. The slightly lower vehicle mileage in the MOPS-data (cf. Section 3.2.2), the differentiation of vehicle sizes as well as the changing battery sizes¹¹¹ may influence the PEV garage ownership as well as the share of PHEVs. Also garage ownership for PEV users could be correlated to the WTPM and hence cause an error. Yet, these effects have not been quantified up to now.

¹¹¹The simulation of the German data (MOP, REM2030) was performed with a battery size of 27 kWh in 2015 for medium sized BEV increasing to 40 kWh in 2020 and thereafter.

Several assumptions for charging points were discussed in Section 5.3.3 in which the influence of charging power, charging infrastructure availability, SOC-limits for users to recharge or another initial charging price were discussed. Also different strategies for the CPO for CP deconstruction and setup as well as different battery sizes were investigated. The minimum number of public charging points also influences results as the first charging points to be built are created based on the (driving and) parking behavior of conventional vehicles. It could be better to focus on the parking of PEVs, however in the beginning the limited number of PEVs is not expressive for the zone occupancy and a CPO would focus on the congestion and zone occupancy of conventional vehicles. With a certain number of PEVs his focus would change to PEVs. Another option would be to combine the two setup strategies, however no good mechanism was found to combine the occupancy of a small number of PEVs (e.g. 0.5 million in 2020) and a large number of conventional vehicles (45 million conventional vehicles), since their occupancy rates differ largely. A different initial charging infrastructure could also be used, e. g. a randomly distributed. Yet, the current public CPs in the observation area are already in place and should not be neglected. Moreover, in all variations most of this initial charging infrastructure is deconstructed.

Further, the more general assumption every user needs one home-charging point may be doubted. Yet, in Section 5.2.3, it was shown that charging at work-only would decrease the number of PEVs by 50%. Since vehicles park even less in public than at work, a frequently used charging point seems to be a valid presumption. Combining investment and variable cost to a TCO for public charging points might be discussible as well. Especially when investments are much higher than running cost, this might be true, although the reproduction of diesel market shares in Section 4.4.1 contradicts this hypothesis. However, at least for slow charging points the operating expenditure often exceeds the annual capital expenditure which also supports the assumption to switch off charging stations when they do no economize.

Apart from all assumptions that would change results for public CP market diffusion, all public slow charging points would have to be subsidized until 2030. The question is: Can a positive correlation of the timely evolution between public CPs and PEVs be triggered with such a subsidy? Figure 5.24 shows the correlation of the timely evolution between public CPs and PEVs with respect to the total subsidies to public CPs until 2030. To compare small and large values of the market diffusion of PEVs and public CPs, the logarithm to the base of 10 of values for PEV stock $S_{m,s}(t)$ (with $s \in \{BEV, PHEV\}$) and public CP stock $n_{CP}(t)$ is used for the correlation. The Pearson correlation coefficient is applied with $\rho = \operatorname{cov}(x, y) / (\sigma_x \cdot \sigma_y)$ with $x(t) = \log_{10}(S_{m,s}(t))$ and $y(t) = \log_{10}(n_{CP}(t))$ (see e.g. [Efron and Tibshirani, 1994]). The correlation coefficient for all subsidy scenarios and variations are displayed with 90%-confidence intervals¹¹² and the color indicates the duration of subsidies: no subsidies with green, subsidies until 2020 with red and subsidies until 2030 with blue markers. There are two main findings to be observed: (1) No positive correlation can be found when public CPs are not subsidized until 2030. This is visible for all subsidy scenarios and variations that are close to the abscissa in Figure 5.24. (2) Even when public CPs are subsidized, a positive or negative correlation may result from the simulations. There are several subsidy scenarios and variations where a subsidy is granted for every public CP until 2030, yet the stock of public CPs decreases to zero. Also, (3) the

¹¹²For the determination of confidence bands, the Fisher-transformation is used with $m = \operatorname{arctanh}(\rho)$, $SE = 1/(\sqrt{(n-3)})$ and z = (x-m)/SE (see e.g. [Efron and Tibshirani, 1994]).



Figure 5.24: Relation between subsidy to public charging points and correlation of timely evolution of PEV and public CP diffusion. Total subsidies on ordinate, correlation of PEV and public CPs on abscissa. Results for all subsidy scenarios and variations of Section 5.3 shown with different markers and whisker for confidence intervals. Subsidy scenarios and variations without subsidies with green, with subsidies until 2020 with red and until 2030 with blue markers.

broad confidence intervals indicate a large uncertainty in the correlation and thus to the effect of subsidies granted (cf. [Schroeder and Traber, 2012] for a similar evaluation).

In earlier works [Plötz et al., 2013, Gnann et al., 2015b, Plötz et al., 2015], the effect of other policy measures was analyzed of which some results are briefly repeated here to set the subsidies for charging points into context. Three of the policy options analyzed in [Gnann et al., 2015b] are shown in Table 5.8: (P1) A flat-rate subsidy given to users which started with 1,000 EUR in 2013 and decreased linearly to 300 EUR in 2020. (P2) A decrease of the interest rate for private users from 5% to 4% and (P3) a tax exemption for PHEVs. For these policy options the market diffusion of PEVs was analyzed until 2020 with slightly different parameters than throughout this thesis. Yet, the relative difference to the medium scenario and the range of subsidies of policy options should be comparable.

The total subsidies that had to be paid for these policy options until 2020 (744 million EUR for P1, 2,493 million EUR for P2 and 68 million EUR for P3) are generally higher than those for public charging points (26 million EUR in S3 and 211 million EUR in Power1).¹¹³ As public charging points are subsidized until 2030 the subsidy per PEV user is obviously lower, yet the the PEV stock does not change when public charging points are subsidized. For an uptake of PEV market diffusion the three other policy options would offer a more effective solution, even when their total subsidies are considerably higher. Which policy option is best for an introduction of PEVs is subject of a controversial public and scientific discussion [Srivastava et al., 2010, Diamond, 2009, Mock and Yang, 2014, NPE, 2014, Gass et al., 2014, Bakker and Trip, 2013, Jin et al., 2014].

¹¹³Also Dong et al. (2014) state that subsidies to public charging points are generally low compared to other policy options [Dong et al., 2014].

Policy option	PEV stock	total subsidy	subsidy per PEV
Units	PEVs 2020	million EUR	EUR/PEV
Medium scenario	631,000	-	-
(P1) Flat-rate subsidy of	$1,\!118,\!000$	744	1,529
EUR $1,000$, decreasing			
(P2) Lowering private	$1,\!143,\!000$	$2,\!493$	4,873
interest rate on investment			
(P3) Tax exemption for	$685,\!000$	68	$1,\!248$
PHEV, REEV			
Units	PEVs 2030	million EUR	EUR/PEV
(S1) No subsidies to 3.7 kW	$3,\!991,\!000$	-	-
chargers			
(S3) Subsidies to $3.7 kW$	$3,\!896,\!000$	26	7
chargers until 2030			
(Power1) Subsidies to 22 kW	$3,\!860,\!000$	211	55
chargers until 2030			

Table 5.8: Financial policy options modeled. Shown are the PEVs resulting from the policy options as well as necessary subsidies in total and per user. All subsidies in EUR_{2014} .

For fast charging a rough estimate showed that this charging option might be more promising for CPOs and the PEV market diffusion than slow charging points. This estimate could certainly be extended in further research, since it also covers a general criticism of charging simulations based on driving profiles: the possible interruption of a trip to recharge that is not covered in the PEV simulation of the proposed model.

These results are based on the model proposed in Chapter 4 in which public charging infrastructure is modeled from a mostly techno-economical point of view. In Section 5.3.1, a zone-specific minimum number of public charging points was defined which has to be in place for a vehicle buyer to consider it in his buying decision (see Section 4.2.1). Although minimum number of charging points symbolizes a potential barrier to the adoption of PEVs, the option to potentially recharge could have a greater influence on model results. However, the lack of data in this field of research does not permit a more detailed integration into the vehicle buying decision.

To the best of the author's knowledge, there are no studies treating the interaction of PEVs and their charging infrastructure that could serve for comparison. The studies analyzed in Section 2.2 treated different AFV-types, yet several results could be generalized to all AFVs. Köhler et al. (2010) found that only a small subsidy is needed for the initial refueling infrastructure and infrastructure was not a major barrier for the diffusion of FCEVs [Köhler et al., 2010]. While the small subsidy to public charging can be confirmed with this analysis, the present study shows that public charging infrastructure is not necessary for a PEV market diffusion when PEV users buy a private or commercial home-charging point. This primary charging point is necessary for a market diffusion of PEVs as stated by [Schwoon, 2007, Stephan and Sullivan, 2004, Melaina, 2003].

5.3.6 Summary

In this subsection the co-diffusion of PEVs and public charging infrastructure has been studied. A 3.7 kW charging option with three subsidy options was studied in detail in Section 5.3.2 with respect to the number of PEVs and CPs, several energy related aspects as well as locations of charging points and PEV users homes. Thereafter a number of variations was analyzed to test the robustness of results. A brief analysis of fast charging points and a discussion rounded up this section. The following findings can be extracted from this analysis:

- No impact of public slow charging points on the market diffusion of PEVs can be determined in any subsidy scenario or variation analyzed. The number of PEVs on the other hand influences the profitability of public charging points. Until a sufficient number of PEVs has diffused into the vehicle stock, a subsidy to charging points is needed to cover the expenses. A tipping point will most likely be beyond 2020.
- The energy charged at public charging points will not have a significant impact on the energy system, since the load shift potential from private load to public charging stations is low as both are most occupied at the same times. Also the energy charged at public charging points is only around 3% of the total energy charged by PEVs.
- Charging points with the highest use rates and PEV owners' homes are located 10–30 km away from the city center.
- An increase of power, a variation of the minimal SOC before recharging and the initial public charging price have no influence on the PEV market diffusion.
- Decreasing the battery sizes of PEVs or offering additional charging points at work to private users increases the number of PEVs in stock. However, this increase is not connected with public slow charging infrastructure.
- First calculations indicate that public fast charging points could increase the number of PEVs and offer a more economic public charging point option. More effort needs to be spend on their investigation.

5.4 Synopsis of simulation results

This chapter aimed at analyzing the market diffusion of PEVs and their charging infrastructure using the model proposed in Chapter 4. Three parts were presented with a rising model complexity: the market potential of PEVs as private or commercial vehicles (Section 5.1), the market evolution of PEVs with non-public infrastructure (Section 5.2) and the co-diffusion of PEVs and their infrastructure (Section 5.3). Here, results are compared to each other and their implications are discussed.

Commercial fleet vehicles are the most important user group for PEV sales, followed by private and lastly company cars. Commercial fleet vehicles show a higher market potential for PEVs than private vehicles. The larger shares of commercial vehicles in new registrations in Germany as well as the reimbursement of VAT and depreciation allowances for commercial vehicles increase the market potential for PEVs in general. The analysis of commercial fleet driving profiles collected for this thesis also shows a favoring higher mileage and regularity compared to private driving profiles. This results in higher PEV market shares in registrations for fleet vehicles than for private vehicles. That also influences the need for public charging infrastructure, as commercial PEVs charge less often in public.

Framework conditions have the largest impact on PEV diffusion, followed by battery sizes and charging options at work. Public (slow) charging points have no impact on PEV market diffusion. While market shares are larger for commercial fleet vehicles, the stock is dominated by private users in which PEVs diffuse through the second-hand car market. The market diffusion of PEVs is strongly dependent on framework conditions, especially on energy and battery prices. Results for three scenarios range from 2 to 10 million PEVs in Germany in 2030. This is also visible on the left panel of Figure 5.25 which shows a comparison of results for the market diffusion of PEVs that were shown in Section 5.2. Non-monetary factors like a WTPM or the cost for individual charging infrastructure have an influence on PEV market diffusion in the beginning (cf. Section 5.2.2 and [Gnann et al., 2015b]), yet it decreases over time (as assumed ex ante). Apart from framework conditions, only additional charging points at work have a considerable influence on the diffusion of PEVs in the simulations without public charging infrastructure.

The interaction of PEVs and their charging infrastructure is complex, since charging options for PEVs have to be divided into three groups: domestic or commercial, work and public charging points. Section 5.3 especially focused on the interaction of public CPs and PEVs. The right panel of Figure 5.25 shows all simulation results of Section 5.3 with the number of public CPs in 2030 on the ordinate and the number of PEVs in stock in 2030 on the abscissa. This figure again demonstrates that most subsidy scenarios and variations are on a horizontal line, indicating that the number of public charging points has no influence on the PEV diffusion. Only two options increase the PEVs in stock in 2030: A decrease of BEV battery sizes (raises the PEV stock by about 2.5 million additional PEVs) and additional charging points at work (about 700,000 PEVs). Also the decrease of BEV battery sizes raises the PEV stock independent of the number of charging points.

For a PEV diffusion, most important are domestic and commercial charging points, followed by charging points at work and lastly in public. Results show that the availability of domestic or commercial charging points has the most impact on market diffusion of PEVs which is cut to half when only charging at work is possible and to zero with public



Figure 5.25: Comparison of simulation results. *Left panel:* Results of PEV market diffusion without public charging infrastructure from Section 5.2. Framework scenarios with blue crosses and sensitivity calculations with red diamonds. *Right panel:* Results of PEV and public CP market diffusion in 2030. Calculations of Section 5.3 shown with green crosses.

charging points only. The number of PEVs can be increased with additional charging points at work, also when users have to pay for it. Public slow charging points ($\leq 22 \text{ kW}$) can not increase the number of PEVs, independent of their exact configuration, while public fast charging points ($\geq 22 \text{ kW}$) indicate slight increases in a rough estimate. Lin and Greene (2011) come to similar conclusions for charging power (cf. [Lin and Greene, 2011]). The market diffusion of PEVs has a noteworthy impact on the diffusion of charging points on the other hand. For the assumed one-to-one allocation of individual charging points this connection is predefined, yet the number of PEVs is also decisive for the diffusion of public charging points. Public (slow) charging infrastructure needs to be subsidized until reaching their tipping point. In all calculations made in this thesis, this tipping point is beyond 2020.

The problem of a limited range of BEV users can best be addressed by a switch to PHEVs, then by a increase of battery sizes. Additional charging options and rental cars are less important from a techno-economical point of view. Resuming the possibilities to increase the assumed range of BEVs in Section 5.2.3, four possibilities were discussed: (1) switch from BEV to PHEV, (2) use substitute vehicles, (3) raise charging options and (4) increase a BEVs battery size. In Section 5.2.3, the alternative use of rental cars for long-distance trips was discussed (case 2), which might help in the beginning of a market diffusion, yet not in the long term. Additional charging options (case 3) would help to increased the BEV range if created at work and increase the PEV stock in 2030. Also public fast charging points may be an option, yet slow charging in public does not increase the PEV stock in 2030. Increasing the battery size (case 4), to simply gain range, may not be affordable for a large number of vehicles (cf. Section 5.3.3). Thus, offering affordable BEVs with an user-accepted range will be the key task for the automotive industry in the next years. Switching to a PHEV (case 1) does not increase the range of the BEV, but it may be the favorite option for a large number of potential PEV customers during the next years. Almost all results throughout this thesis confirm this statement.

Chapter 6

Conclusions and further research

6.1 Summary and conclusions

Plug-in electric vehicles are a means to reduce greenhouse gas emissions and the dependency on fossil fuels from the transportation sector. However, their diffusion depends, among other factors, on the prevalence of a charging infrastructure and vice versa. Hence, the aim of this thesis was to answer the following question: "How do the diffusion of plugin electric vehicles and the diffusion of their charging infrastructure mutually influence each other?"

For this purpose, an agent-based model has been developed that simulates potential PEV buyers individually with the existing charging infrastructure and jointly simulates PEV users and their interaction at public charging points for Germany until 2030. A charging infrastructure agent decides about the setup or deactivation of public charging points based on the publicly charged energy and associated revenues. The individual analysis permits to examine charging at different infrastructure facilities - at home (domestic or commercial), at work and in public - which is decisive since the usage of one charging facility influences the other [Schroeder and Traber, 2012].

The model is based on driving profiles with an observation period of at least one week to cope with the variation in driving between users and days. Two household travel surveys with an observation period of one week were used to extract the driving behavior of private vehicles. As for commercial vehicles, driving profiles with an observation period of more than one day were not publicly available, more than 500 commercial vehicle driving profiles with an observation period of three weeks have been collected for this thesis. Further, two types of PEVs are distinguished in the analyses - PHEV and BEV - that have different needs for the charging infrastructure since PHEV can refuel conventionally as well.

Three scenarios were defined with different prices for the most important framework conditions: electricity, fuel and battery prices. A scenario with favorable conditions for PEVs (pro-EV) and a scenario with unfavorable conditions (contra-EV) were completed by a medium scenario in between the two. None of the scenarios makes extreme assumptions, e.g. the oil price lies between 115 $_{2014}$ /bbl and 180 $_{2014}$ /bbl in 2030. The following scenario-independent findings respond to the initially proposed research questions (Chapter 1). They are divided into contentual and methodological results and conclusions.

There is no lock-in in the co-diffusion of PEVs and their charging points for countries

with a large availability of home charging points. Model results show that, a domestic or commercial charging infrastructure has a strong influence on PEV diffusion results and covers the needs of most PEV users. A simulation of charging only at work showed that the PEV stock was cut by half and charging only at public charging points would return even fewer PEVs since parking durations at these sites are too short. Thus, charging at home is a prerequisite for a large number of PEVs. Additional charging at work could increase the number of PEVs since this would effectively increase PEVs' electric ranges. Additional charging at public slow charging points ($\leq 22 \text{ kW}$) has no influence on market diffusion results of PEVs which was demonstrated in multiple scenarios. Parking times are either not sufficient or public charging is not necessary to be able to drive PEVs. Additional calculations with a slightly adapted approach for public fast charging points (>22 kW) indicate a potential stimulation of PEV diffusion, though further research on this topic is needed.

The diffusion of public charging points follows the diffusion of PEVs. The influence of PEVs on home and work charging points was not analyzed since their one-to-one allocation (one home and/or work charging point per vehicle) was a prerequisite of the model. For public charging points a strong influence of the number of PEVs was found. The number of PEVs and their usage determines the profitability of public charging points. Until there is a sufficient number of vehicles charged with a good amount of public energy charged, public charging points have to be subsidized in large measure. Yet, the most important charging infrastructure type for PEV diffusion are charging points at home (domestic or commercial), followed by charging points at work and lastly by public charging points.

Framework conditions are more important than public charging points. With charging infrastructure only at domestic and commercial sites, a PEV market diffusion of 2-10 million PEVs, depending on assumptions, is possible for Germany until 2030. However, the diffusion of PEVs is largely influenced by framework conditions, such as oil, electricity and battery prices. While the influence of a PEV diffusion on energy prices is supposedly low until 2030, potential policies to foster PEV market diffusion should be dynamically adaptable to react to changing framework conditions. Further, the configuration of vehicles (especially battery sizes) influences the total number of PEVs. Thus, car makers have to find the right balance between vehicles with high ranges and their affordability to the customer. All simulations show a higher share of PHEVs and commercial vehicles which both have a limited need for a public charging infrastructure. It can be confirmed that "inadequate recharging availability will not be the key barrier in holding back the near-term penetration of BEVs-PHEVs" [Lin and Greene, 2011]. This statement may also apply to public charging points in the medium-term if battery technology keeps developing and electric ranges are extending.

In countries with home-charging options, an extensive public charging infrastructure roll-out is not necessary from a techno-economical point of view. As a result, policy makers should focus on facilitating the access to home charging points for potential PEV buyers with monetary or non-monetary instruments. Similarly, regulatory barriers for charging at work may be reduced, e. g. a tax exemption for charging at work which could be considered a fringe benefit to the employee (see also [Trigg et al., 2013]). For public charging points the focus should lie on fast charging which is not an unusual situation to the user because of the similarity to conventional refueling stations and fewer charging points are needed. Here, the terms of billing, like the exact measurement of energy charged, could be facilitated. Investments in public charging infrastructure remain a risk though [Schroeder and Traber, 2012]. In EU directive 2014/94/EU, which deals with the deployment of an alternative fuels infrastructure, 150,000 public charging points in Germany by 2020 were proposed [EC, 2014], while the BMWE suggested 35,000 public charging points by 2020 [BMWE, 2015]. The model results contradict these numbers from a techno-economical point of view as the availability of home charging options and actual user behavior should to be taken into account.

The distinction of different access types for charging infrastructure - at home, at work, in public - is obligatory for a model development. The analysis of the co-diffusion of PEVs and their charging infrastructure is different to studies for other alternative drive trains, since the usage of different charging options depends on each other. Especially the public charging infrastructure diffusion should not be analyzed independently of home charging infrastructure diffusion which covers high shares of charging needs.

Driving behavior differs between drivers, days of individual drivers and user groups. While most studies focus on the typical mileage of a user on an average day, this is neither sufficient to identify early niche markets, nor does it reflect the requirements for PEVs to become economically viable. High annual mileage paired with regular driving within the electric range require a detailed analysis of individual driving behavior in different user groups. This requires driving profiles of more than one day.

The differentiation between BEVs and PHEVs is important. While BEVs are limited to the use of (public) charging points, PHEVs can also use refueling stations for range extension. The use of public slow charging points by PHEVs will most likely be opportunity charging only, while it may be a requirement for some BEVs. The large share of PHEVs throughout all model results indicates the necessity for their consideration.

Agent-based simulation models fit best for these various modeling requirements. The previously mentioned aspects make agent-based models the most promising approach to gain insights into the co-diffusion of PEVs and their charging infrastructure, since agent-based models are useful "when the population is heterogeneous, when each individual is (potentially) different." [Bonabeau, 2002]

Studies which analyze driving profiles individually should use driving profiles with long observation periods. Driving profiles are often used to cope with the variation in driving between different vehicle users. When driving profiles are used for an individual analysis of PEVs, a long observation period is decisive. This lowers an eventual overestimation of PEV-potentials as it covers the variation in driving.

Commercial vehicles should be analyzed individually when they have significant market shares. In Germany, commercial users account for more than half of the vehicle market and thus represent an important market segment for PEVs. Since special rules apply to their buying decision which differ from private vehicles, e. g. the exemption from VAT, they should be modeled separately. This applies for other vehicle markets as well if market shares have a relevant size or accounting rules differ between private and commercial vehicles.
6.2 Discussion and further research

Discussion

Different approaches to the diffusion of PEVs and the diffusion of their charging infrastructure are conceivable, e.g. modeling of coupled differential equations in a top-down approach. However, as mentioned in the last section, charging infrastructure should be divided into subgroups and their usage depends on each other. Also a distinction between PHEVs and BEVs is necessary, since the ability of PHEVs to refuel conventionally incorporates another option to refuel. The diversity of users and the task to identify a niche with convenient attributes for PEVs make such top-down approaches inadequate for this research question. Since no overall objective function can be formulated for this research question (PEVs could be maximized, however the number of public charging infrastructure may be maximized or minimized), an optimization model does not seem useful either. The individual rules that can be formulated for this large number of differently behaving users favor the use of a simulation model which is confirmed by previous modeling approaches (cf. Chapter 2). Further, the analysis is a model-based assessment and no probabilities of occurrence of the three scenarios can be given. Hence, conclusions are drawn related to the influencing factors of the diffusion of PEVs and their charging infrastructure, but not on their absolute number.

Concerning data, driving profiles with a "long" observation period were used. Although driving profiles with an observation period of one week or three weeks are more reliable for an individual analysis, they still do not cover all aspects of driving behavior, e.g. long-distance travels. This issue was addressed with an estimated number of days that exceed the electric range of PEVs in Section 5.2.3. Further, the number of driving profiles is limited, yet their representativeness was discussed in Chapter 3 and some results are shown with confidence bands due to the sample size (see Section 5.2.1). Like most studies, the analyses performed in this thesis could be improved with larger data sets. This especially applies to commercial fleet vehicle and company car profiles, but also to the magnitude and evolution of the WTPM. Since the joint simulation was performed with data for the region of Stuttgart, their results should not be directly transferred to less populated areas. However, the transferability to Germany was discussed in Section 5.3.5, finding comparable results for the market diffusion of PEVs for Germany. A projection to other countries might be a further field of research (see e.g. [Pasaoglu et al., 2014, Seixas et al., 2015]), but was beyond the scope of this thesis.

Moreover, results are based on scenarios and assumptions that were discussed thoroughly throughout this thesis. Yet, not all aspects influencing this topic could be incorporated into the model and some important facets should be mentioned: The simulations for the co-diffusion of PEVs and their charging infrastructure are performed for Germany from 2015 until 2030. This is an early market phase of PEVs in a highly industrialized country, and, e.g., not comparable to a complete replacement of conventional vehicles by plug-in electric vehicles. Results might differ for countries in which the electricity grid is not as elaborated and stable as for Germany. Several aspects could affect results in the long term, but are not incorporated into the model: Improvements in battery technology [Nykvist and Nilsson, 2015] are currently unknown and results for studies vary largely. Three evolutions were chosen that were discussed thoroughly with a groups of experts in this field [Plötz et al., 2013]. An inclusion of fuel cell electric vehicles like in [Propfe et al., 2012a] could be a model extension for the future, yet the focus of this study was to analyze the connection between PEVs and their charging infrastructure. In addition, an increasing use of car-sharing programs could affect the diffusion of PEVs in the long term, but until 2030 the number of PEVs in car-sharing is expected to not be the main driver for overall PEV diffusion. Rebound effects [Frondel et al., 2012] or a change in mobility behavior would both influence results, yet data on both effects is not publicly available in the desired granularity. Further on, feedback loops for the costs of batteries and vehicles representing economies of scale could also be incorporated into a later version of the model. Further, aspects like the ban on driving in cities due to air pollution, an intensification of European directive 443/2009 [EC, 2009], which treats the regulations of CO₂ emissions in the vehicle fleet, after 2020 or a tax on electricity charged for driving could affect PEV diffusion by large, yet they are beyond the scope of this study. Also, the psychological effect of a (public) charging infrastructure which might increase the need for public charging infrastructure was not the focus of this study.

Further research

This thesis aimed at analyzing the co-diffusion of PEVs and their charging infrastructure. While the author focused on answering this question, several further fields of research could be identified.

Modeling of public charging infrastructure in this thesis focused on techno-economical aspects. The psychological need for charging points for potential users is often stated, yet to the best of the author's knowledge unquantified. Retrieving quantitative data on the vehicle buyer's utility of charging infrastructure and integrating it into his buying decision would certainly improve the model. Also survey data about neighboring effects would ameliorate the interaction in the model and the switch to a multi-agent simulation could be possible. More research is needed that combines psychological and techno-economical aspects in general and for PEVs and their charging infrastructure in particular.

Further, initial calculations were made for the influence of fast charging points on PEVs in this thesis. Results showed that the profitability of fast charging points was higher and also their contribution to PEV diffusion is noticeable. These calculations were made with a different approach since staying connected to a charging point as long as vehicles are parked is appropriate for slow charging, yet not for the interruption of longdistance trips for fast charging. However, this analysis did not consider the simultaneity of charging events. Further research on the interaction of PEV diffusion and fast charging points could be an interesting research topic.

Moreover, energy-related aspects like load shifting, buffering energy for intermittent renewable energy or energy storage for PEVs have a minor relevance for Germany until 2030 [Dallinger and Wietschel, 2012, Heinrichs, 2014]. Yet, these could become relevant in local networks or with a faster increasing number of PEVs which may include additional benefits that make public charging points more profitable. In connection with a public charging infrastructure, this could be a further field of research as well.

Finally, the publicly available information about company cars is scarce. Their buying decision is more complex than for fleet vehicles and private cars, since the company (with a pre-selection) and the private user (with the final choice) are involved. Also data on driving behavior and their WTPM is hardly available. Understanding their buying behavior and their potential to adopt PEVs is an uncovered field of research.

Appendix A

Vehicle usage data

A.1 Extraction of private driving profiles

The private data sets MOP [MOP, 2010] and MOPS [Hautzinger et al., 2013] are both household travel surveys which contain the trips of all members of participating household.¹¹⁴ Since in this analysis the movement of vehicles is of concern, all vehicle trips have to be assigned to vehicles where unambiguously possibly. The following assumptions are used (see also [Kley, 2011, Chlond et al., 2014]):

- 1. If there is only one vehicle in the household, trips of all household members as vehicle driver are assigned to the vehicle. In this case the socio-demographic information of the first driver is assigned to the vehicle profile.
- 2. If the number of vehicles exceeds the number of drivers, the trips of the first household member are assigned to the first vehicle, those of the second driver to the second vehicle until the last driver's trips are assigned. This might overestimate the driving of single vehicles.
- 3. If the number of vehicles is smaller than the number of household members, the vehicles will be exempted from the further analysis, since the allocation of trips to vehicles is unknown.

These assumptions are valid for both private data sets used in the analysis, MOP and MOPS, while details for both are explained in the following.

Extraction of vehicle driving profiles from MOP

The part of MOP used in this thesis contains mobility behavior of the years 1994 until 2010. For every year, there are four tables in MOP (XX indicates the year):

- *hhXX*: Household table which contains information about the household, e.g. household size, household income, city size, parking situation of vehicles in the household.
- pXX: Person table contains information about all members of participating households, e.g. their age, sex or education.

 $^{^{114}}$ For more details refer to Section 3.2. Note, that MOPS is a synthetic data set, yet the structure of tables is similar to MOP.

- wXX: Trip table that comprises all trips performed in the observation period and the means of transport.
- $tank(XX+1)^{115}$: Refueling table which includes all information about the vehicles and their refueling behavior.

All files can be connected with their primary keys [MOP, 2010]. To retrieve the above mentioned cases in which an allocation of trips of persons to vehicles is possible, all trips are extracted with a vehicle as means of transport and the person as driver (wXX.vmdiw=4). For plausibility reasons, also the possession of a driving license is checked (pXX.fspkw=1). Then all households with only one vehicle (case 1) are extracted and all trips are assigned to the vehicle (see Listing A.1). Thereafter, all trips of persons in households with more vehicles than driver (case 2) are assigned to the vehicles while trips of the first stating driver are assigned to the first stated vehicle (pkwXX.persnr = wXX.persnr, see Listing A.1).

Listing A.1: MOP vehicle driving profiles extraction part 1

SELECT

```
pkw10.hhid, pkw10.persnr, w10.jahr AS jahr, w10.wotag, w10.bertag,
  w10.datum, w10.abzeit, w10.zweck, w10.anzeit, w10.km, p10.sex,
  p10.gebjahr, p10.schulab, p10.beruf, hh10.raumtyp, hh10.kreis,
  hh10.ewzahl, hh10.hhtyp, hh10.einko,
  CASE tank11.pkwordnr
    WHEN 1 THEN hh10.parkstr1
    WHEN 2 THEN hh10.parkstr2
    WHEN 3 THEN hh10.parkstr3
    ELSE hh10.parkgar+1
  END AS garage, tank11.privpkw, tank11.nutzung, tank11.marke,
  tank11.baujahr, tank11.hubraum, tank11.benzin, tank11.kmjahr,
  tank11.hub_klas, pkw10.persnr AS pkwno, hh10.hhgro AS hhgro
FROM
  (SELECT
    wege.hhid, 1 AS persnr
  FROM
    (SELECT
      DISTINCT hhid, persnr
    FROM
      (SELECT
          id AS hhid, persnr
      FROM
          w10
      WHERE
          vmdiw = 4
      )
    ) wege,
    (SELECT
        DISTINCT pp.hhid, pp.persnr, pp.hh_pot_drivers, hh.pkwhh
    FROM
      (SELECT
          DISTINCT p1.hhid, p1.persnr, p2.hh_pot_drivers
      FROM
        (SELECT
            id AS hhid, persnr
```

¹¹⁵The household, person and trip table use the same year for one survey wave while the refueling table has the following year as connection: tank11 belongs to hh10.

```
FROM
            p10
        WHERE
            fspkw = 1
        ) p1
        LEFT OUTER JOIN
        (SELECT
            id AS hhid, COUNT(persnr) AS hh_pot_drivers
        FROM
            p10
        WHERE
            fspkw = 1
        group by id
        ) p2
        on (p1.hhid = p2.hhid)
      ) pp, hh
      WHERE
          (hh_pot_drivers > pkwhh AND pkwhh = 1)
      AND
          pp.hhid = hh.id
    ) personen
    WHERE
      wege.hhid = personen.hhid
    AND
      wege.persnr = personen.persnr
  ) pkw10
 p10, hh10, w10, tank11
WHERE
  pkw10.hhid = p10.id
AND
  pkw10.persnr = p10.persnr
AND
  pkw10.hhid = hh10.id
AND
  pkw10.hhid = w10.id
AND
  pkw10.hhid = tank11.idhh
AND
  pkw10.persnr = tank11.pkwnr
```

Listing A.2: MOP vehicle driving profiles extraction part 2

SELECT

```
pkw10.hhid, pkw10.persnr, w10.jahr AS jahr, w10.wotag, w10.bertag,
w10.datum, w10.abzeit, w10.zweck, w10.anzeit, w10.km, p10.sex,
p10.gebjahr, p10.schulab, p10.beruf, hh10.raumtyp, hh10.kreis,
hh10.ewzahl, hh10.hhtyp, hh10.einko,
CASE tank11.pkwordnr
WHEN 1 THEN hh10.parkstr1
WHEN 2 THEN hh10.parkstr2
WHEN 3 THEN hh10.parkstr3
ELSE hh10.parkgar+1
END AS garage, tank11.privpkw, tank11.nutzung, tank11.marke,
tank11.baujahr, tank11.hubraum, tank11.benzin, tank11.kmjahr,
tank11.hub_klas, pkw10.persnr AS pkwno, hh10.hhgro AS hhgro
FROM
(SELECT
```

```
wege.hhid, wege.persnr
 FROM
    (SELECT
      DISTINCT hhid, persnr
   FROM
      (SELECT
          id AS hhid, persnr
     FROM
          w10
      WHERE
          vmdiw = 4
      )
    ) wege,
    (SELECT
        DISTINCT pp.hhid, pp.persnr, pp.hh_pot_drivers, hh.pkwhh
   FROM
      (SELECT
          DISTINCT p1.hhid, p1.persnr, p2.hh_pot_drivers
      FROM
        (SELECT
            id AS hhid, persnr
       FROM
            p10
        WHERE
            fspkw = 1
        ) p1
        LEFT OUTER JOIN
        (SELECT
            id AS hhid, COUNT(persnr) AS hh_pot_drivers
        FROM
            p10
        WHERE
            fspkw = 1
        group by id
        ) p2
        on (p1.hhid = p2.hhid)
      ) pp, hh
      WHERE
          hh_pot_drivers <= pkwhh
      AND
          pp.hhid = hh.id
    ) personen
   WHERE
      wege.hhid = personen.hhid
   AND
      wege.persnr = personen.persnr
  ) pkw10
, p10, hh10, w10, tank11
WHERE
  pkw10.hhid = p10.id
AND
  pkw10.persnr = p10.persnr
AND
  pkw10.hhid = hh10.id
AND
  pkw10.hhid = w10.id
```

AND pkw10.persnr = w10.persnr AND pkw10.hhid = tank11.idhh AND pkw10.persnr = tank11.pkwnr

Since the data structure changed over the years, this allocation has to be adapted for every year. Table A.1 shows the differences between the years with respect to the data of 2010.

Field	Problem	Concerns	Solution
		years	
hhXX.Parkstr1	not	1994-2002	hhXX.parkgar+1 as garage
	available		
hhXX.Parkstr2	not	1994-2002	hhXX.parkgar+1 as garage
	available		
hhXX.Parkstr3	not	1994-2002	hhXX.parkgar+1 as garage
	available		
tankXX.privpkw	v not	1995 - 1999	NULL as privpkw
	available		
tankXX.nutzung	; not	1995 - 1999	NULL as nutzung
	available		
tankXX.typ_k_a	different	1995, 1996,	case tankXX.typ_k_a when 2 then 3
	coding	2000-2002	when 3 then 5 else tankXX.typ_k_a
			end as benzin
tankXX.typ_k_a	different	1997-1999	case tankXX.typ_k_a when 2 then 5
	coding		when 3 then 4 when 4 then 2 else
			tankXX.typ_k_a end as benzin
tankXX.kmjahr	not	1995-2003	NULL as kmjahr
	available		
tankXX.hub_kla	snot	1995, 1996,	NULL as hub_klas
	available	2002	
tankXX.id	Combination	1995-2001	hh94.id = tank95.id/10 (div);
	of HHID		$p94.pkwnr = tank95.id \%10 \pmod{0}$
1 3737 1	and PersNr	1005 0001	
tankXX.pkwnr	not	1995-2001	tankXX.pkwnr = tankXX.id %10
1 37 37 1	available	2002	(modulo)
tankXX.pkwnr	not	2002	tankAA.pkwnr = tankAA.pkwnrhh
	available	1004	1. 1.1 2010
tank AA .marke	ASCII-	1994	coaing like 2010
	coding		

Table A.1: Changes between observation waves in data set MOP.

After the extraction of all driving profiles, trips with 0 km are deleted (three cases) and trips with arrival times prior to departure (two cases). Since the field $tankXX.hub_klas$, which contains the vehicle size, is not completely filled, the size of the vehicles is determined by their cubic capacity: small (< 1,400 ccm), medium (1,400 ccm \leq CC < 2,000 ccm) and large (CC \geq 2,000 ccm). This yields to 6,339 vehicles with 172,978 vehicle trips.

Extraction of vehicle driving profiles from MOPS

A similar procedure is performed for MOPS. This data set is slightly different from MOP as it contains no information about the vehicles used and less socio-demographic information about the households and persons. Thus, there are three tables:

- Haushalt: Contains the information about households, i.e. the four attributes: household-id, zone of the household, household size (persons) and vehicles in household.
- Personen: Information about the persons (nine attributes): household-id, person-id, sex, employment status, possession of a transit pass, possession of a driving license, zone of working place, zone of apprentice position.
- Wege: Information on all trips performed (twelve attributes): household-id, personid, day of week, departure time, arrival time, duration of trip, starting zone, stopping zone, distance, means of transport, trip purpose, duration of following activity.

Since the number of vehicles is only available in the household table, the allocation of person trips to vehicles runs as in Listing A.1.

Listing A.3: MOPS vehicle driving profiles extraction

```
CREATE TABLE mops_weg_miv AS
(SELECT
FROM
  mops_weg
WHERE
  modus LIKE 'MIV')
CREATE TABLE mops_pkw_hh AS
(SELECT
  hh.hhid AS hhid.
  hh.hh_groesse AS pkw_in_hh,
  count(miv.pid) AS pkw_driving
FROM
  mops_haushalt hh,
  mops_weg_miv miv
WHERE
  hh.hhid = miv.hhid
GROUP BY
  miv.hhid);
CREATE TABLE mops_weg_pkw AS
SELECT
  alltrips.hhid,
  CASE feasible.pkw_in_hh
    WHEN 1 THEN 1
    ELSE alltrips.pid
  END AS pkwid,
  alltrips.wochentag,
  alltrips.abzeit,
  alltrips.anzeit,
```

```
alltrips.dauer,
  alltrips.vz_ab,
  alltrips.vz_an,
  alltrips.strecke,
  alltrips.modus,
  alltrips.zweck,
  alltrips.dauer_aktivitaet
FROM
    SELECT
    FROM
      mops_pkw_hh
    WHERE
      pkw_in_h = 1
    OR (
       pkw_in_h > 1
      AND pkw_driving <= pkw_in_hh
    )
  ) feasible,
  mops_weg_miv alltrips
WHERE
  feasible.hhid = alltrips.hhid;
```

In the initial data set, there are 21,949,597 vehicle trips available performed by 1,585,271 vehicles. The allocation reduces the data sample to 20,514,826 vehicle trips of 1,312,817 vehicles. For practical reasons, especially computing time, the time intervals of trips were reduced to 15 min sections, which further decreases the number of trips to 19,100,429. A further partition is described in Section 4.3.6.

A.2 Further information about MOPS

An important advantage of MOPS in comparison to MOP is the availability of geographical information. This information is necessary for the joint simulation of PEVs at public charging points. Yet, the geographic information in the driving profiles are no coordinates, but geographic zones in GIS-format. A map of these zones is shown in Figure 3.2 in Section 3.2, yet the 1,173 zones are hardly visible on the map. For a better interpretation of results, Figure A.1 shows the connection of zone area and its distance to the city center.

Figure A.1 shows the area of the zones on the ordinate and the distance to the city center on the abscissa. Both axes use a logarithmic scale to be able to compare small to large values and every dot corresponds to one zone. It is clearly visible that the area of zones rises with the distance to the city center which was already mentioned in Section 3.2. Also two groups are visible for zones with a distance of 1 to 10 km distance to the city center (inner city zones) and from 10 to 100 km distance (outer city zones).



Figure A.1: Distance and area of zones in observation area of MOPS. Both axes with logarithmic scales. Every point corresponds to one zone.

A.3 Collection of commercial vehicle usage data

The data collection of REM2030 was performed from June 2011 until June 2014. The goal was to collect data of commercial vehicles for at least three weeks and to be representative for the distribution of vehicles to commercial branches in commercial vehicle registrations. The data collection was carried out in 13 survey waves with 50 GPS-trackers of which 45 on average were used during the surveys and 40 sent usable data (see next paragraph for data correction). Companies were asked by phone to participate in a data collection for the application of PEVs as commercial vehicles and the rate of participation was at about 20% on average. GPS-trackers were sent to companies who distributed them to vehicle users who were willing to participate in the data collection personally¹¹⁶. The vehicle movements were audited in an online portal and companies were contacted in case the cars were not moving. All vehicle trips including their starting and stopping time and geographical coordinates were collected. After 4-5 weeks, participants had to fill out a small questionnaire indicating the vehicles' sizes (small, medium, large, LCVs) and the company size (<10, 11-50, 51-100, 101-250, 251-1,000, 1,001-5,000, >5,000). Since 2014 also the common parking spot ("own parking spot on company site", "varying spots on company site", "no parking on company site"), the number of car users (one/multiple) and the vehicle usage (as fleet vehicle or company car) were requested. These questionnaires were digitized and reports for every company with vehicle-specific recommendations about the PEV-replaceability based on the driving behavior were sent to companies. Additionally the city size of the company's main site, the commercial branch and commercial segment according to [Eurostat, 2008] were collected as additional information.

This procedure allows to pursue the representativity regarding the commercial branches in vehicle registrations. However, incorrect reporting of additional information in the

¹¹⁶Participants had to agree personally because of data privacy.

questionnaire cannot be precluded. Several plausibility checks of the trips were performed to correct some technical issues. Since the data recording with GPS-trackers suffers from dead spots of GPS and radio communication, the recorded trips were tested for plausibility and, if necessary, adapted or additional trips were inserted. There were two types of trips that had to be adapted: (1) the distance between starting and stopping point was shorter than the minimal distance extracted from Google-maps and (2) the length of the trips resulted in an average speed below 1 km/h or the duration was below one minute. The first case of incorrect distances concerned 2,585 trips with an mean distance of 4.2 km (SD: 30.3 km). Incorrect driving times (case 2) were found in 2,397 cases with a mean distance of 1.1 km (SD: 13.4 km). Furthermore, additional trips were inserted if the destination of trip *i* differed from the origin of the trip i + 1. Then an additional trip was determined using Google-maps, which concerned 2,597 trips with a mean value of 15.6 km (SD: 67.6 km).

Appendix B

Techno-economical parameters

This annex holds the parameters used for calculations and their references were discussed in Section 4.3. Table B.1 holds the parameters for small vehicles, Table B.2 for medium sized vehicles, Table B.3 for large vehicles and Table B.4 for light commercial vehicles. For all BEVs a maximum depth of discharge of 90% is used, while PHEVs may use 80% of their battery capacity [Plötz et al., 2013].

The limited availability of PEVs as described in Section 4.3.5 is shown in Table B.5. Lastly, the cost and subsidies to public charging points with 22 kW and 50 kW charging power used in Section 5.3.3 are shown in Table B.6.

Parameter	unit	2015	2020	2025	2030
battery capacity BEV^a	kWh	20	20	20	20
battery capacity $PHEV^a$	kWh	8	8	8	8
conventional consumption $Gasoline^b$	l/km	0.058	0.054	0.051	0.048
conventional consumption Diesel^b	l/km	0.046	0.043	0.041	0.038
conventional consumption $PHEV^b$	l/km	0.054	0.051	0.048	0.045
electric consumption BEV^b	kWh/km	0.164	0.155	0.146	0.138
electric consumption $PHEV^b$	$\rm kWh/km$	0.153	0.144	0.135	0.127
O&M cost Gasoline ^{c}	$\mathrm{EUR/km}$	0.026	0.026	0.026	0.026
O&M cost $Diesel^c$	$\mathrm{EUR/km}$	0.026	0.026	0.026	0.026
O&M cost $PHEV^c$	$\mathrm{EUR/km}$	0.023	0.023	0.023	0.023
$O\&M \text{ cost } BEV^c$	$\mathrm{EUR/km}$	0.018	0.018	0.018	0.018
net list price $Gasoline^d$	EUR	10,477	$10,\!699$	11,033	11,403
net list price $Diesel^d$	EUR	$12,\!666$	$12,\!888$	$13,\!222$	$13,\!592$
net list price PHEV w/o battery ^{d}	EUR	$14,\!991$	$14,\!556$	$14,\!556$	$14,\!556$
net list price BEV w/o battery ^d	EUR	10,923	$10,\!480$	$10,\!480$	$10,\!480$
vehicle tax $Gasoline^e$	$\mathrm{EUR/yr}$	65	50	50	50
vehicle tax Diesel^e	$\mathrm{EUR/yr}$	139	126	126	126
vehicle tax $PHEV^e$	$\mathrm{EUR/yr}$	26	26	26	26
vehicle tax BEV^e	EUR/yr	0	0	0	0

Table B.1: Parameters for small vehicles. All prices and costs without VAT in EUR₂₀₁₄.

a: [Hacker et al., 2011b, Gnann et al., 2012a, Linssen et al., 2012, Pfahl, 2013]

b: [Helms et al., 2011]; c: [Propfe et al., 2012b]; d: [Pfahl, 2013]; e: [BMF, 2014]

Parameter	unit	2015	2020	2025	2030
battery capacity BEV^a	kWh	27	40	40	40
battery capacity $PHEV^a$	kWh	10	10	10	10
conventional consumption $Gasoline^b$	l/km	0.072	0.066	0.062	0.057
conventional consumption Diesel^b	l/km	0.057	0.053	0.049	0.046
conventional consumption $PHEV^b$	l/km	0.066	0.062	0.058	0.055
electric consumption BEV^b	kWh/km	0.201	0.190	0.180	0.170
electric consumption \mathbf{PHEV}^b	kWh/km	0.189	0.179	0.168	0.159
$O\&M \text{ cost Gasoline}^c$	$\mathrm{EUR/km}$	0.048	0.048	0.048	0.048
O&M cost $Diesel^c$	$\mathrm{EUR/km}$	0.048	0.048	0.048	0.048
O&M cost $PHEV^c$	$\mathrm{EUR/km}$	0.043	0.043	0.043	0.043
O&M cost BEV^c	$\mathrm{EUR/km}$	0.033	0.033	0.033	0.033
net list price $Gasoline^d$	EUR	17,298	$17,\!698$	18,298	18,965
net list price $Diesel^d$	EUR	$19,\!485$	$19,\!885$	$20,\!485$	$21,\!152$
net list price PHEV w/o battery ^{d}	EUR	$21,\!677$	$21,\!116$	$21,\!116$	21,116
net list price BEV w/o battery ^d	EUR	$17,\!613$	$17,\!042$	$17,\!042$	$17,\!042$
vehicle tax $Gasoline^e$	$\mathrm{EUR/yr}$	125	101	101	101
vehicle tax $Diesel^e$	EUR/yr	226	209	209	209
vehicle tax $PHEV^e$	$\mathrm{EUR/yr}$	34	34	34	34
vehicle tax BEV^e	EUR/yr	0	0	0	0

Table B.2: Parameters for medium sized vehicles. All prices and costs without VAT in EUR₂₀₁₄.

a: [Hacker et al., 2011b, Gnann et al., 2012a, Linssen et al., 2012, Pfahl, 2013]

b: [Helms et al., 2011]; c: [Propfe et al., 2012b]; d: [Pfahl, 2013]; e: [BMF, 2014]

	I.			-	-2014
Parameter	unit	2015	2020	2025	2030
battery capacity BEV^a	kWh	59	80	80	80
battery capacity $PHEV^a$	kWh	13	13	13	13
conventional consumption $Gasoline^b$	l/km	0.095	0.087	0.080	0.074
conventional consumption Diesel^b	l/km	0.071	0.066	0.061	0.056
conventional consumption $PHEV^b$	l/km	0.084	0.078	0.072	0.067
electric consumption BEV^b	kWh/km	0.216	0.204	0.193	0.183
electric consumption $PHEV^b$	$\rm kWh/km$	0.204	0.193	0.182	0.171
$O\&M \text{ cost Gasoline}^c$	EUR/km	0.074	0.074	0.074	0.074
O&M cost $Diesel^c$	EUR/km	0.074	0.074	0.074	0.074
$O\&M \text{ cost } PHEV^c$	EUR/km	0.066	0.066	0.066	0.066
O&M cost BEV^c	$\mathrm{EUR/km}$	0.051	0.051	0.051	0.051
net list price $Gasoline^d$	EUR	30,755	$31,\!355$	$32,\!255$	$33,\!255$
net list price $Diesel^d$	EUR	$32,\!987$	$33,\!587$	$34,\!487$	$35,\!487$
net list price PHEV w/o battery ^{d}	EUR	$34,\!986$	$34,\!351$	$34,\!351$	$34,\!351$
net list price BEV w/o battery ^{d}	EUR	30,932	30,232	30,232	30,232
vehicle tax $Gasoline^e$	EUR/yr	229	193	193	193
vehicle tax Diesel^e	EUR/yr	349	325	325	325
vehicle tax $PHEV^e$	EUR/yr	46	46	46	46
vehicle tax BEV^e	EUR/yr	0	0	0	0

Table B.3: Parameters for large vehicles. All prices and costs without VAT in EUR_{2014}

a: [Hacker et al., 2011b, Gnann et al., 2012a, Linssen et al., 2012, Pfahl, 2013]

b: [Helms et al., 2011]; c: [Propfe et al., 2012b]; d: [Pfahl, 2013]; e: [BMF, 2014]

Parameter	unit	2015	2020	2025	2030
battery capacity BEV^a	kWh	32	32	32	32
battery capacity $PHEV^a$	kWh	16	16	16	16
conventional consumption $Gasoline^b$	l/km	0.119	0.110	0.102	0.095
conventional consumption Diesel^b	l/km	0.089	0.083	0.077	0.072
conventional consumption $PHEV^b$	l/km	0.104	0.098	0.092	0.086
electric consumption BEV^b	kWh/km	0.324	0.306	0.289	0.274
electric consumption $PHEV^b$	kWh/km	0.301	0.285	0.269	0.255
$O\&M \text{ cost Gasoline}^c$	EUR/km	0.050	0.050	0.050	0.050
$O\&M \text{ cost Diesel}^c$	EUR/km	0.050	0.050	0.050	0.050
O&M cost $PHEV^c$	EUR/km	0.044	0.044	0.044	0.044
$O\&M \text{ cost } BEV^c$	EUR/km	0.034	0.034	0.034	0.034
net list price $Gasoline^d$	EUR	38,000	38,600	39,500	40,500
net list price $Diesel^d$	EUR	40,200	40,800	41,700	42,700
net list price PHEV w/o battery ^{d}	EUR	42,853	$42,\!171$	$42,\!171$	$42,\!171$
net list price BEV w/o battery ^{d}	EUR	38,215	$37,\!477$	$37,\!477$	$37,\!477$
vehicle tax $Gasoline^{e}$	EUR/yr	161	161	161	161
vehicle tax Diesel^e	EUR/yr	161	161	161	161
vehicle tax $PHEV^e$	EUR/yr	161	161	161	161
vehicle tax BEV^e	EUR/yr	0	0	0	0

Table B.4: Parameters for LCVs. All prices and costs without VAT in EUR₂₀₁₄.

a: [Hacker et al., 2011b, Gnann et al., 2012a, Linssen et al., 2012, Pfahl, 2013]

b: [Helms et al., 2011]; c: [Propfe et al., 2012b]; d: [Pfahl, 2013]; e: [BMF, 2014]

Table B.5: Limited availability of PEVs.

Parameter	unit	2015	2020	2025	2030
limited availability BEV small ^{a}	-	0.482	0.845	0.970	0.995
limited availability PHEV small ^{a}	-	0.037	0.650	0.989	1.000
limited availability BEV medium ^{a}	-	0.480	0.876	0.982	0.998
limited availability PHEV medium ^{a}	-	0.539	0.982	1.000	1.000
limited availability BEV $large^{a}$	-	0.043	0.184	0.532	0.852
limited availability PHEV $large^a$	-	0.273	0.889	0.994	1.000
limited availability BEV LCV^a	-	0.260	0.632	0.893	0.976
limited availability PHEV LCV^a	-	0.018	0.461	0.976	0.999

a: Assumptions based on press announcements summed up in [Plötz et al., 2013, Ch. 7.4].

Table B.6: Cost and subsidies for public CPs with a charging power of 22 kW and 50 kW. All costs in EUR_{2014} without VAT.

scenario	option	2015	2020	2025	2030
Power1 (22 kW)	assumed CP annuity	100	100	466	883
	annual subsidy	$1,\!254$	$1,\!114$	557	-
Power2 (50 kW)	assumed CP annuity	100	100	$1,\!840$	$3,\!580$
	annual subsidy	$6,\!693$	$5,\!196$	$2,\!444$	-

Danksagung

Die vorliegende Arbeit entstand zwischen 2011 und 2015 während meiner Tätigkeit am Fraunhofer Institut für System- und Innovationsforschung ISI. In diesen vier Jahren bin ich von einer Vielzahl an Personen unterstützt worden, denen ich hierfür sehr dankbar bin.

An vorderster Stelle möchte ich meinem Doktorvater, Prof. Dr. Martin Wietschel, danken, der mir immer für Fragen zur Verfügung stand und mich häufig von der Detailebene zurück zu den Kernaussagen führte. Im gleichen Atemzug ist die außerordentliche Unterstützung meines Betreuers, Dr. Patrick Plötz, zu nennen, dem ich nicht nur für sein stetiges inhaltliches Hinterfragen, sondern auch für sein großes Engagement sehr dankbar bin. Eine bessere wissenschaftliche und persönliche Betreuung hätte ich mir nicht wünschen können.

Zudem möchte ich mich bei den Mitgliedern der Prüfungskommission bedanken: Prof. Dr. Christof Weinhardt, Prof. Dr. Maxim Ullrich und vor allem bei Prof. Dr. Wolf Fichtner für die Übernahme des Korreferats. In meiner Arbeit habe ich Mobilitätsdaten aus dem Großraum Stuttgart verwendet, die ich dankenswerterweise von Dr. Martin Kagerbauer und Michael Heilig vom Institut für Verkehrswesen des KIT zur Verfügung gestellt bekommen habe. Für die Unterstützung bei der Datenaufbereitung mit GIS bin ich Dr. Martin Pudlik zu Dank verpflichtet. Desweiteren konnte ich von den Abschlussarbeiten von Florian Zischler, Sebastian Dziubiel, Michael Haag und Jan Hoffmann profitieren, aber auch von der Zuarbeit zahlreicher wissenschaftlicher Hilfskräfte.

Für das Korrekturlesen meiner Arbeit danke ich Patrick Plötz, Martin Wietschel, Vicki Duscha, Julia Michaelis und Rebekka Volk, für die Englischkorrektur Barbara Sinnemann und meiner Mutter Claudia Choleva-Gnann. Meinen ISI-Kollegen, insbesondere Julia Michaelis, Vicki Duscha, Uta Schneider, Judit Kockat, Simone Steinhilber, Simon Funke und André Kühn, möchte ich für ihre Hilfsbereitschaft, Freundschaft und die oftmals notwendige Ablenkung danken. Meiner Freundin, Rebekka Volk, danke ich vor allem für ihre Geduld und ihr offenes Ohr bei zahlreichen sachlichen und unsachlichen Gesprächen rund um meine Dissertation.

Karlsruhe, im August 2015

List of Abbreviations and Variables

ABS	Agent based Simulation
ADS	Agent based Model
	Alternating Current
AU	Alternating Current
	Arcentarive Fuel Venicle
	Alternative Automobiles Diffusion and Infractive
ALADIN	(ALADIN-model)
ASTRA	Assessment of Transport Strategies (ASTRA-model)
BEV	Battery Electric Vehicle
BMWE	Federal Ministry of Economics Affairs and Energy
	(Bundesministerium für Wirtschaft und Energie; former
	Federal Ministry of Economic Affairs and Technology,
	Bundesministerium für Wirtschaft und Technologie)
BR	Brazil
CAGR	Compound Annual Growth Rate
CH	Switzerland
CNG	Compressed Natural Gas
CO_2	Carbon Dioxide
CP	Charging Point
CPO	Charging Point Operator
CV	Conventional Vehicle
DC	Direct Current
DE	Germany
EC	European Commission
ENTD	National survey on transport and movements (Enquête
	Nationale des Transports et des Déplacements)
EREV	Extended-range Electric Vehicle (see also REEV)
FCEV	Fuel Cell Electric Vehicle
GHG	Greenhouse Gas
GPS	Global Positioning System
HEV	Hybrid Electric Vehicle
IT	Italy
KBA	Federal Motor Transport Authority
	(Kraftfahrt-Bundesamt)
KiD	Motor Traffic in Germany (Kraftfahrzeugverkehr in
	Deutschland)
LCV	Light Commercial Vehicle
U V	Light Commercial venicle

LPG	Liquefied Petroleum Gas
LPGV	Liquefied Petroleum Gas Vehicle
MAS	Multi-agent Simulation
MiD	Mobility in Germany (Mobilität in Deutschland)
MOP	German Mobility Panel (Deutsches Mobilitätpanel)
MOPS	Mobility Panel for the region of Stuttgart
NGV	Natural Gas Vehicle
NHTS	National Household Travel Survey
NL	Netherlands
PEV	Plug-in Electric Vehicle
PHEV	Plug-in Hybrid Electric Vehicle
REEV	Range-Extended Electric Vehicle (see also EREV)
REM2030	Regional Eco Mobility commercial vehicle driving profiles
REM2030S	Regional Eco Mobility commercial vehicle driving profiles
	for the region of Stuttgart
SCMD	Swedish Car Movement Data
SD	Standard Deviation
SOC	State of Charge
TCO	Total Cost of Ownership
VAT	Value Added Tax
VKT	Vehicle Kilometers Traveled
VRI	Vehicle to Refueling station Index
WTPM	Willingness To Pay More

Symbol	Unit	Explanation
a	years	Vehicle age
$a_{cp}(t)$	EUR/a	Annuity for charging point in year t
$a_{i,s}^{\mathrm{CI,opex}}(t)$	EUR/a	Annuity of operating expenditure for
-) -		individual charging infrastructure in year t
$a_{i,s}^{\text{CP,opex}}(t)$	EUR/a	Annuity of operating expenditure for public
-,		charging point in year t
$a_{i,s}^{\text{veh,capex}}(t)$	EUR/a	Annuity of capital expenditure for vehicle
$a_{is}^{\text{veh,opex}}(t)$	EUR/a	Annuity of operating expenditure for vehicle
$ACD_i(t)$	km	Average commuting distance of user i in year
. ,		t
A_z	$\rm km^2$	Area of zone z
b_i	-	Brand of the vehicle i
$c_{r,s}^c(t)$	l/km	Conventional vehicle consumption
$c_{r,s}^{e'}(t)$	$\rm kWh/km$	Electric vehicle consumption
$C_i^e(t)$	EUR/a	Annual cost for energy charged by user i in
		year t
$C_{r,s}(t)$	kWh	Net capacity of vehicle with vehicle size r
		and propulsion technology s in year t
CPN_z	$\rm CP/km^2$	Minimal number of charging points in zone z
$d(\Delta \tau)$	km	Distance driven in time period $\Delta \tau$

$d_{i,s}^{el}(t)$	km	Kilometers driven in electric mode by user i
	_	with propulsion system s in year t
d_i	km	Kilometers driven by user i
$dep_{i,s}^{\text{ven,capex}}(t)$	EUR/a	Depreciation for vehicle capital expenditure of user i for propulsion system s
$dep_{i}^{\text{veh,opex}}(t)$	EUR/a	Depreciation for vehicle operating
··· <i>F</i> 1,8	_ 0 _ 0/ 0	expenditure of user i for propulsion system s
DR	-	Depreciation rate
$f_{m,a}(t)$	-	Share of driving profiles of users of group m
J 111,8 (*)		and propulsion system s in year t
$f_{ims}(t)$	_	Share propulsion system s in driving profile i
J 1,111,5 (*)		and group m in year t
$q_{i}^{\mathrm{veh}}(t)$	EUR/a	Fringe benefit tax for company cars for user i
51,8 (1)	/	and propulsion system s in year t
i	_	Index for driving profile
$I^{CI}(t)$	EUR	Investment for individual charging
- (*)		infrastructure
$I^{CP}(t)$	EUR	Investment for public charging points
$ITR_i(t)$	-	Income tax rate of user i in year t
$k_{r}^{c}(t)$	EUR/l	Costs for conventional fuel
$k_{n,s}^{\text{OM}}(t)$	EUR/km	Costs for operations and maintenance for
<i>T</i> , <i>S</i> (1)	/	vehicle of size r and propulsion system s in
$1 \tan(1)$		year t
$k_{r,s}^{\mathrm{tax}}(t)$	EUR/a	Vehicle tax for vehicle of size r and
1		propulsion system s in year t
l	-	Location type index; $l \in \{\text{domestic}, $
T()		commercial, work, public}
L(a)	-	Vehicle survival probability until year a
$lim_{m,s}(t)$	-	Limited availability of vehicle in group m for
$IDG(\mu)$	EUD	propulsion system s in year t
$LP_{r,s}^{\odot}(t)$	EUR	Gross list price of vehicle of size r and
$\mathbf{ID}(\mathbf{i})$	EUD	propulsion technology s in year t
$LP_{r,s}(t)$	EUR	Net list price of vehicle of size r and
$\mathbf{T} \mathbf{D} PEV(\mathbf{u})$	EUD	propulsion technology s in year t
$LP_{r,s}^{r,s}(t)$	EUR	Net list price of vehicle of size r and
		PEV
$n_{cp}(t)$	-	Number of public charging points in year t
$n_{CP,z}(t)$	-	Number of public charging point in year t
		and zone z
$n_m(t)$	-	Vehicle registrations of group m in year t
$N_{m,s}(t)$	-	Vehicle registrations of group m and
		propulsion technology s in year t
m	-	Index for vehicle group; $m \in \{\text{private small}, $
		private medium, private large, fleet small,
		fleet medium, fleet large, fleet LCV, company
		small, company medium, company large}

OCC_z	Vehicle min $parked / (lm^2)$	Specific zone occupancy of zone z
$occ_{z,PEV}(t)$	Vehicle min	Specific zone occupancy of zone z by PEVs
	$parked/(km^2 \cdot v)$	wk)
$p_s^{batt}(t)$	EUR/kWh	Battery price for vehicle of propulsion
-car(1)	FUD	technology s in year t
$p_{r,s}\left(t\right)$	LUK	and propulsion technology s in year t
$p_{cn}(t)$	EUR	Price for a public charging point in year t
$p_{cl}(t)$	EUR/kWh	Price for electricity at a public charging
Pet(0)	2010/11/11	point in year t
$p_{i,l}(t)$	EUR/kWh	Charging price for user i at location l in year
		t
$p_{pc}(t)$	EUR/kWh	Price for public charging in year t
$P_l(\tau, t)$	kW	Power for charging at the location l where
		vehicle i was parked at τ and year t
$P_{p,z}(t)$	kW	Power for public charging in zone z at time t
$P^{\mathrm{scrap}}(a)$	-	Probability of scrapping a vehicle of age a
r	-	Index for vehicle size; $r \in \{\text{small, medium,}\}$
		large, LCV}
s	-	Index for propulsion technology;
		$s \in \{\text{Gasoline, Diesel, PHEV, BEV}\}$
$s_{i,s}(t)$	_	Electric driving share of vehicle <i>i</i> with
1,3 (1)		propulsion system s
$S_{m,s}(t)$	-	Vehicle stock of vehicle group m and
~ 111,3 (*)		propulsion technology s in year t
$SOC_i(\tau, t)$	kWh	Battery's state of charge of vehicle i at time
		τ in year t
$SP_{is}(t)$	EUR	Resale price of vehicle i and propulsion
		technology s in year t
t	vears	Year of observation
$T^{CI}(t)$	vears	Investment horizon of charging infrastructure
-u (°)	J 00120	for user group u in year t
T^{dlim}	vears	Maximum depreciation period of vehicles
$T^{veh}(t)$	vears	Investment horizon of vehicles for user group
u (°)	Jeans	u in year t
$TCO^{a, veh}(t)$	EUR/a	Annual total cost of ownership for vehicle i
1,8 (*)	_ 0 _ 0/ 0	and propulsion technology s in year t
$TCO^{a,CI}(t)$	EUR/a	Annual total cost of ownership for individual
$1 \bigcirc 0_{i,s}$ (c)	LOIt/ a	charging infrastructure for vehicle i and
		propulsion technology s in year t
$TE^{PEV(t)}$	EUB	Amount of tax exemption for PEVs of size r
$L_{r,s}$ (b)		and propulsion technology s in yoar t
21	_	Index for user group: $u \in \int \text{private floot}$
u		$\begin{array}{l} \text{matrix for user group, } u \in \{\text{private, neet,} \\ \text{company} \} \end{array}$
$u^a(t)$	EUR /a	Utility value of propulsion technology a for
$u_{i,s}(v)$	LUIL/a	user i in year t
		user e m year e

$use_z(t)$	kWh/yr	Usage of public charging points in zone z and year t
VKT_i	km/a	Annual vehicle kilometers traveled by vehicle i
$W_{i,l}(t)$	kWh/a	Annual amount of energy charged by vehicle i at location l in year t
$W_{i,l}(\tau, \tau+1, t)$	kWh/a	Amount of energy charged by vehicle i at location l in year t between τ and $\tau + 1$
$W_{pc}(t)$	kWh/a	Total amount of energy charged at public charging points in year t
$\operatorname{WTPM}_{i,s}^{a}(t)$	EUR/a	Annuity of willingness to pay more by the owner of vehicle i for propulsion technology s in year t
$\operatorname{wtpm}_{i,s}(t)$	-	Willingness to pay more by the owner of vehicle i for propulsion technology s in year t as a percentage compared to a conventional vehicle
z	-	Index for zones
$z_u(t)$	-	Interest rate for user group u in year t
α_1	-	Parameter for calculation of residual values
β_1	-	Parameter for calculation of residual values (age)
β_2	-	Parameter for calculation of residual values (mileage)
β_3	-	Parameter for calculation of residual values (list price)
β	-	Parameter for stock calculation (shape)
Δn_{cp}	-	Difference between charging points between two periods of time t
$\Delta \tau$	min	Time period between τ_1 and τ_2
$\kappa_{r,s}(t)$	kWh	Gross battery capacity of vehicle with size r and propulsion technology s in year t
au	min	Time section in observation period
$ au_i^0$	min	Initial point in time of observation of vehicle i
$ \begin{array}{c} \tau_i^{max} \\ \theta \end{array} $	min -	Last point in time of observation of vehicle i Parameter for stock calculation (scale)

List of Figures

2.1	Alternative fuel vehicles considered in this thesis.	6
2.2	Characteristics of charging infrastructure considered in this thesis	9
2.3	Availability of garages in German vehicle stock.	10
2.4	Stock market share of different gas vehicles and corresponding refueling	
	stations for different countries and years	12
2.5	Classification of models	16
3.1	Comparison of different driving profile data sets	29
$3.2 \\ 3.3$	Observation area of MOPS	32
	data sets	33
3.4	Comparison of daily VKT in different commercial driving profile data sets	37
4.1	Overview of the model ALADIN	45
4.2	Criteria ranked first in the decision making process for passenger car pur-	
	chase differentiated by vehicle size	46
4.3	Example for battery simulation	48
4.4	Algorithm for the inclusion of the limited availability	53
4.5	Example for PEV stock simulation	54
4.6	Algorithm for the construction and deconstruction of public charging points.	56
4.7	Composition of vehicle costs for different drive trains in 2020	57
4.8	Limited availability of PEVs	65
4.9	Electric driving shares over annual VKT for medium sized vehicles of five	
	different samples	69
4.10	Diesel market shares within different commercial branches in Germany	71
5.1	Mean and variation of log-normal distribution for private driving profiles	
	(MOP)	80
5.2	Mean and variation of log-normal distribution for company car driving	
	profiles (MOP)	80
5.3	Mean and variation of log-normal distribution for commercial fleet driving	
	profiles (REM2030)	80
5.4	Differences between individual utilities of best plug-in electric and conven-	
	tional vehicle	85
5.5	Results for the three PEV market diffusion scenarios for Germany	91
5.6	Simulation results for vehicle registrations in 2015–2030 in the medium	
	scenario	93
5.7	Different inventory calculations of individual charging point stock	94
5.8	Sensitivities to parameter variation	95

5.9	Monte-carlo-simulation for variation of scenario parameters
5.10	Influence of non-monetary factors on PEV market diffusion
5.11	PEV stock development including substitute vehicles for long-distance trips.100
5.12	Influence of individual charging infrastructure on PEV market diffusion 101
5.13	Specific zone occupancy and minimum number of public charging points in
	different zones in the region of Stuttgart
5.14	Simulation results for Germany for PEV and public CP stock with different
	subsidies
5.15	Simulation results for the region of Stuttgart for energy consumed at public
	CPs and public CP stock with different subsidies
5.16	Simulation results for the region of Stuttgart for energy consumed by PEV
	stock over one week in 2030
5.17	Simulation results for the region of Stuttgart for public CP stock over time
	and energy consumed at public CPs in 2030 with respect to distance to
	city center
5.18	Simulation results for the region of Stuttgart for PEV and total vehicle
	stock with respect to distance of home (private) or company (commercial)
F 10	to city center
5.19	Simulation results for Germany for PEV and public CP stock with variation
F 00	of charging power
5.20	Simulation results for Germany for PEV and public CP stock with variation
5.91	of infrastructure availability
0.21	of recharging level or initial public charging price
ຮ່າງ	Simulation regults for Cormony for DEV and public CD stock with variation
0.22	of charging point solup strategies
5 93	Simulation results for Cormany for PEV and public CP stock with variation
0.20	of BEV battery sizes
5 24	Belation between subsidy to public charging points and correlation of timely
0.24	evolution of PEV and public CP diffusion 123
5.25	Comparison of simulation results.
J. _ J	
A.1	Distance and area of zones in observation area of MOPS

List of Tables

2.1	International PEV adoption in major car markets.	7
2.2	Overview of analyzed models for AFVs and their refueling infrastructure .	17
2.3	Analysis of different models for AFVs and their refueling infrastructure	21
2.4	Analysis of models for AFVs and their refueling infrastructure - part 1	24
2.5	Analysis of models for AFVs and their refueling infrastructure - part 2	25
3.1	Privately and commercially licensed passenger cars in Germany	30
3.2	Overview of private driving profiles for Germany	30
3.3	Geographic zones of observation area of MOPS.	32
3.4	Annual VKT in different vehicle sizes of private driving profiles for Germany.	34
3.5	Common parking spot of private vehicles for Germany	35
3.6 3.7	Overview of commercial driving profiles for Germany	35
	tiated by vehicle sizes.	38
3.8	Driving distances distinguished by commercial branches	39
4.1	Scenario-specific parameters used in the model ALADIN	58
4.2	Cost for charging infrastructure options.	59
4.3	Vehicle registrations and general cost parameters	60
4.4	Definition of private adopter groups and their willingness to pay more	63
4.5	Vehicle registrations in region of Stuttgart and corresponding driving profiles.	66
4.6	Comparison of assignment of company cars in MOPS and MOP	67
4.7	Comparison of electric driving shares and annual VKT	69
5.1	Consideration of scenarios in results.	78
5.2	Summary statistics for daily VKT of private and commercial vehicles	81
5.3	Statistical comparison of driving behavior in different user groups	81
5.4	PEV market potential of different user groups	82
5.5	PEV stock in 2030 differentiated by scenario, user group, vehicle size and PEV type.	91
5.6	Cost and subsidies for public CPs with a charging power of 3.7 kW in three	
	subsidy scenarios.	107
5.7	Comparison of results with data for Germany and Stuttgart	121
5.8	Financial policy options modeled.	124
A.1	Changes between observation waves in data set MOP	139
B.1	Parameters for small vehicles.	145
B.2	Parameters for medium sized vehicles.	146

B.3	Parameters for large vehicles
B.4	Parameters for LCVs
B.5	Limited availability of PEVs
B.6	Cost and subsidies for public CPs with a charging power of 22 kW and
	50 kW

Bibliography

- [ADAC, 2013] ADAC (2013). ADAC Autokosten-Rechner. Munich, Germany. 61
- [AGEB, 2015] AGEB (2015). "Energieverbrauch dank milder Witterung deutlich gesunken", press announcement No. 1/2015. Arbeitsgemeinschaft Energiebilanzen, Berlin, Germany. 109
- [Al-Alawi and Bradley, 2013] Al-Alawi, B. M. and Bradley, T. H. (2013). Review of hybrid, plug-in hybrid, and electric vehicle market modeling studies. *Renewable and Sustainable Energy Reviews*, 21:190–203. 2, 15, 77, 79, 88
- [Allen, 1988] Allen, D. (1988). New telecommunications services: Network externalities and critical mass. *Telecommunications Policy*, 12(3):257–271. 4, 19
- [Amjad et al., 2011] Amjad, S., Rudramoorthy, R., Neelakrishnan, S., Sri Raja Varman, K., and Arjunan, T. V. (2011). Impact of real world driving pattern and all-electric range on battery sizing and cost of plug-in hybrid electric two-wheeler. *Journal of Power Sources*, 196:3371–3377. 2, 27, 77
- [ANP, 2012] ANP (2001-2012). Anuario esta estatistico brasileiro do petroleo, gas natural e biocombustiveis. 11, 12
- [Auto21, 2011] Auto21 (2011). Monitoring driving behaviour for Plug-in Hybrid Electric Vehicle (PHEV) research. Technical report, Auto21, Winnipeg, Canada. 3, 28, 29
- [Babrowski et al., 2014] Babrowski, S., Heinrichs, H., Jochem, P., and Fichtner, W. (2014). Load shift potential of electric vehicles in Europe. *Journal of Power Sources*, 255(0):283 – 293. 8
- [Bain and Englehardt, 1991] Bain, L. and Englehardt, M. (1991). Statistical analysis of reliability and life-testing models: theory and methods, volume 115. CRC Press, Boca Raton, US. 52
- [Bakker, 2010] Bakker, S. (2010). The car industry and the blow-out of the hydrogen hype. *Energy Policy*, 38(11):6540 – 6544. Energy Efficiency Policies and Strategies with regular papers. 1
- [Bakker and Trip, 2013] Bakker, S. and Trip, J. J. (2013). Policy options to support the adoption of electric vehicles in the urban environment. Transportation Research Part D: Transport and Environment, 25(0):18 – 23. 123
- [Ball et al., 2009] Ball, M., Weindorf, W., Bünger, U., and Wietschel, M. (2009). Hydrogen distribution. In *The Hydrogen Economy: Opportunities and Challenges*, pages 322–347. Cambridge University Press, New York, USA. 18

- [Batley et al., 2004] Batley, R. P., Toner, J. P., and Knight, M. J. (2004). A mixed logit model of UK household demand for alternative-fuel vehicles. *International Journal of Transport Economics*, 31(1):55–77. 89
- [BCG, 2009] BCG (2009). The Boston Consulting Group: The comeback of the electric car How real, How soon, and What Must Happen Next. Technical report. 2, 77
- [BCG, 2013] BCG (2013). Boston Consulting Group (BCG): Trendstudie 2030+ Kompetenzinitiative Energie des BDI. Study of the Boston Consulting Group in behalf of the Federation of German Industry (BDI). München. 58
- [BDEW, 2014] BDEW (2014). German Association of Energy and Water Industries (BDEW): BDEW-Strompreisanalyse Juni 2014 Haushalte und Industrie. 58
- [Becker, 2009] Becker, T. A. (2009). Electric Vehicles in the United States A New Model with Forecasts to 2030. 8
- [Behrends and Kott, 2009] Behrends, S. and Kott, K. (2009). Zuhause in Deutschland Ausstattung und Wohnsituation privater Haushalte. Technical report, German Statistical Office, Wiesbaden, Germany. 8, 34
- [Berg, 1985] Berg, M. R. (1985). The potential market for electric vehicles: results from a national survey of commercial fleet operators. *Transportation Research Record*, (1049). 79, 86
- [BFS, 2013] BFS (2013). Bundesamt für Statistik der Schweizerischen Eidgenossenschaft: PC-Axis databases, Switzerland. www.pxweb.bfs.admin.ch/Dialog/statfile.asp?lang=1. 11, 12
- [Björnsson and Karlsson, 2015] Björnsson, L.-H. and Karlsson, S. (2015). Plug-in hybrid electric vehicles: How individual movement patterns affect battery requirements, the potential to replace conventional fuels, and economic viability. *Applied Energy*, 143(0):336 – 347. 102
- [Blum, 2014] Blum, A. (2014). Electro-mobility: Statistical analysis of human mobility patterns. Master's thesis, Infernum, Wuppertal. 28
- [BMF, 2001] BMF (2001). Federal Ministry of Finance (BMF): Tables for depreciation (AfA-Tabelle): IV D 2 - S 1551 - 498/01, BStBl 2001 I S. 860. 50
- [BMF, 2014] BMF (2014). Federal Ministry of Finance (BMF): Taxation of passenger cars. (Übersicht zur Kraftfahrzeugsteuer für Personenwagen.). 61, 145, 146, 147
- [BMF, 2015] BMF (2015). Federal Ministry of Finance (BMF): Taxation of company cars: EStG6 Bewertung, EStG8 Einnahmen. 51, 61
- [BMWE, 2015] BMWE (2015). Verordnungsentwurf des Bundesministeriums für Wirtschaft und Energie vom 9. Januar 2015: "Verordnung über technische Mindestanforderungen an den sicheren und interoperablen Aufbau und Betrieb von öffentlich zugänglichen Ladepunkten für Elektromobile (Ladesäulenverordnung LSV)". Berlin, Germany. 2, 3, 120, 131

- [BMWi and BMU, 2010] BMWi and BMU (2010). Energy Concept for a environmentally compatible, sustainable and affordable energy supply. (Energiekonzept für eine umweltschonende, zuverlässige und bezahlbare Energieversorgung). Technical report, Federal Ministry of Economic Affairs and Technology (BMWi), Federal Ministry for the Environment, Nature Conservation and Nuclear Safety (BMU), Berlin, Germany. 1, 2
- [Bočkarjova et al., 2014] Bočkarjova, M., Rietveld, P., Knockaert, J., and Steg, L. (2014). Dynamic Consumer Heterogeneity in Electric Vehicle Adoption. *Innovation*, 3:4. 89
- [Bonabeau, 2002] Bonabeau, E. (2002). Agent-based modeling: Methods and techniques for simulating human systems. Proceedings of the National Academy of Sciences of the United States of America, 99(Suppl 3):7280–7287. 22, 43, 44, 72, 131
- [Bradley and Daly, 1991] Bradley, M. A. and Daly, A. J. (1991). Estimation of logit choice models using mixed stated preference and revealed preference information. In Les methodes d'analyse des comportements de deplacements pour les annes 1990 - 6e conference internationale sur les comportements de deplacements, chateau bonne entente, Quebec, 22-24 May 1991, vol. 1. 62
- [Brown et al., 2001] Brown, L. D., Cai, T. T., and DasGupta, A. (2001). Interval estimation for a binomial proportion. *Statistical Science*, 16:101–117. 90
- [Brown, 2013] Brown, M. (2013). Catching the PHEVer: Simulating Electric Vehicle Diffusion with an Agent-Based Mixed Logit Model of Vehicle Choice. Journal of Artificial Societies and Social Simulation, 16(2):5. 89
- [Brownstone et al., 2000] Brownstone, D., Bunch, D. S., and Train, K. (2000). Joint mixed logit models of stated and revealed preferences for alternative-fuel vehicles. *Transportation Research Part B: Methodological*, 34(5):315 – 338. 89
- [Bruce et al., 2012] Bruce, I., Butcher, N., and Fell, C. (2012). Lessons and Insights from Experience of Electric Vehicles in the Community. In *Proceedings of Electric Vehicle* Symposium 26 (EVS 26), Los Angeles, US. 2
- [Calmon, F., 2013] Calmon, F. (2013). Brasil chega aos 20 milhões de motores flex, diz Anfavea. 6
- [Capros et al., 2013] Capros, P., Vita, A. D., Tasios, N., et al. (2013). EU Energy Trends to 2050. Luxembourg. 58
- [Chan, 2007] Chan, C. C. (2007). The state of the art of electric, hybrid, and fuel cell vehicles. In *Proceedings of the IEEE*, volume 95, pages 704–718. 1, 3, 5
- [Chen et al., 2013] Chen, T. D., Kockelman, K. M., Khan, M., et al. (2013). Locating Electric Vehicle Charging Stations – Parking-Based Assignment Method for Seattle, Washington. In *Transportation Research Board of the National Academies, Washington* D.C., volume 340, pages 28–36. 2, 77

- [Chlond et al., 2014] Chlond, B., Weiss, C., Heilig, M., and Vortisch, P. (2014). Hybrid Modeling Approach of Car Uses in Germany on Basis of Empirical Data with Different Granularities. *Transportation Research Record: Journal of the Transportation Research Board*, 2412(-1):67–74. 31, 135
- [Collantes and Melaina, 2011] Collantes, G. and Melaina, M. W. (2011). The co-evolution of alternative fuel infrastructure and vehicles: A study of the experience of Argentina with compressed natural gas. *Energy Policy*, 39(2):664–675. 13
- [Cont, 2001] Cont, R. (2001). Empirical properties of asset returns: stylized facts and statistical issues. *Quantitative Finance*, 1(2):223–236. 10
- [Continental, 2011] Continental (2011). Continental-Mobilitätsstudie 2011. Continental AG, Hanover, Germany. 14
- [Dagsvik et al., 2002] Dagsvik, J., Wennemo, T., Wetterwald, D., and Aaberge, R. (2002). Potential demand for alternative fuel vehicles. *Transportation Research Part B: Method-ological*, 36(4):361–384. 2, 3, 27, 77
- [Dallinger et al., 2013] Dallinger, D., Gerda, S., and Wietschel, M. (2013). Integration of intermittent renewable power supply using grid-connected vehicles - A 2030 case study for California and Germany. *Applied Energy*, 104(0):666–682. 7
- [Dallinger and Wietschel, 2012] Dallinger, D. and Wietschel, M. (2012). Grid integration of intermittent renewable energy sources using price-responsive plug-in electric vehicles. *Renewable and Sustainable Energy Reviews*, 16(5):3370 – 3382. 7, 8, 58, 111, 133
- [DAT, 2011] DAT (2011). DAT report 2011. Technical report, Deutsche Automobil Treuhand (DAT), Ostfildern, Germany. 30
- [Dataforce, 2011] Dataforce (2011). Elektrofahrzeuge in deutschen Fuhrparks Zur künftigen Bedeutung von Elektrofahrzeugen in deutschen Flotten. Technical report, Dataforce Verlagsgesellschaft für Business Informationen, Frankfurt a.M., Germany. 46, 64, 73, 85
- [de Haan et al., 2009] de Haan, P., Mueller, M. G., and Scholz, R. W. (2009). How much do incentives affect car purchase? Agent-based microsimulation of consumer choice of new cars–Part II: Forecasting effects of feebates based on energy-efficiency. *Energy Policy*, 37(3):1083–1094. 2, 44
- [Deneke, 2005] Deneke, K. (2005). Nutzungsorientierte Fahrzeugkategorien im Strassenwirtschaftsverkehr. Number 53 in Schriftenreihe des Instituts für Verkehr und Stadtbauwesen, TU Braunschweig, Germany. 29
- [Dexheimer, 2003] Dexheimer, V. (2003). Hedonic methods of price measurement for used cars. Technical report, German Statistical Office, Wiesbaden, Germany. 50
- [Diamond, 2009] Diamond, D. (2009). The impact of government incentives for hybridelectric vehicles: Evidence from US states. *Energy Policy*, 37(3):972 – 983. 123

- [Dieckhoff et al., 2014] Dieckhoff, C., Appelrath, H.-J., Fischedick, M., Grunwald, A., Höffler, F., Mayer, C., and Weimer-Jehle, W. (2014). Zur Interpretation von Energieszenarien. Energiesysteme der Zukunft, Project of National Academy of Sciences Leopoldina, acatech - German National Academy of Science and Engineering, Union of the German Academies of Sciences and Humanities. 56
- [Dimitropoulos et al., 2011] Dimitropoulos, A., Rietveld, P., and van Ommeren, J. N. (2011). Consumer valuation of driving range: A meta-analysis. Technical report, Tinbergen Institute Discussion Paper. 63
- [Doebling et al., 1996] Doebling, S. W., Farrar, C. R., Prime, M. B., and Shevitz, D. W. (1996). Damage identification and health monitoring of structural and mechanical systems from changes in their vibration characteristics: a literature review. Technical report, Los Alamos National Lab., NM (United States). 5
- [Dong et al., 2014] Dong, J., Liu, C., and Lin, Z. (2014). Charging infrastructure planning for promoting battery electric vehicles: An activity-based approach using multiday travel data. *Transportation Research Part C: Emerging Technologies*, 38:44–55. 2, 40, 77, 123
- [Dreher, 2001] Dreher, M. (2001). Analyse umweltpolitischer Instrumente zur Förderung der Stromerzeugung aus regenerativen Energieträgern im liberalisierten Strommarkt: eine Untersuchung unter technischen, ökonomischen und umweltrelevanten Gesichtspunkten am Beispiel der Region Baden-Württemberg. PhD thesis, Karlsruhe, Univ., Diss., 2001. 16
- [Duleep et al., 2011] Duleep, G., Van Essen, H., Kampan, B., and Grünig, M. (2011). Impacts of electric vehicles - deliverable 2: Assessment of electric vehicle and battery technology. Technical Report 11.4058.04, ICF, CE Delft, Ecologic, Delft, Netherlands. 89
- [Dütschke et al., 2013] Dütschke, E., Paetz, A.-G., and Wesche, J. (2013). Integration Erneuerbarer Energien durch Elektromobilität? Inwieweit sind Konsumenten bereit, einen Beitrag zu leisten? *uwf UmweltWirtschaftsForum*, 21:233–242. 63
- [Dütschke et al., 2011a] Dütschke, E., Schneider, U., Peters, A., Paetz, A.-G., and Jochem, P. (2011a). Moving towards more efficient car use – what can be learnt about consumer acceptance from analysing the cases of LPG and CNG? In *Proceedings of the* 2011 ECEEE summer study, Toulon, France. 13, 14, 62
- [Dütschke et al., 2011b] Dütschke, E., Schneider, U., Sauer, A., Wietschel, M., Hoffmann, J., and Domke, S. (2011b). Roadmap zur Kundenakzeptanz - Zentrale Ergebnisse der sozialwissenschaftlichen Begleitforschung in den Modellregionen. Technical report, Fraunhofer ISI, Federal Ministry of Transport, Building and Urban Development (Bundesministerium für Verkehr, Bau und Stadtentwicklung) (BMVBS), Berlin, Germany. 2, 7, 14, 62, 73, 77
- [Easterly and Levine, 2001] Easterly, W. and Levine, R. (2001). What have we learned from a decade of empirical research on growth? It's Not Factor Accumulation: Stylized Facts and Growth Models. *The World Bank Economic Review*, 15(2):177–219. 10

- [EC, 2009] EC (2009). European Commission: Regulation (EC) No 443/2009 of the European Parliament and of the Council of 23 April 2009 setting emission performance standards for new passenger cars as part of the Community's integrated approach to reduce CO_2 emissions from light-duty vehicles. 1, 60, 61, 133
- [EC, 2014] EC (2014). European Comission: Directive 2014/94/EU of the European Parliament and of the Council of 22 October 2014 on the deployment of alternative fuels infrastructure. 2, 3, 131
- [Ecotality and INL, 2012] Ecotality and INL (2012). The EV Project Q1 2012 Report. Technical report, Ecotality Inc. and Idaho National Lab, US. 2
- [Efron and Tibshirani, 1994] Efron, B. and Tibshirani, R. J. (1994). An introduction to the bootstrap. CRC press, Boca Raton, US. 122
- [Egbue and Long, 2012] Egbue, O. and Long, S. (2012). Barriers to widespread adoption of electric vehicles: An analysis of consumer attitudes and perceptions. *Energy Policy*, 48(0):717 – 729. Special Section: Frontiers of Sustainability. 2, 77
- [eldia.com, 2011] eldia.com (2011). La cantidad de estaciones de servicio se redujo mas de 30%. http://www.eldia.com.ar/edis/20110326/la-cantidad-estaciones-servicio-redujomas-20110326111347.htm. 11, 12
- [Elgowainy et al., 2012] Elgowainy, A., Zhou, Y., Vyas, A., Mahalik, M., Santini, D., and Wang, M. (2012). Impacts of Charging Choices for Plug-In Hybrid Electric Vehicles in 2030 Scenario. 2287:9–17. 89
- [ENTD, 2009] ENTD (2009). Enquête nationale transports et déplacements (ENTD) 2008. Technical report, Ministry of Ecology, Sustainable Development and Energy (Ministère de l'Ecologie, du Dévelopment durable et de l'Energie). 3, 28, 29
- [Eppstein et al., 2011] Eppstein, M. J., Grover, D. K., Marshall, J. S., and Rizzo, D. M. (2011). An agent-based model to study market penetration of plug-in hybrid electric vehicles. *Energy Policy*, 39(6):3789–3802. 2, 62, 85, 89
- [Erdöl-Vereinigung der Schweiz, 2008] Erdöl-Vereinigung der Schweiz (2008). Jahresbericht. Zurich, Switzerland. 11, 12
- [ESMT, 2011] ESMT (2011). Marktmodell elektromoblität schlussbericht. Technical report, European School of Management and Technology(ESMT), Berlin, Germany. 57, 73
- [europia, 2012] europia (2005-2012). European Petroleum Industry Association Annual Reports. Brussels, Belgium. 11, 12
- [Eurostat, 2008] Eurostat (2008). NACE Rev. 2 Statistical classification of economic activities in the European Community. Technical report, Eurostat - Statistical Office of the European Communities, Luxembourg. 38, 71, 142
- [Eurostat, 2013] Eurostat (2013). Passenger Transport Statistics Statistics Explained. http://epp.eurostat.ec.europa.eu/statistics_explained/index.php/ Passenger_transport_statistics, Luxembourg. 11, 12

- [Fahrmeir et al., 2011] Fahrmeir, L., Künstler, R., Pegeot, I., and Tutz, G. (2011). Statistik – Der Weg zur Datenanalyse. Springer, Berlin, Germany. 34, 82, 84, 90
- [Fischer et al., 2009] Fischer, M., Werber, M., and Schwartz, P. V. (2009). Batteries: Higher energy density than gasoline? *Energy Policy*, 37(7):2639 – 2641. 8
- [Fleiter et al., 2011] Fleiter, T., Worrell, E., and Eichhammer, W. (2011). Barriers to energy efficiency in industrial bottom-up energy demand models-a review. *Renewable* and Sustainable Energy Reviews, 15(6):3099–3111. 15, 16, 17
- [Flynn, 2002] Flynn, P. C. (2002). Commercializing an alternate vehicle fuel: lessons learned from natural gas for vehicles. *Energy Policy*, 30(7):613–619. 12
- [Follmer et al., 2010] Follmer, R., Gruschwitz, D., and Jesske, B. (2010). Mobilität in Deutschland 2008 Ergebnisbericht. infas Institut für angewandte Sozialwissenschaft, Institut für Verkehrsforschung des Deutschen Zentrums für Luft und Raumfahrt e.V., Berlin, Germany. 73
- [Fraunhofer ISI, 2014] Fraunhofer ISI (2014). REM2030 Driving Profiles Database V2014-07. Technical report, Fraunhofer Institute of Systems and Innovation Research ISI, Karlsruhe, Germany. 28, 29, 35, 36, 37, 38, 39, 59, 64, 80, 81, 86
- [Frondel et al., 2012] Frondel, M., Ritter, N., and Vance, C. (2012). Heterogeneity in the rebound effect: Further evidence for Germany. *Energy Economics*, 34(2):461 – 467. 133
- [Frühauf, 2012] Frühauf, K. (2012). Was kostet Autofahren im Vergleich zu 1980 wirklich?, Progenium GmbH & Co.KG, Berlin, Germany. 61
- [Funke et al., 2015] Funke, S., Gnann, T., and Plötz, P. (2015). Addressing the different needs for charging infrastructure: an analysis of some criteria for charging infrastructure set-up. In Leal, W. and Kotter, R., editors, *E-Mobility in Europe Trends and Good Practice.* Springer, London, UK. In press. 55, 105, 106
- [Funke and Plötz, 2014] Funke, S. and Plötz, P. (2014). A Comparison of Different Means to Increase Daily Range of Electric Vehicles - The Potential of Battery Sizing, Increased Vehicle Efficiency and Charging Infrastructure. In Proceedings of the Vehicle Power and Propulsion Conference 2014 (IEEE-VPPC 2014), Coimbra, Portugal. 61, 98
- [Garcia and Calantone, 2002] Garcia, R. and Calantone, R. (2002). A critical look at technological innovation typology and innovativeness terminology: a literature review. *Journal of product innovation management*, 19(2):110–132. 5
- [Gass et al., 2014] Gass, V., Schmidt, J., and Schmid, E. (2014). Analysis of alternative policy instruments to promote electric vehicles in austria. *Renewable Energy*, 61(0):96 101. World Renewable Energy Congress, 8-13 May 2011, Linköping, Sweden. 123
- [Geels, 2002] Geels, F. W. (2002). Technological transitions as evolutionary reconfiguration processes: a multi-level perspective and a case-study. *Research policy*, 31(8):1257– 1274. 16
- [Geroski, 2000] Geroski, P. (2000). Models of technology diffusion. Research Policy, 29(45):603-625. 15, 64, 88
- [GFK, 2012] GFK (2012). Hohe Kaufbereitschaft für Elektroautos. Studie der GfK Panel Services Deutschland zu Einstellungen und Akzeptanz von Elektrofahrzeugen. GfK, Landsberg a. Lech, Germany. 62
- [Glerum et al., 2013] Glerum, A., Stankovikj, L., Thémans, M., and Bierlaire, M. (2013). Forecasting the demand for electric vehicles: Accounting for attitudes and perceptions. *Transportation Science*, 48(4):483–499. 89
- [Globisch and Dütschke, 2013] Globisch, J. and Dütschke, E. (2013). Anwendersicht auf Elektromobilität in gewerblichen Flotten. Ergebnisse aus den Projekten mit gewerblichen Nutzern von Elektrofahrzeugen im Rahmen des BMVBS-Vorhabens "Modellregionen für Elektromobilität 2009-2011". Technical report, Fraunhofer ISI, Karlsruhe, Germany. 44, 70, 85
- [Globisch et al., 2013] Globisch, J., Schneider, U., and Dütschke, E. (2013). Acceptance of electric vehicles by commercial users in the electric mobility pilot regions in Germany. In *Proceedings of the 2013 ECEEE summer study*, Toulon, France. 7, 14
- [Gnann, 2010] Gnann, T. (2010). Thermische Vorkonditionierung von Elektrofahrzeugen. Diploma thesis, Karlsruhe Institute of Technology (KIT) and Fraunhofer ISI, Karlsruhe, Germany. 61
- [Gnann and Plötz, 2015] Gnann, T. and Plötz, P. (2015). A review of combined models for market diffusion of alternative fuel vehicles and their refueling infrastructure . *Renewable and Sustainable Energy Reviews*, 47(0):783 – 793. 10
- [Gnann et al., 2015a] Gnann, T., Plötz, P., Funke, S., and Wietschel, M. (2015a). What is the market potential of plug-in electric vehicles as commercial passenger cars? A case study from Germany. Transportation Research Part D: Transport and Environment, 37(0):171 – 187. 35, 61, 79, 86, 98
- [Gnann et al., 2013] Gnann, T., Plötz, P., and Haag, M. (2013). What is the future of public charging infrastructure for electric vehicles? - A techno-economic assessment of public charging points for Germany. In *Proceedings of the 2013 ECEEE summer study*, Toulon, France. 34, 51, 66
- [Gnann et al., 2012a] Gnann, T., Plötz, P., and Kley, F. (2012a). Vehicle charging infrastructure demand for the introduction of plug-in electric vehicles in Germany and the US. In *Proceedings of Electric Vehicle Symposium 26 (EVS 26)*, Los Angeles, US. 28, 36, 61, 68, 86, 89, 98, 102, 145, 146, 147
- [Gnann et al., 2015b] Gnann, T., Plötz, P., Kühn, A., and Wietschel, M. (2015b). Modelling market diffusion of electric vehicles with real world driving data German market and policy options. *Transportation Research Part A: Policy and Practice*, 77(0):95 – 112. 43, 47, 61, 73, 83, 88, 95, 97, 123, 126

- [Gnann et al., 2012b] Gnann, T., Plötz, P., and Wietschel, M. (2012b). Range Limits of Electric Vehicles: Invest in Charging Infrastructure or Buy Larger Batteries: A Techno-Economic Comparison. In ENERDAY 7th Conference on Energy Economics and Technology, Dresden, Germany. 98
- [Gnann et al., 2015c] Gnann, T., Plötz, P., and Wietschel, M. (2015c). How to address the chicken-egg-problem of electric vehicles? Introducing an interaction market diffusion model for EVs and charging infrastructure. In Accepted paper at the 2015 ECEEE summer study, Toulon, France. 105
- [Gnann et al., 2012c] Gnann, T., Plötz, P., Zischler, F., and Wietschel, M. (2012c). Elektromobilität im Personenwirtschaftsverkehr - eine Potenzialanalyse. Working paper, Fraunhofer ISI, Karlsruhe, Germany. 28, 79, 86
- [Golob et al., 1997] Golob, T. F., Torous, J., Bradley, M., Brownstone, D., Crane, S. S., and Bunch, D. S. (1997). Commercial fleet demand for alternative-fuel vehicles in California. *Transportation Research Part A: Policy and Practice*, 31(3):219 – 233. 70, 79
- [Götz et al., 2011] Götz, K., Sunderer, G., Birzle-Harder, B., and Deffner, J. (2011). Attraktivität und Akzeptanz von Elektroautos: Ergebnisse aus dem Projekt OPTUM Optimierung der Umweltentlastungspotenziale von Elektrofahrzeugen. Number 18 in ISOE-Studientexte. 7, 14, 73
- [Greene, 1985] Greene, D. L. (1985). Estimating daily vehicle usage distributions and the implications for limited-range vehicles. *Transportation Research Part B: Methodological*, 19(4):347 – 358. Special Issue Economic Models of Automobile Demand. 5, 28, 40
- [Greene, 1996] Greene, D. L. (1996). Survey evidence on the importance of fuel availability to the choice of alternative fuels and vehicles. *Energy studies review*, 8:215–231. 5, 13
- [Grube, 2004] Grube, M. (2004). Tankstellengeschichte in Deutschland. http://www.geschichtsspuren.de/artikel/34-verkehr/138-tankstellengeschichte.html. 3
- [Grübler, 1990] Grübler, A. (1990). The rise and fall of infrastructures: dynamics of evolution and technological change in transport. Physica Verlag, Heidelberg, Germany. 19
- [GVR, 2012] GVR (2012). The GVR Gas Vehicle Report. NGV Journal. 11, 12
- [Hacker et al., 2011a] Hacker, F., Harthan, R., Hermann, H., Kasten, P., Loreck, C., Seebach, D., Timpe, C., Zimmer, W., Leppler, S., and Möck, A. (2011a). Betrachtung der Umweltentlastungspotenziale durch den verstärkten Einsatz von kleinen, batterieelektrischen Fahrzeugen im Rahmen des Projekts "E-Mobility". Technical report, Öko-Institut e.V. Berlin, Germany. 79
- [Hacker et al., 2011b] Hacker, F., Harthan, R., Kasten, P., Loreck, C., and Zimmer, W. (2011b). Marktpotenziale und CO2-Bilanz von Elektromobilität - Arbeitspakete 2 bis 5 des Forschungsvorhabens OPTUM. Technical report, Öko-Institut, Freiburg, Berlin, Germany. 2, 29, 57, 61, 77, 89, 103, 145, 146, 147

- [Hacker et al., 2009] Hacker, F., Harthan, R., Matthes, F., and Zimmer, W. (2009). Environmental impacts and impact on the electricity market of a large scale introduction of electric cars in europe-critical review of literature. ETC/ACC technical paper, 4:56–90.
- [Hacker et al., 2015] Hacker, F., von Waldenfels, R., and Mottschall, M. (2015). Wirtschaftlichkeit von Elektromobilität in gewerblichen Anwendungen. Technical report, Öko-Institut, Berlin, Germany. 79
- [Haendel et al., 2015] Haendel, M., Gnann, T., and Plötz, P. (i.p.,2015). Optimierung und Potenziale für Elektrofahrzeuge in Fuhrparks. Report within the project "Get eReady", Fraunhofer ISI, Karlsruhe, Germany. 86, 98
- [Hare and Deadman, 2004] Hare, M. and Deadman, P. (2004). Further towards a taxonomy of agent-based simulation models in environmental management. *Mathematics and Computers in Simulation*, 64(1):25 – 40. MSSANZ/IMACS 14th Biennial Conference on Modelling and Simulation. 44
- [Hartmann and Frigg, 2006] Hartmann, S. and Frigg, R. (2006). Models in science. In Zalta, E., editor, *The Stanford Encyclopedia of Philosophy.* Stanford. 68
- [Hautzinger et al., 2013] Hautzinger, H., Kagerbauer, M., Mallig, N., Pfeiffer, M., and Zumkeller, D. (2013). Mikromodellierung für die Region Stuttgart - Schlussbericht. Technical report, INOVAPLAN GmbH, Institute for Transport Studies at the Karlsruhe Institute of Technology (KIT), Institut für angewandte Verkehrs- und Tourismusforschung e.V., Karlsruhe, Heilbronn, Germany. 30, 32, 33, 65, 81, 135
- [Heinrichs, 2014] Heinrichs, H. U. (2014). Analyse der langfristigen Auswirkungen von Elektromobilität auf das deutsche Energiesystem im europäischen Energieverbund, volume 5. KIT Scientific Publishing. 8, 133
- [Helms et al., 2011] Helms, H., Jöhrens, J., Hanusch, J., Höpfner, U., Lambrecht, U., and Pehnt, M. (2011). UMBReLA Umweltbilanzen Elektromobilität. Technical report, ifeu – Institute for Energy and Environmental Research, Heidelberg, Germany. 61, 145, 146, 147
- [Heupel et al., 2010] Heupel, T., Krol, B., and Stender, T. (2010). FOM-Umfrage: Elektromobilität Kauf- und Mobilitätsverhalten in Bezug auf Elektroautomobile. FOM Hochschule für Ökonomie und Management, Essen, Germany. 62
- [Higgins et al., 2012] Higgins, A., Paevere, P., Gardner, J., and Quezada, G. (2012). Combining choice modelling and multi-criteria analysis for technology diffusion: An application to the uptake of electric vehicles. *Technological Forecasting and Social Change*, 79(8):1399 – 1412. 89
- [Hu and Green, 2011] Hu, H. and Green, R. (2011). Making markets for hydrogen vehicles: Lessons from LPG. International Journal of Hydrogen Energy, 36(11):6399–6406. 17, 20, 21

- [Huang et al., 1997] Huang, J.-C., Haab, T. C., and Whitehead, J. C. (1997). Willingness to pay for quality improvements: Should revealed and stated preference data be combined? *Journal of Environmental Economics and Management*, 34(3):240 – 255. 62
- [Huétink et al., 2010] Huétink, F. J., van der Vooren, A., and Alkemade, F. (2010). Initial infrastructure development strategies for the transition to sustainable mobility. *Technological Forecasting and Social Change*, 77(8):1270–1281. 3, 17, 20, 21
- [Huston et al., 1988] Huston, M., DeAngelis, D., and Post, W. (1988). New computer models unify ecological theory. *BioScience*, pages 682–691. 44
- [ICCT, 2014] ICCT (2014). European vehicle market statistics. The International Council on Clean Transportation (ICCT). 7
- [IEA, 2010] IEA (2010). International Energy Agency (IEA): Energy Technology Perspectives 2010. Technical report, Paris, France. 1, 5, 77
- [IEA, 2013] IEA (2013). International Energy Agency (IEA): World Energy Outlook 2013. Paris, France. 58
- [IEA, 2014] IEA (2014). International Energy Agency (IEA): World Energy Outlook 2014. Paris, France. 1
- [infas and DLR, 2002] infas and DLR (2002). Mobilität in Deutschland (MiD) 2002. Technical report, infas Institut für angewandte Sozialwissenschaft GmbH, Deutsches Zentrum für Luft- und Raumfahrt e. V. (DLR), Bonn, Berlin, Germany. 8, 10, 30, 33, 34, 35, 66, 67, 121
- [infas and DLR, 2008] infas and DLR (2008). Mobilität in Deutschland (MiD) 2008. Technical report, infas Institut für angewandte Sozialwissenschaft GmbH, Deutsches Zentrum für Luft- und Raumfahrt e. V. (DLR), Bonn, Berlin, Germany. 30, 33, 34, 35, 103
- [IVS et al., 2002] IVS et al. (2002). Kraftfahrzeugverkehr in Deutschland 2002 (KiD2002). Technical report, IVS Institut für Verkehr und Stadtbauwesen, Technische Universität Braunschweig, Braunschweig, Germany. 30, 35, 37, 38, 70, 86
- [Jakobsson et al., 2014] Jakobsson, N., Plötz, P., Gnann, T., Sprei, F., and Karlsson, S. (2014). Are electric vehicles better suited for multi-car households? In *Proceedings of the European Electric Vehicle Conference EEVC*, Brussels, Belgium. 98, 99
- [Janssen et al., 2006] Janssen, A., Lienin, S. F., Gassmann, F., and Wokaun, A. (2006). Model aided policy development for the market penetration of natural gas vehicles in Switzerland. *Transportation Research Part A: Policy and Practice*, 40(4):316–333. 3, 8, 11, 12, 17, 18, 19, 20, 21, 22
- [Jebaraj and Iniyan, 2006] Jebaraj, S. and Iniyan, S. (2006). A review of energy models. Renewable and Sustainable Energy Reviews, 10(4):281 – 311. 15

- [Jin et al., 2014] Jin, L., Searle, S., and Lutsey, N. (2014). Evaluation of state-level US electric vehicle incentives. The International Council on Clean Transportation (ICCT) White Paper. 123
- [Jochem et al., 2014] Jochem, P., Kaschub, T., and Fichtner, W. (2014). *How to integrate electric vehicles in the future energy system?* Springer, Berlin, Germany. 9
- [JP-StatBureau, 2013] JP-StatBureau (2013). Japanese Statistics Bureau: Japan Statistical Yearbook, Chapter 12 Transport and Tourism, Statistics of Motor Vehicles Owned. Tokyo, Japan. 2
- [Jun, 2012] Jun, S.-P. (2012). A comparative study of hype cycles among actors within the socio-technical system: With a focus on the case study of hybrid cars. *Technological Forecasting and Social Change*, 79(8):1413 – 1430. 1
- [Kagerbauer, 2010] Kagerbauer, M. (2010). Mikroskopische Modellierung des Außenverkehrs eines Planungsraums, volume 70. KIT Scientific Publishing, Karlsruhe, Germany. 40
- [Kahn Ribeiro et al., 2007] Kahn Ribeiro, S., Kobayashi, S., Beuthe, M., Gasca, J., Greene, D., Lee, D. S., Muromachi, S., Newton, P. J., Plotkin, S., Sperling, D., Wit, R., and Zhou, P. J. (2007). Transport and its infrastructure". Technical report, Working Group III to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change [B. Metz, O.R. Davidson, P.R. Bosch, R. Dave, L.A. Meyer (eds)], Cambridge, United Kingdom and New York, NY, USA. 1, 5, 77
- [Kaldor, 1957] Kaldor, N. (1957). A model of economic growth. The Economic Journal, (268):591–624. 10
- [Kalhammer et al., 2007] Kalhammer, F. R., Kopf, B. M., Swan, D. H., Roan, V. P., and Walsh, M. P. (2007). Status and prospects for zero emissions vehicle technology. *Report* of the ARB Independent Expert Panel, 1(1):12–36. 2, 46, 77
- [Karlsch and Stokes, 2003] Karlsch, R. and Stokes, R. G. (2003). Faktor Öl Die Mineralölwirtschaft in Deutschland. C.H. Beck, Munich, Germany. 3
- [Karlsson, 2013] Karlsson, S. (2013). The Swedish car movement data project, Final Report. Technical report, Chalmers University of Technology, Gothemburg, Sweden. 3, 28, 29
- [Karlsson and Kullingsjö, 2013] Karlsson, S. and Kullingsjö, L.-H. (2013). GPS measurement of Swedish car movements for assessment of possible electrification. In *Proceedings* to EVS27, Barcelona, Spain. 27, 72
- [KBA, 2012a] KBA (2012a). Federal Motor Transport Authority (KBA): Registrations of passenger cars 2001 - 2012 distinguished by segments and models (FZ11). Flensburg, Germany. 60
- [KBA, 2012b] KBA (2012b). Federal Motor Transport Authority (KBA): Registrations of trucks 2002 2011 distinguished by total weight allowed. Flensburg, Germany. 60

- [KBA, 2013a] KBA (2013a). Federal Motor Transport Authority: Vehicle stock distinguished by fuel types. www.kba.de/cln_033/nn_269000/ DE/Statistik/Fahrzeuge/Bestand/EmissionenKraftstoffe/ b_emi_z_teil_2.html, Flensburg, Germany. 11, 12
- [KBA, 2013b] KBA (2013b). Federal Motor Transport Authority: Vehicle stock in Germany. www.kba.de/cln_033/nn_191172/ DE/Statistik/Fahrzeuge/Bestand/FahrzeugklassenAufbauarten/ b_fzkl_zeitreihe.html, Flensburg, Germany. 11, 12
- [KBA, 2013] KBA (2013). Vehicle registrations in 2013 sorted by available subgroups within commercial branches and vehicle sizes. Special analysis for Fraunhofer ISI. Technical report, German Federal Motor Transport Authority (KBA), Flensburg, Germany. 40
- [KBA, 2014a] KBA (2014a). Fahrzeugzulassungen (FZ) Bestand am 01.01.2014 an Kraftfahrzeugen und Kraftfahrzeuganhängern nach Haltern, Wirtschaftszweigen (FZ23). Technical report, German Federal Motor Transport Authority (KBA), Flensburg, Germany. 10, 30, 65, 66
- [KBA, 2014b] KBA (2014b). Fahrzeugzulassungen (FZ) Neuzulassungen 2013 an Kraftfahrzeugen und Kraftfahrzeuganhängern nach Haltern, Wirtschaftszweigen (FZ24). Technical report, German Federal Motor Transport Authority (KBA), Flensburg, Germany. 2, 29, 30, 38, 39
- [KBA, 2014c] KBA (2014c). Federal Motor Transport Authority (KBA): Vehicle stock (01/01/2014) distinguished by vehicle registrations areas (FZ1). Flensburg, Germany. 65, 66, 67
- [Keles et al., 2008] Keles, D., Wietschel, M., Möst, D., and Rentz, O. (2008). Market penetration of fuel cell vehicles analysis based on agent behaviour. *International Journal of Hydrogen Energy*, 33(16):4444–4455. 2, 27, 77, 89
- [Kemfert, 1998] Kemfert, C. (1998). Makroökonomische Wirkungen umweltökonomischer Instrumente. Lang, Bern, Switzerland. 16
- [Kemp and Volpi, 2008] Kemp, R. and Volpi, M. (2008). The diffusion of clean technologies: a review with suggestions for future diffusion analysis. *Journal of Cleaner Production*, 16(1):S14–S21. 5
- [Ketelaer et al., 2014] Ketelaer, T., Kaschub, T., Jochem, P., and Fichtner, W. (2014). The potential of carbon dioxide emission reductions in german commercial transport by electric vehicles. *International Journal of Environmental Science and Technology*, 11(8):2169–2184. 38, 79
- [Khan and Kockelman, 2012] Khan, M. and Kockelman, K. M. (2012). Predicting the market potential of plug-in electric vehicles using multiday GPS data. *Energy Policy*, 46(0):225 – 233. 28, 104
- [Kihm and Trommer, 2014] Kihm, A. and Trommer, S. (2014). The new car market for electric vehicles and the potential for fuel substitution. *Energy Policy*, 73(0):147 – 157. 79, 89, 103

- [Kley, 2011] Kley, F. (2011). Ladeinfrastrukturen für Elektrofahrzeuge Analyse und Bewertung einer Aufbaustrategie auf Basis des Fahrverhaltens. Fraunhofer Publishing, Karlsruhe, Germany. 2, 8, 9, 27, 29, 31, 59, 86, 98, 102, 135
- [Kley et al., 2010] Kley, F., Dallinger, D., and Wietschel, M. (2010). Assessment of future EV charging infrastructure. International Advanced Mobility Forum (IAMF) 2010 Full Paper, Geneva, Switzerland. 8
- [Kley et al., 2011] Kley, F., Lerch, C., and Dallinger, D. (2011). New business models for electric cars - a holistic approach. *Energy policy*, 39(Nr.6):3392–3403. 8, 9
- [Klöckner, 2014] Klöckner, C. A. (2014). The dynamics of purchasing an electric vehicle a prospective longitudinal study of the decision-making process. Transportation Research Part F: Traffic Psychology and Behaviour, 24(0):103 – 116. 44
- [Knie, 1999] Knie, A. (1999). Die Neuerfindung urbaner Automobilität: Elektroautos und ihr Gebrauch in den USA und Europa. Ed. Sigma, Berlin, Germany. 62, 73
- [Köhler et al., 2010] Köhler, J., Wietschel, M., Whitmarsh, L., Keles, D., and Schade, W. (2010). Infrastructure investment for a transition to hydrogen automobiles. *Technological Forecasting and Social Change*, 77(8):1237–1248. 17, 18, 19, 20, 21, 22, 27, 89, 124
- [Konrad et al., 2012] Konrad, K., Markard, J., Ruef, A., and Truffer, B. (2012). Strategic responses to fuel cell hype and disappointment. *Technological Forecasting and Social Change*, 79(6):1084 – 1098. Contains Special Section: Actors, Strategies and Resources in Sustainability Transitions. 1
- [Kreyenberg et al., 2013] Kreyenberg, D., Wind, J., Devries, J., and Fuljahn, A. (2013). Assessing the customer value of electric vehicles. *Auto Tech Review*, 2(4):42–46. 46
- [Kuby et al., 2013] Kuby, M. J., Kelley, S. B., and Schoenemann, J. (2013). Spatial refueling patterns of alternative-fuel and gasoline vehicle drivers in Los Angeles. Transportation Research Part D: Transport and Environment, 25(0):84 – 92. 32
- [Kurani et al., 1994] Kurani, K. S., Turrentine, T., and Sperling, D. (1994). Demand for electric vehicles in hybrid households: an exploratory analysis. *Transport Policy*, 1(4):244 – 256. Special Issue Sustainable transportation and electric vehicles. 98
- [KVV, 2006] KVV (2006). Karlsruher Verkehrverbund: Nahverkehrsplan 2006. Engelhardt und Bauer, Karlsruhe, Germany. 106
- [Lam et al., 2013] Lam, A., Leung, Y.-W., and Chu, X. (2013). Electric Vehicle Charging Station Placement: Formulation, Complexity, and Solutions. arXiv preprint arXiv:1310.6925. 2
- [Lamberson, 2008] Lamberson, P. J. (2008). The diffusion of hybrid electric vehicles. Technical report, Center for the Study of Complex Systems, University of Michigan, Michigan, USA. 2, 27, 77, 89

- [Laroche et al., 2001] Laroche, M., Bergeron, J., and Barbaro-Forleo, G. (2001). Targeting consumers who are willing to pay more for environmentally friendly products. *Journal of Consumer Marketing*, 18(6):503–520. 62, 70, 85
- [Lawless, 1982] Lawless, J. F. (1982). Statistical methods and model for lifetime data. Wiley&Sons, New York. 52
- [Lemnet, 2014] Lemnet (2014). Data on German Charging Stations. Lemnet.org. Ilmenau, Germany. 59, 67
- [Lin et al., 2012] Lin, Z., Dong, J., Liu, C., and Greene, D. (2012). PHEV Energy Use Estimation: Validating the Gamma Distribution for Representing the Random Daily Driving Distance. *Transportation Research Board 91st Annual Meeting*, 2468(12). 40, 79
- [Lin and Greene, 2011] Lin, Z. and Greene, D. L. (2011). Promoting the market for plugin hybrid and battery electric vehicles. *Transportation Research Record: Journal of the Transportation Research Board*, 2252(1):49–56. 2, 9, 77, 120, 127, 130
- [Linden and Reddy, 2002] Linden, D. and Reddy, T. (2002). Handbook of Batteries. Mcgraw-Hill Professional, New York. 61
- [Linssen et al., 2012] Linssen, J., Bickert, S., Hennings, W., et al. (2012). Netzintegration von Fahrzeugen mit elektrifizierten Antriebssystemen in bestehende und zukünftige Energieversorgungsstrukturen-Advances in Systems Analyses 1, volume 1. Forschungszentrum Jülich, Jülich, Germany. 61, 145, 146, 147
- [Linz et al., 2003] Linz, S., Dexheimer, V., and Kathe, A. (2003). Hedonische Preismessung bei Gebrauchtwagen. Wirtschaft und Statistik, page 538. 50
- [Ma et al., 2014] Ma, T., Zhao, J., Xiang, S., Zhu, Y., and Liu, P. (2014). An agentbased training system for optimizing the layout of afvs' initial filling stations. *Journal* of Artificial Societies and Social Simulation, 17(4). 2, 77
- [Mahler and Rogers, 1999] Mahler, A. and Rogers, E. M. (1999). The diffusion of interactive communication innovations and the critical mass: the adoption of telecommunications services by German banks. *Telecommunications Policy*, 23(1011):719–740. 4, 19
- [manager magazin, 2013] manager magazin (2013). "elektroautoprojekt better place gibt auf". Hamburg, Germany. 8
- [Marklines, 2015] Marklines (2015). xEV sales 2004-2014. www.marklines.com. 6
- [Massiani, 2010] Massiani, J. (2010). Modelling and Evaluation of the diffusion of electric vehicles: existing models, results and proposal for a new model for policy in european countries. European School of Management and Technology (ESMT), Berlin, Germany. 64
- [McKinsey, 2009] McKinsey (2009). Roads toward a low-carbon future: Reducing CO₂ emissions from passenger vehicles in the global road transport system. Technical report, New York, US. 1, 7

- [McKinsey, 2011] McKinsey (2011). McKinsey & Company: A portfolio of power-trains for Europe: a fact-based analysis - The role of Battery Electric Vehicles, Plug-in Hybrids and Fuel Cell Electric Vehicles. Technical report. 2, 73, 77, 85
- [McKinsey, 2012] McKinsey (2012). McKinsey & Company: Die Energiewende in Deutschland - Anspruch, Wirklichkeit und Perspektiven. 58
- [McManus and Senter Jr., 2009] McManus, W. and Senter Jr., R. (2009). Market models for predicting PHEV adoption and diffusion. 89
- [Meade and Islam, 2006] Meade, N. and Islam, T. (2006). Modelling and forecasting the diffusion of innovation - a 25-year review. *International Journal of Forecasting*, 22(3):519–545. 64
- [Melaina, 2003] Melaina, M. W. (2003). Initiating hydrogen infrastructures: preliminary analysis of a sufficient number of initial hydrogen stations in the US. International Journal of Hydrogen Energy, 28(7):743-755. 3, 17, 18, 20, 21, 124
- [Meyer and Winebrake, 2009] Meyer, P. E. and Winebrake, J. J. (2009). Modeling technology diffusion of complementary goods: The case of hydrogen vehicles and refueling infrastructure. *Technovation*, 29(2):77–91. 3, 17, 18, 20, 21, 22, 27, 89
- [Michaelis et al., 2013a] Michaelis, J., Junker, J., Gnann, T., and Plötz, P. (2013a). The potential use of hydrogen produced from surplus electricity of renewable sources in the german transport sector. In *Elektrik/Elektronik in Hybrid- und Elektrofahrzeugen und elektrisches Energiemanagement IV*, Haus der Technik Fachbuch; 130, pages 572–581. Expert Verlag, Renningen, Germany. 61
- [Michaelis et al., 2013b] Michaelis, J., Plötz, P., Gnann, T., and Wietschel, M. (2013b). Vergleich alternativer Antriebstechnologien Batterie-, Plug-in Hybrid- und Brennstoffzellenfahrzeug. In Alternative Antriebskonzepte bei sich wandelnden Mobilitätsstilen: Tagungsbeiträge vom 08. und 09. März 2012 am KIT, Karlsruhe, KIT Scientific Publishing, Karlsruhe, Germany, pages 51–80. Jochem, Patrick and Poganietz, Witold-Roger and Grunwald, Armin and Fichtner, Wolf. 6
- [Mock, 2010] Mock, P. (2010). Entwicklung eines Szenariomodells zur Simulation der zukünftigen Marktanteile und CO₂-Emissionen von Kraftfahrzeugen (VECTOR21).
 PhD thesis, Universität Stuttgart, Stuttgart, Germany. 57, 62, 73, 85
- [Mock et al., 2013] Mock, P., German, J., Riemersma, I., Ligterink, N., and Lambrecht, U. (2013). From laboratory to road – A comparison of official and real-world fuel consumption and CO₂ values for cars in Europe and the United States. ICCT. Technical report, Beijing, Berlin, Brussels, San Francisco, Washington. 60, 61
- [Mock et al., 2009] Mock, P., Hülsebusch, D., J., U., and A., Schmid, S. (2009). Electric vehicles - a model based assessment of future market prospects and environmental impacts. In *Proceedings of Electric Vehicle Symposium 24 (EVS 24)*, Stavanger, Norway. 2, 3, 77, 89, 103

- [Mock and Yang, 2014] Mock, P. and Yang, Z. (2014). Driving electrification: A global comparison of fiscal incentive policy for electric vehicles. The International Council on Clean Transportation (ICCT). URL http://www. theicct. org/sites/default/files/publications/ICCT_EV-fiscal-incentives_20140506. pdf. Last accessed, 22(6):2014. 7, 102, 123
- [MOP, 2010] MOP (2010). "Mobilitätspanel Deutschland" 1994-2010. Technical report, Projektbearbeitung durch das Institut für Verkehrswesen der Universität Karlsruhe (TH). Verteilt durch die Clearingstelle Verkehr des DLR-Instituts für Verkehrsforschung: www.clearingstelle-verkehr.de, Karlsruhe, Germany. 3, 8, 28, 29, 30, 33, 34, 35, 59, 80, 81, 135, 136
- [Motavalli, J., 2012] Motavalli, J. (2012). Flex-Fuel Amendment Makes for Strange Bedfellows. The New York Times, New York, US. 6
- [Mueller and de Haan, 2009] Mueller, M. G. and de Haan, P. (2009). How much do incentives affect car purchase? Agent-based microsimulation of consumer choice of new cars–Part I: Model structure, simulation of bounded rationality, and model validation. *Energy Policy*, 37(3):1072–1082. 2, 44
- [MWV, 2013] MWV (2013). Association of the German Petroleum Industry (MWV): Entwicklungen des Tankstellenbestandes in Deutschland. Berlin, Germany. 11, 12
- [Namdeo et al., 2014] Namdeo, A., Tiwary, A., and Dziurla, R. (2014). Spatial planning of public charging points using multi-dimensional analysis of early adopters of electric vehicles for a city region. *Technological Forecasting and Social Change*, 89(0):188 – 200.
- [Naumer et al., 2010] Naumer, H.-J., Nacken, D., and Scheurer, S. (2010). Allianz Global investors: The sixth Kondratieff - long waves of prosperity. Frankfurt, Germany. 1
- [Naumer et al., 2013] Naumer, H.-J., Nacken, D., and Scheurer, S. (2013). Allianz Global investors: The "green" Kondratieff or why crises can be a good thing. Frankfurt, Germany. 1
- [Nelson et al., 2009] Nelson, P. A., Santini, D. J., and Barnes, J. (2009). Factors determining the manufacturing costs of lithium-ion batteries for PHEVs. In *Proceedings of Electric Vehicle Symposium 24 (EVS 24)*, Stavanger, Norway. 58
- [Nemry and Brons, 2010] Nemry, F. and Brons, F. (2010). Plug-in hybrid and battery electric vehicles - market penetration scenarios of electric drive vehicles. Technical report, European Commission, Joint Research Centre, Institute for Prospective Technological Studies, Sevilla, Spain; Luxemburg. 2, 3, 27, 77, 89
- [NEP, 2009] NEP (2009). Federal Government of Germany: National Development Program E-Mobility (Nationaler Entwicklungsplan Elektromobilität der Bundesregierung). Technical report, Berlin, Germany. 1, 2
- [Neubauer et al., 2012] Neubauer, J., Brooker, A., and Wood, E. (2012). Sensitivity of battery electric vehicle economics to drive patterns, vehicle range, and charge strategies. *Journal of Power Sources*, 209(0):269 – 277. 2, 27, 28, 77, 86, 89

- [Newman et al., 2014] Newman, D., Wells, P., Donovan, C., Nieuwenhuis, P., and Davies, H. (2014). Urban, sub-urban or rural: where is the best place for electric vehicles? *International Journal of Automotive Technology and Management*, 14(3):306–323. 113
- [NGV-Journal, 2015] NGV-Journal (2015). The NGV-Journal: Worldwide NGV statistics. http://www.ngvjournal.com/worldwide-ngv-statistics/. 6
- [NPE, 2011a] NPE (2011a). Nationale Plattform Elektromobilität (NPE): Anhang des zweiten Berichts der Nationalen Plattform Elektromobilität. Technical report, Federal Government's Joint Unit for Electric Mobility (GGEMO), Berlin, Germany. 60, 61
- [NPE, 2011b] NPE (2011b). Nationale Plattform Elektromobilität (NPE): Zweiter Bericht der Nationalen Plattform Elektromobilität. Technical report, Federal Government's Joint Unit for Electric Mobility (GGEMO), Berlin, Germany. 60, 61, 73
- [NPE, 2012] NPE (2012). Nationale Plattform Elektromobilität (NPE): Fortschrittsbericht der Nationalen Plattform Elektromobilität (Dritter Bericht). Technical report, Federal Government's Joint Unit for Electric Mobility (GGEMO), Berlin, Germany. 2, 60, 77
- [NPE, 2014] NPE (2014). Nationale Plattform Elektromobilität (NPE): Fortschrittsbericht 2014 - Bilanz der Marktvorbereitung. Technical report, Federal Government's Joint Unit for Electric Mobility (GGEMO), Berlin, Germany. 2, 7, 123
- [Nykvist and Nilsson, 2015] Nykvist, B. and Nilsson, M. (2015). Rapidly falling costs of battery packs for electric vehicles. *Nature Clim. Change*, 5(4):329–332. 1, 58, 60, 101, 132
- [Offer et al., 2010] Offer, G. J., Howey, D. A., Contestabile, M., Clague, R., and Brandon, N. R. (2010). Comparative analysis of battery electric, hydrogen fuel cell and hybrid vehicles in a future sustainable road transport system. *Energy Policy*, 38:24–29. 18
- [OICA, 2014] OICA (2014). International Organization of Motor Vehicle Manufacturers (OICA): 2014 Q2 Production Statistics. Paris, France. 1
- [Orbach and Fruchter, 2011] Orbach, Y. and Fruchter, G. E. (2011). Forecasting sales and product evolution: The case of the hybrid/electric car. *Technological Forecasting* and Social Change, 78(7):1210–1226. 27
- [Pasaoglu et al., 2014] Pasaoglu, G., Fiorello, D., Martino, A., Zani, L., Zubaryeva, A., and Thiel, C. (2014). Travel patterns and the potential use of electric cars - Results from a direct survey in six European countries. *Technological Forecasting and Social Change*, 87(0):51 – 59. 132
- [Pearre et al., 2011] Pearre, N. S., Kempton, W., Guensler, R. L., and Elango, V. V. (2011). Electric vehicles: How much range is required for a days driving? *Transportation Research Part C: Emerging Technologies*, 19(6):1171 – 1184. 28
- [Peters et al., 2011a] Peters, A., Agosti, A., Popp, M., and Ryf, B. (2011a). Electric mobility - a survey of differnt consumer groups in germany with regard to adoption. Proceedings of the 2011 ECEEE summer study, Toulon, France. 44, 62, 63, 66, 72

- [Peters et al., 2011b] Peters, A., Agosti, A., Popp, M., and Ryf, B. (2011b). Elektroautos in der Wahrnehmung der Konsumenten - Zusammenfassung der Ergebnisse einer Befragung in Deutschland. Technical report, Fraunhofer ISI, Karlsruhe, Germany. 14, 46, 62, 63
- [Peters and de Haan, 2006] Peters, A. and de Haan, P. (2006). Der Autokäufer seine Charakteristika und Präferenzen. Ergebnisbericht im Rahmen des Projekts "Entscheidungsfaktoren beim Kauf treibstoffeffizienter Neuwagen". Technical report, ETH Zurich, Zurich, Switzerland. 44, 46, 73
- [Peters et al., 2012] Peters, A., Doll, C., Kley, F., Möckel, M., Plötz, P., Sauer, A., Schade, W., Thielmann, A., Wietschel, M., and Zanker, C. (2012). Konzepte der Elektromobilität und deren Bedeutung für Wirtschaft, Gesellschaft und Umwelt. Büro für Technikfolgen-Abschätzung beim Deutschen Bundestag (TAB), Berlin, Germany. 73
- [Peters et al., 2013] Peters, A., Doll, C., Plötz, P., Sauer, A., Schade, W., Wietschel, M., and Zanker, C. (2013). Konzepte der Elektromobilität und deren Bedeutung für Wirtschaft, Gesellschaft und Umwelt - Studie des Büros für Technikfolgen-Abschätzung beim Deutschen Bundestag. Technical report, Fraunhofer ISI, Karlsruhe. 1, 7
- [Peters and Dütschke, 2014] Peters, A. and Dütschke, E. (2014). How do Consumers Perceive Electric Vehicles? A Comparison of German Consumer Groups. Journal of Environmental Policy & Planning, 16(3):359–377. 62, 63, 66, 72, 73
- [Pfahl, 2013] Pfahl, S. (2013). 4. Alternative Antriebskonzepte: Stand der Technik und Perspektiven-Die Sicht der Automobilindustrie. In Alternative Antriebskonzepte bei sich wandelnden Mobilitätsstilen: Tagungsbeiträge vom 08. und 09. März 2012 am KIT, Karlsruhe, KIT Scientific Publishing, Karlsruhe, Germany, pages 81-108. Jochem, Patrick and Poganietz, Witold-Roger and Grunwald, Armin and Fichtner, Wolf. 29, 45, 49, 57, 58, 60, 61, 65, 66, 74, 145, 146, 147
- [Pfahl et al., 2013] Pfahl, S., Jochem, P., and Fichtner, W. (2013). When will Electric Vehicles Capture the German Market? And Why? In Proceedings of Electric Vehicle Symposium 27 (EVS 27), Barcelona, Spain. 2, 77, 85
- [Plötz, 2014] Plötz, P. (2014). How to estimate the probability of rare long-distance trips. Working Paper Sustainability and Innovation, Fraunhofer ISI, Karlsruhe, Germany. 66, 73, 79, 99, 104, 120
- [Plötz et al., 2015] Plötz, P., Funke, S., and Jochem, P. (2015). Real-world fuel economy and CO₂ emissions of plug-in hybrid electric vehicles. Working Paper Sustainability and Innovation, Fraunhofer ISI, Karlsruhe, Germany. 68
- [Plötz and Gnann, 2013] Plötz, P. and Gnann, T. (2013). Who should buy electric vehicles? - the potential early adopter from an economical perspective. In *Proceedings of* the 2013 ECEEE summer study, Toulon, France. 73
- [Plötz et al., 2012] Plötz, P., Gnann, T., and Wietschel, M. (2012). Total Ownership Cost Projection for the German Electric Vehicle Market with Implications for its Future

Power and Electricity Demand. In ENERDAY 7th Conference on Energy Economics and Technology, Dresden, Germany. 52, 70, 73

- [Plötz et al., 2014a] Plötz, P., Gnann, T., and Wietschel, M. (2014a). Modelling market diffusion of electric vehicles with real world driving data – part i: Model structure and validation. *Ecological Economics*, 107(0):411 – 421. 27, 28, 36, 43, 47, 49, 61, 62, 64, 68, 70, 83, 84, 86, 90, 91, 97
- [Plötz et al., 2013] Plötz, P., Gnann, T., Wietschel, M., and Kühn, A. (2013). Market evolution scenarios for electric vehicles - detailed version. Commissioned by acatech -German National Academy of Science and Engineering and Working Group 7 of the German National Platform for Electric Mobility (NPE). Fraunhofer ISI, Karlsruhe, Germany. 52, 57, 59, 61, 64, 65, 86, 89, 120, 123, 132, 145, 147
- [Plötz et al., 2015] Plötz, P., Gnann, T., Wietschel, M., and Ullrich, S. (2015). How to foster electric vehicle market penetration? - a model based assessment of policy measures and external factors. In Accepted paper at the 2015 ECEEE summer study, Toulon, France. 123
- [Plötz et al., 2014b] Plötz, P., Schneider, U., Globisch, J., and Dütschke, E. (2014b). Who will buy electric vehicles? Identifying early adopters in Germany. *Transportation Research Part A: Policy and Practice*, 67(0):96 – 109. 62, 73, 113
- [Potoglou and Kanaroglou, 2007] Potoglou, D. and Kanaroglou, P. S. (2007). Household demand and willingness to pay for clean vehicles. *Transportation Research Part D: Transport and Environment*, 12(4):264 – 274. 89
- [Propfe et al., 2012a] Propfe, B., Kreyenberg, D., Wind, J., and Schmid, S. (2012a). Market Penetration Analysis of Electric Vehicles in the German Passenger Car Market Towards 2030. Technical report, German Aerospace Center (DLR), Daimler AG, Stuttgart, Kirchheim/Teck-Nabern, Germany. 89, 103, 133
- [Propfe et al., 2012b] Propfe, B., Redelbach, M., Santini, D. J., and Friedrich, H. (2012b). Cost analysis of Plug-in Hybrid Electric Vehicles including Maintenance & Repair Costs and Resale Values. In *Proceedings of Electric Vehicle Symposium 26 (EVS 26)*, Los Angeles, USA. 61, 145, 146, 147
- [Redelbach et al., 2013] Redelbach, M., Sparka, M., Schmid, S., and Friedrich, H. E. (2013). Modelling customer choice and market development for future automotive powertrain technologies. In *Proceedings of the Electric Vehicle Symposium (EVS27)*, Barcelona, Spain. 2, 77, 79, 103
- [RegProg, 2011] RegProg (2011). Federal Government of Germany: Government program E-Mobility (Regierungsprogramm Elektromobilität). Technical report, BMWi, BMVBS, BMW, BMBF, Berlin, Germany. 1, 2
- [Reinke, 2014] Reinke, J. (2014). Bereitstellung öffentlicher Ladeinfrastruktur für Elektrofahrzeuge: eine institutionenökonomische Analyse. PhD thesis, TU Berlin, Berlin, Germany. 8

- [Richter and Lindenberger, 2010] Richter, J. and Lindenberger, D. (2010). Potenziale der Elektromobilität bis 2050 - Eine szenarienbasierte Analyse der Wirtschaftlichkeit, Umweltauswirkungen und Systemintegration. Endbericht, Energiewirtschaftliches Institut an der Universität zu Köln (EWI), Cologne, Germany. 89
- [Rogers, 1962] Rogers, E. M. (1962). Diffusion of innovations. Free Press of Glencoe, New York, US. 14, 16, 18, 46, 62, 64
- [Rousseau et al., 2012] Rousseau, A., Badin, M., Redelbach, M., Kim, A., Da Costa, D., Santini, D., Vyas, A., Le Berr, F., and Friedrich, H. (2012). Comparison of Energy consumption and costs of different HEVs and PHEVs in European and American context. In *Proceeding of European Electric Vehicle Congress (EEVC)*, Brussels, Belgium. 58
- [Rune, 2001] Rune, J. (2001). Mobile telecommunications network and method for implementing and identifying hierarchical overlapping radio coverage areas. US Patent 6,275,706. 106
- [Santini and Vyas, 2005] Santini, D. and Vyas, A. (2005). Suggestions for a New Vehicle Choice Model Simulating Advanced vehicle Introduction Decisions (AVID): Structure and Coefficients. Technical report, Argonne National Laboratory, Chicago, US. 2, 3, 27, 29, 77
- [Santini et al., 2010] Santini, D. J., Gallagher, K. G., and Nelson, P. A. (2010). Modeling of manufacturing costs of lithium-ion batteries for HEVs, PHEVs, and EVs. In *Proceedings of Electric Vehicle Symposium 25 (EVS 25)*, Shenzhen, China. 58
- [Sathaye and Kelley, 2013] Sathaye, N. and Kelley, S. (2013). An approach for the optimal planning of electric vehicle infrastructure for highway corridors. *Transportation Research Part E: Logistics and Transportation Review*, 59(0):15 – 33. 2
- [Schade, 2008] Schade, W. (2008). ASTRA Assessment of Transport Strategies Ein Systemdynamik-Modell zur integrierten Bewertung von langfristigen Politikstrategien - Kurzübersicht des ASTRA Modells. Technical report, Fraunhofer ISI, Karlsruhe, Germany. 27
- [Schade, W., 2010] Schade, W. (2010). Reducing Greenhouse Gas Emissions of Transport beyond 2020: Linking R&D, Transport Policies and Reduction Targets. Technical report, Fraunhofer ISI, Karlsruhe. 1, 7
- [Schlesinger et al., 2011] Schlesinger, M., Lindenberger, D., and Lutz, C. (2011). Energieszenarien. Study on Behalf of the German Ministry for Economy and Technology. 58
- [Schraven et al., 2011] Schraven, S., Kley, F., and Wietschel, M. (2011). Induktives Laden von Elektromobilen Eine techno-ökonomische Bewertung. Zeitschrift für Energiewirtschaft, 35(3):209–219. 8
- [Schroeder and Traber, 2012] Schroeder, A. and Traber, T. (2012). The economics of fast charging infrastructure for electric vehicles. *Energy Policy*, 43(0):136 – 144. 123, 129, 131

- [Schühle, 2014] Schühle, F. (2014). Die Marktdurchdringung der Elektromobilität in Deutschland: Eine Akzeptanz-und Absatzprognose. Rainer Hampp Verlag, Mering, Germany. 2, 29, 77, 79
- [Schwoon, 2006] Schwoon, M. (2006). Simulating the adoption of fuel cell vehicles. Journal of Evolutionary Economics, 16(4):435–472. 3, 17, 19, 20, 21
- [Schwoon, 2007] Schwoon, M. (2007). A tool to optimize the initial distribution of hydrogen filling stations. Transportation Research Part D: Transport and Environment, 12(2):70-82. 17, 19, 20, 21, 124
- [Seixas et al., 2015] Seixas, J., Simes, S., Dias, L., Kanudia, A., Fortes, P., and Gargiulo, M. (2015). Assessing the cost-effectiveness of electric vehicles in European countries using integrated modeling. *Energy Policy*, 80(0):165 – 176. 132
- [Sensfuss, 2008] Sensfuss, F. (2008). Assessment of the impact of renewable electricity generation on the German electricity sector: An agent-based simulation approach. Number 188 in 16. VDI-Verlag, Düsseldorf, Germany. 16, 17, 44
- [Shafiei et al., 2012] Shafiei, E., Thorkelsson, H., Asgeirsson, E. I., Davidsdottir, B., Raberto, M., and Stefansson, H. (2012). An agent-based modeling approach to predict the evolution of market share of electric vehicles: A case study from Iceland. *Technological Forecasting and Social Change*, 79(9):1638 – 1653. 29, 89
- [Shepherd et al., 2012] Shepherd, S., Bonsall, P., and Harrison, G. (2012). Factors affecting future demand for electric vehicles: A model based study. *Transport Policy*, 20(0):62–74. 2, 3, 27, 77, 89
- [Sierzchula, 2014] Sierzchula, W. (2014). Factors influencing fleet manager adoption of electric vehicles. Transportation Research Part D: Transport and Environment, 31(0):126 – 134. 44, 85
- [Skinner et al., 2010] Skinner, I., van Essen, H., Smokers, R., and Hill, N. (2010). Towards the decarbonisation of EU's transport sector by 2050. European Commission, Brussels, Germany. 1, 7
- [Smith et al., 2011] Smith, R., Shahidinejad, S., Blair, D., and Bibeau, E. (2011). Characterization of urban commuter driving profiles to optimize battery size in light-duty plug-in electric vehicles. *Transportation Research Part D: Transport and Environment*, 16(3):218 – 224. 2, 27, 72, 77
- [Sperling and Kitamura, 1986] Sperling, D. and Kitamura, R. (1986). Refueling and new fuels: An exploratory analysis. *Transportation Research Part A: General*, 20(1):15 – 23. 13
- [Sperling and Kurani, 1987] Sperling, D. and Kurani, K. S. (1987). Refueling and the vehicle purchase decision: the diesel car case. Technical report, Transportation Research Group, Departments of Environmental Studies and Civil Engineering, Univ. of California, Davis, US. 5, 13, 18

- [Srivastava et al., 2010] Srivastava, A. K., Annabathina, B., and Kamalasadan, S. (2010). The challenges and policy options for integrating plug-in hybrid electric vehicle into the electric grid. *The Electricity Journal*, 23(3):83 – 91. 123
- [Steinbach, 2015] Steinbach, J. (2015). Modellbasierte Untersuchung von Politikinstrumenten fur Förderung erneuerbarer Energien und Energieeffizienz im Gebäudebereich. PhD thesis, Karlsruhe Instritute of Technology (KIT), Karlsruhe, Germany. 44
- [Steinhilber et al., 2013] Steinhilber, S., Wells, P., and Thankappan, S. (2013). Sociotechnical inertia: Understanding the barriers to electric vehicles. *Energy Policy*, 60(0):531 – 539. 2, 77
- [Steinmeyer, 2007] Steinmeyer, I. (2007). Definition und Bedeutung des Personenwirtschaftsverkehrs. Ein Sachstandsbericht aus dem Jahr 2006. Schriftenreihe A des Instituts für Land- und Seeverkehr, (44). 29
- [Stephan and Sullivan, 2004] Stephan, C. and Sullivan, J. (2004). An agent-based hydrogen vehicle/infrastructure model. In Congress on Evolutionary Computation, 2004. CEC2004, volume 2, pages 1774 – 1779. 3, 17, 19, 20, 21, 124
- [Sullivan et al., 2009] Sullivan, J. L., Salmeen, I. T., and Simon, C. P. (2009). PHEV marketplace penetration: An agent based simulation. 89
- [Tate et al., 2008] Tate, E., Harpster, M. O., and Savagian, P. J. (2008). The electrification of the automobile: From conventional hybrid, to plug-in hybrids, to extended-range electric vehicles. Detroit, US. 2, 46
- [Thiel et al., 2010] Thiel, C., Perujo, A., and Mercier, A. (2010). Cost and CO₂ aspects of future vehicle options in Europe under new energy policy scenarios. *Energy Policy*, 38(11):7142 – 7151. Energy Efficiency Policies and Strategies with regular papers. 2, 77, 85
- [Thielmann, 2011] Thielmann, A. (2011). Energiespeicher-Monitoring f
 ür die Elektromoblit
 ät (EMOTOR) - Trendbericht Teil 1. Technical report, Fraunhofer ISI, Karlsruhe, Germany. 1
- [Thielmann et al., 2014] Thielmann, A., Fan, C., Friedrichsen, N., Gnann, T., Hettesheimer, T., Hummen, T., Marscheider-Weidemann, F., Reiss, T., Sauer, A., and Wietschel, M. (2014). Energiespeicher für die Elektromobilität - Deutschland auf dem Weg zum Leitmarkt und Leitanbieter? Fraunhofer ISI, Karlsruhe. 8
- [Thielmann et al., 2012] Thielmann, A., Isenmann, R., and Wietschel, M. (2012). Technologie-Roadmap Energiespeicher für die Elektromobilität 2030. Technical report, Technologie-Roadmapping am Fraunhofer ISI: Konzepte-Methoden-Praxisbeispiele, Karlsruhe, Germany. 1, 8, 102
- [Train, 2009] Train, K. E. (2009). Discrete Choice Methods with Simulation. Cambridge University Press, Cambridge, UK, 2nd edition. 88
- [Tran and Daim, 2008] Tran, T. A. and Daim, T. (2008). A taxonomic review of methods and tools applied in technology assessment. *Technological Forecasting and Social Change*, 75(9):1396 – 1405. 15

- [Trigg et al., 2013] Trigg, T., Telleen, P., Boyd, R., Cuenot, F., DAmbrosio, D., Gaghen, R., Gagné, J., Hardcastle, A., Houssin, D., Jones, A., et al. (2013). Global EV outlook: Understanding the electric vehicle landscape to 2020. *IEA*, *Paris, France.* 130
- [Turrentine and Kurani, 2007] Turrentine, T. S. and Kurani, K. S. (2007). Car buyers and fuel economy? *Energy Policy*, 35(2):1213 1223. 73
- [UK DoT, 2013] UK DoT (2013). UK Department for Transport: Vehicle Licensing Statistics: 2013. London, UK. 2
- [UP, 2012] UP (2012). Unione petrolifera: Statistiche economiche, energetiche e petroliefere, Rome, Italy. 11, 12
- [US-DoT, 2009] US-DoT (2009). 2009 National Household Travel Survey. Technical report, U.S. Department of Transportation, Federal Highway Administration, Washington, D.C., US. 3, 28, 29
- [van der Vooren and Alkemade, 2010] van der Vooren, A. and Alkemade, F. (2010). The diffusion of infrastructure dependent technologies. a simple model. Karlsruhe, Germany. 3, 17, 19, 20, 21, 64
- [VCD, 2008] VCD (2008). CO₂-basierte Dienstwagenbesteuerung. Technical report, Verkehrsclub Deutschland (VCD), Berlin, Germany. 30
- [VDA, 2010] VDA (2006, 2010). International Auto Statistics. German Association of the Automotive Industry (VDA) Berlin; Frankfurt, M., Germany. 11, 12
- [Ventosa et al., 2005] Ventosa, M., Baillo, A., Ramos, A., and Rivier, M. (2005). Electricity market modeling trends. *Energy policy*, 33(7):897–913. 16
- [Voltstat, 2014] Voltstat (2014). Mobility data of Chevrolet Volt. 68, 69, 70
- [Vyas et al., 2009] Vyas, A., Santini, D., and Johnson, L. (2009). Plug-in hybrid electric vehicles; potential for petroleum use reduction: issues involved in developing reliable estimates. In 88th Annual Meeting of the Transportation Research Board, Washington D.C., US. 9
- [Wansart and Schnieder, 2010] Wansart, J. and Schnieder, E. (2010). Modeling market development of electric vehicles. In 2010 4th Annual IEEE Systems Conference, pages 371 –376, San Diego, US. 2, 27, 77, 89
- [Watabe and Mori, 2011] Watabe, T. and Mori, M. (2011). Deutsche bank global market research: LiB materials industry - automotive LiB materials get set for growth phase in 2011. Technical report, Deutsche Bank FITT Research. 7
- [Weiss et al., 2014] Weiss, C., Chlond, B., Heilig, M., and Vortisch, P. (2014). Capturing the Usage of the German Car Fleet for a One Year Period to Evaluate the Suitability of Battery Electric Vehicles A Model based Approach. *Transportation Research Procedia*, 1(1):133 – 141. Planning for the future of transport: challenges, methods, analysis and impacts - 41st European Transport Conference Selected Proceedings. 99

- [Wesche, 2013] Wesche, J. (2013). Nutzerakzeptanz von Stromladetarifen im Bereich der Elektromobilität. Master's thesis, Kassel University and Fraunhofer ISI, Karlsruhe, Germany. 63
- [Wietschel et al., 2011] Wietschel, M., Dallinger, D., Doll, C., Gnann, T., Held, A., Kley, F., Lerch, C., Marscheider-Weidemann, F., Mattes, K., Peters, A., Plötz, P., and Schröter, M. (2011). Gesellschaftspolitische Fragestellungen der Elektromobilität. Technical report, Fraunhofer ISI, Karlsruhe, Germany. 6
- [Wietschel et al., 2012] Wietschel, M., Dütschke, E., Funke, S., Peters, A., Plötz, P., Schneider, U., Roser, A., and Globisch, J. (2012). Kaufpotenzial für Elektrofahrzeuge bei sogenannten "Early Adoptern". Endbericht, Fraunhofer ISI, IREES GmbH, Karlsruhe, Germany. 46, 62, 63, 66, 72, 73
- [Wietschel et al., 2014a] Wietschel, M., Gnann, T., Plötz, P., and Ullrich, S. (2014a). Impact of policy options on the market evolution of electric vehicles. Commissioned by acatech - German National Academy of Science and Engineering and Working Group 7 of the German National Platform for Electric Mobility (NPE). Fraunhofer ISI, Karlsruhe, Germany. 90
- [Wietschel et al., 2009] Wietschel, M., Kley, F., and Dallinger, D. (2009). Eine Bewertung der Ladeinfrastruktur für Elektrofahrzeuge. ZfAW Zeitschrift für die Wertschöpfungskette Automobilwirtschaft, 3:S. 33–41. 8
- [Wietschel et al., 2014b] Wietschel, M., Plötz, P., Kühn, A., and Gnann, T. (2014b). Market evolution scenarios for electric vehicles - summary. Commissioned by acatech - German National Academy of Science and Engineering and Working Group 7 of the German National Platform for Electric Mobility (NPE). Karlsruhe, Germany. 79
- [WLPGA, 2014] WLPGA (2014). World LP Gas Association: Autogas incentive policies. Neuilly-sur-Seine, France. 6
- [Wöhe and Döring, 2002] Wöhe, G. and Döring, U. (2002). *Einführung in die allgemeine Betriebswirtschaftslehre*. Vahlen, Munich, Germany. 49
- [Wooldridge and Jennings, 1995] Wooldridge, M. and Jennings, N. R. (1995). Intelligent agents: Theory and practice. *The knowledge engineering review*, 10(02):115–152. 44
- [Wu et al., 2015] Wu, G., Inderbitzin, A., and Bening, C. (2015). Total cost of ownership of electric vehicles compared to conventional vehicles: A probabilistic analysis and projection across market segments. *Energy Policy*, 80(0):196 – 214. 2, 77
- [WVI et al., 2010] WVI, IVT, DLR, and KBA (2010). Kraftfahrzeugverkehr in Deutschland 2010 (KiD2010). Technical report, WVI Prof. Dr. Wermuth Verkehrsforschung und Infrastrukturplanung GmbH, Braunschweig, IVT Institut für angewandte Verkehrsund Tourismusforschung e. V., Heilbronn, DLR Deutsches Zentrum für Luft- und Raumfahrt - Institut für Verkehrsforschung, Berlin, KBA Kraftfahrt-Bundesamt, Flensburg, Germany. 28, 29, 35, 36, 37, 38, 39, 86, 103

- [Yabe et al., 2012] Yabe, K., Shinoda, Y., Seki, T., Tanaka, H., and Akisawa, A. (2012). Market penetration speed and effects on CO₂ reduction of electric vehicles and plug-in hybrid electric vehicles in Japan. *Energy Policy*, 45(0):529 – 540. 89
- [Yeh, 2007] Yeh, S. (2007). An empirical analysis on the adoption of alternative fuel vehicles: The case of natural gas vehicles. *Energy Policy*, 35(11):5865–5875. 12
- [Zeigler, 1976] Zeigler, B. P. (1976). Theory of modelling and simulation. Wiley, New York, US. XXII, 435 S. 15
- [Zhang et al., 2013] Zhang, L., Brown, T., and Samuelsen, S. (2013). Evaluation of charging infrastructure requirements and operating costs for plug-in electric vehicles. *Journal* of Power Sources, 240(0):515 – 524. 104
- [Zhang et al., 2011] Zhang, T., Gensler, S., and Garcia, R. (2011). A study of the diffusion of alternative fuel vehicles: An agent-based modeling approach. *Journal of Product Innovation Management*, 28:152–168. 89
- [Zischler, 2011] Zischler, F. (2011). Potentialanalyse Elektromobilität für Gemeinden in Baden-Württemberg. Diploma thesis, Karlsruhe Institute of Technology (KIT) and Fraunhofer ISI, Karlsruhe, Germany. 86, 98

How do market diffusions of plug-in electric vehicles and their charging infrastructure interact? And what is a sufficient number of public charging stations to overcome a potential lock-in? In this thesis, an agent-based simulation model is introduced to answer this question for Germany until 2030. The inclusion of different charging options and electric drive trains allows a realistic picture of the future market diffusion.

The Fraunhofer ISI analyzes the framework conditions for innovations. We explore the short- and long-term developments of innovation processes and the societal impacts of new technologies and services. On this basis, we provide our clients from industry, politics and science with policy recommendations and perspectives for key decisions. Our expertise lies in a broad scientific competence as well as an interdisciplinary and systemic research approach.



FRAUNHOFER VERLAG



